Measuring the impact of visual impairment during childhood and adolescence: Development of vision-specific patient-reported outcome measures for children and young people living with visual impairment.

Alexandra O. Robertson

Submission of thesis for the degree of Doctor of Philosophy

University College London

I, Alexandra Robertson confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

#### Abstract

Patient-reported outcome measures (PROMs) are increasingly becoming the gold standard for reporting on the quality of healthcare and are used to capture information about the subjective impact of disability or impairment. In the UK, it is estimated that 2 per 1000 children are visually impaired. Despite this, there is a dearth of psychometrically robust and age-appropriate PROMs designed for children and young people.

To address this, a pre-established theoretical and methodological framework was applied to explore the day-to-day impact of visual impairment during childhood and adolescence and, in doing so, develop a suite of age-appropriate, vision-specific PROMs. Specifically, this thesis documents the development of PROMs designed for young people aged older than 15 years. The parallel work comprising final psychometric evaluation of PROMs designed for children aged younger than 10 years is also reported demonstrating calibration of the instruments to allow for their use longitudinally in paediatric ophthalmology contexts.

One hundred and twenty-nine young people (aged 13-19 years) living with visual impairment took part in four phases of instrument development, comprising semistructured interviews, cognitive interviews, and national postal surveys. A further 86 children (aged 7-13 years) took part in the final phase of psychometric validation for the child-instrument versions.

Qualitative analysis revealed fluctuation in the self-reported impact of visual impairment during childhood and adolescence, and results have implications for developing and administering interventions to promote self-reported outcomes. The final suite of ageappropriate instruments is grounded in the perspectives of children and young people living with visual impairment and psychometrically robust for use independently, and simultaneously, within clinical practice. Follow-up studies are needed to generate the evidence required to understand children's, young peoples', parents' and clinicians' attitudes towards using the PROMs, as well as the feasibility of implementing the developed instruments within ophthalmology contexts, and with attention to novel, vision-specific approaches.

#### Acknowledgements

There are a number of people without whom this work would simply not have been possible, and to whom I would like to express my profound gratitude. It has been a wonderful journey, filled with tremendous highs and equally terrific lows. When I began this PhD, I did not comprehend the number of opportunities that it would afford, and I am extremely grateful for the consistent support and encouragement I have received throughout. Firstly, I would like to sincerely thank all of the children/young people and their families who took part in the study, and opened my mind to the debilitating and very personal experience of living with visual impairment. The willingness of some children/young people to take part in the research was, at times, overwhelming, and has had a long-lasting impact on the way I approach research and now reflect upon my experience as a PhD student.

I am extremely grateful for the support of our funders, Fight for Sight and the Child Health Research Appeal Trust (CHRAT) and feel very privileged to have completed this work within the excellent infrastructure of UCL and the NIHR BRCs at Moorfields Eye Hospital and Great Ormond Street Hospital. I would like to thank staff at these centres for their help and guidance, especially during the recruitment phases. I would also like to thank staff working at a number of external centres throughout the UK, who were extremely helpful in identifying potential participants and generously offered their time and support.

I cannot express my gratitude towards my two supervisors deeply enough. I thank Professor Jugnoo Rahi for sharing with me her wisdom and expertise, and patiently guiding me through the process. I'd like to thank Val Tadić for her enthusiasm and encouragement, and for consistently offering to read (and re-read) my work. I would also like to thank Professor Gillian Lewando Hundt for her input during the early stages of the project. A special mention goes to Professor Mario Cortina Borja who patiently and selflessly committed a lot of time to ensure the project succeeded, and made sure

that my timeline remained realistic. I feel very lucky to have worked with such an excellent combination of academics, and will treasure the skills you have taught me.

I am grateful to my MPhil/PhD examiners, Phillippa Cumberland, Rachel Knowles, Chris Morris and Chris Lloyd for their time and input and would also like to thank Mike Linacre for his prompt and reliable responses to my Rasch-related queries. It is amazing how valuable advice coming from the other side of the world can be!

Finally, I would like to thank all my family and friends for their consistent enthusiasm, encouragement, and for picking me up when things were hard. A special thanks goes to my parents for their overwhelming love and support, and reassurance that I am capable of more than I believe. Words do not express how grateful I am for all your wholehearted efforts to make the last three years as easy as possible. My sincere apologies for the necessary neglect and general absence (falling asleep on the sofa) and broken promises to come home early. I'd also like to thank Peter for your subtle career advice and (less subtle) encouragement not just throughout the past three years but also throughout my journey through higher education.

## Contents

	18
Chapter 2 Background	20
2.1 Childhood and adolescence	20
2.1.1 Adolescence as a key transitional life stage	23
2.2 Childhood disability	26
2.3 Measuring the impact of disability or illness in healthcare	28
2.3.1 Patient-reported outcome measures (PROMs) and patient-reported exp measures (PREMs)	perience 31
2.3.2 Development and application of PROMs for paediatric settings	34
2.4 PROM content, design and development	36
2.4.1 Involving children in questionnaire development	38
2.5 Visual Impairment	41
2.5.1 Impact of visual impairment upon development during childhood	42
2.5.2 Impact of visual impairment during adolescence	46
2.6 Measuring the impact of visual impairment	49
2.6.1 Paediatric vision-specific PROMs	51
2.6.2 Conceptual and theoretical framework for the development of paediatri specific PROMs	c vision- 64
2.7 Research aims	65
2.8 Summary	66
	60
Chapter 3 Psychometric measurement in nealthcare	
3.1 Change in the nature of measurement in healthcare	69
3.1 Change in the nature of measurement in healthcare 3.2 Classical Test Theory	69 71
3.1 Change in the nature of measurement	69 71 73
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76 78
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76 78 78
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76 78 78 78
<ul> <li>3.1 Change in the nature of measurement in nealthcare</li></ul>	69 71 73 74 75 76 76 78 78 78
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76 76 78 78 78 78 78
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 75 76 76 78 78 78 78 78 78
<ul> <li>3.1 Change in the nature of measurement in nearthcare</li></ul>	68 71 73 73 74 75 76 78 78 78 78 78 78 78 78 78 78
<ul> <li>3.1 Change in the nature of measurement in healthcare</li></ul>	69 71 73 74 74 75 76 78 78 78 78 78 78 78 78 78 78 78 78

4.6 Phase 1: Item development and adaptation	83
4.6.1 Participants and recruitment method	83
4.6.2 Identifying the issues relevant to YP with visual impairment	84
4.6.2.1 Topic guide development	84
4.6.3 Data collection: In-depth semi-structured interviews	99
4.6.4 Qualitative data analysis	99
4.6.4.1 First stage data analysis (exploring the impact of visual impairment du childhood and adolescence)	uring . 100
4.6.4.2 Second stage data analysis (item generation)	. 101
4.7 Phase 2: Pre-testing	.104
4.7.1 Participants and recruitment method	. 104
4.7.2 Data collection: Cognitive interviews	. 105
4.7.3 Data analysis	. 106
4.8 Phase 3: Piloting	.107
4.8.1 Participants and recruitment method	. 107
4.8.2 Data collection: Postal survey	. 109
4.8.3 Data analysis	. 110
4.9 Phase 4: Formal Validation	.110
4.9.1 Participants and recruitment method	. 111
4.9.2 Data collection: Postal survey	. 112
4.9.3 Data verification	. 114
4.9.4 Data analysis	. 114
4.9.4.1 Preliminary item reduction	. 114
4.9.4.2 Testing for unidimensionality	. 115
4.9.4.3 Formal psychometric item reduction and validation (Rasch analysis)	. 116
4.9.4.4 Calibration of the child- and YP-VQoL and FV instruments	. 120
4.9.4.5 Assessment of construct validity	. 122
4.9.4.6 Analysis of unidimensionality using the final instrument versions	. 124
Chapter 5 Growing up with visual impairment: Results from an in-dep	th
qualitative investigation	.125
5.1 Phase 1 Participation Rate	.126
5.2 Findings/themes: Challenges, changes, and coping strategies	.127
5.2.1 Acceptance of and adaptation to visual impairment	. 127
5.2.1.1 Differences between children/YP with early and late onset visual impairment	. 128
5.2.1.2 Differences between children/YP with differing severity of visual impairment	100
	120
5.2.2 Outlat Environment	104
5.2.2.1 Relationships with aduits	. 131

5.2.2.2 Romantic relationships	. 132
5.2.3 Developing and asserting independence and responsibility	. 132
5.2.3.1 Barriers to developing independence	. 132
5.2.4 Future: Developing plans in light of visual impairment	. 134
5.2.5 Transitions in Education	. 135
5.2.5.1 Social life at school	. 136
5.2.5.2 VI-specific educational support and demands	. 136
5.2.6 Coping strategies	. 137
5.2.6.1 Perseverance	. 137
5.2.6.2 Humour	. 138
5.2.6.3 Functional support from others	. 138
5.2.7 Key time points of change	. 139
5.2.7.1 Transition from Primary to Secondary education	. 139
5.2.7.2 Mid-way through Secondary education	. 139
5.2.7.3 Transition from Secondary education into higher education	. 140
5.3 Transition from paediatric to adult ophthalmology services: Views, experiences, and transition-related needs	.140
5.3.1 Confidence to self-manage healthcare in the future as an adult	141
5.3.1.1 Communication with professionals within clinical contexts	. 141
5.3.1.2 Environment	. 142
5.3.2 Emotional Attachments to Paediatric Ophthalmology Services	. 143
5.4 Summary	.143
Chapter 6 Results from psychometric evaluation	.145
6.1 Study sample characteristics	145
6.1.1 Relationship between demographic characteristics and participation	147
6.2 Participation rate	150
6.3 Development of the vision-related quality of life (VOol.) instruments	157
6.3.1 Phase 1: Item development and adaptation	157
6.3.1.1 Outcomes from the expert consensus meeting concerning the VOol	. 137
instrument development	. 172
6.3.2 Phase 2: Pre-testing	. 173
6.3.2.1 Self-reported feasibility of the VQoL instrument	. 173
6.3.2.2 VQoL item modification	. 174
6.3.2.3 Changes to response options	. 177
6.3.2.4 Outcomes from the expert consensus meeting concerning the VQoL instrument development	. 177
6.3.3. Phase 3: Piloting	. 177
6.3.3.1 Qualitative feedback	. 177
6.3.3.2 Quantitative feedback	. 178

6.3.3.3 Response distribution in the VQoL instrument	179
6.3.4 Phase 4: Validation	182
6.3.4.1 Data verification	187
6.3.4.2 Missing data	187
6.3.4.3 Use of response categories	189
6.3.4.4 Response distribution	189
6.3.4.5 Assessment of unidimensionality	192
6.3.4.6 Rasch analysis	197
6.3.4.7 Rasch analysis using the 28-item Child-VQoL instrument	198
6.3.4.8 Rasch analysis using the 35-item YP-VQoL instrument	204
6.3.4.9 Calibration of the Child- and YP-VQoL instruments	215
6.3.4.10 Testing construct validity	224
6.3.4.11 Final assessment of unidimensionality	226
6.4 Development of the functional vision (FV) instruments	227
6.4.1 Phase 1: Item development and adaptation	227
6.4.1.1 Outcomes from the expert consensus meeting concerning the FV	004
Instrument development.	234
6.4.2 Phase 2: Pre-lesting	235
6.4.2.1 Self-reported reasibility of the FV Instrument	235
6.4.2.2 FV item modification	235
6.4.2.1 Qualitative feedback	238
6.4.2.2 Quantitative feedback	200
6.4.2.2 Quantitative reedback	239
6.4.4 Phase 4: Volidation	240
6.4.4.1 Data varification	242
6.4.4.2 Missing data	247
6.4.4.3 Lise of response estagories	240
6.4.4.4 Posponso distribution	249
6.4.4.5 Assessment of unidimensionality	251
6 4 4 6 Rasch analysis	255
6.4.4.7 Rasch analysis using the 30-item Child-FV instrument	255
6.4.4.8 Rasch analysis using the 40-item YP-FV instrument	263
6.4.4.9 Calibration of the Child- and YP-FV instruments	270
6.4.4.10 Testing construct validity	
6.4.4.11 Final assessment of unidimensionality	
6.5 Directions for use of instruments	282
6.6 Summary	283
Chapter 7 Discussion	205 201
	204

7.1 Strengths
7.1.1 Instrument calibration
7.1.2 Age-range of users of the developed PROMs
7.1.3 Psychometric properties of the final instruments
7.2 Limitations
7.2.1 Participation rate and sample size
7.2.1.1 Predictors of participation: Index of multiple deprivation and severity of visual impairment
7.2.2 Influence of others upon self-report 292
7.3 The final suite of vision-specific PROMs
7.3.1 The Child (8-12 years) and YP (13-17 years) versions of the Vision-Related Quality of Life (VQoL) instrument
7.3.1.1 Definition of VQoL (as defined by children and YP)
7.3.1.2 Comparison to existing vision-specific PROMs
7.3.1.3 Construct validity of the VQoL instrument(s)
7.3.2 The Child (8-12 years) and YP (13-17 years) versions of the Functional Vision (FV) instrument
7.3.2.1 Definition of FV (as defined by children and YP)
7.3.2.2 Construct validity of the FV instrument(s)
7.4 Experience of visual impairment during childhood and adolescence: Key
findings in relation to existing knowledge
7.4.1 Age- and vision-specific transitions and changes during childhood and early adolescence
7.4.2 Transition from paediatric- to adult-ophthalmology services: Implications for clinical practice
7.6 Application of the finalised PROMs
7.7 The role of PROMs within the current NHS: Policy and practice
7.8 Summary
Chapter 8 Future Work
8.1 Preferences and attitudes towards using PROMs
8.1.1 Patient preferences
8.1.2 Parent preferences
8.1.3 Clinician preferences
8.2 Promoting positive attitudes to using PROMs within paediatric ophthalmology contexts
8.3 Electronic methods to administer PROMs
8.3.1 Combining auditory and visual techniques: A feasible method of administering PROMs?
8.4 Capacity of services to address psychologically-related outcomes
8.5 Summary
,

Chapter 9 References	321
Chapter 10 Appendices	347
10.1 Appendix I – External NHS Trusts included as patient identification cer (PICs) in phase 3 and phase 4	ntres 347
10.2 Appendix II – GP letter	348
10.3 Appendix III – Parent information sheets	350
10.3.1 Parents of children aged 13-15 years	350
10.3.2 Parents of YP aged 16-19 years	354
10.4 Appendix IV – Parent invitation letter (for children aged 13-15 years ar YP aged 16-19 years)	nd 358
10.5 Appendix V – Parent consent forms	360
10.5.1 Parents of children aged 13-15 years	360
10.5.2 Parents of YP aged 16-19 years	361
10.6 Appendix VI – Participant information sheets	362
10.6.1 Children aged 13-15 years	362
10.6.2 YP aged 16-19 years	363
10.7 Appendix VII – Invitation letters for YP aged 16-19 years	366
10.8 Appendix VIII – Participant consent forms	368
10.8.1 Children aged 13-15 years	368
10.8.2 YP aged 16-19 years	369
10.9 Appendix IX – Family Background Questionnaire	371
10.10 Appendix X – Systematic review search terms	375
10.11 Appendix XI – Feasibility questions and probes used in phase 2 (exp consultations)	ert 376
10.12 Appendix XII – Feasibility questions included in phase 3 questionnair booklets (presented at the end of each instrument)	e 378
10.13 Appendix XIII – Paediatric Quality of Life Inventory Version 4.0 – UK English, Child Report (ages 8-12)	379
10.14 Appendix XIV – Paediatric Quality of Life Inventory Version 4.0 – UK English, Teenager Report (ages 13-17)	381
10.15 Appendix XV – Questionnaire assessing participants' preferred meth administration (presented at the end of phase 4 questionnaire booklets)	od of 383
10.16 Appendix XVI – Child VQoL scaled score-to-measure table	384
10.17 Appendix XVII – YP VQoL scaled score-to-measure table	385
10.18 Appendix XVIII – Child FV scaled score-to-measure table	386
10.19 Appendix XIX – YP FV scaled score-to-measure table	387
10.20 Appendix XX – Peer-reviewed paper	388

# Index of Figures

Figure 1. PRISMA flowchart showing systematic review	88
Figure 2. Flowchart showing the origin of items included in the first draft VQoL	
instrument	169
Figure 3. Percent of YP choosing extreme response categories when answering iter	ms
in the VQoL instrument	181
Figure 4. Screeplot showing actual, average and percentile eigenvalues in the Child	-
VQoL instrument	196
Figure 5. Screeplot showing actual, average and percentile eigenvalues in the YP-	
VQoL instrument	197
Figure 6. Item-person targeting of the final 24-item Child-VQoL scale	202
Figure 7. Category probability curves for the 4 response categories in the forward-	
scored Child-VQoL instrument	203
Figure 8. Category probability curves for the 4 response categories in the reverse-	
scored Child-VQoL items	204
Figure 9. Item-person targeting of the final 26-item YP-VQoL scale	212
Figure 10. Category probability curves for the 4 response categories in the forward-	
scored YP-VQoL instrument	213
Figure 11. Category probability curves for the 4 response categories in the reverse-	
scored YP-VQoL instrument	214
Figure 12. DIF in overlapping VQoL items between children (aged 8-12 years) and V	ΥP
(aged 13-17 years)	220
Figure 13. Fit of a 4 <sup>th</sup> order FP trendline to the score-to-measure values in the Child	-
VQoL instrument	222
Figure 14. Fit of a 4 <sup>th</sup> order FP trendline to the score-to-measure values in the YP-	
VQoL instrument	223
Figure 15. Flowchart showing the origin of items included in the first draft FV	
instrument	231
Figure 16. Percent of YP choosing extreme response categories when answering ite	ems
in the FV instrument	241
Figure 17. Screeplot showing actual, average and percentile eigenvalues in the Chil	ld-
FV instrument	254
Figure 18. Screeplot showing actual, average and percentile eigenvalues in the YP-	FV
instrument	255
Figure 19. Item-person targeting of the final 29-item Child-FV scale	261
Figure 20. Category probability curves for the 29-item Child-FV instrument	262
Figure 21. Item-person targeting of the final 31-item YP-FV scale	269
Figure 22. Category probability curves for the 4 response categories in the 31-item `	YP-
FV instrument	270
Figure 23. DIF in overlapping FV items between children (aged 8-12 years) and YP	
(aged 13-17 years).	276
Figure 24. Fit of a 4 <sup>th</sup> order FP trendline to the score-to-measure values in the Child	-FV
instrument	278
Figure 25. Fit of a 4 <sup>th</sup> order polynomial FP trendline to the score-to-measure values	in
the YP-FV instrument	279

## Index of Tables

Table 1. Classification of severity of visual impairment	42
Table 2. Updated description of all-cause generic visual impairment and eve disorder	er-
specific PROMs	54
Table 3. Categories used to rank severity of visual impairment	82
Table 4. Domains of QoL, literature which supports the inclusion of each domain in	the
topic quide, and modifications which were made to the topic quide in response to ea	ach
naper identified	91
Table 5. Example VOol, and EV item developed in the foundation research <sup>5</sup> and	
modified version in the current development	102
Table 6. Example of item adaptation for Parent booklets	102
Table 7. Missing data values assigned to VOol, responses in phase 4 data analysis	103
	113
Table 8 Missing data values assigned to EV responses in phase 4 data analysis	113
Table 0. Nonserraphic and divides assigned to PV responses in phase 4 data analysis	115
nationated compared to these who did not participate	116
Table 10. The exefficients and significance of variables (demographic and slipical	140
Table TO. The coefficients and significance of variables (demographic and clinical	1 1 0
Table 44. Demographic and clinical characteristics of VD who participated in each	149
Table 11. Demographic and clinical characteristics of YP who participated in each	4 - 4
Table 42. Demographic characteristics of children who participated in phase 4	151
Table 12. Demographic characteristics of children who participated in phase 4	155
Table 13. Areas of overlap and discrepancy between VQoL and FV reported by 16-	19
year old YP and 10-15 year old children	158
Table 14. Items included in the first draft VQoL instrument	170
Table 15. Minor linguistic modifications made to VQoL items during the expert	
consensus meeting	173
Table 16. Items which were modified as a result of phase 2: pre-testing	176
Table 17. Items included in the Child (8-12 years) and YP (13-17 years) version of t	he
VQoL instruments entered into phase 4	183
Table 18. Missing data per item in the VQoL instrument, separated by age-group	188
Table 19. Z-score skewness, Z-score kurtosis, floor and ceiling effects for each item	ו in
the VQoL instrument	190
Table 20. Items excluded as a result of response distribution	191
Table 21. Outcome from FA conducted to assess dimensionality in the VQoL	
instruments	193
Table 22. Average of the first five eigenvalues produced by parallel analysis for eac	h
VQoL dataset	194
Table 23. Problematic items in the Child-VQoL instrument due to fit to the Rasch	
measurement scale	198
Table 24. Fit and DIF statistics of items in the final 24-item Child-VQoL scale	200
Table 25. Problematic items in the YP-VQoL instrument due to fit to the Rasch	
measurement scale	206
Table 26. Fit and DIF statistics of items in the final 26-item YP-VQoL scale	210
Table 27. Items included in the final 24-item Child- and 26-item YP-VQoL instrument	nts
and labels used for calibration	216
Table 28. Table showing correlation coefficients between VQoL logit scores, PedsQ	۱۲
summary and sub-scale scores for children and YP aged 8-17 years	225

Table 29. Table showing correlation coefficients between VQoL logit scores and VA for children and XP aged 8-17 years
Table 30 Figenvalues for the largest factor, percent of variance explained, variation
between eigenvalues for datasets with imputed missing data, number of eigenvalues
and items not loading onto factors in the final $EA$ 226
Table 31 Areas of overlap and discrepancy between EV reported by 16-19 year old VP
and 10-15 year old children 228
Table 32 Items included in the first draft EV instrument 232
Table 33. Minor linguistic modifications made to FV items during the expert consensus
meeting
Table 34. Items which were modified as a result of phase 2: pre-testing
Table 35. Items included in the Child (8-12 years) and YP (13-17 years) version of the
FV instrument entered into phase 4
Table 36. Missing data per item in the FV instrument, separated by age-group 248
Table 37. Z-score skewness, z-score kurtosis, floor and ceiling effects for each item in
the child- and YP-FV instruments250
Table 38. Outcome from FA conducted to assess dimensionality in the FV
instruments
Table 39. Average of the first five eigenvalues produced by parallel analysis for each
dataset
Table 40. Problematic items in the Child-FV instrument due to fit to the Rasch
measurement scale
Table 41. Fit and DIF statistics of items in the final 29-item Child-FV scale
Table 42. Problematic items in the YP-FV instrument due to fit to the Rasch
measurement scale
Table 43. Fit and DIF statistics of items in the final 31-item YP-FV scale
Table 44. Items included in the final 29-item Child- and 31-Item YP-FV instruments and
labels used for calibration of the two instruments
Table 45. Correlation coefficients comparing FV logit scores and VA for children and
YP aged 8-17 years
i able 46. Eigenvalues for the largest factor, percent of variance explained, variation
between eigenvalues for datasets with imputed missing data, number of eigenvalues
and items not loading onto factors

### Abbreviations

- BL Blind
- CTT Classical test theory
- DDA Disability Discrimination Act
- DIF Differential item functioning
- FA Factor analysis
- FP Fractional polynomial
- FV Functional vision
- **GP** General Practitioner
- HRQoL Health-related quality of life
- ICF International Classification of Functioning, Disease, and Health
- IMD Index of multiple deprivation
- IQR Interquartile range
- IRT Item response theory
- LogMAR Logarithm of the minimum angle of resolution
- LV Low vision
- MNSQ Mean-square
- NHS National Health Service
- PCA Principal component analysis
- PIC Patient identification centre
- PID Participant identifier
- PREM Patient-reported experience measure
- PROM Patient-reported outcome measure
- QoL Quality of life
- RCT Randomised control trial
- RMSE Root-mean-square standard error
- RMT Rasch model theory
- SE Rasch model-based standard error of the measure, not adjusted for misfit
- SES Socio-economic status
- SVI Severe visual impairment
- ToM Theory of mind
- UCL University College London
- UK United Kingdom

- VA Visual acuity
- VI Visual impairment
- VQoL Vision-related quality of life
- WHO World Health Organisation
- YP Young people
- ZSTD Standardised z-score

#### Chapter 1 Introduction

Childhood and adolescence are critical periods of life spanning birth to the age of 19 years<sup>1</sup> and characterised by learning, development, and maturation. When a child is visually impaired, childhood is likely to be disrupted in many ways. Within clinical practice and research concerning visually impaired children, there is growing interest in understanding the broader impact of visual impairment in the hope of improving vision-related quality of life (VQoL). This interest is reflected in the recent emergence of patient-reported outcome measures (PROMs) which are designed to capture self-reported and assessed burden. Despite a growth in the number of vision-specific PROMs designed for adults,<sup>2-4</sup> there are few robust vision-specific PROMs which have been designed, and are suitable, for children and young people (YP) with visual impairment. In 2011, Rahi and colleagues<sup>5-7</sup> developed a conceptual and methodological framework for the development of vision-specific PROMs suitable for children and YP, resulting in development of two complementary, age-appropriate, self-report measures: one assessing VQoL and the other functional vision (FV) in children aged 10-15 years.

The framework established by Rahi et al.<sup>5-7</sup> provides a methodological and theoretical starting point in which to develop age-appropriate, vision-specific PROMs capturing VQoL and FV which are suitable for use by a wider age range of children and YP with visual impairment. This thesis reports the extension and adaptation of these PROMs for use by children and YP aged 8-17 years, with particular focus on the development of PROMs designed for YP aged 15 years and older living with visual impairment. Adopting a life course perspective to capture VQoL and FV among YP living with visual impairment affords the opportunity to explore, in depth, YP's experiences of visual impairment, the everyday impact of visual impairment during adolescence, and the experience of growing up with visual impairment, as reported in this thesis.

This thesis begins with a discussion of relevant literature identifying the dynamic impact of visual impairment during childhood, the role of PROMs within generic paediatric healthcare and, later, within paediatric ophthalmology services. A brief introduction to psychometric measurement within healthcare, classical test theory (CTT) and modern test theory as popular methodologies follows (Chapter 3). The 4-phase method used in the current project is detailed in Chapter 4. Both Chapters 5 and 6 report the results from the analysis; Chapter 5 details the qualitative findings from the first phase of instrument development and Chapter 6 presents the results from the psychometric analysis of PROMs, including the outcome of unidimensionality assessments, and Rasch analysis. The findings (both qualitative and psychometric) are discussed in Chapter 7, and finally in relation to future work needed to assess the acceptability and feasibility of implementing the developed PROMs within routine paediatric ophthalmology services (Chapter 8).

#### Chapter 2 Background

#### 2.1 Childhood and adolescence

Childhood is a transient period of early life course defined as the period between birth and adolescence.<sup>1</sup> Growth and development during childhood takes place at a rate that exceeds that of any other period of life course, as important neurological changes occur<sup>8,9</sup> in parallel and association with biological and behavioural changes such as motor<sup>10, 11</sup> and muscular development.<sup>12</sup> Despite aspects of growth and development which are universal and occur within all humans regardless of gender and ethnicity, definitions of childhood are largely reliant on cultural perspective and constructed in a wider social world in which assumptions are made and shared.<sup>13</sup> For example, among Western economically developed societies, childhood is accepted as a time for leisure and learning, which is free from major responsibility and commitment.<sup>14</sup> However, Robson<sup>15</sup> argues that the notion of children as dependent in this context fails to acknowledge the work children do in contributing to household survival and well-being. Indeed, children in developing communities play an important role in contributing to household production, for example by performing labour-intensive agricultural tasks. Within these cultures, childhood is conceptualised as a brief period of life characterised by biological and physical growth before the onset of formal employment and financial responsibility.

Throughout history, scientists have tried to make sense of growth during childhood by developing theories describing and explaining change. Most early theories were centred upon competing forces of nature and nurture. Greek philosophers such as Plato and Aristotle, for example, believed that parents should nurture their children carefully to avoid the negative force of human nature from dominating, and causing children to become rebellious and unruly.<sup>16, 17</sup> Today, perspectives span many aspects of development, and are often focused upon providing details of specific aspects of development which take place in a broader social and cultural context. In parallel to

physical growth, children are expected to develop cognitive capacities allowing them to think critically about the world that they live in, regulate and control their behaviour, and the behaviour of others; language skills for comprehension and speech production; and social skills through which they can establish and maintain meaningful relationships with others. There are several dominant theories within the field of childhood development, each emphasising factors such as biological and sexual drives.<sup>18</sup> rewards and punishments<sup>19, 20</sup> and, socialization.<sup>21</sup> Many of these perspectives describe stages of development, viewing childhood as multi-phasic, and thus propose that children of a given age show broad similarities across many situations. Piaget's theory of cognitive development<sup>22, 23</sup> remains particularly influential and has many reallife applications in modern day society, for example, within the field of education. Despite evolution of developmental theories over time, competing forces of nature and nurture remain theoretically influential, viewing children as evolving bio-psychological human organisms in the context of external environments - both immediate and remote. Developmental milestones during childhood are heavily influenced by the culture in which an individual grows up, and the wider social world in which they are nurtured.

At its simplest, adolescence is a term used to describe the *pinnacle* of childhood; a time when most developmental milestones have been achieved, but adulthood has not yet begun.<sup>24</sup> Similar to definitions of childhood, adolescence is a period which is socially constructed and the roles or duties of adolescents within society are largely dependent on cultural perspective. The timing of onset of adolescence is variable and difficult to define in precise terms; based upon multiple factors such as timing of physical and biological growth, growth of abstract thinking,<sup>25-27</sup> increased autonomy and independence,<sup>28, 29</sup> and growth of understanding about the community and broader world in which one lives.<sup>30, 31</sup>

One of the most transparent ways to identify the timing of adolescence is through reference to puberty. Regardless of the culture in which they live, all individuals face a

significant degree of biological change during adolescence as a result of puberty. These changes are essential for reproduction in the future. The timing of onset and duration of puberty is variable, and largely reliant on genetic factors, body mass, nutritional state, and general health.<sup>32, 33</sup> Puberty also occurs at significantly different points for girls and boys.<sup>34-36</sup> Despite occurring at different times, the experience of puberty is new and unusual for both males and females, triggering events such as the onset of pubic hair growth and growth spurts. It is not a single event but a process which takes time and, like the majority of theories of childhood and adolescent development, can be broken down into multiple phases.

Another factor complicating identification of the timing of onset of adolescence is cultural variation due to differences between national laws which set the minimum age thresholds for participating in activities traditionally considered to be reserved for adults. These activities include voting, marriage, property ownership and alcohol consumption. In most Western, developed societies, the age at which individuals gain the right to these activities is 18 years but in other countries this threshold varies. For example, girls in Iran are recognised as adults when they are just 9 years old.<sup>36</sup>

Definitions of adolescence are also influenced by the historical context in which they are developed and the dominant features of society at a specific time point. Early or emerging adulthood has recently been discussed as an extension of adolescence in modern day society as individuals increasingly struggle to achieve and maintain economic independence and security, and thus, delay the onset of adulthood. Greater frequency of college enrolment, time spent exploring future career choices, and later age at marriage and childbearing are currently trending.<sup>37, 38</sup> This is in stark contrast to definitions of adolescence which were used during the 17<sup>th</sup> and 18<sup>th</sup> centuries as children became skilled in specialist trades during childhood, and went straight into adulthood.<sup>39</sup> In modern day society, YP aged 18-25 increasingly pursue multiple paths and relations before pursuing serious, long-term stable relationships.<sup>40</sup> The term

'millennial' is increasingly used to describe the role of YP in modern day Western society.<sup>37</sup>

Whilst there is profound variations in the timing of adolescence, the United Nations define adolescents as individuals aged 10-19 years old.<sup>36</sup> For clarity this wide definition of adolescence, which is based on contemporary Western perspectives, is endorsed throughout the remainder of this thesis.

#### 2.1.1 Adolescence as a key transitional life stage

Although puberty and development of reproductive function is often considered a physiological accomplishment occurring at the level of sexual organs, specialised neurons in the brain and hormones secreted by the pituitary gland play a critical role in governing reproductive function.<sup>41, 42</sup> A complex interplay between neurotransmitters such as norepinephrine, dopamine, serotonin, glutamate, neuropeptide Y, galanin, GABA, and corticotropin-releasing hormone during adolescence is required to trigger physical reproductive growth. Thus, hormonal changes are robust during adolescence and may be as great as eighteen-fold increase in males and eight-fold in females.<sup>43</sup> From a physiological perspective, hormones are thought to act on the brain to affect behaviour through their impact upon peripheral and neural-based processes.<sup>44</sup> These physiological interactions are likely to play a role in subjective, self-reported well-being, and are linked to emotional and behavioural characteristics of adolescence such as mood swings, changes in energy levels, and restlessness.<sup>45-47</sup> Some evidence suggests associations between hormonal changes and increased family conflict,<sup>48</sup> and risky behaviour<sup>49</sup> particularly when it results in high rewards.<sup>50, 51</sup> Impulsive, irrational behaviour gradually decreases in parallel with increased grey matter in the occipital lobe.8, 52

Adolescence can also be described as a period succeeding the stages of growth outlined by theories of childhood development. For example, from the perspective of cognitive development, at the age of 12, children are thought to enter the stage of

formal operations, in which they are able to think abstractly and hypothetically, taking into account the influence of multiple variables.<sup>26</sup> From a different perspective, children the same age are thought to develop understanding of other people's perspectives, by comparing themselves and others to a 'generalized other', assessing whether the view of one person is shared between others.<sup>21</sup> Completion of the stages outlined by theories of development has likely led to the definition of adolescence as a 'summit' or final stage of development: a prerequisite for adulthood.

In addition to change during adolescence and progression through the stages outlined by theories of development, an adolescent living in the developed Western world will likely experience a number of formal, policy driven transitions. These primarily take place within education. Children who are raised in the UK and enter mainstream education will experience at least two formal transitions during their childhood and early adolescence: one when they enter primary education at the start of term following their 5<sup>th</sup> birthday<sup>53</sup> and another when they enter secondary education at the age of 11 years.<sup>54</sup> Those who wish to extend their education beyond the statutory minimum requirement of completing secondary education at the age of 16 years<sup>53</sup> may experience numerous further transitions as they enter sixth form, college, or University. With each transition in education, there will be substantial change in environment (both physical and social).

It is also likely that individuals will experience a number of informal transitions during adolescence which, in comparison to formal transitions in education, are a result of changes driven primarily by growth, development and psychological maturation, and may occur at flexible time points which are driven by individual differences. Specifically, transitions in social interactions and the nature of friendships have been highlighted as characteristic of adolescence. For example, during adolescence, friendships with peers and family members have been shown to aid preparation for future independence, socialisation and intimate or romantic interactions,<sup>55-57</sup> and be important to the development of autonomy and a sense of identity as an adult.<sup>58</sup>

One major feature of adolescence is the experience of challenges, difficulties and distress. This may be the consequence of a combination of the biological, physiological, and policy-driven changes which occur during this period. Aristotle stated that during adolescence, individuals 'are heated by Nature as drunken men by wine'. Similarly, Socrates characterised adolescents as inclined to 'contradict their parents' and 'tyrannize their teachers'. Hall (1904) was the first to use the term 'storm and stress' to characterise adolescence<sup>59</sup> and subsequently paved the way for scientific research exploring the experience and cause of the apparent developmental 'crisis' which occurs during adolescence. Stress during adolescence was traditionally conceptualised in terms of conflict with parents, mood disruptions, and increased risky behaviour.<sup>59</sup> A greater number of negative life events related to peers, school and family are encountered during this period and co-occurrence of developmental events make it hard to pinpoint the cause of behavioural change. Negative emotions occur when there is discrepancy between what is expected and what really happens<sup>60, 61</sup> and may be a result of changes in perceived social role during adolescence, challenging ones view of themselves and their value in the lived world.<sup>62</sup> Despite use of terminology depicting adolescence as turbulent, it is increasingly recognised that not all adolescents face such challenges and stresses.<sup>63</sup> From this perspective, storm and stress is not something written into the human life course, but attributable to individual differences in the experience of adolescence. Research documenting the age-related challenges of adolescence in relation to visual impairment is discussed in Section 2.5.2 (pg. 46).

In summary, childhood and adolescence constitutes a critical period for profound growth and development. Adolescence is a period of life course occupied by growth which is unique from that which takes place during childhood. In addition to biological, physiological, and neurological development, adolescents are faced with a number of transitions or changes which inevitably make life more difficult. The social construction

of modern-day Western society means that YP may face a number of economic and financial challenges in addition to social, emotional, and psychological disruption.

#### 2.2 Childhood disability

When a child grows up with a disability or illness, childhood and adolescence is likely to be qualitatively different from that of healthy and non-disabled children. The Disability Discrimination Act (DDA)<sup>64</sup> defines disabled children as those with a 'physical or mental impairment which has substantial and long-term adverse effects on their ability to carry out normal day-to-day activities'. In this definition disability is described as a broad category, stemming from no particular origin. In contrast, other definitions of childhood disability take checklist-type approaches: using umbrella terms such as learning difficulties, traumatic brain injury and autism, and those who need special education and related services.<sup>65</sup> Thus the population of children with disabilities is both substantial and complex in terms of their health-related needs, and may be difficult to characterise.

The social model of disability regards disability as a social construct, existing only as a result of the broader society's failure to accommodate differences brought by disability.<sup>66, 67</sup> This provides stark contrast to early research which tended to ignore the disabled child, and focus on the views and perceptions of those living with the child (e.g. their parents and families). Whilst portraying disabled children as vulnerable and inferior, this type of research focused upon the negative impact of disabled children upon their families and broader society, incurring economic difficulties and strained family relationships.<sup>66</sup> The voices of disabled children themselves were frequently ignored.<sup>66, 69, 70</sup> More recently, and with help from the social model of disabled children as valuable in their own right. It is increasingly accepted that disabled children are treated the same as their non-disabled counterparts, and that any adjustments or adaptations they require are put in place to allow for participation. This attitude is legally enforced

by definitions of childhood disability, such as the DDA's definition which specifies individuals' ability to carry out 'normal day-to-day activities'.

A second model demonstrating potential to change the conceptualisation of disability is the International Classification of Functioning, Disease and Health (ICF).<sup>71</sup> This model lists a number of environmental factors which account for the broader social context in which disability occurs. Instead of labelling individuals as 'disabled', this model views disability as a universal attribute, as every human being will experience a decrement in health at some point during their life course.<sup>72</sup> The ICF framework is based on interplay between two models of disability which are thought to be only partially valid: the medical model and the social model. Neither model is thought to be adequate in itself, based on narrow conceptualisations of disability confined to physical health (the medical model) and social construction (the social model). As a result, the ICF distinguishes between three levels of human functioning; bodily functions & structure, activity and participation, and environmental and personal factors, providing a mid-point between medical and social perspectives.

From social perspectives, it is not merely the presence of disease or impairment which renders a child 'disabled' but rather the interaction between physical afflictions and day-to-day culture which is socially constructed. Disability is something which is imposed, whereas impairment is a private, internalised attribute.<sup>73</sup> Some evidence shows how social frameworks have potential for improving disabled children's participation in everyday life, for example, by increased awareness, and removal, of barriers towards participation in education.<sup>74</sup> Thus, a new landscape in which to conceptualise disability is beginning to show potential for improving disabled children's participation in everyday life. Whilst still in development and undergoing modification, social models of disability are the first to allow the 'voices' of disabled children to be heard and thus led to recognition of the broader impact of childhood disability.

#### 2.3 Measuring the impact of disability or illness in healthcare

Information has always been fundamental for healthcare services to ensure good outcomes for patients, and provide high quality care. Traditionally, the impact of disease and illness is measured using assessments and observations of physical function. These include impairment-based measures such as clinical examinations and functional capacity evaluations. Using these assessments, health professionals can gain insight as to the immediate impact of disease or illness on physical health and functional ability, and make inferences about the impact at an individual level and upon a patient's daily life. For example, a patient who reports shoulder pain may likely experience some form of difficulty when brushing their hair or taking a shower. Functional assessments of health also play a critical role in the formal diagnosis, and treatment of disease.

It is only in the last 10 years and with advances in information technology that measurement in health care has exploded in terms of quantity, quality and accessibility. It is now easier than ever before for health professionals to collect, analyse, and report data relating to multiple outcomes. Alongside this growth, it is increasingly apparent that functional assessments of health are not sufficient to capture the impact of disability which, by definition, considers restricted functional ability in 'the manner considered normal for a person'.<sup>75</sup> This is in line with social models of disability such as the social model and ICF which describe individuals living with disability as active, valuable members of society as opposed to oppressed and inferior.

In 1948, the World Health Organisation (WHO) defined health as 'a state of complete physical, social, and mental well-being, and not merely the absence of disease or infirmity'.<sup>76(p1)</sup> At the time, this definition was ground-breaking in terms of its breadth and ambition, overcoming restrictions to disease and including broader domains of physical, mental and social well-being. Subsequently, and likely as a result of the broad definition of health, Quality of Life (QoL) emerged as a multi-dimensional construct which can be used to measure the broader impact of disability or illness. Definitions of

QoL have varied over time and are difficult to establish due to the notion that each individual has their own unique QoL which depends on their experiences, lifestyles, hopes, attitudes and beliefs. It is not surprising that definitions of QoL also vary between contexts and perspectives.<sup>77, 78</sup> The range of factors included under the umbrella term 'QoL' can be placed upon a continuum, ranging from purely subjective, such as an individual's experience or appraisal of their daily lifestyle, to objective appraisals of physical function.<sup>79</sup> The 1995 WHO Quality of Life Group defined QoL as:

*'an individual's perception of their position in life in the context of the culture and value system in which they live and in relation to their goals, standards, and concerns.'* <sup>80(p1)</sup>

This definition is perhaps the most encompassing and accepted definition within existing literature and has therefore been adopted for the purpose of this thesis.

Interest in measuring QoL within healthcare has emerged over time and in response to recognition of the need to provide patients with treatment that at least makes them feel better when no cure is available. A substantial level of research highlighting the broader impact of disability, disease and illness upon psychosocial issues such as psychological well-being, and social relationships has supported the WHO definition of health and been influential in that there have been notable changes in clinical practice. For example, instead of focusing upon treating, and providing care for the purely physical impact of disease, specialist nurses have been trained in order to provide social support,<sup>81</sup> and psychological therapies such as cognitive behavioural therapy have been designed to enhance adaptation in those diagnosed with terminal illnesses.<sup>82</sup> Psychosocial support for individuals facing long-term disability and/or illness is increasingly emphasised by a number of health-related charities, such as Macmillan Cancer Support,<sup>83</sup> and Samaritans.<sup>84</sup>

When measured among patient populations, a common assumption is that good QoL is associated with good health<sup>85</sup> and if an individual is unable to perform a normative task, QoL is compromised.<sup>86</sup> However, this is discordant with empirical evidence which suggests that, in practice, perceived health, well-being and life satisfaction are often

different from objective health status and disability.<sup>87</sup> In response to this discordance, the 'disability paradox' was developed to explain variance between health, disability and quality of life.<sup>88</sup> Instead of being dependent upon interactions between subjective and objective health outcomes, this theory emphasises a balance between the body, mind and spirit as essential for high levels of QoL among all individuals regardless of disability. Thus capturing QoL within healthcare is conceptually difficult as it is underpinned by a number of subjective evaluations, each involving some degree of individual difference. This difficulty has led to the development of a number of conceptual models and measures of QoL but with very little uniform consensus on what is included and what should be measured.<sup>89,90</sup> In response to this ambiguity, it has been suggested that three different aspects of health should be assessed within routine healthcare: objective measures such as clinical indices which patients would not necessarily use or be aware of (such as blood pressure), functional performance (the ability to perform certain activities which are part of daily lifestyles), and patients' own evaluation of the experience of living with illness or disability.<sup>77</sup> In relation to the latter, health-related quality of life (HRQoL) has emerged as an adjunct to measure QoL in health and has evolved from a loosely integrated body of research on health status. functional performance and social well-being. Despite ongoing evolution and development, to date, experts have disagreed on the specific definition of HRQoL. For example, in a systematic review, a large number of HRQoL models were found. however more than one-quarter of authors did not define HRQoL using the main concepts which are used by others, and some did not define HRQoL at all.<sup>91</sup> Similarly, the extent to which HRQoL is distinct from QoL has been debated, as some researchers have concluded that aspects of health and functioning are only some of the life domains that may impact QoL.<sup>92</sup>

Despite disagreement which has, to date, impeded the development of a universal definition of HRQoL, there is some agreement that HRQoL includes the same factors included in QoL definitions: subjective factors such as health perception, and emotional

state, and objective factors such as physical function and symptoms, but the factors are framed by the experience of disease, illness or disability in the broader social and cultural context.<sup>93</sup> Insertion of HRQoL into the ICF model of disability has been recommended in the view that disability is not a fixed state, but is fluid and changeable and therefore a general feature of the human experience.<sup>94</sup>

# 2.3.1 Patient-reported outcome measures (PROMs) and patientreported experience measures (PREMs)

As life expectancy increases worldwide,<sup>95</sup> more individuals are surviving into old age with chronic and co-morbid conditions. As a result, policies within healthcare increasingly prioritise patients' *quality*, as opposed to *quantity*, of life. Concurrently, decision-making within healthcare is increasingly based upon patient satisfaction and preference. In 2008, Lord Darzi<sup>96</sup> stated three components to quality of care: patient safety, patient experience, and effectiveness of care. Patient-reported outcome measures (PROMs) can be used to measure the final component: effectiveness of care, by providing a platform for patients to report on the subjective impact of clinical care/treatment/interventions. As a consequence of this report, PROMs have played a mandatory role in healthcare in the UK since 2009. Hospitals providing four specific surgical procedures; hip or knee replacement, varicose vein surgery and groin hernia repair have been legally required to collect data from patients using PROMs before and after surgery to reflect the impact upon patients' subjective well-being.<sup>97-100</sup>

The second component of Lord Darzi's statement, that is, patient experience, can be measured using patient-reported experience measures (PREMs), which are distinct from PROMs because of their focus on the process of, as opposed to the *impact* of healthcare. Lord Darzi's report stated that payments to hospitals should be based on the quality of care they provide as opposed to the quantity of patients who are treated. Thus, PREMs are important to improving quality in healthcare: a requirement which was put forward by the NHS Operating Framework in 2012.<sup>101</sup>

Patient-reported outcome measures (PROMs) come in the format of questionnaire instruments which are used within clinical practice and research and allow individuals with disability or illness to self-report on the everyday impact of their condition, including their health status, preferences, condition-specific symptoms, pain intensity, and HRQoL. Self-report on the impact of disability or illness is the defining feature of PROMs and using PROMs, patients can communicate this impact to others, primarily clinical professionals.

The majority of PROMs incorporate some form of HRQoL as the primary outcome and response formats such as Likert-type scales which require the user to indicate, using numbers, images or diagrams their feelings, attitudes and beliefs about their health. Each category or image is assigned a value and summary scores can be calculated to represent a subjective evaluation of disease impact. Thus, PROMs incorporating assessment of HRQoL can be used to quantify the complex and subjective burden of living with a disease, illness or disability by reducing patients' self-report experience into a single, or series, of numeric values (see Chapter 3, pg. 68 for more background information on measurement in healthcare). There are a number of benefits of using PROMs in this way to quantify subjective outcomes, namely that outcomes can be captured quickly and with minimal support from healthcare professionals. Outcomes can also be compared between patients living with different degrees, or forms of disability. In this way, PROMs aid decision making based on prioritisation of interventions, treatment, or medication. In some cases, users are required to reflect on their subjective state within a specific time frame e.g. over the last week/month, giving healthcare professionals insight as to the ongoing, longitudinal nature of disease impact. PROMs are also inexpensive, and flexible instruments as they can be administered both manually using pen and paper and electronically, using computers or tablets, with patients completing measures in their own time outside of clinical settings.

PROMs which are currently available for use in healthcare can be divided into genericand disease-specific measures. Generic PROMs are used to compare outcomes across different health conditions and interventions,<sup>102</sup> and are particularly beneficial when assessing cost-effectiveness within healthcare.<sup>103, 104</sup> However, the development of these measures has undoubtedly been affected by ambiguity surrounding the definition of HRQoL which is discussed in Section 2.3 (pg. 28). For example, health status has frequently been placed under the umbrella term of HRQoL<sup>105</sup> when it has previously been defined as reflecting an individual's relative level of wellness and illness which does not take into account individuals' perspectives of disease or treatment; a domain covered by the majority of HRQoL definitions.<sup>106, 107</sup> When a health status measure is used to quantify HRQoL, misleading conclusions can be made which may have serious implications such as misinterpretation of the value of an intervention upon HRQoL.

Alternatively, condition-specific PROMs have been developed within specific fields of health such as cancer,<sup>108-110</sup> heart disease,<sup>111-113</sup> and liver disease.<sup>114, 115</sup> These measures can be used to detect the condition-specific impact of a disease or disability which may not necessarily be covered by generic measures and thus discriminate more sensitively between patients with the same condition. Disease-specific measures can be particularly valuable when comparing the subjective well-being of patients living with varying manifestations of the same disease or when assessing the burden of disabilities with unique physical symptoms.

To date, PROMs have primarily been used in clinical trials,<sup>116-118</sup> national audits,<sup>119</sup> and clinical registers<sup>120</sup> but are increasingly widespread – at the level of both primary and secondary healthcare.<sup>121, 122</sup> In a systematic review of evidence supporting the use of PROMs, improvements were demonstrated in processes which took place in primary care (such as advice, education and counselling, and diagnoses given by consultants).<sup>123</sup> PROMs have also been shown to increase patient-provider communication about aspects of HRQoL,<sup>124</sup> enhance consultants' ability to detect

psychological and functional problems,<sup>125</sup> increase emotional support offered by consultants<sup>124</sup> and enhance 'patient centricity' within clinical care: an asset which is highly valued by patients.<sup>103</sup> However, the direct impact of PROMs upon health and well-being has been less convincingly demonstrated, with the majority of randomised control trials suffering from methodological limitations, such as use of multiple instruments and lack of consensus on the relevance of a single PROM.<sup>123</sup>

Although evidence supports that validity and reliability of PROMs are largely comparable to routinely used clinical measures, there are a number of barriers towards using PROMs routinely. These include, for example, the need for clinicians to be able to receive and interpret data promptly and communicate findings in a way which is understandable by patients.<sup>126, 127</sup> Clinicians may also be sceptical about using PROMs in the view that information gathered can force discussions into areas that consultants have little control over.<sup>123</sup> To date, this type of literature has been largely confined to adult populations, and little is known about the benefits of and barriers towards using PROMs within paediatric (and adult) ophthalmology services.

#### 2.3.2 Development and application of PROMs for paediatric settings

To date, a number of PROMs have been developed for use by children living with a specific form of disability or illness.<sup>128-133</sup> In addition to containing items designed to capture the condition-specific symptoms and impact, PROMs which are designed for use by children need to include items which capture factors that contribute to QoL during childhood as these will likely differ from those attributable to QoL during adulthood. Definitions of QoL emphasise experiences, lifestyles, attitudes and beliefs contributing to overall well-being or happiness. It is therefore incorrect to assume a child, who has only a fraction of the experience, and a very different lifestyle from an adult, can use an adult-centred PROM to report on the impact of disease or disability.<sup>134</sup> Nevertheless, in the past, when child-specific PROMs have been unavailable, HRQoL measures for adults have been used in paediatric populations.<sup>135</sup> This approach is largely problematic because instruments not only contain items which

do not sensitively reflect children's priorities, beliefs, and HRQoL, but they are also framed in a way which is difficult for children to understand and interpret<sup>79</sup> (see Section 2.1, pg. 20 for a brief discussion of cognitive and intellectual development during childhood).

In the past, when age-appropriate PROMs have been unavailable for children, and PROMs developed for adults have been deemed inappropriate, measures of HRQoL have been completed by adults who have a substantial amount of everyday contact with the child, with the assumption that such 'proxy' reports provide reliable and valid inferences as to the HRQoL of the child. The most obvious proxy, or informant, used within clinical practice is a parent or guardian given the amount of contact they have with the child, and subsequent insight as to the child's subjective appraisal of HRQoL in multiple contexts e.g. at different times of the day, in different locations, and when playing or engaging with other people. However, any adult who has everyday contact with a child and/or sufficient knowledge of the potential impact of disease or illness can provide a proxy report. In some cases, proxies can be teachers and clinicians.<sup>136</sup> The value of proxy reports provided by adults with varying relationships to the child has been discussed in extant literature, as although mothers may be more involved with their child's healthcare, teachers and clinicians may be more impartial, and have views based upon comparisons of children with varying degrees of disability.<sup>136</sup>

Proxy reports can be particularly useful when a child is unable to self-report due to disability or age, or may be simply unwilling. For example, the youngest age at which children have been shown to reliably and validly self-report on complex outcomes such as HRQoL is 5 years<sup>137</sup> but this may be higher, depending on the nature of impact of disease or illness upon cognitive capacities, such as reading or interpreting items. Moreover, the true reliability and validity of outcomes provided by children as young as 5 years is questionable given that higher reliability is associated with increasing age.<sup>137</sup>

Importantly, when results from self- and proxy-reports are compared, outcomes are likely to be quantitatively and qualitatively different. Proxy reports represent an adult's

perception of the child's experience and are influenced by demographic characteristics such as the age, sex and socio-economic status of the respondent as well as their relationship to the child, and any emotional impact of disease or illness upon their own subjective well-being.<sup>138</sup> Empirical evidence demonstrates parents often underestimate their child's HRQoL<sup>139</sup> and are better at rating externalizing problems such as physical symptoms and somatic distress<sup>140, 141</sup> compared to internalizing problems such as social or emotional issues.<sup>142-144</sup> Thus, when both self- and proxy-reports are available, they must be used as complementary outcomes which provide unique perspectives. Outcomes should never be aggregated and one cannot rely solely on proxy-reports in the absence of PROMs which are suitable for children and sensitive to the impact of disability during childhood.<sup>145</sup> As a last resort, the International Society for Pharmacoeconomics and Outcomes Research recommend that, when an age-appropriate self-report measure is truly unavailable, a proxy-report measure based upon observable content such as behaviour may be used to give insight as to HRQoL.<sup>146</sup>

Identification of the need for child-specific PROMs is implicitly recommended by contemporary perspectives on disability during childhood. Instead of treating the purely physical, functional impact of disability, age-appropriate PROMs can be used to give children and YP a 'voice' and active role within healthcare e.g. in decision making regarding the management of their disability or disease.

### 2.4 PROM content, design and development

Methods used to develop PROMs and questionnaires in general need to be high quality, controlled, and rigorous if users of these instruments are to be certain that outcomes truly reflect the measurement constructs and that they are reliable. If questionnaires are poorly designed, data will be misinterpreted, and irrelevant and/or incorrect conclusions will be made.
Questionnaire development usually involves a series of standard stages in which ideas about content and format are formulated, and tested and analysed. There are many factors to consider when creating a questionnaire, including question type, wording and order, response options, and method of administration. The early stages of questionnaire design are usually exploratory and qualitative in nature as researchers need to gain information about the measurement outcome, and, in the case of PROM development, the construct of interest, the patient population and their perception of the impact of disability or disease. As discussed in Section 2.3 (pg. 28), PROMs can comprise a number of varying features. Factors such as patients' physical and psychological capability to complete PROMs independently and with ease need to be considered during the development.

The most common way to conceptualise the patient population, their self-reporting needs and the impact of disease or disability is to conduct lengthy, in-depth and often unstructured interviews with a sample of representative patients.<sup>147</sup> Interviews should give a rough shape of the enquiry and inform the content of the new PROM.<sup>148, 149</sup>

Once a questionnaire has been drafted, a quantitative approach is usually taken to ensure that the measure is reliable, valid and psychometrically robust, that is, capable of capturing the intended construct accurately and consistently over time. In 2011, the consensus-based standards for the selection of health measurement instruments (COSMIN) committee developed a checklist which can be used to calculate the quality of PROMs,<sup>150, 151</sup> thereby advocating a systematic, rigorous approach to instrument development which is grounded in psychometric evaluation. Types of validity featured in the checklist include *content* (i.e. that the measure provides coverage of all aspects of the outcome), and *criterion* (i.e. how well the instrument can be used to predict an outcome for another measure) validity. *Face* validity (i.e. that the measure appears to be, at face value, a good measurement tool) and *construct* validity (i.e. the instruments ability to measure the intended outcome) are psychometric properties which can be used as evidence of whether an instrument measures what it purports to measure, and

internal consistency reliability (i.e. the certainty that all items/questions in a questionnaire that are intended to probe the same area will produce similar results) indicates homogeneity among items and the instruments capacity for measurement.<sup>152</sup>

Classical Test Theory (CTT) approaches such as factor analysis (FA) and factor rotation<sup>153</sup> or modern psychometric approaches such as Item Response Theory (IRT)<sup>154, 155</sup> and Rasch analysis<sup>156</sup> are usually applied in the later stages of questionnaire development<sup>157</sup> in order to test the psychometric properties of the questionnaire and ensure that the measure is accurate and precise. The underlying assumption of these approaches is that the outcome of interest is not directly observable (e.g. wellbeing), and therefore one must collect data using mechanisms through which the outcome may be assessed, such as self-report. These mechanisms give us conclusions which can be made about unobservable factors. Using the principles provided by these approaches, one can determine how successful the mechanisms are when estimating unobservable variables.<sup>158</sup> A detailed description of these psychometric approaches, their strengths, weaknesses and current consensus on which approach is optimal is presented in Chapter 3 (pg. 68).

# 2.4.1 Involving children in questionnaire development

Unlike research conducted in the 1970s which focuses on Piagetian models of development through experiments *with* children, today, researchers are beginning to draw on sociological theories such as theories of childhood and disability.<sup>14</sup> This is evidenced, for example, by direct participation by children in the traditional stages of questionnaire development. In contrast to being the *object* of study, children are increasingly viewed as competent and worthy of a voice in an adult-ist world.<sup>159</sup> Information stemming from children's opinions, attitudes and beliefs is increasingly valued. As a consequence, methodologies such as those used to develop questionnaires, are increasingly designed in a way that allows for children's active participation through, for example, in-depth interviews. Historically, researchers would

have turned to the parents, teachers or paediatricians to represent children within research, and the views of children themselves would have been ignored.

Involving children in the development of questionnaires is important to ensure the questionnaire has content validity (see Section 2.4, pg. 36) by containing questions or items which are meaningful and relevant for the child and, in the context of PROMs, shed light on the subjective experience of living with an illness or disability. It is particularly important for children who are living with illness or disability to take part in instrument development given that outcomes from proxy measures are often discordant from children's own self-report.<sup>139, 160</sup>

When conducting research, all researchers are required to protect their participants from harm. This is a legal requirement which is enforced by ethical guidelines and regulations outlined by the National Institutes of Health<sup>161</sup> and World Medical Association.<sup>162</sup> Whilst the standards outlined by these institutes are applicable to research involving all human subjects, regardless of age, when research involves children, researchers must follow an additional subset of guidelines which are developed with the aim of protecting children. For example, researchers are required to gather informed assent from both the child and at least one parent when a child is under the age of 16 years.<sup>161, 163</sup>

Researchers must also consider ethical standards and research guidelines in relation to the child's developmental capacity and ability to take part in the research. Involving children in interviews and focus groups, essential first steps in the development of questionnaires, may be challenging. Children might not understand what an interview or focus group is, or what is expected of them<sup>164</sup> and may conform to their perception of the interviewer's expectations, resulting in unusually high levels of agreeableness.<sup>164,</sup> <sup>165</sup> High levels of imagination during childhood may also cause researchers difficulty when interpreting children's true feelings or beliefs.<sup>164</sup> Children's voices may be interpreted by adult researchers in a way which conforms to the expectations or

understanding of adults, and researchers may add meaning to the perceptions and views of children.<sup>166</sup>

Particular ethical considerations arise when children are disabled, ill or in some way vulnerable. Gathering true self-reported burden of a disease or disability during childhood is likely to be difficult as the child may require some form of assistance from a parent or caregiver, such as reading questions aloud, writing down answers or physical manipulation of a questionnaire. As a consequence, children and YP may avoid disclosing sensitive or confidential information for fear that a parent or guardian may find out.

Institutional research ethics committees working alongside researchers have a duty to protect participants from potential harm, even if this means that modifications have to be made to the research design. This is particularly prominent when involving adolescents in research, given the developmental characteristics of adolescence. Ongoing changes in hormonal, neuroanatomical, and psychosocial well-being<sup>167</sup> make adolescents more susceptible to stress and negative emotions such as anger and worry.<sup>168, 169</sup> Distress could occur if the adolescent is triggered to remember a negative event, or asked about activities or behaviours which they are ashamed about or trying to keep secret. Discussions of sensitive issues such as the functional impact of disease or disability upon, for example, personal hygiene may cause distress and embarrassment.

Ensuring confidentiality is another important factor consistently highlighted in ethical regulations and standards. This is a legal requirement,<sup>161, 162</sup> and if researchers are unable to assure participants that the information they provide will be kept confidential, they may be reluctant to participate.<sup>170</sup> However, ensuring confidentiality is somewhat problematic as researchers also have a duty to pass on information if they feel an individual is 'at risk'.<sup>171</sup> A common way to overcome this is to ensure that participants are informed prior to participation of the potential need to link sensitive and identifiable data. However, ensuring children or adolescents living with some form of disability are

informed of this possibility may be particularly difficult if, for example, they have specific learning difficulties which impact their understanding of the questions in assent or consent forms. To ensure data remain confidential throughout the course of a research project, most researchers will anonymise the data early to avoid data linkage. However, if data cannot be linked to the participant, researchers are unable to disclose sensitive information when a participant has revealed that they are at risk of abuse or harm.

## 2.5 Visual Impairment

Visual impairment is a form of childhood disability characterised by reduced visual function commonly measured by visual acuity (VA). VA has been traditionally measured using a Snellen notation, but the modern gold standard is a chart using the logarithm of the minimum angle of resolution (logMAR) notation to quantify the distance at which an individual is capable of reading printed text of various sizes. Globally, 285 million people are estimated to be visually impaired, of whom 39 million are blind (the latter classified according to the WHO taxonomy of severity of visual impairment as logMAR worse than 1.3).<sup>172</sup> In the UK, it is estimated that 2 in every 1000 children are visually impaired.<sup>173, 174</sup> The majority of children and YP with severe visual impairment and blindness (SVI/BL) have very early onset impairment, and grow up without useful vision. Severity of visual impairment can range from moderately reduced VA to complete blindness (see Table 1, pg. 42).<sup>175-178</sup> The most common causes of visual impairment in childhood are cerebral visual impairment, disorders of the optic nerve, and disorders of the retina.<sup>179</sup>

Table 1. Classification of severity of visual impairment.

Category	Presenting	g distance VA			
	Worse than:	Equal or better than:			
0 Mild or no visual		0.46			
impairment		6/18			
1 Moderate visual	0.46	1.00			
impairment	6/18	6/60			
2 Severe visual	1.00	1.30			
impairment	6/60	3/60			
3 Blindness	1.30	1.77			
	3/60	1/60*			
4 Blindness	1.77	Light perception			
	1/60*	Light perception			
5 Blindness	No light	perception			
9	Undetermined or unspecified				

\* or counts fingers at 1 metre.

Whilst onset of SVI/BL is likely to be early in life, a number of disorders such as Stargardt's disease and neurological disorders<sup>180</sup> can cause onset of visual impairment or SVI/BL later during adolescence. Impaired vision can either be stable or progressive in nature at presentation and can manifest in a number of ways, affecting VA, visual field function, and contrast sensitivity in addition to oculomotor performance and visual perceptual abilities.<sup>181</sup>

Three quarters of children with SVI/BL have conditions which are neither preventable, nor treatable.<sup>173</sup>

# 2.5.1 Impact of visual impairment upon development during

# childhood

From a developmental perspective, the impact of reduced visual sensory input during childhood and early life course is significant, and likely to impact in a number of ways.

A substantial amount of literature supports the notion that communication between individuals begins long before an infant begins to speak. In the 1970s, researchers coined two types of intersubjectivity: primary and secondary. In the first 9 months after birth, infants are thought to be capable of primary intersubjectivity; that is, communicating to others about their immediate desires and emotions. This type of communication is largely driven by expressive behaviour such as crying which is intentionally used to get some kind of reward e.g. food or comfort.<sup>182, 183</sup> After the age of 9 months, infants are thought to be capable of secondary intersubjectivity: the ability to communicate with others about aspects of the wider environment such as surrounding objects, people and events.<sup>184</sup> In this dominant theory, vision is essential for communicating that an object exists and which qualities are particularly interesting.<sup>22, 185</sup>

Children who have visual impairment or blindness have been shown to compensate for the lack of visual stimuli in the development of primary and secondary intersubjectivity, by using auditory clues such as verbal instructions which are provided by others<sup>186, 187</sup> or tactile or sonorous stimulation.<sup>187-190</sup> For example, mothers have been shown to communicate effectively with their visually impaired infants through stroking, blowing or imitating the sounds they made.<sup>191</sup> Others have suggested that spacing plays a critical role among children with visual impairment<sup>192</sup> and these children are able to use their own bodies and the space they take up, as reference for the location, size, and physical attributes of objects.<sup>192, 193</sup>

In contrast to these findings, development of social communication, including joint attention, in children with visual impairment has been likened in the past to the development of children diagnosed with autism in that delay is often seen in specific difficulties such as social avoidance, social approach, and anxiety during social interactions.<sup>194</sup> Theory of mind (ToM) is an important developmental milestone during infancy which is defined as the ability to conceive mental states; that is, understand about the connections between other people's knowledge, desires, feelings and their

actions.<sup>195</sup> ToM is a prerequisite to meaningful communication with others. Critical to the development of ToM is the observation of other people's behaviour and evaluation of the intention of the behaviour. When observed in sighted infants, this milestone is thought to emerge at the age of 2 years, and is enhanced by pretend/or imaginative play.<sup>196</sup> When assessed in children with visual impairment, research has suggested that the development of ToM is universally delayed to some degree until the age of 12 years.<sup>197-199</sup>

Language development is another developmental capacity thought to be delayed in infants with visual impairment. Again, this form of delay is not surprising given that in sighted populations, vocabulary growth is associated with secondary intersubjectivity, and reliant upon observing objects in the wider physical environment.<sup>200</sup> A degree of social communication is also reliant upon non-verbal cues, such as facial expressions, body postures and gestures, which convey important information about how an individual uses language e.g. whether they are lying. Without vision, a child will miss out on these important non-verbal cues. Research documenting this delay, however, has mixed conclusions. Some researchers have shown that children with visual impairment develop language in the same way as sighted children<sup>201</sup> and others have shown delay in relation to limited experience with other people and the environment.<sup>202,</sup> <sup>203</sup> It is also likely that children with differing degrees of visual impairment severity experience differences in terms of language development.<sup>194</sup> The origin of this form of delay is difficult to pinpoint due to the complex interface between biological, environmental, and social influences. For example, research documenting developmental delay in language acquisition may be related to the responses of others, such as parents, who may view their child as helpless and therefore incapable of communicating for themselves.<sup>204, 205</sup>

Despite the possibility that visually impaired infants may be capable of compensating somewhat in some sensory domains, it is likely that there is some degree of developmental delay among most children living with visual impairment. The extent to

which this prevails into adolescence and early adulthood is debateable as some literature suggests that visually impaired children eventually catch up with their sighted peers in many aspects of development,<sup>198</sup> and have some cognitive strengths in relation to age-matched controls.<sup>206</sup> When discussed in relation to ToM, for example, developmental gain may be a result of greater opportunity for conversational interactions as visually impaired children get older.<sup>207, 208</sup> In a study of high-functioning adolescents aged 10-16 years living with SVI/BL (versus visual impairment), participants were found to score significantly higher on a battery of neuropsychological tests than age-matched expectations based on test norms, with particular strengths in auditory and working memory.<sup>206</sup> These findings may be explained well by superior cortical capacities, including neural plasticity,<sup>209</sup> and a greater allocation of resources for auditory processing,<sup>210</sup> leading to specific cognitive strengths.<sup>211</sup> Thus, conflicting evidence for the nature and magnitude of the relationship between visual impairment and developmental delay has, to date, prevented development of a strong empirical foundation.

The impact of visual impairment at the level of activity and participation, the latter components of disability outlined by the ICF model, can be examined through self-reported experiences of children and YP living with visual impairment. Despite favouring the same activities as sighted children, children with visual impairment have reported being unable to participate in some activities, such as reading, watching live sports or concerts, or playing outdoor games<sup>212</sup> and subsequently unable to pursue the activities that they enjoy the most. However, self-reported accounts of the impact of visual impairment are often placed within a broader context in which experience of society, cultural norms, other people, and their reactions to visual impairment are influential. For example, children with visual impairment have reported difficulty socialising freely with other children, travelling independently, and choosing academic and sporting activities without input from a parent or guardian.<sup>213</sup> Maintaining a healthy social life may be difficult for children and YP with visual impairment as a consequence

of the functional impact of visual impairment upon social skills such as assertiveness skills or establishing eye contact.<sup>214-216</sup>

Determining the impact of visual impairment during childhood is further complicated by variation in the manifestation of impairment, nature of visual deterioration and onset of impairment. Some research suggests that individuals who are blind from birth have an advantage over those who develop visual impairment at a later stage during childhood in relation to arithmetic skills,<sup>217</sup> working memory,<sup>217</sup> and auditory spatial processing.<sup>218</sup> This may be attributable to these children's ability to compensate for the lack of visual input through re-organisations of neural networks<sup>219, 220</sup> as previously discussed (pg. 42). In comparison, individuals with later onset, acquired visual impairment may have missed the critical period for re-organisation of neural networks. Despite a wealth of empirical attention paid towards investigating differences in brain activity, particularly in the visual cortex, among those with early onset or congenital visual impairment and late onset deterioration in vision,<sup>219-226</sup> differences between the two populations in relation to the self-reported experience of visual impairment are yet to be formally investigated. Given that a substantial amount of learning comes from vision and takes place during early childhood, it is likely that differences between the populations extend beyond these variations.

### 2.5.2 Impact of visual impairment during adolescence

As previously discussed (see Section 2.1.1, pg. 23), adolescence is widely recognised as a time for growth and development in which individuals experience a number of formal transitions. There are also a number of informal changes that occur during adolescence, as children develop the cognitive and social skills necessary for progression in education and the nature of social interactions change. Development of independence is widely recognised as a defining characteristic of adolescence as individuals learn to perform tasks associated with everyday living such as travelling, cooking, or cleaning, without help from a parent or guardian. Adolescence is a period of life course recognised as qualitatively different from both childhood and adulthood,

when viewed in terms of healthy and sighted individuals. Thus, it is important to consider the impact of visual impairment as equally unique during this period. As children develop into adolescents, the negative impact of visual impairment can be viewed as dynamic and changing: increasing as individuals encounter new environments and increased opportunities for independence, and decreasing as individuals develop effective coping strategies, and adapt to their disability.

At the level of psychological well-being, over time individuals who are living with a chronic illness or disability, such as visual impairment, may become increasingly familiar, and subsequently better able to cope with the everyday impact or burden of their condition. Resilience and coping are intrinsic processes which are often studied in children living with long-term impairment or disability with a view to explain positive adjustment.<sup>227</sup> Resilience is defined as the process of identifying or developing resources and strengths to manage stressors flexibly and gain positive outcomes.<sup>228</sup> Thus, it is likely that an adolescent who has been living with visual impairment since an early age will undergo behavioural-motivational, cognitive and emotional adjustments<sup>229</sup> during childhood and adolescence, which may reduce the negative impact of visual impairment.

Despite the possibility for positive adaptation over time, a number of environmental barriers likely increase the impact of visual impairment during adolescence. As YP develop, they will likely encounter new situations and face a range of new environments. At the same time, they may experience a desire to become more independent and, fitting with the expectations, norms and attitudes of their peer group, resist functional support from others. Evidence documents a large proportion of YP with visual impairment aged 16-25 are not likely to be in employment, education or training (NEET) and are twice as likely to be NEET as the general population of 16-25 year olds.<sup>230</sup> During adolescence, individuals with visual impairment will be faced with realisations of the impact of visual impairment in the future, and may be worried about gaining a job in light of biased appraisals by future employers, and their increased

independence and responsibility in the future.<sup>231</sup> Thus, evidence suggests that as a child develops into adolescence, the impact of visual impairment is increasingly related to external environmental factors such as participation in everyday life events, in contrast to intrinsic, personality-related factors such as adjusting and coping with the functional impact of reduced vision.

Similar to their sighted peers, adolescents with visual impairment will likely experience a number of transitions or changes which will need to be overcome in light of living with visual impairment. Such changes may alter the impact of visual impairment, introducing new restrictions, or the need for new adaptations. To date, the existing literature documents only one normative transition experienced by children with visual impairment as they develop into early adulthood: from full-time education into formal employment. This empirical focus is likely to reflect knowledge of poor post-educational outcomes among YP with visual impairment<sup>230</sup> and the subsequent need to support YP as they transition into competitive economic and occupational climates. Similar to sighted YP, good educational outcomes among YP living with visual impairment are associated with greater rates of employment<sup>232, 233</sup> and social support from parents and broader social networks.<sup>233, 234</sup> Several transition-related programmes have been developed specifically for use by adolescents with visual impairment. These place heavy emphasis upon independence training,<sup>235, 236</sup> work experience,<sup>237</sup> transition planning,<sup>237, 238</sup> and early tuition of career-related skills.<sup>239</sup> Despite a number of informal changes documented among sighted adolescence, namely related to the value of social interactions and friendships, the range and experience of these informal changes during adolescence of YP living with visual impairment is yet to be formally investigated. Although independence is a key characteristic of adolescence, literature exploring the development of independence among YP with visual impairment is largely confined to improving post-educational outcomes. The impact of limited functional ability upon independence has received limited empirical attention and, when it does exist, is primarily discussed in developments of vision-related quality of life

outcome measures,<sup>231, 240</sup> and as a component of the broader, daily impact of visual impairment.

# 2.6 Measuring the impact of visual impairment

In ophthalmology, quantifications of visual impairment traditionally consist of purely objective, functional assessments such as VA. These measures provide necessary insight as to visual function and are critical for clinical management. However, these measures do not capture the *subjective* impact of visual impairment and are thus not sufficient to capture the interaction between levels and forms of impact of visual impairment as specified by social models of disability such as the ICF.

Attempts have been made to overcome the shortfall of measurement in ophthalmology, by incorporating generic HRQoL measures into both adult,<sup>241, 242</sup> and paediatric ophthalmology practice.<sup>243</sup> However, the full impact of vision-related problems cannot be adequately described, or measured, using generic HRQoL measures, as some QoL issues reported by visually impaired individuals are likely to be specific to the experience of living with impaired vision,<sup>244</sup> hence the need for measuring vision-specific QoL issues i.e. vision-related Quality of Life (VQoL).

Measurement of the impact of visual impairment upon daily functioning, i.e. the individual's ability to participate in activities which are meaningful and enjoyable, is a second outcome which measures of VA do not adequately address. Functional vision (FV) is defined as vision that can be used to perform tasks requiring vision and differs from VA in that it comprises a self-reported evaluation of the difficulty of completing everyday tasks in real everyday environments (e.g. navigating around an unfamiliar room, boiling a pan of water, or locating a bus stop). As with most questionnaires, items included in measures of FV should be developed with a particular population in mind, referencing activities which may only be relevant and meaningful for individuals of the same age or living within the same culture. Thus, functional vision incorporates elements of VA into a broader physical, social, and cultural context.

In the past, and perhaps as a product of issues surrounding conflation of HRQoL and functional status and ability, items relating to FV have been included in measures designed to capture VQoL among the population of adults living with visual impairment.<sup>245, 246</sup> For example, the Refractive Status and Vision Profile (RSVP) contains 8 subscales which include items relevant to both symptoms and physicalsocial functioning, and expectations and concerns.<sup>247</sup> Similarly, the Impact of Vision Impairment (IVI) measures 'handicap' which is defined as an interaction between limitation or activity and an individual's needs. This measure incorporates items which can be categorized into five domains: emotion, leisure, mobility, social, and household.<sup>248</sup> The multi-dimensional nature of these approaches is potentially problematic as condensing multiple outcomes relating to both VQoL and FV into one summary score, clinicians risk making inaccurate inferences about the effectiveness of treatment and intervention upon different aspects of the impact of visual impairment. To comprehensively capture the impact of visual impairment in light of individual differences such as needs, emotions, and preferences for example, would involve development and administration of VQoL and FV PROMs as complementary, yet distinct measures.

Similar to use of generic HRQoL PROMs (see Section 2.3.2, pg. 34), in the past, when child-appropriate vision-specific PROMs have been unavailable, vision-specific PROMs which were designed for adult populations have been used as alternatives.<sup>249, 250</sup> Whilst these contain items which may be more specific to the impact of visual impairment than items contained in generic HRQoL measures designed for children, a number of factors render them inappropriate for use in paediatric ophthalmology services. Firstly, the items included in these instruments are not developmentally sensitive, meaningful or valuable to the aspects of everyday life which may comprise VQoL and FV among paediatric populations.<sup>107, 251, 252</sup> For example, they contain items that are irrelevant to activities of childhood and adolescence, such as driving or occupation.<sup>253, 254</sup> Secondly, these instruments may contain language or response-categories which are difficult for

children or YP to understand.<sup>255, 256</sup> Finally, there are a number of ophthalmic conditions which present during childhood for which vision-specific adult PROMs may not be applicable.<sup>254</sup> Thus, PROMs which are both age- and vision-specific are needed for use within paediatric ophthalmology, to capture the dynamic impact of visual impairment during childhood and adolescence.

#### 2.6.1 Paediatric vision-specific PROMs

The impact of visual impairment during early life course is complex and dynamic, impacting an individual at the level of bodily functions, activity and participation as outlined by the ICF. The impact may also be prone to qualitative change as children develop physical capacities to adapt to the functional restrictions of visual impairment and psychological maturity to cope with visual impairment. Transitions or changes during childhood and adolescence mean that YP frequently have to adapt to the impact of visual impairment in varying social and cultural contexts, for example, when facing the transition from education to employment. Vision-specific PROMs which are suitable for use by children and YP are therefore important if ophthalmic professionals want to pick up on the dynamic and changing impact, and provide the most appropriate treatment for children and YP living with visual impairment. Despite the need for ageappropriate vision-specific PROMs, to date there are few vision-specific PROMs which are both age-appropriate and suitable for detecting developmental change, and psychometrically robust for use by children and YP.

There are several more reasons why PROMs capturing the broader impact of visual impairment are needed. Firstly, vision-specific PROMs constitute an important methodological tool in the development and evaluation of new treatments or interventions. For instance, outcomes can be measured before and after administration of a particular intervention to evaluate changes which are attributable to the intervention. For example, both VQoL and FV may be valuable outcomes when measured before and after low vision rehabilitation services and clinical interventions as subjective outcomes which complement traditional measures of VA. PROMs may

also be valuable in randomised control trials designed to assess the impact of clinical treatment upon subjective outcomes.

Vision-specific PROMs can also be useful in assessing the longitudinal impact or disease course of an ophthalmic disorder by assessing the progressive impact of visual impairment across multiple developmental milestones and time-points. This measurement allows practitioners and health professionals to assess temporal and societal factors associated with change in the impact of visual impairment such as a change in education.

To date, several vision-specific PROMs have been developed for children with visual impairment. In a recent systematic review of PROMs specifically designed for use by children with visual impairment, 17 eligible instruments were identified.<sup>257</sup> However, the development of these instruments has been influenced substantially by methodological challenges such as defining VQoL and FV, and ensuring the measures conform to psychometric standards. The identified instruments differ substantially in the outcomes that they capture. Six measures were generic, capturing the impact of visual impairment in children irrespective of eye condition/disease and 11 were ophthalmiccondition specific. Seven instruments were designed to measure VQoL and three to measure FV or visual ability. The remaining instruments measured varying outcomes related to visual impairment, including the impact of treatment for amblyopia, symptoms, psychosocial impact, and well-being. Others such as parents, adult stake holders, and sighted children often had input in the item development stage rendering the instrument items potentially inappropriate for children with visual impairment. Only a small number of instruments identified had shown sufficient psychometric properties such as evidence of different types of validity and reliability<sup>151</sup> (see Section 2.4, pg. 36).

Since that review was published, five new vision-specific instruments have been developed for use by children. Table 2 (pg. 54) shows an updated description of all-cause generic visual impairment and eye disorder-specific instruments. Three of these more recent instruments are generic and report to capture activity and participation,<sup>258</sup>

vision-related difficulties<sup>259</sup> and functional vision.<sup>6</sup> However, only one of these conforms to psychometric standards. This instrument was developed during the foundation research, and designed to capture FV among children aged 10-15 years. The remaining two instruments are eye disorder-specific (designed for use by children with amblyopia<sup>260</sup> and refractive error<sup>261</sup>) and one does not differentiate between VQoL and FV.<sup>260</sup> Thus, development of PROMs which are both vision-specific and age-appropriate is an emerging, yet methodologically challenging field of research.

Table 2. Updated description of all-cause generic visual impairment and eye disorder-specific PROMs.

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								domains	
Cardiff Visual Ability Questionnaire for Children (CVAQC) <sup>262-264</sup>	2010, UK	English, Chinese, Turkish	Visual impairment	5-18 years	Child	Children	25	Visual ability	Rasch-derived unidimensional scale capturing visual ability/ functional vision
Children's Visual Function Questionnaire (CVFQ) <sup>265-267</sup>	2004, USA	English, German	Visual impairment	0-7 years	Proxy	Researchers	35 (age <3); 40 (age 3- 7)	Vision-related quality of life	FA derived multi- dimensional instrument with following subscales: (1) General health, (2) General vision, (3) Competence, (4) Personality, (5) Family impact and (6) Treatment
The Impact of Vision Impairment on Children (IVI_C) <sup>213,</sup> <sup>268</sup>	2011, Australia	English	Visual impairment	8-18 years	Child	Children, parents and teachers	24	Vision-related quality of life	Rasch-derived unidimensional scale capturing the impact of visual impairment on participation in daily activities
LV Prasad- Functional Vision Questionnaire (LVP- FVQ) <sup>269</sup>	2003, India	English, Hindi, Telugu	Visual impairment	8-18 years	Child	Children, parents, clinicians and researchers	20	Functional vision	Rasch-derived unidimensional scale capturing visual ability/ functional vision

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
LV Prasad-	2012, India	English,	Visual	8-16	Child	Children and	23	Functional	Rasch-derived
Functional Vision		Hindi,	impairment	years		parents		vision	unidimensional scale
Questionnaire		Telugu							capturing visual ability/
Second Version									functional vision
(LVP-FVQ II) <sup>270</sup>									
	0011			40.45			05		Develo la fonda este
Vision-related	2011, UK	English		10-15	Child	Children	35	Vision-related	Rasch derived scale
Quality of Life of			impairment	years				quality of life	capturing vision-related
Children and Young									quality of life
Amblyopia	2001. USA	English	Amblyopia	3-13	Child and	Parents.	20	Impact of	Parent version: FA
Treatment Index	,	5 -		vears	Proxv	clinicians and	(proxv	amblvopia	derived multi-dimensional
(ATI) <sup>271</sup>				,	,	researchers	version,	treatment	scale with the following
							age 3-6)		subscales relating to
							19 (child		treatment of amblyopia:
							version,		(1) Adverse effects, (2)
							age 7-13)		Treatment compliance
									and (3) Social stigma
									Child version: FA derived
									multi-dimensional scale
									with following subscales
									relating to treatment of
									amblyopia: (1) Adverse
									effects, (2) Treatment
									compliance and

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
									The Amblyopia Treatment
									Index (ATI) continued
									(3) Functioning at near
Children's	2011, UK	English	Amblyopia	5-7 years	Child	Children	3 items	Quality of life	Impact of amblyopia upon
Amblyopia							(across 7		daily life (not derived
Treatment Quality of							different		psychometrically)
Life Questionnaire							scenarios)		
(CAT-QoL) <sup>272, 273</sup>									
Convergence	1999, USA	English	Convergence	9-18	Child	Children and	15	Symptoms	Unidimensional scale
Insufficiency			insufficiency	years		researchers			capturing symptom
Symptom Survey									severity (not derived
(CISS) <sup>2/4</sup>									psychometrically)
Effects of	2010. USA	Enalish	Juvenile	8-18	Child	Children and	23 (age 8-	Vision-related	Unidimensional scale
Youngsters'	,	5	idiopathic	vears		clinicians	15);	quality of life	capturing visual
Evesight on Quality			arthritis-	5			26 (age 16-	and/or visual	ability/functional vision
of Life (EYE-Q) <sup>275-277</sup>			associated				18)	function	(not derived
			uveitis				,		psychometrically)
Emotional Impact of	2004, UK	English	Amblyopia	<67	Proxy	Parents and	15	Impact of	Multi-dimensional scale
Amblyopia				months		clinicians		amblyopia	capturing the impact of
Questionnaire								treatment	amblyopia treatment in
(EIAQ) <sup>278</sup>									form of following
									subscales (not derived
									psychometrically):

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
									Emotional Impact of
									Amblyopia Questionnaire
									(EIAQ) continued
									(1) Child's experience, (2)
									Family's experience and
									(3) Child's well-being
Intermittent	2010, USA	English	Intermittent	5-17	Child and	Children and	12	Health-	Child version:
Exotropia			exotropia	years	Proxy	parents		related	Unidimensional scale
Questionnaire								quality of life	capturing quality of life of
(IXTQ) <sup>279-281</sup>									children (not derived
									psychometrically)
									Parent version: FA
									derived multi-dimensional
									scale with following
									subscales: (1) Function,
									(2) Psychosocial effects
									and (3) Surgery
Nasolacrimal Duct	2006, USA	English	Nasolacrimal	6-48	Proxy	Parents and	29	Symptoms	A priory determined two-
Obstruction			duct	months		clinicians		and quality of	dimensional scale with
Questionnaire			obstruction					life	the following subscales:
(NLDO) <sup>202</sup>									(1) Symptoms and (2)
									Child's health-related
									quality of life

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
Pediatric Refractive	2006, USA	English	Refractive	8-18	Child	Not reported	26	Vision-related	Multidimensional scale
Error Profile			error	years				functional	with following subscales
(PREP) <sup>283, 284</sup>								and well-	(not derived
								being	psychometrically): (1)
									Overall vision, (2) Near
									vision, (3) Far vision, (4)
									Symptoms, (5)
									Appearance, (6)
									Satisfaction, (7) Activities,
									(8) Academics, (9)
									nandling and (10) Peer
									perception
Perceived	2002 11K	English	Amblyonia	3-5 vears	Proxy	Parents and	10	Perceived	Item details not available
Psychosocial	2002, 010	Linghon	/ inbiyopia	o o youro	1 loxy	clinicians	10	nsvchosocial	
Questionnaire						Chinicitano		well-being of	
(PPQ) <sup>285, 286</sup>								child	
Psychological	2006, UK	English	Amblyopia,	16-18	Child	Children and	8 items	Psychological	Unidimensional scale
Impact			strabismus,	years		researchers	(across 4	impact	capturing psychosocial
Questionnaire			or refractive				scenarios)		impact of strabismus (not
(PIQ) <sup>287</sup>			errors						derived psychometrically)
Quality of Life in	2007, Italy	Italian	Allergic	5-12	Child	Children,	16	Quality of life	FA derived two-
Children with Vernal			conjunctivitis	years		parents,			dimensional scale with
Keratoconjunctivitis						clinicians and			following subscales: (1)
(QUICK) <sup>200</sup>						researchers			Symptoms and

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
									Quality of Life in Children
									with Vernal
									Keratoconjunctivitis
									(QUICK) continued
									(2) Daily activities
Children's Vision for	2013, Saudi	English	Amblyopia	5-12	Child	Children,	28	Vision-related	Rasch-derived
Living Scale	Arabia			years		paediatric		quality of life	unidimensional scale
(CVLS) <sup>200</sup> *						eye care			capturing: (1) Mood, (2)
						professionals			Self-esteem, (3) Social
						and			relations, (4) Functional
						researchers			vision, (5) Visuo-motor
									function and, (6)
									Academic performance
				10.00					
Student Refractive	2014, USA	English	Refractive	10-20	Child	Multi-	38	Vision-related	Rasch-derived
Error and			error	years		disciplinary		quality of life	unidimensional scale
Eyeglasses						research			capturing the impact of
Questionnaire						team			uncorrected and
(SREEQ)									corrected refractive error
									on vision-related quality
									of life

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
Participation and	2017, The	Dutch	Visual	0-25	Proxy (ages	Professionals	44 (age 0-	Activity and	Instruments with the
Activity Inventory for	Netherlands		impairment	years	<7 years)	of low vision	2);	participation	following sub-scales:
Children and Youth					and	rehabilitation	62 (age 3-		0-2 years: (1) Bonding,
(PAI-CY) and Young					Child/Young	centres,	6);		(2) Incentive processing,
Adults (PAI-YA) <sup>258</sup> *					adult (ages 7-	parents,	55 (age 7-		(3) Visual attention, (4)
					25 years)	young adults,	12);		Sensorial functioning, (5)
						children	58 (age 13-		Orientation, (6) Play, (7)
							17);		Mobility, (8)
							141 (age		Communication (not
							18-25)		derived psychometrically)
									3-6 years: (1) Bonding,
									(2) Incentive processing,
									(3) Visual attention, (4)
									Sensorial functioning, (5)
									Orientation, (6) Motor
									functioning, (7) Reading
									and writing, (8) Play, (9)
									Self-reliance, (10)
									Mobility, (11)
									Communication, (12)
									Social relationships, (13)
									Day-care/school/study
									(not derived
									psychometrically)
									7-12 years: (1) Play, (2)
									Self-reliance, (3)
									Finances, (4) Mobility,

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
									Participation and Activity
									Inventory for Children and
									Youth (PAI-CY) and
									Young Adults (PAI-YA)
									continued
									(5) Communication, (6)
									Social relationships, (7)
									Day-care/school/study,
									(8) Leisure time, (9)
									Acceptance/self-
									consciousness (not
									derived psychometrically)
									13-17 years: (1) Self-
									reliance, (2) Finances, (3)
									Mobility, (4)
									Communication, (5)
									Social relationships, (6)
									Day-care/school/study,
									(7) Leisure time, (8)
									Acceptance/self-
									consciousness (not
									derived psychometrically)
									18-25 years: (1) Mobility,
									(2) Communication, (3)
									Social relationships, (4)
									Day-care/school/study,
									(5) Leisure time,

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	construct	and/or domains
								and/or	
								domains	
									Participation and Activity
									Inventory for Children and
									Youth (PAI-CY) and
									Young Adults (PAI-YA)
									continued
									(6) Acceptance/self-
									consciousness, (7)
									Reading and visual aids,
									(8) Household living, (9)
									Finances, (10) Self-care,
									(11) Computer skills, (12)
									Intimate/romantic
									relationships, (13) Peer
									contact, (14) Holiday and
									going out, (15)
									Information/regulations,
									(16) Applying, (17) Work
									(not derived
									psychometrically)
Vision-related	2013,	Japanese	Visual	Details	Child and	Details not	39	Vision-related	Details not available
symptom and	Japan		dysfunction	not	proxy	available		difficulties	
performance				available					
checklist for children									
(VSPCL) <sup>200</sup> *									

Instrument	Year and	Languages	Eye	Age	Respondent	Input into	Number of	Intended	Captured construct
	country		conditions	range		items	items	and/or	and/or domains
								domains	
Functional Vision Questionnaire for Children and Young People with Visual Impairment (FVQ_CYP) <sup>6</sup> *	2013, UK	English	Visual impairment	10-15 years	Child	Children	36	Functional vision	Unidimensional Rasch- derived scale capturing Functional Vision

\*new, vision-specific instruments developed for children published since the original review

PROMs which are developed for use by paediatric populations need to be accessible in a format which can be accessed by children with disability, reflecting the emphasis on participation in the ICF. Within the context of visual impairment, for example, PROM development needs to consider different modes of administration with consideration of patients' limited VA for reading printed text, and subsequent participation in this kind of measurement. The same principles apply when children and YP with visual impairment take part in the development stages of PROMs as, by law, this population may require specialist assistance or adaptations such as information sheets and consent forms to be enlarged or printed in Braille, which can add further resource demands on researchers. Thus, ethical considerations applicable to the participation of healthy, sighted children in research (see Section 2.4.1, pg. 38) are somewhat magnified when children with visual impairment are required to take part.

# 2.6.2 Conceptual and theoretical framework for the development of paediatric vision-specific PROMs

To address the scarcity of psychometrically robust vision-specific PROMs available for use by children and YP with visual impairment, Rahi et al. recently reported research using a child-centred approach to developing two novel PROMs, one assessing VQoL and the other FV (see Section 2.6.1, pg. 51) in children and YP with visual impairment (as per the WHO criteria shown in Table 1, pg. 42).<sup>5-7</sup> The instruments were designed as distinct but complementary measures for capturing different aspects of living with visual impairment: the impact upon functional ability (FV) and the psychosocial-emotional impact (VQoL). In the absence hitherto of an established conceptual framework and child-centred methodology in this area, these measures were developed in the first instance for use by visually impaired children aged 10-15 years. This age group was chosen to ensure children were developmentally capable of self-reporting on the complex issues included in definitions of QoL, HRQoL and VQoL.

This 'foundation' research programme demonstrated the feasibility and benefits of involving children with visual impairment in all standard stages of questionnaire design, an approach which is concordant with the values outlined by the ICF. The researchers ensured the items in the PROMs reflected meaning and value for the paediatric population by conducting individual in-depth interviews with children and YP with visual impairment. During interviews, participants were asked to discuss their everyday life in relation to 6 topics: school, home life, activities and socializing, life skills and independent living, eye problems and the eye clinic, and the future. In keeping with definitions of QoL as dependent on individual differences, experiences, lifestyles, hopes, attitudes and beliefs (see Section 2.3, pg. 28) the content of interview discussions were largely child/young person-centred. The second phase involved pretesting the draft instruments by consulting an expert-reference group of visually impaired children/YP, which was supplemented by an expert consensus meeting of the professionals involved in the development. During consultations, children and YP were asked to reflect on item relevance and comprehensibility, layout of the questions, and administration methods, thus ensuring that completion of the PROMs would be both possible and feasible. The developed instruments were then piloted with an aim of identifying any immediate problems with the questionnaires (e.g. in terms of ceiling or floor effects on particular questions), and finally validated with a nationally representative sample of children and YP with visual impairment through a postal survey. This research demonstrated the feasibility and value of involving visually impaired children and YP in the process of instrument development by grounding the conceptual and theoretical framework for the instruments in their voices and lived experience.

### 2.7 Research aims

The overarching objective of the research reported in this thesis is to explore the dayto-day impact (both immediate and broader) of visual impairment during childhood and

adolescence and use this knowledge to develop a suite of age-appropriate, visionspecific, complementary PROMs of VQoL and FV for use by adolescents with visual impairment. This research builds directly upon the foundation research by Rahi et al. with 10-15 year olds<sup>5-7</sup> and addresses the gap in the VQoL and FV instrument provision for YP older than 15 years. The reported research uses a conceptual and theoretical framework which is grounded in the voices of YP living with visual impairment. Concordant with social theories of disability, participation of YP with visual impairment is integral to the instrument development, and vital for ensuring the instruments are psychometrically robust for use by YP with visual impairment within ophthalmology services and potentially in the future as outcome measures within clinical trials. The research stages outlined in Section 2.6.2 (pg. 64) afford the opportunity to explore, in detail, the experience of growing up with visual impairment, adopting a life course perspective and drawing on the views of visually impaired adolescents (through indepth individual qualitative interviews) who are well-placed to retrospectively reflect on their experience of visual impairment. In parallel to the current research project, instruments designed for children younger than 10 years were developed with a view to address the entire spectrum of children and YP living with visual impairment for which age-appropriate, vision-specific PROMs are not readily available.

### 2.8 Summary

Biological and psychological perspectives demonstrate the dynamic nature of growth, development, learning, and change during childhood and adolescence. Specifically, adolescence is a period associated with a number of age-specific issues as YP experience changes; both intrinsic (related to psychosocial and emotional well-being), and extrinsic (related to the broader social and cultural world in which they develop). When a child or adolescent is living with some form of disability, such as visual impairment, everyday life is likely to be influenced in a number of ways. Despite evidence for the dynamic impact of visual impairment during childhood and

adolescence, there are limited, vision-specific, age-appropriate and psychometrically robust PROMs which are readily available for improving outcomes in paediatric ophthalmology services. In particular, adolescence is a period of life course described as qualitatively unique from both childhood and adulthood. Thus, age-appropriate PROMs that are grounded in the experiences, lifestyles, attitudes, and beliefs of YP living with visual impairment are needed to accurately capture the impact of visual impairment during this period. The aim of the work reported in this thesis is to address this gap, adding to the broader understanding of the impact of visual impairment during childhood and adolescence, and enhancing the conceptualisation and measurement of the impact of visual impairment within paediatric ophthalmology.

### Chapter 3 Psychometric measurement in healthcare

As discussed in Section 2.3 (pg. 28), measurement has always been a fundamental component of healthcare and is important for improving the delivery of healthcare, and allowing policy makers to assess major health problems and set goals. The aim of this chapter is to outline the underlying principles of measurement which must be considered when developing PROMs as measures of the subjective impact of disability. The strengths and weaknesses of each approach will be highlighted as a basis for explaining the psychometric approach used in this research.

The earliest evidence of measures of population health in England and Wales date back to mortality rates which were collected during the 17<sup>th</sup> century as a result of the plague.<sup>289, 290</sup> However, as previously discussed, alongside and likely as a result of growth and development in information technology, measurement in healthcare has exploded in terms of the quantity of measurement instruments available, the ways measurement can be implemented, and the outcomes which can be captured. Today, electronic health records are increasingly implemented within both primary and secondary healthcare settings worldwide.<sup>291, 292</sup> Electronic databases such as the Hospital Episode Statistics<sup>293</sup> contain vast amounts of clinical data collected from the population and are free to access via the internet. Although delivered electronically, with vast improvements in the documentation and indexing of data, the content of these databases (i.e. numeric quantifiable data relating to mortality rates and hospital admissions) is largely comparable to data which were collected in the 17<sup>th</sup> century. Using electronic databases containing this type of data, scientists and statisticians can use sophisticated statistical methods to make inferences and associations which are essential for monitoring health at the level of the population. Today, these methods can be performed quickly and with ease, using statistical software such as SPSS, STATA or R.

### 3.1 Change in the nature of measurement

In the past 172 years (since the first statistical representation of life expectancy was constructed<sup>294</sup>), life expectancy has increased in all countries and in 2011, reached almost double of that in 1841.<sup>295</sup> As a result of this increase, data conceptualising aspects of health, such as QoL (versus length of life) are increasingly valued, and there is growing appreciation within healthcare, that treatment methods and interventions should be aimed at enhancing QoL among patients suffering from disease or illness. To measure QoL, we need to capture psychological outcomes which extend beyond numeric, quantifiable data used for the management of public health. This approach represents changing focus from the population to the individual, and in doing so, views patients as the gatekeepers to the *experience* of disease or illness. To systematically capture the experience of disease or illness, one might use a self-report questionnaire.

The purpose of such self-report questionnaires within healthcare has changed over time. Their earliest use focused upon exploring patients' attitudes and analysing how a range of attitudes was distributed within a population.<sup>296</sup> These questionnaires placed little value on scoring or quantitative analyses, and instead focused on capturing patients' verbal answers to open-ended questions, analysing the meaning behind their responses, and assessing their linguistic choices. Whilst these measures are essential for gaining rich insight as to the experience of disease or illness, statistical analyses are necessary to make comparisons between individuals or populations, and cannot be used to analyse the data provided by them. Thus, to make self-report outcomes, such as PROMs, optimal for use within healthcare, researchers involved in instrument development must resolve two incompatible notions: a) that data captured should represent complex, multidimensional, and subjective outcomes, such as HRQoL, which are theoretically underpinned by individual experience (see Section 2.3, pg. 28) and b) that the data should be captured in a way which is clinically beneficial, allowing practitioners and researchers to quantify outcomes, make associations, and compare individuals and populations.

The question 'Is psychological measurement possible?' is one which has been debated since the 1930s by scientists such as Norman Campbell<sup>297</sup> and Stanley Stevens<sup>298</sup> who argue as to whether measurement, which is a key component of physical science, can be applied to the domains of social science. Additive measurement, specifically, is a term used to describe measurement which is based upon equal units. For example, a scientist can easily apply additive measurement when measuring a patient's height or weight. Here, addition of one inch or one pound clearly involves adding one to the measurement number e.g. the patient's height in inches or weight in pounds. However, adding one unit in physical science (using height, weight, temperature or distance, for example) is conceptually different from adding the same amount in different disciplines e.g. social science.<sup>297</sup> Adding the same amount within social science involves first defining what the constant amount is. When measuring attitudes or experience of illness or disease, scientists and researchers must define what one unit of attitude or experience is. This was a major challenge for most social scientists in the 1930s, and attempts to answer the question 'Is psychological measurement possible?' subsequently failed.

Since the debate in the 1930s, science has progressed to the point to which we now understand more about how to measure psychological outcomes such as QoL. Today, PROMs (as described in Section 2.3.1, pg. 31) contain response formats which convert subjective attitudes or emotions into quantifiable outcomes. Likert-type scales are often used, in which verbal response categories are assigned numeric values (e.g. strongly disagree = 1, disagree = 2, agree = 3 etc.). Responses given by patients to single items are usually added to produce a summary score or summary index. Measurement of subjective self-report in this way has a number of benefits in healthcare which are outlined in Section 2.3.1 (pg. 31). The most important benefit is that outcomes derived from subjective self-report, can be collected, handled and analysed as though they are numeric, quantifiable data. This confers a number of statistical benefits, namely the use

of stronger inferential statistics and more powerful analyses to assess trends between individuals and populations.

However, in order for PROMs to be used in this way, they must be developed using a controlled, systematic approach, following the criteria outlined by Classical Test Theory (CTT),<sup>299</sup> Item Response Theory (IRT),<sup>300</sup> or Rasch Model Theory (RMT).<sup>301</sup> If the criteria outlined by these approaches are met, researchers can be sure that the questionnaire is reliable, valid (see Section 2.4, pg. 36),<sup>158</sup> and responsive to change in the unobservable characteristic (e.g. attitude or experience). Most importantly, researchers can use these approaches to define the constant amount or, in the context of social science, what one unit of attitude or experience is, and thus ensure that the questionnaire conforms to additive measurement.

# 3.2 Classical Test Theory

Classical Test Theory (CTT) is a traditional approach to questionnaire development which was first developed in the early 1900s,<sup>158, 302</sup> and is used to ensure instruments such as questionnaires are high quality (i.e. valid, reliable and responsive to change). There are several key features of CTT but the most important feature is the ability to detect scales which exist within a questionnaire by grouping items. Items are grouped by assessment of the correlation between pairs of items. As a result of item grouping, scale reliability is either confirmed or refuted.

FA<sup>303-306</sup> is the primary statistical method associated with CTT to assess the degree to which groups of items share a common core of information about the true score.<sup>153</sup> In doing so, FA indicates how many scales (i.e. groups of items correlating highly with each other) exist within a single questionnaire and thus indicates dimensionality of the instrument. In order to make inferences about a unit of measurement on an unobservable, psychological outcome, such as HRQoL, the items which make up the instrument must be unidimensional (i.e. they must all measure the same underlying construct). FA works by identifying the characteristics along which items differ

substantially and therefore identifies similarity between items. Maximum likelihood is a statistical method that fits FA models to multivariate observations to determine how many factors (or scales) exist within one instrument. Despite there being some form of statistical criteria for identifying the number of factors present in one instrument, researchers usually interpret outcomes from FA subjectively and iteratively, i.e. judge whether the number of factors determined by FA is conceptually valid and in line with their pre-specified hypotheses.

Throughout the history of questionnaire development, CTT has been a very popular approach to measurement theory and thus, the development of PROMs. This may be because it is relatively easy to understand by those who have some basic understanding of statistics. For example, the coefficient alpha which is used to represent reliability in FA is widely encountered in other statistical methods and may therefore be easy for researchers to interpret in a meaningful way. There are also many readily accessible statistical packages which can be used to perform FA.

Despite CTT being a popular, traditional approach to instrument development, validation, or reduction, it is increasingly recognised that this approach does not address the underlying issue with measurement in psychology; that is, FA *assumes* that the outcome can be measured using an additive model and that responses are linear and merely assesses the similarity *between* items.<sup>307</sup> Thus, FA gives little indication as to whether the items or responses follow a hierarchical structure. Some researchers have proposed that popularity of CTT peaked at a time when researchers focused more on improving the quality of questionnaires, and addressing properties such as validity and reliability, than on the measurement construct itself.<sup>296</sup> Additionally, several authors have argued that, rather than representing true reliability, the Cronbach's alpha value used in CTT for internal consistency reliability, which simply reveals the average degree of 'interrelatedness' between items, leaves little indication of the true unidimensionality of the measure, that is whether the items are similar in terms of the factors they are assigned to.<sup>308</sup>
Perhaps one of the most well documented shortcomings of CTT, is use of Cronbach's alpha as an indicator of the instrument's strength. The underlying assumption in CTT is that the activities/constructs which are captured by individual items in a questionnaire are of equal difficulty. As a consequence, variance between outcome 'summary' scores or indexes are attributed solely to variations between levels of ability among respondents.<sup>307, 309</sup> In reality, PROMs are often designed to capture outcomes such as self-report functional ability and consequently include items which differ in how easy or hard they are for respondents of different ability. This is necessary to ensure PROMs are applicable to patients who have a range of functional abilities, including those who are severely limited, and also those who have only minor difficulties. Thus, variation in respondent ability is unlikely to be the single cause of variation in responses. Assuming that the items are of equal difficulty introduces some measurement bias and may lead to incorrect conclusions within clinical practice.

## 3.3 Modern Test Theory

Identification of the shortcomings of CTT, and enduring debate surrounding the theoretical application of measurement in social science, has triggered application of a new approach to measurement, representing growth in understanding of theory and methods underlying measurement. Two important developments in test theory appeared in the 1960s. One was set forward by Rasch<sup>156, 301</sup> and the other by Lord and Novick.<sup>300</sup> The latter is termed Item Response Theory (IRT), and the former Rasch Measurement Theory (RMT). Within current literature, there is a broad understanding that CTT, IRT and RMT are fundamentally different from each other, each embodying different paradigms in terms of theory of measurement and techniques used to ensure an instrument truly measures an unobservable construct, such as QoL.

Unlike CTT which *assumes* linearity, modern test theory (both IRT and RMT) articulates the conditions under which interval level data can be estimated using the measurement instrument,<sup>310</sup> and can therefore be used to *construct* linearity and

subsequently ensure that the developed instruments can be used for additive measurement. Additionally, modern test theory specifies a probabilistic model which, unlike the assumption made by Cronbach's alpha in CTT, is based on the understanding that not all items included in a questionnaire are of equal difficulty.<sup>311</sup> In this way, modern test theory addresses the shortcoming of CTT when summarising performance on a measure and attributing it exclusively to variation in person ability.

When used for the development of PROMs, modern test theory has been shown to provide more information about the abilities of the full range of the population (with CTT only showing floor and ceiling effects), the suitability of the response options (for which CTT does not provide any formal analysis), and validity of the instrument, showing fit statistics, expected responses and local independence i.e. the item scoring bias resulting from similar items being included in the same instrument (with CTT merely showing inter-item correlations).<sup>312</sup> Thus, modern test theory (including both IRT and RMT) is broadly accepted as a theoretical and methodological extension of CTT which is increasingly applied within PROM developments.<sup>309, 313</sup>

## 3.3.1 Different approaches in modern test theory

Whilst similar in terms of contribution to current understanding surrounding test theory, and sharing many of the benefits they bring to instrument development, the techniques which fall under the umbrella term of modern test theory, must be considered as two distinct approaches with several pertinent differences.

The primary difference between IRT and RMT is in their underlying paradigm.<sup>311, 314</sup> When using IRT, researchers search for a model which best describes the pre-defined data. Often, a model with a greater number of parameters is chosen first (as the best fitting model). Statistical checks are then performed to try and reduce the model parsimoniously to have a smaller number of parameters and a comparable predictive power.<sup>314</sup> This paradigm is often taken for granted as the most appropriate approach within any form of statistical analysis but is increasingly endorsed within the context of instrument development. When used to validate an instrument however, this paradigm may encourage researchers to neglect post-hoc analyses to modify the data (i.e. improve the instrument by re-wording or removing items).<sup>314</sup>

In contrast, the paradigm underlying RMT is that individual responses to the ratings included in PROMs, that is, the data collected, are analysed according to the model of measurement. The theory or model of measurement was first defined by Rasch<sup>311</sup> as allowing for the separation of person and item parameters and thus sufficiency is a key characteristic. Using the notion of sufficiency, Rasch developed models which could be used to provide a probability distribution of responses given by an individual of a specific ability which is dependent only upon the difficulty of the item<sup>301</sup> i.e. the model of measurement. In practice, if data do not fit the model of measurement, post hoc adjustments are usually made iteratively to explore the measurement construct in more detail, and to assess which items need to be removed or amended, thus improving the instrument.

## 3.3.2 Characteristics of Rasch analysis

In practice, an additional quality of RMT or Rasch analysis is that it can be used to perform assessments of the quality of measurement instruments which are beyond those which can be explored using CTT or IRT.<sup>314</sup> This makes Rasch analysis increasingly favoured within the domain of instrument development. For example, using Rasch analysis, researchers can analyse the response-scales, -thresholds, and - categories to ensure they are optimal for capturing outcomes which are sensitive to the needs of respondents. In doing so, researchers can be sure that the ordering of the response categories is empirical. Additionally, Rasch analysis allows for assessment of differential item functioning (DIF): an indication that respondents with comparable levels of ability are responding differently to an item, and that a particular item should be removed. Thus, Rasch analysis can be used as a robust measure of construct validity, reliability (person, and item) and responsiveness<sup>313</sup> (see Section 2.4, pg. 36)

which is grounded in strong foundations from measurement, test, and item-response theory.

There are two models which can be used in Rasch analysis, offering additional flexibility for researchers developing questionnaires which differ in the nature of response options. The Andrich rating scale model is used when instruments contain items for which patients respond using the same response scale across all items.<sup>311</sup> The partial credit model differs in that it can be used when items vary in terms of their response format.<sup>315</sup> Similarly, there are dichotomous and polytomous versions of Rasch analysis which can be used when instruments contain items with only two response options (dichotomous) and items which contain more than two response options (polytomous). Several statistical packages exist for performing Rasch analysis (e.g. RUMM, Winsteps, R), allowing researchers to tailor the analysis to the needs of the research project. There are also a number of guidelines which can be used by researchers when improving the quantitative functioning and measurement performance of the rating scale. The most pertinent guideline refers to the amount of numerical information which indicates to what extent the data produce coherent raw scores e.g. raw scores that support the Rasch model of measurement. At least 10 observations of each response category is recommended to ensure the step calibration is stable when using the Andrich Rating Scale model with polytomous items.<sup>316, 317</sup>

# 3.4 Current consensus on the optimal approach

Currently, there is broad acceptance by those who specialise in measurement as a scientific discipline that a direct comparison of the techniques involved in CTT, IRT and RMT, such as FA and Rasch analysis, for the purpose of examining unidimensionality is not very useful. This is because these comparisons encourage researchers to ignore the theoretical differences underpinning the three approaches i.e. the conceptualisation of measurement, and instead search for a method which best describes their data. This approach is often termed the 'explain the data' paradigm, which is in contrast to

modifying the data or the measurement instrument to fit the philosophy of measurement.<sup>314</sup> Current consensus is that both IRT and RMT approaches provide more information than using CTT alone,<sup>312</sup> however, when selecting an approach, researchers should carefully address the aims and objectives of the research study, as well as the theoretical underpinnings of CTT, IRT and RMT and the inferences which can be made when using each technique.

The method used in the current development of vision-specific, age-appropriate PROMs suitable for YP is underpinned by a combination of both CTT and RMT, with the former being used as purely an indication of the feasibility of applying Rasch analysis in the later stages of evaluation to assess the psychometric properties of the instruments. The exact statistical techniques which are applied are discussed in more detail in Section 4.9.4 (pg. 114). In summary, FA will be performed first to give an indication as to whether unidimensionality exists. If unidimensionality does exist, Rasch analysis will be performed as a stronger psychometric evaluation of the instruments. If unidimensionality is not demonstrated using FA, items showing floor or ceiling effects will be iteratively removed with the aim of preparing the instrument for Rasch analysis. This approach is concordant with the theoretical underpinnings of modern test theory, and the iterative process in which measurement instruments such as PROMs are tailored for optimal use.

# Chapter 4 Method

# 4.1 Ethics approval

This research project was approved by the NHS Research Ethics Committee for Essex and East of England (ref: 12/EE/0455) and adhered to the tenets of the Declaration of Helsinki.

# 4.2 Phases of instrument development

The research was undertaken in 4 standard stages of instrument development (as outlined in Section 2.6.2, pg. 64) comprising in-depth interviews conducted with the aim of identifying the content of instrument items (phase 1), expert consultations to explore participants' understanding and interpretation of the draft instruments (phase 2), a small-scale postal survey to identify any immediate problems with the instruments (phase 3), and a large-scale postal survey to validate the final instruments (phase 4).

# 4.3 Eligibility criteria, patient identification, and recruitment

# 4.3.1 Eligibility criteria

Disorders leading to visual impairment during childhood can be classified using the following characteristics:

- Degree of visual impairment e.g. severity (VI, SVI, or Blind)<sup>178</sup>
- Timing of onset of impaired sight (early onset at ≤ 2 years of age, late onset at ≥ 3 years of age)
- Stability of visual deterioration (stable impairment or progressive)

The overarching aim of this study was to understand the *experience* of children and adolescents living with visual impairment, and quantify the impact of all-cause visual impairment using vision-specific PROMs. This is in contrast to disease-specific PROMs

which can be used to quantify the impact of a specific, or range of specific, ophthalmic disorder(s). Additionally, patients were required to be capable (developmentally, cognitively, and physically) of self-reporting on VQoL and FV. Therefore, patients were eligible if they met the following criteria:

a) Visually impaired, severely visually impaired, or blind (VA in the better-seeing eye logMAR equal to or worse than 0.5 or Snellen worse than 6/18)

b) No other significant sensory, learning or motor impairment which should impact upon ability to self-report

c) Aged between 13 and 18 years on date of recruitment.

Given the probability of some developmental delay in this population (see Section 2.5, pg. 41), a 'stage'- as opposed to 'age'-approach may be more suitable for children and YP living with long-term disability or impairment. Thus, precise age at the time of data collection was subject to discretionary flexibility. Flexibility was also deemed appropriate with regard to manifestation of visual impairment, and medical histories and nature of visual impairment were evaluated in each individual patient identified to determine suitability for inclusion in the study. Thus, some patients who had visual impairment classified as mild visual impairment (such as severely reduced visual fields) were invited to take part. This was with a view that YP living with visual impairment in the UK, and fulfilling the eligibility criteria, are a numerically small population and the developed instruments may be used flexibly within clinical contexts. Before participants were invited to take part in the research, the family GP was contacted to ask if there were any reasons why the research team should not contact the family (see Appendix II, pg. 348).

### 4.3.2 Sampling sources

The research was carried out in 4 standard phases of questionnaire development (see Section 2.6.2, pg. 64) and the potential participants were sampled across the phases from two main sources described below.

#### Source 1:

The Department of Ophthalmology at Great Ormond Street Hospital, and the Paediatric Glaucoma Service and Genetic Eye Disease Service at Moorfields Eye Hospital, London, UK were the primary source of eligible patients used across all research phases. Patient attendance lists containing the hospital numbers of all patients who had attended relevant paediatric ophthalmology clinics were scrutinised at the outset to identify patients as per the eligibility criteria outlined in Section 4.3.1 (pg. 78).

#### Source 2:

During the latter stages of the research project (phase 3 and phase 4), a number of external NHS Trusts were included in the research as patient identification centres (PICs). A total of 20 PICs were identified and recruited through the Paediatric Ophthalmology network and NIHR Clinical Local Research Networks in the UK (see Appendix I, pg. 347). The centres represented an even geographic spread of England, Ireland and Wales. Every centre was required to obtain local research governance approval prior to their participation as a PIC. The purpose of involvement of these centres in the identification and recruitment of eligible patients for phases 3 and 4, where the largest sample sizes were anticipated, was both to ensure a nationally representative sample as well as a sufficient sample size.

All consenting patients identified through these external centres were reported as NIHR patient accrual for the identifying PIC.

# 4.3.3 Sampling framework

One sampling framework was developed comprising patients from source 1.

Source 1: The clinical and demographic details of eligible patients were collated into a sampling framework for the study. The patient identification from source 1 was continuous throughout the project with the relevant details systematically updated using electronic health records from Great Ormond Street Hospital and Moorfields Eye Hospital, London, UK. Particular attention was paid to recording the outcome of the most recent assessment of VA. Demographic details such as date of birth, and contact details such as current/most recent address and contact telephone number were also recorded. Each entry in the framework was assigned a random number using the random number generator function in Excel, and the entire sampling framework was organised in order of random number (ranging from smallest to greatest). When organised in this way, each entry was assigned a participant study ID or 'PID', and the sampling framework was then re-organised in terms of PID (from smallest to largest). The PID was obtained throughout the course of the research project, and used to identify patients in documents such as consent and assent forms. This was compliant with data protection regulations and necessary to ensure no identifiable information was used when the patients and their families were discussed within the research team.

Source 2: In phases 3 and 4, lists of eligible patients identified at each PIC were collated by the collaborating local clinical team and kept on site as per research governance regulations.

Recruitment procedures across these phases are outlined in more detail in Sections 4.6.1, (pg. 83 (phase 1)), 4.7.1, (pg. 104 (phase 2)), 4.8.1, (pg. 107 (phase 3)), and 4.9.1, (pg. 111 (phase 4)).

## 4.4 Sample size

An anticipated participation rate of YP with visual impairment based on prior similar studies in which children and YP with visual impairment participated,<sup>5, 240</sup> of approximately 30% was applied to inform the number of patients invited to participate

in each stage of instrument development. To ensure that results were representative of the wider population of visually impaired adolescents, sample sizes in phases 1 and 2 were calculated based on the principles of data saturation during qualitative research methods.<sup>318</sup> Throughout the course of the research project and 4 phases, caution was taken to avoid inviting too many YP to take part in each individual phase in light of the numerically small population, moderate participation rates and with a view to conserve the majority of the sample for the final validation phases.

# 4.5 Participation analysis

A logistic regression model was fitted to examine the characteristics associated with participation in the 4-phases of instrument development using age, gender, severity of visual impairment, ethnicity, and index of multiple deprivation (IMD) as potential predictors. Severity of visual impairment was categorised based on the WHO classification (see Table 1, pg. 42) but included 'low vision' to account for patients who had VA better than logMAR 0.46 but with other manifestation of visual impairment (e.g. significantly reduced visual fields)(see Table 3, pg. 82). Univariable analyses were performed on each predictor variable individually. Variables which were shown to be significant (p < .05) were then entered into multivariable analyses.

Visual impairment category	Presenting corrected distance VA (logMAR) in the better eye
Low vision (LV)	Better than logMAR 0.48
Visual impairment 1 (VI1)	0.48 – 0.70
Visual impairment 2 (VI2)	0.72 – 1.00
Severe visual impairment (SVI)	1.02 – 1.30
Blind	Worse than logMAR 1.30

Table 3	Categories	used to	rank se	veritv of	f visual im	pairment.
	Categories	u300 i0	Tarik SC	venty of	visuarin	pannen.

## 4.6 Phase 1: Item development and adaptation

The aim of this phase was to explore the lived experience of visual impairment during childhood and adolescence and identify the issues relevant to YP aged >15 years with a view to develop age- and vision-specific questionnaire items suitable for inclusion in the measures of VQoL and FV. Concordant with the most recent, psychometrically robust, conceptual and theoretical development of vision-specific PROMs, a series of in-depth interviews was conducted with adolescents.

## 4.6.1 Participants and recruitment method

Participants were recruited for participation using source 1 (see Section 4.3.2, pg. 80). A required sample size of approximately 15-20 participants was calculated based on principles of data saturation in qualitative research.

Thus, 44 eligible patients were identified using a systematic random sampling approach. Participants aged 16 years or above were selected from the sampling framework based on the premise that interviews with YP aged 13-15 years had already been conducted during the 'foundation' research<sup>5, 6, 240</sup> and the interview transcripts from these interviews were already available. A stratified random sampling approach was used to ensure an even spread of patients were invited to take part in this phase with regard to age and gender. When ordered in terms of PID (ranging from smallest to largest), the first patient was reviewed in terms of age and gender for inclusion in the sample. The same method was used to identify subsequent patients, with each patient who fulfilled the pre-established sample characteristics for the phase (in terms of age and gender) recorded. If the patient in the sampling framework did not fulfil the sample requirements for the phase (e.g. violated the even spread of age and gender), the next patient was assessed for suitability, and so on.

Identified patients were sent invitation packs by post inviting them to take part only in this research phase. Each invitation pack was addressed to the parent of the young

person, and included a separate individually sealed letter addressed to the young person themselves. Both the parent and the young person invitation pack contained:

- a) an invitation letter
- b) a study information sheet
- c) a consent form
- d) a pre-paid envelope

Parent invitation packs also included a family background questionnaire to collect data about the family context including parent background, occupation, and details of the family structure e.g. single/two-parent family, and any other siblings (see Appendix IX, pg. 371).

All invited families were followed-up by telephone 2 weeks later. If patients were willing to be interviewed arrangements were made first over the phone with a parent or guardian, and then confirmed in writing. A reminder telephone call was made the day before each interview to ensure that each family was aware they would receive a visit from a researcher the following day.

The families of all patients whom the researcher had been unable to contact via telephone were sent a postal reminder 4 weeks following the initial invitation. No additional reminders were sent, in accordance with the ethics approval.

4.6.2 Identifying the issues relevant to YP with visual impairment

#### 4.6.2.1 Topic guide development

The objective of phase 1 was to identify the issues relevant to adolescents living with visual impairment with a view to inform the development of age- and vision-specific items for the VQoL and FV instruments. Thus a flexible topic guide for interviewing YP to capture their needs and concerns was developed. This drew on a systematic review of the literature with the research question 'What factors contribute to the broader impact of living with a visual impairment during adolescence?' This review was

supplemented by consultations with professionals working within paediatric ophthalmology.

### 4.6.2.1.1 Identification of literature

A literature review was conducted using the matrix method to collect and organise findings.<sup>319</sup> PubMed, PsycINFO and EMBASE databases were searched systematically with the main concepts being a) vision, visual impairment or eye disease, b) adolescents/young adults and c) Quality of Life, daily life activity or life satisfaction (see Appendix X, pg. 375 for the search strategy and MeSH terms). The following limits were applied to each search:

- Articles available in English language only
- Humans only

In addition to the main concepts, the limit 'age group' (Child: 6-12 years, Adolescent: 13-18 years, and Young Adult: 19-24 years) was applied in PubMed, and 'human age groups' (School Child <7 to 12 years>, and Adolescent <13 to 17 years>) in EMBASE. This is in keeping with the WHO definition of adolescence as the period between the ages of 10-19 years (see Section 2.1, pg. 20). No age-related limits were applied in PsycINFO due to the small number of results.

The title and abstract of each reference was reviewed for relevance to YP living with visual impairment and relevant citations were exported to EndNote citation manager. Duplicates were removed.

The search was carried out in March 2015.

### 4.6.2.1.2 Screening of literature

The full text of 284 articles was screened to assess the empirical value of the results. Three systematic reviews were excluded from the search (2 because findings were not applicable to the topic guide<sup>257, 320</sup> and 1 because the author failed to identify any relevant papers<sup>321</sup>) and 3 editorials were excluded.<sup>322-324</sup> One study protocol was identified and excluded.<sup>325</sup> A grey literature search was conducted using the reference lists of all remaining eligible articles, including the excluded systematic reviews, editorials and study protocol. Thirteen articles were identified through grey literature, and 5 additional articles referenced in the study protocol were added.

#### 4.6.2.1.3 Eligibility of literature

The aim of the systematic review was to identify, as broadly as possible, factors related to QoL and everyday life of adolescents living with visual impairment. Thus, the review focused upon literature which is both vision- and age-specific. This is in line with using the topic guide for interviewing YP aged 10-15 years<sup>5, 6</sup> as a foundation for developing new questions and probes for YP aged older than 15 years. Despite including adolescents/young adults as a main concept in the review (see Appendix X, pg. 375), and using the search limits to focus upon age-range, a substantial number of articles were removed during the eligibility stage because they did not focus specifically upon individuals within the specified age range of 10-19 years. The majority of these included adolescents or YP who were classified as 'adults' and placed within the age range of 18 years +. In these cases, attention was paid to any analyses between age-groups. Those which did not discriminate between age-groups were excluded. After assessing eligibility in terms of the age of participants, three articles were excluded because participants did not have a visual impairment.

During the second stage of eligibility assessment, attention was paid to identifying papers which used qualitative techniques, such as in-depth interviews or focus groups, which yielded data based on self-report. Some papers used measurement instruments which provided quantitative summary scores indicating overall QoL, or subscales of QoL, which were used to compare groups of YP with varying demographic characteristics (e.g. age, gender, severity of visual impairment). In this case, each article was assessed in terms of the contribution it made towards distinguishing specific aspects of VQoL or FV which could be probed during interviews with YP. Since probing

QoL itself (e.g. "Tell me about your QoL") would likely be confusing for YP, 29 were excluded as shown in Figure 1 (pg. 88).

Articles were assessed in terms of the relevance of findings to VQoL and FV as outcomes, aspects of everyday life for YP living with visual impairment, and suitability for contributing to questions and probes in the topic guide. As a result of being unrelated to these factors, 15 articles were excluded. A further 14 articles were excluded because participants had co-morbidities which would have likely impacted upon their perspectives of QoL and experience of everyday life.

A number of articles documenting the development of instruments designed to measure QoL in varying populations were identified. In these cases, attention was paid to any articles which focused upon the content of the items included in the instruments. Those which focused purely upon the psychometric evaluation of these instruments were excluded.

Formal quality assessment of the remaining articles was not deemed appropriate. This was because the aim of the review was to identify, as broadly as possible, all the issues potentially relevant to the lives of adolescents with visual impairment in order to inform the content of the topic guide. Since the content of individual interviews was intended to be largely patient-driven, the purpose of the systematic review was to guide the interviews and ensure all relevant areas were covered, rather than to develop an evidence base to apply the findings e.g. inform policy and decision-making per se. Nevertheless, quality of the identified papers was assessed in part by judging the sample size and outcomes. One article was excluded due to methodological weaknesses indicating low quality.

Due to the inclusive nature of the systematic review, no further exclusion criteria were applied.

### Figure 1. PRISMA flowchart showing systematic review.



#### 4.6.2.1.4 Synthesis of findings and application to the topic guide

The articles were categorized according to VQoL or FV domain using the six domains included in the topic guide used to develop PROMs for children aged 10-15 years<sup>5</sup> as a foundation (see Table 4, pg. 91). Attention was paid to identifying any new domains, and modifying the content of existing domains to make the topic guide sensitive to the attitudes, preferences, and everyday lifestyles of YP with visual impairment aged older than 15 years. As a result, the following 6 domains were further developed: home life, education (school/college/university), leisure life, independence, social life, and future – aspirations and fears. The domain 'clinical care' was added. Based on the overarching aim of the research project, that is, exploring the experience of growing up with visual impairment, adopting a life course perspective which encourages retrospective reflection on experience during childhood, a battery of probes was developed to encourage participants to reflect upon the differences between their attitudes, beliefs and lifestyles at the time of the interview and those during early childhood.

Several articles contributed to the development of questions or probes within multiple domains. These articles provided descriptive accounts of the everyday challenges and experiences of YP living with visual impairment.

Literature relating to functional difficulties during adolescence was initially included as a separate domain, but later incorporated into the topic guide in the form of probes which were used within each domain. This ensured participants reflected upon their FV in a number of different contexts e.g. home, school, leisure etc. Similarly, literature relating to psychological and emotional difficulties during adolescence was incorporated into the 6 domains of the topic guide as probes (e.g. How does that make you feel?) rather than as a distinct domain. Literature contributing to the topic guide is shown in Table 4 (pg. 91).

As a result of the systematic literature review, several changes were made to the foundation topic guide that had been developed for YP aged 10-15 years. In the foundation topic guide, only one question about social life (Where do you like to

go/what do you like to do with your friends?) was included under the umbrella term of 'activities and socialising'. However, due to a substantial amount of literature demonstrating age-specific difficulties in social communication in YP with visual impairment, a number of new questions were developed and the domain 'social life' was separated from 'leisure life'. New probes touched upon aspects such as friendship groups/dynamics and feelings of exclusion or fitting in. Development of romantic relationships was also probed sensitively.

The majority of identified literature supported the core questions included in each domain of the existing topic guide, but led to the development of a number of new probes. Most pertinent additions were related to experiences of vision-specialist education, attitudes towards teaching assistants, and experience of transitions within healthcare. Probes about exercise were added with the aim of providing participants a platform in which to discuss psychological aspects of living with visual impairment, such as self-image and self-esteem. Table 4. Domains of QoL, literature which supports the inclusion of each domain in the topic guide, and modifications which were made to the topic guide in response to each paper identified.

Domain and supporting	Potential contributors to QoL	Existing questions (bold) and	New age-appropriate questions
references	during adolescence	probes taken from the foundation	(bold) and probes
		topic guide	
Home life <sup>5, 6, 240, 268, 280, 326-340</sup>	Relationships with parents, relationships with other family	Who do you live with? Do you have any siblings?	Do you feel like your parents/siblings/other family members
	members (siblings and extended	What do you like to do with your	understand what it's like for you?
	family), practical and emotional	siblings?	Do any of your parents/siblings/other
	support from family members, over-	What is your relationship like with	family members also have a visual
	protection from family members,	your parents and siblings?	impairment?
	functional ability when completing	Can you tell me about the sorts of	Do you have to do any kind of chores
	domestic tasks e.g. feeding a pet	things you do when you're at	when you're at home?
	dog, taking care of personal	home?	
	hygiene, chores, mobility at home,	What did you do when you were at	
	privacy from parents, me-time,	home yesterday/last weekend?	
	family members understanding,	Did you enjoy doing that?	
	having a role-model in the family,	Is there anything in particular you	
	activities at home (watching TV,	find difficult doing at home	
	playing games, using the computer,	because of your eyesight?	
	doing homework, listening to music,	Did you find that difficult?	
	reading for fun).	Do you have any pets?	
		Do you need help getting ready in	
		the morning?	
		How about finding your way around	
		the house?	

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
		Home life continued Do you feel like you get enough privacy/me-time when you're at home?	
Education (school/college/ university) <sup>5, 6, 213, 240, 268, 335, 336, 339, 341-345</sup>	Likes and dislikes, participation in different lessons, barriers to participation in different subjects, choosing which subjects to take, exams, social life at school, peer support, bullying and teasing at school, relationships with teachers, teaching assistants and specialist education (assistive devices/technology, classes, specialist schools), having to miss school because of hospital visits, teacher knowledge, understanding and support, getting around at school, inclusion/exclusion, keeping up with others and workload, getting around school/mobility, getting to and from school, asking for help, changing school/classes/teachers.	<ul> <li>What sort of things do you enjoy about school?</li> <li>Can you describe to me what you did at school yesterday?</li> <li>What are your favourite lessons/teachers?</li> <li>What are the classrooms like?</li> <li>Who do you spend most time with at school? What do you like to do together?</li> <li>What things do you not like about school?</li> <li>Do you feel your eyesight gets in the way of things you do at school?</li> <li>Is there anything that you would really like to do at school, but can't because of your eyesight?</li> </ul>	Do you go to school or college or university at the moment? What kind of school is it (mainstream/specialist)? Is there a visual impairment unit? Do you travel to school by yourself? Do you have to do many exams at school? Do you have a teaching assistant? Do you like having a teaching assistant? Do you use any special devices or technology at school because of your eyesight? Do you feel like your teachers understand what it's like to have a visual impairment? Have you ever had to change schools/teachers/classes? Were there any differences?

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
			Education (school/college/university) continued Do you think your life at school has changed as you've got older?
Leisure life <sup>5, 6, 212, 240, 268, 327, 330, 333, 335-337, 339, 340, 342, 345-349</sup>	Likes and dislike, physical activity/exercise, meeting friends outside of school, doing sports, walking, running, getting around outdoors, travelling, shopping, visiting restaurants/coffee shops, getting around new places, commitment to leisure activities, adjusting to new environments and activities, extra-curricular activities, playing musical instruments.	What do you like to do in your free time? Where do you go? What activities do you enjoy? Do you ever go shopping? Are there any activities that you would like to do but can't because of your eyesight? What would be the main way you feel your eyesight affects what you can and can't do in your free- time? How about travelling places?	What do you enjoy about []? Do you enjoy exercise? How often do you exercise and where do you do it? Do you use any vision devices/technology to help when you do things outside the house? How about musical instruments?
Independence <sup>5, 6, 212, 240, 268, 327, 336-340, 342, 349</sup>	Responsibilities at home, responsibilities at school, basic economic transactions (e.g. using cash machines and handling money), using public transport when alone,	Apart from chores, do you have any particular responsibilities when you're at home? Do you ever do that by yourself? Do you ever go outside on your own?	Do you ever go outside in the dark? Have you ever had any mobility training? Do you think you can manage your own time well?

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
	Independence continued getting around outdoors when daylight/night time, using VI- assistive devices such as canes, independence from parents, restraints on physical activity e.g. walking in public, perceived reliance on parents, mobility training, managing time independently, travelling, barriers and wishes, writing and sending applications, work experience.	Independence continued Who comes with you when you go shopping/on the bus/on the train etc.? Do you like them to come with you or would you prefer to be able to go on your own? How do you feel aboutcoming with you? <b>Do you ever travel by yourself?</b> Are there any things you feel you can't do because?	Independence continued Do you think your independence has changed as you've got older?
Social life <sup>6, 240, 268, 280, 327-329, 331, 333- 337, 339, 341, 345, 347, 350-363</sup>	Best friends, friends, peers, enemies, initiating and maintaining friendships, self-esteem, developing romantic relationships, dating, barriers to developing romantic relationships, falling in love, eye contact, visual cues, missing body gestures, facial expressions, social support from friends and peers, making friends despite difficulties e.g. parental over-protection,	Where do you like to go/what do you like to do with your friends?	Do you have a best friend or group of best friends? Do they go to the same college as you? How did you meet? Are all your friends girls/boys? Do any of your friends have a visual impairment? What are the like? Can you tell me a bit more about him/her? How/when did you meet?

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
	Social life continued arguments with friends, meeting friends outside of school, prejudice from others (friends and members of the public), public perceptions, overprotection from peers, shyness in social situations, other people understanding, other people helping, recognising friends in public, inclusion in social activities, social networks, not telling other people about visual impairment and functional difficulties, reaction of others.		Social life continued What do you like to do together? Would you like to have a boy/girlfriend? Is there anything you find difficult to do with your friends because of your eyesight? Is there anything you can't do because of your eyesight? Do you spend more time with your friends or family? Which do you prefer? Do you feel like your friends understand what it's like for you? Do you feel like your friendships have changed at all as you've got older? What about when you changed school?
Future – aspirations and fears <sup>5, 6,</sup> 240, 336, 339, 340, 345, 362	Aspirations, preferences, desires, worries and concerns, fears, future education (e.g. going to University, exams, choosing courses), future physical health, future visual function, future employment, making plans for the future, barriers in the future,	What do you think you might do after you've done your GCSEs/finished school/college/University? Have you thought about what kind of job you'd like to do in the future?	Have you thought about moving away from home in the future? How do you feel about that? What do you think it will be like to live by yourself/without your parents in the future?

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
	Future – aspirations and fears continued future independence and autonomy, future family life, finding a romantic partner and building a family, future living environment and location, moving away from home/parents, fear of falling short of expectations of others.	<ul> <li>Future – aspirations and fears continued</li> <li>Do you have any worries/concerns about the future?</li> <li>What are you most worried about?</li> <li>Have you thought of any things you would like to do in the future, but might not be able to do</li> <li>because of your eyesight?</li> <li>Do you have any concerns about your vision in the future?</li> <li>Have you ever thought about having a family in the future?</li> </ul>	
Clinical care <sup>5</sup>	Going to eye clinics, hospital visits, likes and dislikes, preferences.	Do you still regularly attend clinics? Do you like/not like going? Why?	Where/which hospital do you go to? How often do you have to go? Have you changed to adult clinics now or do you still go to the children's ones? Have you ever had to change consultants? How did you find that process? Would you have done it differently?

Domain and supporting references	Potential contributors to QoL during adolescence	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions (bold) and probes
Functional vision <sup>6, 240, 277, 337, 340, 351, 364, 365</sup>	Balance, ocular discomfort (e.g. eye redness, photophobia), pain, environmental factors (e.g. sunshine) and impact upon FV, sleeping, daytime napping.	Do you have any difficulties doing that? Is that easy for you? How does your eyesight affect that? How do you find that?	-
Psychological and emotional well- being <sup>240, 268, 280, 327-329, 333, 335, 337, 339, 345, 351, 353, 358, 360, 362, 363, 366-372</sup>	Self-concept, self-image, self- esteem, anxiety, social anxiety, school anxiety, hostility, worry and concern, identity development, emotions (nervousness, loneliness, desires, frustrations, anger, depression, happiness, embarrassment), hyperactivity/inattention, acceptance of visual impairment, psychological adjustment to visual impairment, resilience, fulfilling ambitions/doing what you want to do, confidence, self-consciousness, (dis)satisfaction with self-appearance and body image, learning that visual impairment will never go away,	How do you feel about that? How does that make you feel? What do you prefer? Do you enjoy that? Does that bother you? Why do you feel like that? Is there anything else that's really important to you that I haven't asked you about?	How important do you think that is? Why do you think that's important? Have you always felt that way?

Domain and supporting	Potential contributors to QoL	Existing questions (bold) and probes taken from the foundation topic guide	New age-appropriate questions
references	during adolescence		(bold) and probes
	Psychological and emotional well- being continued learning that visual impairment is an evolving process, desires to be different/normal/the same as others, coping strategies, motivation to learn new strategies, feeling different, feelings of freedom and independence.		

### 4.6.3 Data collection: In-depth semi-structured interviews

In-depth semi-structured interviews were conducted between March and June 2015 by one interviewer (AR). The developed topic guide was used flexibly to explore many areas of everyday life. After re-confirmation of consent, an 'ice-breaker' activity preceded each interview. During interviews, participants were encouraged to speak to the interviewer independently of their parents or family members. Due to the nature of interviews, which took place at participants' family homes, this was not always possible, and parents and/or siblings were present during a number of interviews. Generic questions about specific aspects of everyday life (i.e. home, school, free time) were discussed before the impact of visual impairment was probed specifically. Opportunity was given for the YP to lead in opening the topic guide and raise any relevant issues independently. This approach encouraged participants to relax, and feel as though they were engaged in an informal 'chat' as opposed to an interview. To address the overarching aim of the research, that is, to contribute to the current understanding of the experience of growing up with visual impairment, participants were asked within each domain of the topic guide to reflect upon their experiences during childhood and recall any pertinent memories of times when they experienced specific difficulties, challenges, or changes. Open-ended probes were used consistently throughout interviews to encourage participants to take the lead during discussions and talk about things which they felt were most important.

### 4.6.4 Qualitative data analysis

Each interview was digitally recorded, transcribed and imported into NVivo 10. Qualitative data analysis was then conducted in two separate stages with different purposes. The first stage was conducted with the aim of exploring the voices of visually impaired adolescents in depth with reference to the lived experience of visual impairment throughout childhood and adolescence; the second stage was conducted with the purpose of applying the qualitative data to generate meaningful and ageappropriate instrument items.

4.6.4.1 First stage data analysis (exploring the impact of visual impairment during childhood and adolescence)

Qualitative analysis based on the tenets of Grounded Theory<sup>373</sup> was conducted to identify key themes related to participants' experience of growing up with visual impairment. This approach was selected based on source data stemming from spontaneous speech in addition to answers which were further probed. Thus, both inductive and deductive methods were incorporated.

Data analysis comprised an ongoing, circular process of open-coding.<sup>374, 375</sup> This was completed independently by two members of the research team (AR and VT) with a view to ensure reliability of the coding process. The first stage of coding comprised both members of the research team coding a third of the transcribed data with a view to identify codes corresponding to experiences and any pertinent transitions and changes during childhood and adolescence. Coding was completed manually by one researcher and by the other using NVivo software. After the first stage of coding, the two researchers met to compare notes and develop an inventory of codes. An initial codebook was used by both researchers to code a further third of the data. Any new codes which had not already been identified were added to the codebook after discussion about their meaning and value. Definitions of codes were refined, and changes to the codebook were tracked.

A final codebook was developed by adding description, examples, and criteria of use and then used by both researchers to code the entire dataset. Ongoing meetings and communication between researchers ensured rigorous and consistent use of the codebook. Using NVivo, the data were organised into individual codes using a reference sheet detailing clinical and demographic details of each participant to allow for comparisons to be made between participants (e.g. in terms of gender, severity of visual impairment, and timing of onset of visual impairment). Finally, the themes emerging from the data by combining the codes were identified and labelled.

### 4.6.4.2 Second stage data analysis (item generation)

Transcriptions were exported into NVivo 10 and organised into seven nodes reflecting the seven topics included in the topic guide. Each node was searched for data relating to the six higher order themes outlined previously by Rahi et al.<sup>5</sup> with 10-15 year olds, namely: 1) social relationships, acceptance and participation; 2) independence and autonomy; 3) psychological and emotional well-being; 4) future – aspirations and fears; 5) functioning – school, home and leisure; and 6) treatment of eye condition. Using these domains as a starting point for item generation was deemed appropriate due to applicability of the domains to existing instrument versions of VQoL and FV for 10-15 year olds, and with the new age-appropriate instrument versions for older adolescents being an adaptation and extension of these existing instruments. Once the data were organised, each theme was re-read and open-coded.<sup>374, 376</sup>

Each item incorporated into the existing instruments (VQoL and FV) during the foundation research is expressed as a statement with a Likert-type response scale comprising four responses. In the existing VQoL instrument, items are phrased as statements and responses refer to users' endorsement of the statement (i.e. whether, and to what extent, it is true about them). In the FV instrument, items are phrased as partial statements, and users are required to indicate their functional visual ability by choosing the suffix which best refers to the perceived difficulty of completing a task. The design and format of VQoL items is empirically grounded based on consideration of using an 'illustrative' child's perspective during the foundation research.<sup>5</sup> Prior to further analysis, all items developed in the foundation research were modified to remove the 'illustrative' child's perspective, which was shown to lack feasibility when used with children and YP, producing significant ceiling effects<sup>5</sup> (see Table 5, pg. 102 for an example of the modified presentation of VQoL and FV items).

Table 5. Example VQoL and FV item developed in the foundation research<sup>5</sup> and

modified version in the current development.

Example VQoL item and response options used in the foundation research	Example FV item and response options used in the foundation research
Ben has got some good friends. How much are you like Ben?	How easy do you findwatching TV
'Actual' response options:	Response options:
1. Not at all	1. Very easy
2. A little bit	2. Easy
3. Quite a lot	3. Difficult
4. Exactly	4. Very difficult or impossible
How much do you want to be like Ben?	5. This doesn't apply to me/I don't do
'Ideal' response options:	this for other reasons
1. Not at all	
2. A little bit	
3. Quite a lot	
4. Exactly	
Example VQoL item and response options modified for the current development	Example FV item and response options modified for the current development
I have got some good friends.	Because of my eyesight, I find watching TV
Response options:	Response options:
1. Not at all true	1. Very easy
2. A little bit true	2. Easy
3. Mostly true	3. Difficult
4. Completely true	4. Very difficult or impossible

Open-codes were mapped onto items included in the existing VQoL and FV measures developed for 10-15 year olds in the foundational research<sup>5-7</sup> which were then entered into an Excel spreadsheet. To ensure comprehensive coverage of the VQoL and FV

issues discussed by YP, codes were also mapped onto items which were developed during the early phases of the foundation research but subsequently excluded. Issues relevant to VQoL (e.g. psychosocial issues) were collated for the VQoL instrument and the items relevant to function, activities and difficulties completing them were compiled for the FV instrument. Issues which did not map onto items included in the development of existing VQoL and FV instruments were considered as new items reflecting age-specific concerns. VQoL and FV instrument versions applicable to adolescents older than 15 years were drafted.

Following this, an expert consensus meeting, including 4 members of the study research group, took place with the aims of: a) gathering feedback from the study research group as to the relevance/suitability of the items included in the new VQoL and FV age-appropriate versions, b) discussing possible age-appropriate modifications of these existing items and c) discussing the need for, and development of new age-appropriate items for each instrument. Prior to this meeting, each member of the study research group was sent a copy of each instrument and asked to reflect on suitability of items independently. Each draft item was then discussed in terms of relevance to the patient population, wording and presentation order.

During the expert consensus meeting, the age boundaries for the older (and younger) instrument versions were discussed and a decision was made to modify these according to the data collected i.e. a data-driven process. This decision was largely based on the emergence of independent living at the age of 18 years. Some participants aged 18 and 19 years old at the time of participation in interviews had begun attending University, and thus were living away from home. The perspectives and daily lives of these individuals were not comparable with those of younger participants. As a result, the data collected during the foundation research were revisited with a view of determining the correct age boundaries for the new older-age instrument versions. The age-range of 13-17 years was deemed the most appropriate based on the emergence of independent activities away from home, less parental

restriction, and increased awareness about the future at the age of 13 years. In parallel, the age boundary for the younger instrument versions was modified from 6-9 years to 8-12 years.

Following the expert consensus meeting, the drafts of the age-appropriate versions of both the VQoL and FV instruments suitable for older adolescents were finalised. A master item pool was developed which included all items in the first draft instruments and was used to track which stage the items had been developed (e.g. in the foundation research or as a result of the interviews with older adolescents).

# 4.7 Phase 2: Pre-testing

The second phase of instrument development was conducted to operationalise the draft age-appropriate instrument versions developed in phase 1. Specifically, individual cognitive interviews with YP with visual impairment were conducted to identify a) the importance and relevance of items, b) YP's understanding and interpretation of the instructions, response options and items, c) any ambiguity they experienced in the completion of the instruments and d) YP's preferences regarding the wording of the items and any suggestions for re-phrasing.

# 4.7.1 Participants and recruitment method

As described above, the expert consensus meeting led to a decision to operationalise the age-appropriate instrument versions of VQoL and FV instruments for adolescents aged 13-17 years.

Patients were identified, selected and recruited for participation in phase 2 using the same random, stratified sampling technique as in phase 1 (and outlined in Section 4.6.1, pg. 83).

A total of 68 eligible patients aged 13-17 years were invited to take part in phase 2 of the instrument development. They were sent invitation packs containing the relevant recruitment materials as specified in Section 4.6.1 (pg. 83) and shown in Appendix II-IX (pg. 348-371).

The recruitment materials reflected the broader age range of participants invited in this phase, compared to those who were invited in phase 1. Parents/guardians of all older patients (i.e. those aged 16-17 years) were sent invitation packs which were addressed to the parent/guardian of each young person, but included a separate invitation letter addressing the young person. Parents and guardians of younger patients (i.e. aged 13-15 years) were sent invitation packs containing only one invitation letter (addressed to the parent/guardian) and an information sheet, assent, and consent forms suitable for children and parents respectively. The follow-up procedure described in Section 4.6.1 (pg. 83) was used.

## 4.7.2 Data collection: Cognitive interviews

Individual cognitive interviews were conducted between September 2015 and January 2016 by one interviewer (AR). Similar to the data collection procedure outlined in Section 4.6.3 (pg. 99), YP were encouraged to feedback on the individual instruments and items independently, but due to the nature of interviews which took place at participants' family homes this was not always possible.

The new age-appropriate versions of the draft VQoL and FV instruments were used, alongside a standard list of questions used to gauge participants' perceived importance, relevance, and comprehensibility of the items (see Appendix XI, pg. 376). Prior to interviews, consent was re-confirmed and the researcher made clear the intention of the interview e.g. to gather feedback on individual questionnaire items as opposed to gathering feedback about participants' VQoL or FV as outcome values. The presentation of VQoL and FV instruments was rotated with each interview starting with a different instrument. Interviews began with the researcher reading through the instruction page of an instrument. Participants were asked about their understanding of the instructions, and any recommendations they may have on how to make the instructions easier to understand. The researcher then read aloud the four response options, including their numeric value and label. Each individual item was presented to each participant, read aloud by the researcher, and participants were asked to indicate which response they deemed appropriate for themselves, why they had chosen a particular value, and suitability of the other response options. Any ambiguity that participants noted in relation to the instrument items, instructions, or response options was probed and responses were noted. The researcher recorded, using pen and paper, participants' responses to each item, and any additional feedback or comments. The procedure was repeated for each instrument.

# 4.7.3 Data analysis

Data entry was ongoing throughout the course of phase 2. After each interview, the researcher entered participants' responses, including any qualitative verbal feedback, into an Excel spreadsheet which was used to collect and compare participants' feedback. Any difficulties participants had when giving a firm answer to any of the items and reasons why this was difficult were recorded. Participants' recommendations for ways to clarify the instructions or response options, and preferred wording of items, were noted in Excel.

A results document was created using Microsoft Word, showing each item and a summary of participants' responses to the item. All items were evaluated by one researcher (AR) and any action (e.g. change or amendment of an item) which was thought to be necessary was recorded in the same document.

A second expert consensus meeting including three members of the research team was carried out to evaluate the action needed for each item. Consistent with the procedure used in Phase 1, all attendees were asked to reflect on the draft instruments independently before the meeting took place. Times of disagreement between researchers were discussed during the meeting before a mutual decision was reached, taking into account the opinions of all members of the team.

The items included in the second draft of each instrument were entered into the master item pool as the second draft of both the VQoL and FV instrument versions for 13-17 year olds.

## 4.8 Phase 3: Piloting

The aim of phase 3 was to pilot the draft instruments with a small sample of participants in order to identify any immediate difficulties with the instruments or any of the items, such as ambiguity, false interpretation, patterns of missing data and distribution of responses (e.g. ceiling and floor effects). Thus, a small sample of approximately 15-20 participants was deemed adequate, with respect to conserving the majority of the remaining sampling framework for use during the formal psychometric evaluation phase (phase 4) where the largest sample size was required.

## 4.8.1 Participants and recruitment method

Participants were identified, selected and recruited for participation in phase 3 of the instrument development using the same random, stratified sampling technique as in previous phases (see Section 4.6.1, pg. 83). At this stage, recruitment of participants from source 1 was supplemented by recruitment from source 2 (see Section 4.3.2, pg. 80). PICs were contacted to confirm an estimated number of patients they were able to identify who fulfilled the eligibility criteria (see Section 4.3.1, pg. 78). Invitation packs were then sent to PICs, along with a recruitment log in which to record the date of invitation of individual patients. On-going communication between the research team and PICs ensured PICs were also sent reminder invitation packs, which were distributed to invited patients 2 weeks following initial invitation, as well as 'Thank You' letters which were distributed after a patient had participated.

Sixty-seven eligible patients from source 1 were invited to take part in phase 3 of the instrument development. An additional 55 eligible patients from source 2 were invited to take part. Identified patients were sent invitation packs by post inviting them to take

part in phase 3 of the instrument development. The invitation packs for children aged 13-15 years differed from those for YP aged 16-17 years. Each invitation pack was addressed to the parent of the young person and contained:

- a) a parent invitation letter
- b) a parent study information sheet
- c) a parent consent form
- d) a family background questionnaire
- e) a young person study information sheet
- f) a young person assent form

Invitation packs sent directly to YP aged 16-17 years included an invitation letter addressing the young person, consent form and information sheet designed specifically for older participants (see Appendix VI-VIII, pg. 363-366, 369). The parents/guardians of participants aged 16-17 years received an invitation letter, information sheet, consent form and family background questionnaire as described in Section 4.6.1 (pg. 83) and shown in Appendix III-V, IX (pg. 354-358, 361, 371).

The second draft of the new age-appropriate instrument versions (VQoL and FV) were printed in the format of an A4 questionnaire booklet in large print (N18). These were printed separately for the young person and the parent and labelled accordingly. Items in parent booklets were identical to those in the young person booklet, but were re-phrased to encourage parents to reflect upon their child's vision-specific outcomes by adding the prefix '*My child*' before each item in the VQoL instrument and '*Because of his/her eyesight my child finds*' before each item in the FV instrument (see Table 6, pg. 109). Both parent and young person booklets were administered with a view that parents' perspectives are often independent from those of their child (see Section 2.3.2, pg. 34), and to encourage independent self-report from YP. Young person booklets also included a number of simple questions at the end to collect data about ease of completion, understanding of the items, and time taken to complete each questionnaire
(see Appendix XII, pg. 378). Each invitation pack included both a parent and young person questionnaire booklet and a large pre-paid envelope for return of the completed questionnaire booklet, parent consent form, family background questionnaire, and young person assent form. Families of patients who were aged 16-17 were sent two pre-paid envelopes, to encourage YP to participate independently of their parent/guardian.

All invited families were followed-up by telephone 2 weeks later with the aim of answering any questions they had. The families of all patients for whom the researcher had been unable to contact via telephone were sent a single postal reminder 4 weeks following the initial invitation.

YP-VQoL item	Parent-VQoL item
I have got some good friends.	My child has got some good friends.
YP-FV item	Parent-FV item
Because of my eyesight, I find watching TV	Because of his/her eyesight my child finds watching TV

Table 6. Example of item adaptation for Parent booklets.

# 4.8.2 Data collection: Postal survey

Data collection and entry were ongoing throughout the course of phase 3. Upon receiving a completed questionnaire booklet, and consent forms, the researcher documented important clinical and demographic details into both an Excel spreadsheet and SPSS data set. All quantitative data were entered into each spreadsheet at two different time points to enable identification of errors in data entry (no errors were identified). Any qualitative annotations made to the questionnaire booklet were noted in the Excel spreadsheet. Overall qualitative feedback and answers to the feasibility questions were recorded in a separate document in Excel and SPSS.

#### 4.8.3 Data analysis

Exploratory data analysis used both graphical and non-graphical methods. Feasibility items included at the end of each questionnaire booklet were analysed using simple descriptive statistics (Median, interquartile range (IQR), and percentages) calculated in SPSS. The following descriptive statistics were calculated for each item:

- a) percent of missing data (per person and per item)
- b) z-score for skewness
- c) z-score for kurtosis

Bar graphs were created in SPSS to view the frequency of participants' responses to each item. Qualitative feedback given in response to individual items was assessed as complementary to quantitative results. Each item was screened individually. Items were flagged as problematic if they had > 20% missing data, z-score for skewness and zscore for kurtosis values outside the acceptable limits (-2.00 to +2.00)<sup>313</sup> or a response category(ies) which were not endorsed by any participants, or alternatively by all participants (i.e. floor or ceiling effects). Participants with > 25% missing responses were excluded based on criteria used in the foundation research.<sup>6</sup>

A third expert consensus meeting took place following the procedure outlined in Section 4.7.3 (pg. 106) in which three members of the research team met to discuss the findings. Decisions to amend, or remove items in light of the feedback obtained were made with reference to the master item pool, at which stage the items were developed (e.g. during the foundation or current development), and feedback given in phase 2 of the instrument development. As a result of this phase, items were modified and entered into the master item pool for use in phase 4.

## 4.9 Phase 4: Formal Validation

The final phase of instrument development aimed to assess the psychometric properties of the new age-appropriate versions of the VQoL and FV instruments. This

was done through a postal administration of the final draft instruments to a larger, nationally representative sample of YP living with visual impairment.

An intended important feature of the developed instrument suite was that the child- (8-12 year olds) and YP- (13-17 year olds) instruments could be used sequentially within ophthalmic practice to detect longitudinal changes in self-reported outcomes. Thus, at the formal validation stage, both the younger (child) and older (YP) versions were simultaneously piloted with both datasets being managed (separately) by AR. To this end, the younger instrument development was conducted by a different member of the research team (VT).

#### 4.9.1 Participants and recruitment method

Children (aged 8-12 years) and YP (aged 13-17 years) were identified, selected and recruited for participation in phase 4 of the instrument development using both primary and secondary recruitment sources (see Section 4.3.2, pg. 80). Due to the minimum sample size needed for formal psychometric evaluation, including Rasch analysis (see Section 3.3.2, pg. 75), all remaining patients in the sampling framework from source 1 (n = 194) were considered for participation, as well as any patients identified from source 2.

All eligible patients were invited to take part in phase 4 only and as per the same recruitment procedure as in earlier phases.

The questionnaire booklets administered in this phase included the final ageappropriate versions of the two instruments alongside the Pediatric Quality of Life Inventory – Child and Teen versions (PedsQL)<sup>377</sup> (see Appendix XIII, pg. 379 and XIV, pg. 381) which was used to assess construct validity of the VQoL instrument (i.e. whether the VQoL instrument measures what it is intended to measure). A brief, 4-item survey regarding the feasibility of completing the instruments within clinical contexts, as well as participants' preferred method of administration was included at the end of each booklet (results are analysed in a separate research project) (see Appendix XV, pg. 383). Two formats of questionnaire booklet were developed whereby the order of presentation of VQoL and FV was counterbalanced. PedsQL and feedback questions were included at the end. As in phase 3, booklets were also developed for parents to complete independently, and to encourage children/YP to independently self-report. Parent booklets included VQoL and FV questionnaire items which were modified in the same way as in phase 3 (see Section 4.8.1, pg. 107) and presented in the same counterbalanced order as for YP, and Child and Teen – versions of the PedsQL Parent-Proxy report.<sup>377</sup>

Electronic versions of Child, YP and parent booklets were developed by a member of the wider research group (MC) as part of a separate research project exploring the feasibility and design of electronic health records in Ophthalmology. The web-link and instructions for completing the online instruments were provided through the invitation letters sent to participants and their parents/guardians.

Each invitation pack included both a parent and child/YP questionnaire booklet and a large pre-paid envelope for returning the completed questionnaire booklet, parent consent form, family background questionnaire, and child/YP assent form. As in phase 3, families of patients who were aged 16-17 were sent two pre-paid envelopes.

All invited families were followed-up by telephone 2 weeks later with the aim of addressing any queries or concerns they had. The families of all patients for whom the researcher had been unable to contact via telephone were sent a single postal reminder 4 weeks following the initial invitation.

## 4.9.2 Data collection: Postal survey

Data collection and entry were ongoing throughout the course of phase 4. Excel was used to document all data coming from the questionnaire booklets: that is, quantitative responses to individual items and any qualitative annotations which were made on the questionnaire booklets, overall qualitative feedback to the instruments, answers to the ease and feasibility questions, and participants' clinical and demographic details. All quantitative responses to the instrument items were input into SPSS. If a participant indicated more than one answer to any of the VQoL or FV items, both responses were documented in Excel. In SPSS, items were allocated a missing data value shown in Table 7 (VQoL) and 8 (FV) (below) which were developed to reduce missing data and over-estimated the child's VQoL/FV. Since the two mid-scale response options represent contrast outcomes (i.e. easy/true and difficult/not true) opting for one response option over the other by over-estimating the child's VQoL/FV is conceptually problematic, and may lead to data misrepresenting the sample. Thus, responses were recorded as missing. Responses assigned a value of 9 were treated as missing data from this point onwards.

Response	Value assigned in SPSS for positively- phrased items	Value assigned in SPSS for negatively- phrased items
No response	9	9
1 (Not at all true) and 2 (A little bit true)	2	1
2 (A little bit true) and 3 (Mostly true)	9	9
3 (Mostly true) and 4 (Completely true)	4	3

Table 7. Missing data values assigned to VQoL responses in phase 4 data analysis.

Table 8. Missing data values assigned to FV responses in phase 4 data analysis.

Response	Value assigned in SPSS
No response	9
1 (Very easy) and 2 (Easy)	1
2 (Easy) and 3 (Difficult)	9
3 (Difficult) and 4 (Very difficult or impossible)	3

All missing data in the PedsQL instrument (child and teen versions), were assigned a 'missing data' value of 9.

## 4.9.3 Data verification

Data entered into the Excel and SPSS databases were cross-referenced using the subtract command in Excel to examine any errors when entering the data. Ten percent of the data were also entered independently into the Excel spreadsheets by a second researcher, and cross-referenced to identify errors. Any discrepancies were noted, and the original questionnaire booklets were checked to ensure the correct response was recorded.

## 4.9.4 Data analysis

Participation rates of children (aged 8-12 years) and YP (aged 13-17 years) were calculated.

#### 4.9.4.1 Preliminary item reduction

To optimise sample size, percentage of missing data per person and item were calculated per instrument. Consistent with phase 3, participants were excluded from psychometric evaluation of the VQoL and/or FV development if >25% of item responses in a single instrument were missing. Items were excluded if >50% of participant responses were missing, in keeping with published criteria to guide item removal<sup>313</sup> and understanding that items with large amounts of missing data are likely to be ambiguous, or not applicable to respondents.

Preliminary item reduction was guided by the distribution of the individual item responses. Items with z-score skewness and z-score kurtosis values outside the range of -2.00 to +2.00 were flagged as problematic, but only removed if ceiling/floor effects were also found (e.g. >60% or <1% responses in an item end category). These thresholds were developed based on the understanding that Rasch analysis requires at least one case within each response option for each item, and more than 60% responses in an extreme category would indicate substantial skew in an item.<sup>313</sup>

## 4.9.4.2 Testing for unidimensionality

For the sake of testing unidimensionality, responses given to negatively worded (e.g. I feel lonely because of my eyesight) versus positively worded (e.g. I feel confident) VQoL items in both child and YP instruments were re-coded so that all responses aligned in the correct direction of the latent trait.

To preserve the sample size whilst testing for unidimensionality a missing data analysis was conducted on the raw datasets after removing items due to distribution of responses. Missing data were imputed using a multiple-pattern regression method,<sup>378, 379</sup> producing 5 imputed datasets for each instrument. Tests of unidimensionality following this imputation were conducted using each imputed dataset, and the results were averaged to derive a final outcome incorporating all imputed missing data values.

The first stage of unidimensionality assessment was conducted using formal exploratory FA with a view to provide evidence of the suitability of the data for Rasch analysis which assumes unidimensionality (see Section 3.3.2, pg. 75). This analysis was conducted using SPSS. The method of principal components was selected with standardised variables and no rotation. The eigenvalue for each factor was analysed in relation to the percent of variance explained. The component matrix was analysed as best evidence indicating unidimensionality in each instrument. Item loadings were scrutinised for values less than 0.4 and items which loaded onto more than one factor. Because the aim of FA was to assess for unidimensionality, no items were removed at this stage.

Parallel analysis is a technique which involves extracting eigenvalues from random data sets that parallel the actual data set with regard to number of cases and variables, and can aid decision making regarding the number of factors to retain in PCA and FA.<sup>380</sup> Parallel analysis was conducted in addition to exploratory FA with a view to provide further evidence for unidimensionality in the instruments. This analysis was conducted using SPSS, and using the syntax provided by O'Connor.<sup>380</sup> The number of

parallel datasets was specified as 1000, and the 'permutations of the raw data set' function was specified to account for imputations of missing data. Both the matrix and scree plot were used to determine the number of existing constructs within each instrument.

4.9.4.3 Formal psychometric item reduction and validation (Rasch analysis)

Prior to conducting Rasch analysis, the original datasets (containing missing data, and items which had not yet been reverse scored) were prepared by re-scoring observed responses in both the VQoL and FV instruments ranging from 1-4 to values ranging from 0-3, to reflect the true magnitude of each score in terms of additive measurement. This was performed using the subtract command in Excel. The original raw datasets containing missing values were used for Rasch analysis based on the premise that Rasch analysis conducted using Winsteps software is capable of making estimations and imputing missing data based on observed scores.

The Rasch rating scale model was applied to assess the true measurement capacity of the instruments, providing evidence for item removal. The process of item removal was iterative, meaning the items were removed one at a time until the criteria of Rasch analysis were met, and all measurement difficulties were resolved. Rasch analysis was conducted using Winsteps 4.0.1.<sup>381</sup> The measurement properties of each instrument (after each iteration of Rasch analysis) were documented using Winsteps outputs in NotePad and compared to produce the optimum measurement instrument. The following syntax was used to ensure missing data values in the FV instruments were recognised:

CODES = 01239; matches the data NEWSCORE = 0123\*; rescore 9 as missing

In the VQoL datasets, items were coded based on the direction of the measurement scale (e.g. forward or reverse scored) using the following syntax (where N represents a forward scored item, and R a reverse scored item):

IVALUEN = 0123\*; scoring for forward scored items

IVALUER = 3210\*; scoring for reverse scored items

The following syntax was used to ensure the discrete missing data values in the VQoL datasets were detected by Winsteps, and treated as missing:

MISSING-SCORED = -1; missing data treated as non-administered

Item fit can be used to confirm that the summary scores provided by the instruments represent a single underlying construct, indicating unidimensionality and supplementing the findings from FA and parallel analysis. INFIT and OUTFIT statistics indicate how well the items contribute to the measurement outcome (e.g. VQoL or FV) and are measured as mean squared standardized residuals, with the 0.5 to 1.5 range being considered acceptable for productive measurement.<sup>382</sup> OUTFIT is based on a sum of squared standardized residuals modelled to approximate a unit normal distribution and INFIT is an information-weighted form of OUTFIT which reduces the influence of less informative, off-target responses.<sup>383</sup> Low values of the fit statistics indicate observations are too predictable (e.g. the data overfit the measurement model) and high values of these statistics indicate unpredictability (e.g. the data underfit the model). Thus, high values of the fit statistics are conventionally inferred as more harmful to measurement than low values which indicate some redundancy in the responses, but do no harm to the instruments capacity for measurement. Items with values of the fit statistics outside the range of 0.5-1.5 were removed iteratively until all items resulted in fit statistics within the acceptable range.

A fundamental component of Rasch analysis, making it distinct from classical test theory approaches such as FA, is that both persons and items differ in terms of ability on the latent trait. Thus, using Rasch analysis, one can assess the *targeting* of the

items in a particular instrument. In this sense, targeting refers to the ability to examine the relative position of item difficulty to person difficulty. The items in an instrument are deemed to target the sample well if the difference between item and person means is less than 1 logit.<sup>384</sup> The item-person map was used as an indicator of the difference between item and person means for each instrument and the difference is also reported in terms of logits.

The direction of alignment of items on the latent trait was assessed using the item polarity command in Winsteps to ensure that higher person abilities corresponded with higher ratings on each item. Observed point-correlations were screened for negative values and average abilities were screened for abilities which were out of order in terms of person ability.

Measurement precision was assessed in each iteration of Rasch analysis for each instrument version using the summary statistics outputs in Winsteps. The person separation index and reliability values were recorded from each analysis using a separate Excel document which enabled the comparison of values across iterations. The threshold of separation index values ≥2.00 and separation reliability values >0.80 were used as the criteria for indicating accurate measurement precision.<sup>313</sup>

Response scale ordering was examined using the Rasch category probability curves which indicate the likelihood of each response scale being selected over the range of the scale. The increase in logits between response options is considered to be good if it increases by at least 1.4 logits,<sup>317</sup> indicating distinct categories, however an increase of more than 5 logits was interpreted as indicating gaps between the response categories. The rating (partial credit) scale function displays the fit mean squared of each response category. Values greater than 1.5 indicate problems with the response options in the same way as the item fit statistics, using the average of the fit mean-square (MNSQ) values associated with the responses in each category. Participants' use of response categories was also assessed using the position of response options

in relation to the ability of the sample. In each case, the ordering of response options was analysed for linear patterns.

Analysis of differential item functioning (DIF) indicates whether subgroups of respondents who have the same person ability respond differently to items. DIF analyses were used to compare participants stratified by age and gender within each analysis. These demographic variables were chosen based on the assumption that the items in the final instrument versions should not produce different results for participants of different gender or age. In other words, items should not be biased in any way to gender or age group. DIF analyses using participants' categorised level of vision (or ability) were not deemed necessary given the developed instruments are intended to detect these differences. All DIF analyses were conducted after each instrument had seemingly fulfilled the multiple criteria of Rasch analysis, and all problematic items had been removed. The threshold of >1.0 was used as an indication of notable DIF between participants.<sup>385</sup> The probability of both Rasch-Welch and Mantel-Hanzel t-statistics for each item were analysed in parallel to DIF contrast values, indicating the probability of observing the DIF contrast when no real DIF is present. Theoretically, if the data fit the Rasch model of measurement, the Mantel-Hanzel t-statistic should be related to the Rasch-Welch statistic.<sup>386</sup> However, DIF contrasts were used as the primary indicator of DIF. As a result of the relatively small sample size used in phase 4, when conducting DIF analyses by age, participants were stratified to 'Young' and 'Old' age groups within each instrument development (i.e. child and young person) to ensure a large number of participants within each age group, and therefore more accurate indications of DIF.

Items which had DIF values above the threshold of 1.0 were removed iteratively and in order of magnitude (from greatest to smallest). After the removal of an individual item due to DIF, all remaining items were entered into a new iteration which was reanalysed using the full Rasch criteria outlined above.

#### 4.9.4.4 Calibration of the child- and YP-VQoL and FV instruments

An important feature of the developed instruments is that they can be used in parallel to measure VQoL and FV across the broader population of children and YP aged 8-17 years. Thus, when used in collaboration summary scores from individuals of different age groups (and given in response to different instrument age-versions) should be comparable in terms of outcome on the latent variable. To ensure instruments are calibrated in this way, and capable of measuring the same outcome among children and YP of varying ages, a final DIF analysis was conducted using the final (and psychometrically robust) version of each instrument. Overlapping or 'core' items included in both child and YP instruments were assessed for significant DIF (between children and YP) and removed if DIF was above the threshold of 1 logit.

Following calibration using DIF analyses, items included in each instrument (e.g. both 'core' items and corresponding age-appropriate items) were selected individually using the ISELECT command in Winsteps, with a view to convert raw summary scores into scores which are comparable between instrument versions. The complete score-to-measure table was used to convert raw scores into user-friendly scores ranging from 0 to 100 on the latent variable in each instrument, and thus produce scores which can be compared between instrument versions, despite differences in the number and wording of individual items.

An equation is produced automatically as part of the complete score-to-measure tables produced by Winsteps which can be used to convert the raw scores in each instrument into measure (logit) scores which indicate the respondent's ability based on the latent trait i.e. FV or VQoL. This is an important outcome in the current study as it can be used within clinical practice to convert raw scores into scores which can be compared between different time points along the trajectory of childhood and adolescence.

However, the equation provided by Winsteps represents a least-squares fit regression model which produces a straight line model. In practice, this trendline may not always

fit the raw scores in the data accurately,<sup>387</sup> as relationships between two variables are often non-linear. In this analysis, the fit of the linear trendline was assessed in each instrument using the least-squares fit equation provided by Winsteps. If the variables were found to misfit the trendline, an iterative procedure was used to fit fractional polynomials (FPs): an extended family of curved trendlines,<sup>387</sup> with a view to improve the model's goodness-of-fit. R was used to fit the raw to logit scores to 2<sup>nd</sup>, 3<sup>rd</sup> and 4<sup>th</sup> order FP trendlines using the equations presented below.<sup>387</sup>

FPs can be defined in terms of the Box-Tidwell transformation. Let *m* be the order of the FP and  $\mathbf{P} = (p, ..., p_m)$  be a vector of powers, such that  $p_1 < ... < p_m$ ; and  $\boldsymbol{\mathcal{E}} = (\mathcal{E}_0, ..., \mathcal{E}_m)$  be a vector of parameters in the regression model. A FP of order *m* can be written as:

$$\phi_m(X; \, \varepsilon, \mathbf{p}) = \sum_{j=0}^m \varepsilon_j \, H_j(X) \tag{a}$$

Where **X** is a matrix of a positive-valued covariate:

$$H_0(X) = 1, P_0;$$
 (b)

And

$$H_{j}(X) = \begin{cases} X^{pj} \text{ if } p_{j} \neq p_{j} - 1 \\ H_{j-1}(X) \ln(X) & \text{ if } p_{j} = p_{j-1} \end{cases}$$
(c)

With X<sup>(pj)</sup> defined as the Box-Tidwell transformation

$$X(pj) = \begin{cases} X p_j & \text{if } p_j \neq 0, \\ \ln X & \text{if } p_j = 0, \end{cases}$$
(d)

The order of the FP specifies the model's complexity in terms of the number of functions defining it. FPs up to order *m*=4 were fitted and the FP that maximised the gain in goodness-of-fit, that is the difference in deviance between the null model, measure =  $\varepsilon_0$ , and the model specified by a log and vector of powers, was chosen. The values *p* = [-2, -1, -1/2, 0, 1/2, 1, 2, 3, 4] were considered.

FPs provide very flexible shapes and are adequate to model non-linear shapes. Note that FPs were fitted adding 1 to both the observed scores and the observed measures to transform them to positive values. This doesn't alter the prediction.

#### 4.9.4.5 Assessment of construct validity

Prior to analysis of construct validity, a second missing data analysis was conducted using the raw scores contained in each dataset, after the removal of items according to Rasch analysis. Missing data in both final versions of the VQoL and FV instruments were imputed using the same method as described in Section 4.9.4.2 (pg. 115), using multiple-pattern regression<sup>378, 379</sup> based on data missing at random, and using SPSS. No missing data were imputed in datasets containing raw scores obtained from the PedsQL instruments as per the published scoring instructions for the Child and Teen versions which recommend calculating the sum of items and dividing by the number of items answered.<sup>388</sup>

Summary scores were calculated for each FV and VQoL instrument using each of the 5 imputed datasets. Summary scores were then converted into comparable (logit) scores using the complete score-to-measure tables produced for each instrument, and recorded using SPSS. PedsQL summary scores were calculated using the published scoring instructions for both Child and Teen versions,<sup>388</sup> which recommend removing persons with >50% missing data, and imputing the mean of completed items for remaining persons.

Construct validity refers to an instrument's ability to measure an intended outcome (in this case, VQoL and FV) (see Section 2.4, pg. 36) and can be analysed using a postulated attribute of a population, which is assumed to reflect test performance, as a proxy for the outcome of interest (VQoL or FV). This is a useful and well-established method for determining construct validity.<sup>389</sup> In all cases, bivariate correlations were used. The raw summary scores were transformed to logit scores using the score-to-measure tables (and the process described in Section 4.9.4.4, pg. 120) to analyse

correlations within each suite of age-specific PROMs (e.g. Child-VQoL, Child-FV, YP-VQoL and YP-FV instruments) and also using the Child and YP datasets combined (e.g. VQoL and FV instruments). Consistent with the treatment of missing data in earlier analyses, correlations were conducted using each imputed dataset. Correlation coefficients were then averaged and transformed into z-scores using the following equation.<sup>390</sup>

$$z = (1/2)[\ln(1+r) - \ln(1-r)]$$
(e)

Each transformed correlation coefficient was compared to the critical value of 1.96, with a value > 1.96 indicating a significant correlation at 5% level.<sup>391</sup>

Construct validity of the developed suite of age-appropriate VQoL and FV instruments was assessed using correlations between VQoL and FV logit scores, summary scores on the PedsQL instruments (including the psychosocial subscale summary score), and latest recorded VA obtained from participants' clinical records. Correlations between FV logit scores and VA was used as evidence of construct validity of the FV instrument, demonstrating a relationship between logit scores based on self-report and objective clinical assessments. A positive correlation between the two variables was hypothesised, with FV scores predicting functional ability. Thus a 1-tailed significance value was specified.

Correlations between the VQoL logit and PedsQL summary scores were conducted to determine the relationship between the VQoL instrument and a generic measurement tool designed to measure HRQoL. Both overall scale and psychosocial scale PedsQL summary scores were entered into the analysis to determine the similarity between the VQoL and overall HRQoL, and the emotional, social, and school functioning scales of the PedsQL. A 2-tailed significance value was specified due to ambiguity in the current literature regarding definitions of QoL, HRQoL and VQoL and uncertainty as to the similarity between VQoL as captured in the developed instrument, and generic HRQoL as measured by the PedsQL.

Finally, a correlation analysis was conducted to determine the relationship between VQoL logit scores and participants' VA, to assess the presence of the disability paradox (see Section 2.3, pg. 28) within the population of children and YP recruited in the instrument development. A 2-tailed significance value was specified.

All correlations are presented with respect to the age-specific version of each instrument (e.g. child and YP versions) with a view to indicate construct validity of each individual instrument, and also in relation to the entire sample of 8-17 year old children and YP recruited in this study, with a view to indicate construct validity of the collaborative suite of instruments.

4.9.4.6 Analysis of unidimensionality using the final instrument versions A second final exploratory FA was conducted following item reduction with a view to indicate dimensionality of the final instrument versions. This was conducted using only the final items in each instrument and followed the procedure described in Section 4.9.4.2 (pg. 115). Analysis was conducted using each dataset with missing data imputed and results were averaged to derive a final outcome.

# Chapter 5 Growing up with visual impairment: Results from an indepth qualitative investigation

The theories of childhood development described in Section 2.1 (pg. 20) describe growth during early life course with regard to a specific developmental capacity, and can be used to understand the *nature* of development and learning. Alternatively, theories of life course seek to understand growth and development from the perspective of *progression* through life course, and with respect to lived experience and significant life events.<sup>392</sup> In doing so, these theories place more emphasis on the subjective experience of life course, with respect to individual differences and varying social and environmental structures.<sup>393-395</sup>

As previously discussed, a wealth of literature describes the substantial impact of visual impairment upon multiple aspects of childhood, and can be categorised according to three levels of human functioning. At the level of activity and participation, the subjective components of the ICF model, studies utilising self-report from children living with visual impairment are extremely beneficial when determining the everyday impact of visual impairment, and present a 'snapshot' of the experience at one time point. However, what appears to be missing from this collection of studies is insight as to what it feels like to grow up with visual impairment, with respect to progressing through early life course, encountering significant life events, changes, and overcoming challenges during childhood. Similarly, literature documenting the impact of visual impairment during adolescence is largely focused upon improving educational and occupational outcomes with a view to optimise future occupational and financial success. To date, no literature has documented the progression of these YP through the early stages of adolescence. Understanding the lived experience during childhood and adolescence has important implications for identifying targets for and developing interventions to promote VQoL or FV, and specifically understanding at which time such interventions will be most effective.

The 4-phase method used in the current development of vision-specific PROMs for YP living with visual impairment affords the opportunity to explore, in-depth the experience of visual impairment during childhood and early adolescence, from the perspective of YP who are well placed to reflect upon their experience. Thus, the objective of this subcomponent of the research project was to describe the lived experience of visual impairment during childhood and adolescence. The specific aim was to identify 1) key time points during childhood and adolescence when pertinent changes occurred, and 2) specific strategies employed by YP for dealing with the challenges encountered, with a view to inform the future development of interventions that could promote VQoL or FV. Specifically, findings may be applied to ensuring timely and optimal use of the patient-reported outcome instruments developed in this research.

A second opportunity afforded by the 4-phase method is an exploration of participants' views about, and experiences of transition from paediatric- to adult-centred ophthalmology services. This is important to advancing current knowledge which, despite demonstrating the importance of a timely and successful transition from child to adult services<sup>396-398</sup> is yet to explore the experiences, and transition-related needs of YP living with visual impairment. Since the interviews conducted in phase 1 were conducted with YP at either side of the threshold of transition, the opportunity was taken to probe their experience of clinical care during childhood and more recently as they approached the conventional age of transition. Thus, the data collected in phase 1 were analysed with a third aim of exploring the views about, experiences, and transition-related needs of YP with visual impairment so as to develop an evidence-base to inform transition planning and provision in ophthalmology.

## 5.1 Phase 1 Participation Rate

Seventeen YP aged 16-19 years participated in phase 1 (40.48% response rate). The demographic characteristics of these participants are shown in Table 9 (pg. 146). All participants were recruited from source 1.

Most in-depth interviews were conducted at participants' homes and one interview was conducted in a private room at one of the primary recruitment centres. All participants were encouraged by the researcher to take part in in-depth interviews alone to avoid the influence of other family members however, due to the interviews being conducted at participants' family homes, siblings and parents attended in a number of cases. Nevertheless, efforts were made to address the participant themselves. The duration of in-depth interviews ranged from 40.2-139.36 minutes (Median = 88.09 minutes, IQR = 39.46).

The point of data saturation was judged as the point at which no new themes/issues emerged from interviews and this threshold was judged to have been reached at the 15<sup>th</sup> interview. To ensure results were comprehensive, a further 2 interviews were conducted, one of which incorporated elements of the first draft of both instruments.

## 5.2 Findings/themes: Challenges, changes, and coping strategies

Qualitative analysis revealed 5 overarching themes related to the experience of growing up with visual impairment, the challenges, and changes YP recalled. A sixth theme described participants' coping strategies. Finally, key time points of change were identified.

## 5.2.1 Acceptance of and adaptation to visual impairment

Acceptance of, and adaptation to visual impairment was an aspect of childhood and early life course spontaneously discussed by most participants. Regardless of manifestation of visual impairment, participants viewed their impairment as a significant challenge which had to be overcome. Acceptance was a key component of overcoming the challenge of visual impairment which took time and effort. The study participants were visually impaired due to a range of ophthalmic conditions with different manifestations, and described different stages of the acceptance process.

5.2.1.1 Differences between children/YP with early and late onset visual impairment

Pertinent variations were apparent among those with early and late onset visual impairment. Participants with early onset visual impairment often described visual impairment as a personal attribute, making them unique. This attribute was fixed and had been internalised throughout childhood, as participants had little, or no, previous experience of full vision, and appeared to have endorsed the notion that their vison would not improve in the future. When probed about the impact of visual impairment during childhood, most participants with early onset visual impairment had difficulty pinpointing differences in their experience compared to that of sighted children, describing "not knowing any different" (Male, 16 years, VI, early onset). Those with early onset visual impairment and often as attempts to prove others wrong. The phrase "I just get on with it" (Female, 16 years, VI, early onset) was often used:

*"I think I adapted to it from a young age. Cos when you're born with it, it's not really an adaptation. It's more like that's who you are." (Male, 16 years, SVI/BL, early onset).* 

"It's to prove to people that, no matter what disability you've got, you know what I mean? It was my decision to do a blind-folded bungee-jump. I knew how scary it was. Everyone was like 'she's not gonna jump, not gonna jump'. The teacher said 'she's not gonna jump'. I swan dived off!" (Female, 16 years, VI, early onset).

In comparison, participants with late onset visual deterioration and subsequent prior experience of visual function, described acceptance as a challenging, ongoing process often associated with negative psychological well-being. Late onset visual impairment significantly impacted participants' sense of identity which had been newly established during childhood and adolescence. Participants described coming to terms with their new identity over time and difficulties navigating social encounters in light of this change. Negative responses from others triggered participants' recalibration of expectations and views about themselves. Two participants who had late onset progressive visual deterioration reported treatment for depression and feelings of grief and loss, signalling that they had not yet reached a point of full acceptance. Ongoing changes in functional vision throughout the course of childhood and adolescence meant that further changes had to be made to participants' choice of physical and leisure activities. Being able to find a substitute for an activity which was no longer possible enhanced psychological well-being, but was not always possible:

"You don't want family or friends who you've seen before, and have seen you seeing, noticing that it has got bad." (Male, 16 years, SVI/BL, late onset).

"It really upset me. I lost so many friends because I wasn't Pearl. I was Pearl who just found out she has got an eye disability. So I had to mature like that [snaps fingers]. I had to do things for myself [...] A lot of people thought I was the person who had changed [...] and people didn't like it." (Female, 16 years, VI, late onset).

"My eyes had got worse. And then I had to adapt myself again [...] So it's like losing someone. Cos you're constantly having to, like, like grieving I suppose." (Female, 19 years, SVI/BL, late onset).

*"I used to love trampolining, but I had to stop because of my eyes." (Female, 19 years, SVI/BL, late onset).* 

"Like watching TV, cos they [those with early onset VI] haven't done it. And for them, they don't care. But if you've had sight, that's the first thing. It's like teaching me something new." (Male, 16 years, SVI/BL, late onset).

5.2.1.2 Differences between children/YP with differing severity of visual

#### impairment

Participants who had visual impairment which was classified as severe (SVI/BL) were more resigned to the impact of visual impairment than those who had milder forms of visual impairment. Participants with visual impairment classified as SVI/BL described understanding the restrictive nature of their impairment through experience, and developing realistic expectations of the impact over time. This understanding enhanced acceptance of visual impairment, allowing YP to internalise the functional limitations of visual impairment as an aspect of personality or self-image, and subsequently have positive attitudes towards their limited functional ability. In comparison, and supporting the presence of the disability paradox (see Section 2.3, pg. 28) in the current sample of YP living with visual impairment, participants with milder forms of visual impairment described hopes that their vision may improve to the extent that they may be able to take part in activities which were currently impossible. Being on the threshold of participation meant that participants were not willing to accept the functional impact of visual impairment in the same way as those with visual impairment classified as SVI/BL. For example, whilst all participants expressed some degree of disappointment or frustration about not being able to drive in the future, those with mild forms of visual impairment described hopes that someday their vision may improve to the level at which they would be legally permitted to drive:

"That little niggle in the back of your mind 'oh what happens if my eyesight gets better?" I tried to push myself to try and see a number plate and it wasn't working. I tried my damned hardest to try." (Female, 16 years, VI, early onset).

#### 5.2.2 Social Environment

Social relationships were dynamic and changeable during the course of childhood. During early childhood, relationships were primarily established within family communities. Siblings and cousins were described as key playmates and valuable friendships were developed with others who attended the same nurseries and preschools. Making new friends and losing, or "drifting away" (Male, 16 years, VI, early onset) from old friends is a normative part of childhood for all YP, regardless of visual impairment. As YP with visual impairment developed, appreciation was increasingly paid to 'true' friends who could accept visual impairment as part of their identity and offer functional support, such as guiding in public places, or reading the board at school:

"[When I can't see the board at school] I just put my pen down and wait. I'll give my friend 'the look' and he'll be like 'yeah'." (Male, 17 years, VI, early onset).

As participants developed through childhood, they described increasing awareness of the broader social environment and public perception of disability. Some described experiences of discrimination which impacted willingness to use assistive devices such as magnifiers or white canes when in public. In particular, YP with visual impairment described difficulties making sure that others were aware of their functional limitation e.g. when crossing the road, but at the same time did not patronise them or embarrass them in public.

#### 5.2.2.1 Relationships with adults

Adults such as teachers and parents were increasingly respected and appreciated as *equals* rather than superiors or carers during the course of childhood and adolescence. Some participants identified teachers who they "could have a laugh with" (Male, 17 years, VI, early onset), or whom they had recently identified as "mates" (Male, 16 years, SVI/BL, early onset). YP with visual impairment valued new and increasingly mature relationships with adults and peers on the basis that these relationships enhanced feelings of safety and support. Instead of playing playground games in which they would likely be excluded or may be at risk of harm, YP discussed preferences of sedentary activities such as chatting with friends in the canteen.

Adults such as teachers and parents were increasingly identified as role models throughout the course of adolescence, and contributed to the development of selfesteem among YP with visual impairment, providing a sense of direction in relation to their future and, in particular, the types of careers or activities which would be possible in light of the functional impact of visual impairment:

"When you're at senior school, it's weird cos it changes immediately. You used to play football, rugby, that sort of thing. But now most of the time we chat. Just walk around, have a conversation. Get food, that sort of thing." (Male, 16 years, SVI/BL, early onset).

"My uncle's a lawyer. And he's told me stories about people he's dealt with [...] I like the idea of it and how he's, even though they've had such a rough background, he's helped them. I would like to do that for other people as well." (Male, 16 years, SVI/BL, early onset).

"I did catering in Year 11 and my teacher got me into doing a lot of cooking. And my nan, she used to teach me a lot of cooking as well. So I got it from them really." (Female, 17 years, VI, early onset).

#### 5.2.2.2 Romantic relationships

Having a boyfriend or girlfriend was described by YP with visual impairment as enjoyable and comforting, and preferable in light of having somebody other than a parent, carer, or teacher, to offer functional support. Romantic partners could offer support in a way which was discreet and subsequently eliminated the psychological burden of having to rely on parents. Despite favouring romantic relationships, YP with visual impairment described challenges maintaining relationships during times of progressive visual deterioration:

*"I lost quite a few friends when I found out. I was in a relationship with someone and then they ended up ending the relationship cos they didn't think they could cope with someone with an eye problem." (Female, 17 years, VI, early onset).* 

## 5.2.3 Developing and asserting independence and responsibility

All participants described a growing need for autonomy as they grew up, which was signalled by the development of independence and responsibility throughout childhood. Some participants described work experience they had recently undertaken and how they gained increased responsibility for time management and safety of others. Specifically, adolescents with visual impairment described a shift between forced independence, in which parents would encourage participants to 'brave' new situations or environments independently, and self-guided independence in which participants were willing to try new tasks, engage in new activities, and communicate with others without help from a parent, carer, teacher or peer.

#### 5.2.3.1 Barriers to developing independence

Many environmental barriers were perceived as restricting participants' attempts to develop independence in light of visual impairment. These included menus at cafés or fast food restaurants which were positioned behind the counter, glass obstacles in shops, overhanging plants in walkways, price tags which were out of reach, crowded areas, timetables at public transport stations, and inability to obtain a provisional driving license as proof of age. When faced with environmental barriers, most YP with visual impairment described choosing the 'safe option' e.g. when in restaurants/cafés ordering something they had ordered before and which is within a certain (known) price range. Others described the challenges of 'braving it' in public which often resulted in negative psychosocial consequences such as feeling stupid, embarrassed or different from others:

"[I never ask for help] because sometimes people can be really arse-y towards you. Me and my friend went to SubWay and you know they have the menus on the top bit, neither me or her could see it. My friend asked the lady for help and she said 'can't you see the board?!' and she went 'I can't I'm blind!' Sometimes people don't react well. We're like aliens to them." (Female, 17 years, VI, early onset).

The majority of participants described the inability to drive, or take driving lessons, as an issue which was increasingly salient as they developed into adolescence and detrimental to their growing independence. The majority of YP relied on parents or siblings to courier them between home, school, and social occasions, but were increasingly aware of public perceptions of this, rendering it inappropriate or "baby-ish" (Male, 16 years, SVI/BL, early onset) in light of their age. Participants were intensely frustrated by the imposed barrier towards independence, which was magnified at the time when they and their friends became legally eligible to drive. In attempts to overcome the barrier, three participants had received mobility-related training in which they were taught strategies to travel independently. Establishing and maintaining mobility training however, was described as difficult and time consuming and participants perceived limited resources available.

"I'd rather be able to do it [driving] but not do it, than not be able to do it and want to do it. [...] It's about having that choice. I'd rather have the choice, but I'll never have it, so, just got to deal with it." (Female, 17 years, VI, early onset).

"[Driving] is something which has been building up in me over time and it didn't really affect me till I reached the age of 17 [...] my sister passed her test the year before and at that point I wasn't worried about it. But when my friends started to take it and pass it, it really hit me and I realised that I definitely wouldn't be able to do that. Before it always seemed like something that was gonna happen in the future." (Male, 17 years, VI, early onset).

## 5.2.4 Future: Developing plans in light of visual impairment

YP with visual impairment were apprehensive about their future. Many were excited about anticipated events such as moving away from home or starting University and had positive expectations. However, there are several challenges that YP discussed in relation to their visual impairment; the most pertinent being realisation of the possibility for future visual deterioration:

"I was told by my opticians that I'm gonna reach a point at [age] 40 where it's gonna stay the same and after that it's gonna tail off. Eventually it's gonna get worse. But it's something that I know about now so that by the time I reach that age, hopefully I'll have found ways to work around it." (Male, 17 years, VI, early onset).

Participants discussed how their plans for the future had developed and, in some cases, been amended, adjusted, or abandoned in light of visual impairment. Coming to terms with the impact of visual impairment upon future occupation was an ongoing process during childhood, which took place regardless of manifestation or nature of visual impairment.

Parents played an important role as advocates for plans in the future which are realistic, providing guidance and giving suggestions, and ensuring YP developed skills related to future independent-living. In some cases, YP had developed ambitious and potentially unrealistic plans for the future and parents played an important role in ensuring plans remained grounded. However, YP with visual impairment often perceived input from parents as overprotective and were frustrated about the imposed limitations.

Safety and familiarity were key considerations for YP when making plans and thinking about the future. The majority of participants discussed preferences for living somewhere close to home in the future and familiarity with the local area was essential to developing further independence:

"Then I thought there are a lot of careers [that I can't do]. I can't be a vet [...]" (Male, 16 years, SVI/BL, late onset).

"My dad's a plumber. Later on in life I've got to understand that I can't do manual labour because of my eyesight." (Male, 16 years, VI, late onset).

"My mum's brought me up to be able to cope. [...] she's brought me up to develop strategies and be able to be independent." (Male, 16 years, SVI/BL, early onset).

"I want to go and study abroad for a year. I want to go to China but my mum won't let me go. But I want to do it all [...] I wanted to study in America, but again, my mum, she's really iffy about that." (Female, 17 years, VI, early onset).

"If my university is somewhere that's really busy, and it's like in central London or something, and my flat is in central London [Dad laughs] I'm just saying!" (Male, 16 years, VI, late onset).

## 5.2.5 Transitions in Education

Integral to participants' retrospective account of growing up with visual impairment was the progression through education. Periods of childhood were often recalled in terms of key time points in education e.g. "when I was in Year 7 things started to change". (Male, 16 years, SVI/BL, late onset).

At the time of the interview, all participants had experienced a formal transition in education which occurred at the age of 11 years, and is based on the legal educational requirements in the UK. Two participants had experienced transitions between mainstream and specialist education and one had changed schools across separate counties. Five participants had changed school as a result of inadequate VI-support:

"You've got to make a few modifications so that I can have an education [...] Well, try living with it! All they had to do is print off a couple of pages." (Female, 17 years, VI, early onset).

Not all transitions in education were formal transitions. Changes were experienced between classes, year groups, classrooms, teachers, and friendship groups. With each transition, YP experienced a range of VI-specific challenges such as navigating new environments, informing others about their VI-specific needs, adapting to new classroom layouts, and managing adaptations in increasingly frequent exams. The impact of visual impairment magnified the degree of change brought by each transition. Challenges were overcome with varying success and often associated with changes in personality and ability to cope with visual impairment:

"I was always shy. I always thought no one will like me because I'm partly blind. But then, I got into secondary school, started doing Performing Arts and then my confidence started getting bigger and bigger and then as soon as I started being able to make people laugh it fuels me and I keep going." (Male, 16 years, SVI/BL, late onset).

#### 5.2.5.1 Social life at school

Participants described changes in social groups and norms as they progressed through education. Transitions between schools and colleges often resulted in new peer groups. Friendships with others who also had visual impairment were established when YP transitioned from mainstream to specialist education and were valued highly:

"I asked one of my friends who's blind how to put something on a coat hanger. I said I lay it on my bed and she goes 'I've got a better idea. If you hold it up with your mouth, and then do the top bit up'. I had never thought to do it like that." (Female, 17 years, VI, early onset).

#### 5.2.5.2 VI-specific educational support and demands

The level and nature of VI-specific educational support changed dramatically for most participants as they progressed through the education system. Some described variations between schools. Improvements in support enhanced aspects of psychological well-being such as feelings of belonging. Alternatively, deteriorations in VI-specific support made transitions in education particularly challenging, and increased participants' awareness of their unique educational needs and the difference between themselves and their sighted classmates:

"I didn't realise how much they could help me. I just assumed it was OK, and it was normal to be happening [...] Until it improved I didn't really realise how much better it was." (Male, 17 years, VI, early onset).

"It hadn't been fully set-up for visually impaired children." (Male, 17 years, VI, early onset).

YP with visual impairment noticed developments in technology which took place throughout the course of their education. These were often discussed with excitement, positive perceptions of future developments, and ability to cope with visual impairment. Blackboards changed to whiteboards which then became interactive, meaning YP were increasingly able to use devices such as cameras or CCTV devices to access resources easily. As YP progressed through education, they experienced increasing difficulty keeping up with the teacher and their classmates during lessons. Use of VIspecific devices such as magnifiers was associated with pain and fatigue at the end of the school day:

"When I concentrate on reading and writing I have neck-ache and it's really bad and I have to carry a lot of books behind my back so it also increases the pain." (Male, 16 years, SVI/BL, early onset).

"We're on the computers all day long and like 5 hours, all the time and it's, um, it does drain you at the end of the day. It's very tiring." (Female, 17 years, VI, early onset).

## 5.2.6 Coping strategies

Participants discussed a range of coping strategies that had been developed and used throughout childhood and adolescence. Despite adolescents living with conditions with varying levels of sight impairment, and manifestation of VI, the coping strategies that had been developed were often similar, including both extrinsic (i.e. those based on the availability of assistive devices and functional support) and intrinsic (i.e. those attributable to personality, beliefs and attitudes) strategies.

#### 5.2.6.1 Perseverance

Perseverance was a coping strategy identified among those with early onset visual impairment. Participants discussed attempts to succeed in spite of the impact of visual impairment. YP often described attempts to work around the impact of VI, or find alternatives, implying profound emotional and cognitive investment. Participants with SVI/BL discussed the importance of being organised, knowing where their belongings were, and being well-prepared for new situations:

"Try, try, try again and if that fails, just try again! [...] And after you fail, go to bed!" (Male, 16 years, SVI/BL, early onset).

*"I try to find ways. Solve problems […] It's a problem solving skill." (Male, 16 years, SVI/BL, early onset).* 

"I like to put my things where I remember. In the morning, say if I'm in a rush I might be looking in the wrong place. And if I was looking for something small, then I might not notice it." (Male, 16 years, SVI/BL, late onset).

#### 5.2.6.2 Humour

Humour emerged as an important coping strategy which developed throughout childhood and adolescence, and enabled participants to evoke positive reactions from others when they made mistakes or blundered in social contexts. Participants often described 'making fun' of themselves and their impairment in front of others. When initiated by oneself, humour was an important coping strategy. However, when initiated by others, humour was seen as a barrier towards social inclusion:

"I'm quite lucky, if you can get a laugh out of it, you think why not?!" (Male, 16 years, VI, early onset).

*"I'm always laughing at myself, always tripping up. Cos that's all you can do." (Female, 17 years, VI, early onset).* 

"The teachers understood how far was enough. If it was tasteful [then it was fine]. They're not vindictive." (Male, 16 years, SVI/BL, early onset).

#### 5.2.6.3 Functional support from others

Functional support which could be offered by others was an important extrinsic strategy. At a basic level, all participants received functional support from family members which was largely appreciated. As participants developed, however, functional support from family members, particularly parents, became increasingly less desirable based on the perception or experience of stigma from friends and peers. Functional support which could be provided by friends became increasingly salient for YP, enabling them to engage in social activities outside of family communities:

"When I'm at the high street, my school friends are also there and when my mum holds my hand, I don't like it. Because then they'll mug me when I go to school." (Male, 16 years, SVI/BL, early onset). "The group of friends I've got now, I don't have to ask. They'll come up to me and link arms with me anyway. Even though I don't ask!" (Female, 17 years, VI, early onset).

## 5.2.7 Key time points of change

Three key time-points were identified, signalling times of change in the nature of the impact of visual impairment during childhood and adolescence. All three were identified in relation to education. As YP grew up, change was increasingly related to internal, intrinsic factors such as self-development, as opposed to change which was driven by formal, policy-related transitions:

5.2.7.1 Transition from Primary to Secondary education at the age of 11 years

The transition from primary to secondary education at the age of 11 years was a mandatory transition imposed by legal educational requirements in the UK. Participants recalled expansions of social environments, and increased awareness of new social norms or conventions as outcomes of the transition. For example, differences were identified in activities during break or lunch times, as methods of social engagement transition from physical to communicative and social group norms matured. Navigation around a new educational environment required time and effort as YP established familiarity with their new surroundings.

#### 5.2.7.2 Mid-way through Secondary education

Mid-way through secondary education was a time-point which YP described in terms of increased awareness, and early development, of independence and autonomy. It is during this time when participants experienced increased choice within education, as subjects were dropped or initiated depending on individual interests. For the majority of YP with early onset, non-progressive sight impairment, this was a time when parents permitted visits to shopping centres, cinemas, local parks, and engagement in social activities outside of school and home increased.

#### 5.2.7.3 Transition from Secondary education into higher education

Change during the transition from secondary education into higher education at the age of 16 was increasingly related to dynamic shifts in internal values and attributes, in contrast to social or physical environments. The transition to an increasingly flexible educational institution allowed participants to experiment with personal style (i.e. in the absence of school uniform restrictions), and identify with adults as equals rather than superiors. During this period independence surged as YP were able to approach vision-related difficulties with maturity, and understand the broader social context in which visual impairment is perceived by others.

"I like having independence. A lot of [sixth form] is independent study which I like. You don't have to wear uniform [...] It's nicer having a bit more freedom and a bit more choice in everything." (Male, 17 years, VI, early onset).

## 5.3 Transition from paediatric to adult ophthalmology services:

## Views, experiences, and transition-related needs

At the time of the interview, 8 participants had transitioned from paediatric ophthalmic services: 6 into adult services and 2 into dedicated adolescent services. Only 2 (25%) of these participants preferred their prior paediatric service, due to its more child-centre approach to communication, although pros and cons were identified by all. The two participants now in an adolescent service identified significant positive benefits of this specialist service bridging child and adult care. Only 1 subject (14%) still in paediatric services did not want to transition, attributable to a strong relationship with their managing clinician. Two participants were unsure whether they had transitioned: both had stable visual impairment and had not been reviewed for some years.<sup>399</sup>

Fourteen codes emerged from analysis of interview data, identifying two key components relevant to transition '*Communication with professionals within clinical contexts*' and '*Healthcare environment*'. Both were associated with the overarching theme 'Confidence to self-manage healthcare in the future as an adult'. 'Emotional

*attachments to child-centred care'* was a further sub-theme which influenced participants' self-reported willingness to transition.<sup>399</sup>

## 5.3.1 Confidence to self-manage healthcare in the future as an adult

Participants discussed having increased responsibility for their own healthcare, with most recognising the diminishing role their parents would play once they entered adult services, in some cases describing parents as 'handing over' or encouraging them to take control and build confidence to manage their healthcare independently. They recognised that growing up involved greater maturity and transition into adult care enhanced feelings of autonomy, confidence and control. Nevertheless, attitudes varied, ranging from strong preferences to take control of the transition and subsequent clinical care, to disengagement.

#### 5.3.1.1 Communication with professionals within clinical contexts

Participants who had transitioned described the major differences between paediatric and adult services in relation to communication with their managing clinicians.<sup>399</sup> This reflected, in part, the shorter duration of outpatient appointments and the larger clinical teams in adult services, which meant that participants were not certain of seeing the same clinician(s) at each visit.

"[In child-centred care] you'll be treated like a child but in adult clinics you'll be treated like you would be if you were in an interview." (Male, 16 years, transitioned to adult services).

"[Now I've transitioned] they take a bit more time. They used to treat me like a child. They talk to you that way." (Female, 17 years, transitioned to adult services).

Some participants who had not yet experienced a transition and remained in paediatric services described parents 'taking over' the consultation and communicating on their behalf. Attitudes towards parents 'taking the lead' varied: some felt excluded or embarrassed when parents intervened, whereas others valued their parents' input and disease-specific knowledge.

"My mum will get me involved in the conversation [...] and says she'd rather me speak to them cos obviously it's my eyes and not hers but sometimes it feels like my mum and the doctor are having the conversation and I'm like 'hello, I'm here!'" (Female, 18 years, transitioned to adult services).

"I think my mum knows more about my condition than I do! So I'd rather have her there than kick her out and go 'oh I wanna do this by myself'." (Male, 16 years, not transitioned).

#### 5.3.1.2 Environment

Participants who transitioned noted the different environment in adult services: some welcomed this, in particular the reduced sensory 'overload' of paediatric outpatient play areas but for others this was initially unwelcome and surprising, adversely impacting their feelings of belonging, confidence and involvement in healthcare.<sup>399</sup> However those who had not yet transitioned expressed strong dislike of child-centred environments, which was often the primary cause of desire to move into adult-centred care.

"When I had my first appointment I remember it being so different! I thought 'what is this?!' [Laughs]. Cos it's duller. You just sit there and wait and then get called and go." (Female, 18 years, transitioned to adult services).

"I won't play with the toys or watch cartoons. [...] I'd rather watch some news or sports on the TV and sit quietly." (Male, 16 years, not transitioned).

"There's loads of little kids. You're the big one and you don't feel like you're in the right place anymore." (Female, 18 years, transitioned to adult services).

"I hate little kids! [...] because I'm looking there [gestures at eye level] and they're quite short. Sometimes they're running. And I want to smack them!" (Male, 16 years, not transitioned).

Notably two participants who had transitioned into specific adolescent/young person services valued the new clinical environment, appreciating, in particular, the opportunity for contact with a peer group similar in age, which enhanced their sense of belonging and age-appropriate provision of televisions and computers.

"In the kid's one there were baby things to do there, whereas when you go to the teenage ward there's more grown up things. [...] there's a pool table, TV's and computers." (Male, 16 years, transitioned to adolescent services).

## 5.3.2 Emotional Attachments to Paediatric Ophthalmology Services

Emotional attachment to the managing clinician was cited as a reason to be unwilling to transition by two participants with late onset and/or progressive visual impairment: one participant explained the role of their managing clinician in the process of diagnosis and acceptance of progressive visual deterioration and the desire that this practitioner would be involved in her future health care.<sup>399</sup> The other described losing contact with his paediatric ophthalmologist as causing loss of accessible vision-specific support, which subsequently impacted his acceptance of, and adaptation to late onset visual impairment.

"When I was with Doctor J and they told me that I'd eventually go blind, she came over and gave me a hug [...] I've stayed in children's' clinics simply because of Doctor J. I have a funny feeling that when I move to adult's clinics I will still see Doctor J." (Female, 17 years, not transitioned).

"There was more support. Doctor P was there and she would understand some stuff." (Male, 16 years, transitioned).

"I'm a bit sad that I'm leaving my doctor. Most of the doctors I have known for quite a few years, I've got used to them." (Female, 17 years, not transitioned).

## 5.4 Summary

Findings from this stream of instrument development indicate that the impact of visual impairment during childhood and adolescence is fluid and dynamic, changing as a result of transitions in school, physical growth, and psychological development. YP with visual impairment identified five core elements spanning intrinsic, personality-based changes to broader social contexts and the influence of others during childhood. Critical to development with visual impairment was a process of adaptation and adjustment, for which variability was seen among those with variations in timing of visual impairment onset and severity.

With regards to transitions from paediatric to adult ophthalmology services, findings indicate some variability in the content and timing of current transition processes in the

UK. Aspects of transition valued by YP, such as age-appropriate communication, suitable physical clinical environments, and an age-appropriate peer group also being served by the service, are likely to be associated with effective transition.<sup>399</sup>
# Chapter 6 Results from psychometric evaluation

Results from each phase of instrument development are presented in this chapter including development, modification and validation. After the first phase a master item pool was developed and updated after each subsequent phase.

## 6.1 Study sample characteristics

A total of 383 patients were identified through source 1. All identified patients were invited to take part in one of the 4 phases of instrument development and a total of 115 YP identified through source 1 participated, rendering the overall participation rate for source 1 30.03%.

During phases 3 and 4, 20 external patient identification centres (PICs) were included as a second source of recruitment. Determining the exact number of potentially eligible patients identified by the PICs is problematic due to an absence of a prior sampling framework but rather rough estimates which were provided by each centre regarding the number of eligible patients they could potentially identify within the time frame of the research project. However, a total of 14 YP identified through source 2 participated.

Thus, in total, 129 YP participated in the instrument development; each contributing to one phase only.

The demographic characteristics of the overall sample are shown in Table 9 (pg. 146). As it was not possible to determine the demographic characteristics of the nonparticipators identified and invited directly by PICs, this Table shows only those identified through source 1. Notably, the sample size in each phase of development varied according to the specific aims of the phase, the approaches used to collect data, and the nature of the data collected. The majority of patients included in the sampling framework were conserved for the final phase of data collection, where quantitative statistical analyses were required. Table 9. Demographic and clinical characteristics of the full sample of YP who

participated compared to those who did not participate (where known for source 1).

Demographic characteristic	Participators <i>n</i> = 115, <i>n</i> (%)	Non-participators n = 268, n (%)
Age		
13	13 (11.3)	46 (17.16)
14	22 (19.13)	34 (12.69)
15	20 (17.39)	37 (13.81)
16	21 (18.26)	61 (22.76)
17	32 (27.83)	77 (28.73)
18	6 (5.22)	13 (4.85)
19	1 (0.87)	-
Gender		
Male	62 (53.91)	141 (52.61)
Female	53 (46.09)	127 (47.39)
Ethnicity		
White British/other	82 (71.3)	113 (42.16)
Black Caribbean/African/other	4 (3.48)	20 (7.46)
Asian Indian/Pakistani/Bangladeshi/Chinese/other	20 (17.39)	41 (15.3)
Mixed	3 (2.61)	4 (1.49)
Other	4 (3.48)	9 (3.36)
Unknown/not stated	2 (1.74)	81 (30.22)
Severity of visual impairment		
LV: logMAR ≤ 0.46	2 (1.74)	8 (2.99)
VI 1: logMAR 0.48-0.70	46 (40)	91 (33.96)
VI 2: logMAR 0.72-1.00	38 (33.04)	84 (31.34)
SVI: logMAR 1.02-1.30	15 (13.04)	32 (11.94)
Blind: logMAR ≥ 1.32	14 (12.17)	53 (19.78)

Demographic characteristic	Participators <i>n</i> = 115, <i>n</i> (%)	Non-participators <i>n</i> = 268, <i>n</i> (%)
Index of multiple deprivation quintile rank		
1: most deprived	17 (14.78)	53 (19.78)
2	18 (15.65)	73 (27.24)
3	22 (19.13)	59 (22.01)
4	26 (22.61)	47 (17.54)
5: least deprived	32 (27.83)	36 (13.43)

The demographic characteristics of the sample represent the population of children living with SVI/BL in the UK well in terms of gender and ethnicity but marginally underrepresent children coming from the most deprived quintile of index of multiple deprivation (14.78% in this study compared to 40% in the UK population<sup>173</sup>). The similarity of the participants included in this sample to the overall population of children and YP living with visual impairment (versus SVI/BL) is unknown due to limited information about the frequency of childhood visual impairment (as opposed to SVI/BL) in the UK and internationally.

Notably, the composition of demographic characteristics of the sample varied in each phase (see Table 11, pg. 151), with phase 4 (the final stage of psychometric evaluation requiring the largest number of participants) containing the greatest spread of characteristics, and being most representative of the population of children living with SVI/BL in the UK.<sup>173</sup>

# 6.1.1 Relationship between demographic characteristics and participation

Table 9 (pg. 146) shows a substantial number of participants who did not participate and for whom ethnicity was recorded as unknown. This was likely a result of the method used to identify eligible patients using the data stored in electronic health records. Because the families of the YP who did not participate in the research did not complete a Family Background Questionnaire, these data can be described as missing not at random. Thus, participants with ethnicity recorded as unknown (n = 83) were excluded from the logistic regression model. The sample size of some demographic characteristics (namely, ethnic minorities and LV) was small. Thus, ethnicity was regrouped into 2 categories (White British/other and non-White), and the LV and VI1 categories of severity of visual impairment were merged prior to analysis of the relationship between demographic characteristics and participation in the study.

A test of the full final model against a constant only model was statistically significant (chi-squared = 16.62, p=.001, df = 4). Predicted participation overall was 65.3% (87.7% for non-participated and 28.3% for participated). The Wald criterion demonstrated that only IMD and severity of visual impairment made a significant contribution to prediction (p <.1). Specifically, a one unit increase in IMD (from more to less deprived) means YP are 1.6% times more likely to participate. Having visual impairment classified as Blind, compared to LV/VI1 means participants are 52.7% times less likely to participate.

	Univa	ariable ana	lyses		Multiv	variable an	alysis		Final mu	ltivariable	analys	is
Predictor	Coefficient	Wald's χ²	р	AOR	Coefficient	Wald's χ²	р	AOR	Coefficient	Wald's χ²	р	AOR
Age	-0.39	0.24	0.62	0.96	-	-	-	-	-	-	-	-
Gender (Male constant)	0.13	0.3	0.58	1.14	-	-	-	-	-	-	-	-
Ethnicity (White British/other constant)	-0.55	4.52	0.03	0.58	-0.20	0.51	0.48	0.82	-	-	-	-
Severity of visual impairment (LV/VI1 constant)	-	5.22	0.16	-	-	4.11	0.25	-	-	4.29	0.23	-
VI2	-0.14	0.23	0.63	0.87	-0.16	0.29	0.59	0.85	-0.17	0.33	0.57	0.85
SVI	-0.38	1	0.32	0.68	-0.33	0.73	0.39	0.72	-0.34	0.79	0.38	0.71
Blind	-0.8	4.85	0.03	0.45	0.74	3.92	0.05	0.48	-0.75	4.1	0.04	0.47
Index of multiple deprivation (from higher deprivation to lower deprivation)	0.02	14.52	0	1.02	0.02	10.36	0	1.02	0.02	13.57	0	1.02

Table 10. The coefficients and significance of variables (demographic and clinical characteristics) included in the logistic regression models.

# 6.2 Participation rate

Table 11 (pg. 151) shows the demographic characteristics of the YP who participated in each phase of instrument development, including those who were recruited through source 2. Participation ranged from 26.23% (phase 2) to 40.48% (phase 1). Participation in phase 3 and 4 was 31.34% and 26.4% respectively, excluding those who were invited through source 2.

All but one in-depth interview in phase 1 and expert consultations in phase 2 were conducted at participants' family homes. Three YP took part in phase 2 and two YP took part in phase 4 shortly after their 18<sup>th</sup> birthday and were deemed close enough to the age threshold to be included in analyses.

Demographic characteristic	Phase 1 <i>n</i> = 17, <i>n</i> (%)	Phase 2 <i>n</i> = 16, <i>n</i> (%)	Phase 3 <i>n</i> = 23, <i>n</i> (%)	Phase 4 <i>n</i> = 73, <i>n</i> (%)
Age				
13	*	3 (18.75)	4 (17.39)	8 (10.96)
14	*	2 (12.5)	6 (26.09)	19 (26.03)
15	*	3 (18.75)	4 (17.39)	15 (20.55)
16	7 (41.18)	2 (12.5)	4 (17.39)	14 (19.18)
17	8 (47.06)	3 (18.75)	5 (21.74)	15 (20.55)
18	1 (5.88)	3 (18.75)	-	2 (2.74)
19	1 (5.88)	-	-	-
Gender				
Male	10 (58.82)	8 (50)	13 (56.52)	39 (53.42)
Female	7 (41.18)	8 (50)	10 (43.48)	34 (46.58)
Ethnicity				
White British/other	11 (64.71)	12 (75)	19 (82.61)	54 (73.97)
Black Caribbean/African/other	-	-	-	3 (4.11)
Asian Indian/Pakistani/Bangladeshi/Chinese/other	4 (23.53)	4 (25)	4 (17.39)	11 (15.07)

Table 11. Demographic and clinical characteristics of YP who participated in each phase of instrument development.

Demographic characteristic	Phase 1 <i>n</i> = 17, <i>n</i> (%)	Phase 2 <i>n</i> = 16, <i>n</i> (%)	Phase 3 <i>n</i> = 23, <i>n</i> (%)	Phase 4 <i>n</i> = 73, <i>n</i> (%)
Ethnicity continued				
Mixed	1 (5.88)	-	-	2 (2.74)
Other	1 (5.88)	-	-	-
Unknown/not stated	-	-	-	3 (4.11)
Severity of visual impairment				
LV: logMAR ≤ 0.46	1 (5.88)	-	-	1 (1.37)
VI 1: logMAR 0.48-0.70	8 (47.06)	9 (56.25)	9 (39.13)	20 (27.4)
VI 2: logMAR 0.72-1.00	3 (17.65)	5 (31.25)	7 (30.43)	30 (41.1)
SVI: logMAR 1.02-1.30	2 (11.76)	1 (6.25)	4 (17.39)	8 (10.96)
Blind: logMAR ≥ 1.32	3 (17.65)	1 (6.25)	3 (13.04)	14 (19.18)
Timing of onset of visual impairment	1			
Early (≤2 years)	15 (88.24)	10 (62.5)	21 (91.3)	58 (79.45)
Late	2 (11.76)	6 (37.5)	2 (8.7)	15 (20.55)
Nature of deterioration of visual impairment	1			
Stable	12 (70.59)	5 (31.25)	21 (91.3)	60 (82.19)
Progressive	5 (29.41)	11 (68.75)	2 (8.7)	13 (17.18)

Demographic characteristic	Phase 1 <i>n</i> = 17, <i>n</i> (%)	Phase 2 <i>n</i> = 16, <i>n</i> (%)	Phase 3 <i>n</i> = 23, <i>n</i> (%)	Phase 4 <i>n</i> = 73, <i>n</i> (%)
Index of multiple deprivation quintile rank				
1: most deprived	1 (5.88)	2 (12.5)	1 (4.35)	17 (23.29)
2	2 (11.76)	-	5 (21.74)	14 (19.18)
3	4 (23.53)	4 (25)	4 (17.39)	11 (15.07)
4	8 (47.06)	3 (18.75)	5 (21.74)	12 (16.44)
5: least deprived	2 (11.76)	7 (43.75)	8 (34.78)	19 (26.03)

\* YP aged < 16 years were not recruited to phase 1 (as described in Section 4.6.4.2, pg. 101).

In phase 4, 194 children (aged 8-12 years) were identified as eligible through source 1 and invited to take part. Of these, 60 participated (response rate: 30.93%). Additionally, 26 children who were identified and invited through source 2 participated. Overall, 86 children participated in phase 4. Notably four children completed Child questionnaire booklets: three participated shortly before their 8<sup>th</sup> birthday, and the other was aged 13 years and 3 days old on the date of participation, and were deemed close enough in age to the threshold to be included in analyses. Their characteristics are shown in Table 12 (pg. 155). This sample was closest (in comparison to the sample in phases 1-4 of YP aged 13-17 years) to being representative of the population of children living with SVI/BL in the UK.<sup>173</sup>

Demographic characteristic	Phase 4 <i>n</i> = 86, <i>n</i> (%)
Age	
7	3 (3.49)
8	19 (22.1)
9	22 (25.58)
10	9 (10.47)
11	15 (17.44)
12	17 (19.77)
13	1 (1.16)
Gender	
Male	51 (59.3)
Female	35 (40.7)
Ethnicity	L
White British/other	53 (61.63)
Black Caribbean/African/other	9 (10.47)
Asian Indian/Pakistani/Bangladeshi/Chinese/other	21 (24.42)
Mixed	3 (3.49)
Other	-
Unknown/not stated	-
Index of multiple deprivation quintile rank	L
1: most deprived	21 (24.42)
2	14 (16.28)
3	16 (18.6)
4	15 (17.44)
5: least deprived	17 (19.77)
Missing	3 (3.49)

Table 12. Demographic characteristics of children who participated in phase 4.

Demographic characteristic	Phase 4 <i>n</i> = 86, <i>n</i> (%)
Severity of visual impairment	
LV: logMAR ≤ 0.46	5 (5.81)
VI 1: logMAR 0.48-0.70	37 (43.02)
VI 2: logMAR 0.72-1.00	32 (37.21)
SVI: logMAR 1.02-1.30	5 (5.81)
Blind: logMAR ≥ 1.32	7 (8.14)
Timing of onset of visual impairment	
Early (≤2 years)	73 (84.88)
Late	13 (15.12)
Nature of deterioration of visual impairment	
Stable	55 (63.95)
Progressive	31 (36.05)

Data were collected from both children/YP and their parents or guardians in the form of a proxy report administered in phase 4.

Of the 159 families recruited, data were excluded from a) one YP because the parent report was returned in absence of a completed YP report and b) one YP because they failed to return any consent forms even though they returned completed questionnaire booklets. As a result, a total of 86 children and 71 YP were included in phase 4 analyses.

Only two YP used the electronic version of the instruments. However, one of these participants was excluded prior to further analysis because no consent forms were returned. No children participated using the electronic version of either instrument.

# 6.3 Development of the vision-related quality of life (VQoL) instruments

# 6.3.1 Phase 1: Item development and adaptation

Table 13 (pg. 158) shows an adaptation of the final hierarchy of child-centred data which was developed in the foundation research.<sup>5-7</sup> This table was used to compare the self-reported impact of visual impairment between children (aged 10-15 years old) and YP (aged 16-19 years old), highlighting areas of overlap and discrepancy between the two age-groups and thus justifying the suitability of existing instrument items (taken from the foundation research) and the need for new, age-appropriate items. Areas of overlap and discrepancy were organised according to the six overarching themes in Rahi et al.'s<sup>5</sup> VQoL thematic framework. In developing the YP-VQoL instrument, aspects of functioning – home, school and leisure which were discussed in relation to psychological and emotional well-being were considered for inclusion as items in the instrument, as opposed to aspects of self-reported ability which were considered for inclusion in the YP-FV instrument (see Section 6.4.1, pg. 227).

Table 13. Areas of overlap and discrepancy between VQoL and FV reported by 16-19 year old YP and 10-15 year old children.

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
acceptance and participation	(parents, siblings, and extended family), Friends (at home and at school), Peers, teachers and general public,	treat me any different to anyone else. (Male, 16, VI, early onset) Someone took my glasses, she stole, took my glasses [] and she hid em. [] and I feel lost	others, Unwanted/unhelpful social support and negative attention, Social support in public places/situations,	that I can rely on him. But at the end of the day, he's only 10. [] it makes me feel a bit guilty at times. (Female, 17, VI, late onset) [ <i>One of my teaching</i> <i>assistants is</i> ] really fussy
	Related issues of bullying and teasing, Isolation, fitting in, inclusion and exclusion, Stigma associated with impairment and disability, Current aspirations, Communication.	without my glasses and I got a bit upset. Apparently they were joking but to me it was bullying. (Female, 17, VI, early onset) Cos they're all talking about different things that I couldn't talk about cos I hadn't experienced it. Like driving, for example.	Long-term romantic relationships.	and there are times when I can be just doing [ <i>my work</i> ] and, and she's looking over my shoulder, peering. Or she'll take my textbook, and give it to someone else. (Male, 16, VI, early onset) My mum came up with a system.

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
		Social relationships, acceptance and participation continued (Female, 19, SVI/BL, late onset)		Social relationships, acceptance and participation continued In the dark I'd link arms with her, and she'd squeeze my arm (Female, 17, VI, late onset) I was in a relationship with someone and then they ended it cos they didn't think they could cope with someone with an eye problem. (Female, 17, VI,
Independence and autonomy	Support (e.g. at school, home, leisure, mobility training),	That's irritating but you develop ways around it. (Male, 16, SVI/BL, early onset)	Increased responsibility, Driving as a means to being independent.	late onset) I used to get the train with friends. And then, in the last couple of years I've done university trips on my own []

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Independence and autonomy continued Access to adaptive technologies (at home and school), Having a Learning Support Assistant, Daily independent living activities (e.g. travel, shopping, money, roads, self-care, making food and drinks, housework, telling the time), Mobility issues (e.g. being guided, using a cane),	Independence and autonomy continued I suppose you have responsibility [to do the cleaning] but you can't always do it to the best of your ability. I feel quite independent at home. Cos my mum's brought me up to be able to cope [] to develop strategies and be able to, be independent. (Male, 16, SVI/BL, early onset)		Independence and autonomy continued and gone off to see friends on my own and done my own thing. I've become a lot more independent as I've grown up. (Male, 17, VI, early onset) It's not that easy to get about, so I've had to rely on everyone else to drive me about. (Male, 17, VI, early onset) Yeah, it makes me feel a lot less independent, cos I have to rely on other ways of getting around other than driving myself. (Female, 17, VI, early onset)

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Independence and autonomy continued Vulnerability, and communication (e.g. reading text messages and emails, using mobile phones).			Independence and autonomy continued Um, but then when my friends started to take [ <i>their</i> <i>driving test</i> ] and pass it. That's when it really hit me and I realised that I definitely wouldn't be able to drive. (Male, 17, VI, early onset)
Psychological and emotional well-being	Emotions relating to living with visual impairment, Stigma associated with impairment and disability, Coping, Frustrations,	I used to get a bit frustrated when it was something which I couldn't see. (Male, 17, VI, early onset)		

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Psychological and emotional well-being continued Privacy, Acceptance of visual impairment, Uncertainty, Body and self-image, including appearance of the eye, wearing glasses/contact lenses, identifying as impaired or disabled, Self-esteem.	Psychological and emotional well-being continued My eyes had got worse. And then I had to adapt myself again. [] So it's like losing someone. Cos you're constantly having to adjust, like grieving I suppose. (Female, 19, SVI/BL, late onset) I know that eventually it's gonna get worse and worse.		

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
		Psychological and emotional well-being continued But it's something that I know about now so that by the time I reach that age, hopefully I'll have found ways to work around that. (Male, 17, VI, early onset) I just learn to get on with it. (Male, 16, SVI/BL, late onset)		
Future – aspirations and fears	Challenges, Restrictions, Opportunities for future education and career,	Then I thought, yeah, cos there are a lot of careers I can't do. (Male, 16, SVI/BL, early onset)	Growing acceptance of future impact of visual impairment,	In another life, or if I had vision, I would probably be a Royal Marine. It's one of those things where you can't do it. (Male, 16, SVI/BL, early onset)

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Future – aspirations and fears continued Independent living and mobility (especially driving and reliance on transport), Visual prognosis, Worries, concerns, uncertainties, Future relationships, e.g. dating, marriage and having children.	Future – aspirations and fears continued One day I could just be blind. So I hope for a cure but, at the same time, if it happens, it happens. Nothing I can do about it! (Male, 17, SVI/BL, early onset) I'd like to be a stockbroker or something like that. (Male, 16, VI, early onset)		Future – aspirations and fears continued [My visual impairment] narrows what kind of job prospects I could have. I wanted to join the police force. But that had to be knocked on the head pretty quickly cos you need perfect vision for that. (Male, 18, VI, early onset) Bringing [ <i>children</i> ] up. I'd have to do it on my own, in some respect. (Female, 19, SVI/BL, late onset)
Functioning – school, home and leisure	Feelings relating to activities and functioning.	I get a bit upset and disappointed when I can't		

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
		Functioning – school, home and leisure continued [play football]. I used to do tennis a lot with my friends and then realised I couldn't really play competitively with them. (Male, 17, VI, early onset) I used to play a lot of games, like on PlayStation, and stuff but now, I have tried a few times but it just gets, when you hear the sounds and stuff. And I remember some of the controls. But it's not the same. So that gets frustrating.		

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
		Functioning – school, home and leisure continued Especially if you're at your friends or someone's and they're all playing games and you just sort of, sitting there. (Male, 16, SVI/BL, late onset)		
Treatment of eye condition	Visits to the eye- clinic/hospital (e.g. prolonged waiting times, invasive eye examinations, eye tests, use of eye drops, communication with the doctor),	No I hated going [ <i>to</i> <i>hospital</i> ]. Hated it. (Male, 16, SVI/BL, early onset)	Transition from child- to adult-centred care.	[ <i>My visits to hospital</i> ] slowly tailed off as I got older. I was having an eye test, which then transferred down to just having an eye test at an opticians'. (Male, 17, VI, early onset)

VQoL domain⁵	Domain-pertinent issues identified from interviews previously conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Treatment of eye condition continued Related treatment, operations, and subsequent use of medication, Wider impact, such as missing school and subsequently getting behind with school work.	Treatment of eye condition continued It's annoying cos I'm usually in school and I always have to miss school. I have to go in and out of school. It's annoying cos I miss half the lesson. (Female, 16, VI, early onset) I never used to like [ <i>the</i> <i>eye drops</i> ]. They used to sting. (Male, 16, VI, late onset)		Treatment of eye condition continued I told you I hate children. When they're noisy. When I'm in the waiting room, they're all noisy and I'm 16. (Male, 16, SVI/BL, early onset)

Substantial overlap was found between the self-reported impact of visual impairment, and VQoL issues raised by younger children (aged 10-15 years) in the foundation research and YP (aged 16-19 years) (see Table 13, pg. 158), with both children and YP referencing similar physical and social environments, and daily challenges in light of visual impairment. For example, all children and YP discussed their relationships with friends and family members as valuable aspects of everyday life. However, differences emerged in the way YP described the frequency of these issues and emphasised them as aspects of everyday life and global well-being. For example, it is only during later adolescence (e.g. between the age of 16-19 years) that YP perceived the burden they place upon others when asking for help or needing care. Another example of the differences between child and YP self-report is evident when YP discussed the impact of their vision upon their ability to drive. Many children discussed this as an important barrier in the future. However YP discussed not being able to drive as an aspect of their everyday life which limits independence and inclusion with peers in conversations about driving lessons.

As a result of substantial overlap, the majority of new, age-appropriate codes mapped well onto the existing VQoL items developed in the foundation research. Although most codes reflected some degree of new, age-specific attitude and conceptualisation of visual impairment, the majority of existing VQoL items was deemed suitable for reflecting the underlying issue. In total, codes were mapped onto 31 existing VQoL items. These items covered the VQoL domains: school life, home life, social life, independence, psychological well-being, future aspirations and concerns, and experience of eye clinics. Five items included in the foundation VQoL instrument did not receive any new age-appropriate codes (un-coded) and 12 codes were not mapped onto existing VQoL items, and considered for inclusion as new age-specific items.

Ten VQoL items which had been removed during the foundation instrument development were entered into the first draft instruments based on the relevance of the new, age-appropriate codes. Codes which contributed to the inclusion of these VQoL

168

items primarily centred upon YP's experience of teasing, mocking, or bullying when at school, and awareness of the negative impact of visual impairment upon their future career development (see Figure 2, below).



Figure 2. Flowchart showing the origin of items included in the first draft VQoL instrument.

Using the codes, a draft VQoL instrument for YP aged 13-17 years was developed.

Table 14 (pg. 170) shows the items included in the first draft instrument.

	I have got some good friends
	I make new friends easily, despite my eyesight problems
	I get along with my family
	I can stand up for myself if someone picks on me
	My friends understand about my eyes
	I get treated the same as everyone else
	My friends help me at school
	I feel left out because of my eyesight
	I feel like I fit in
	My friends encourage me to join in their activities
	My teachers understand about my eyesight
	I feel different from other children
(1)	In spite of my eyesight, I am independent
<u> </u>	I am comfortable going places on my own
) sma	I can do most things on my own
g ite	People give me a chance to do things on my own
cistir	People overprotect me because of my eyesight
Û	I am given the freedom to do things on my own
	I am comfortable asking for help
	I have enough time to myself
	I feel frustrated because of my eyesight
	I feel lonely because of my eyesight
	I worry what other people think about my eyes
	I am positive about the future
	I am confident I will be able to look after myself when I'm older
	I worry my eyesight will get worse
	I like to have a go at everything, despite my eyesight
	I can get around on my own
	I like being at school
	1

Table 14. Items included in the first draft VQoL instrument.

	I have to work harder at school because of my eyesight
	I am happy with my social life
pa	I spend enough time with my friends
-cod	I cope well with my eyesight problems
un f	I am treated fairly
sting	I feel confident
Exi	My family encourage me
	I get picked on because of my eyesight
e late	I feel that having extra help at school is embarrassing
rom the ( <i>n</i> = 10)	I prefer being with other children who have eyesight problems
led fi	I have friends who don't have eyesight problems
colud	I have got used to living with my eyesight
ns e) deve	I do not let my eyesight stand in my way
s of e	I worry about what job I will be able to do in the future
atior	I have plans for the future
pund	I find going to the eye clinic helpful
Ъ.	I have a say in what happens to me at the eye clinic
	Strangers/public perceptions and other people understanding
	Resilience, having a go despite likelihood of failing
: 12)	Embarrassment in public
<i>u</i> ) :	Feeling more tired than others
odes	Dealing with limited choices and restrictions
ate c	Having to take extra time and care
propria	Accepting, adapting, adjusting, getting used to the impact of VI
ge-ap	Denial, coming to terms with a diagnosis
ew aç	Helplessness/vulnerability
ž	Guilt
	Pride, overcoming difficulties, succeeding, perseverance
	Missing school/disruption to routine because of eye clinic

Notably, not all of the items included in the foundation VQoL instrument were assigned codes. Despite this, they were entered into the first draft of the new, age-appropriate instruments. This was due to relevant literature and professional opinion indicating that they will likely be relevant items, despite not emerging as highly important for YP during interviews.

# 6.3.1.1 Outcomes from the expert consensus meeting concerning the VQoL instrument development

The first major outcome of the expert consensus meeting was the decision to modify the age-boundary for the new suite of instruments designed for older adolescents. This is described in more detail in Section 4.6.4.2 (pg. 101).

Following this decision, all items shown in Table 14 (pg. 170) were discussed for suitability and inclusion in the second draft instruments. Minor modifications were made to items to reflect the linguistic preferences of YP who took part in the first phase of instrument development. The term 'eyes' as used in the VQoL items 'My friends understand about my eyes' and 'I worry what other people think about my eyes' was replaced with the term 'eyesight' and 'children' as used in VQoL items 'I feel different from other children' and 'Other children pick on me because of my eyesight' was replaced with the term 'young people'. Other linguistic modifications were made to reflect the content of in-depth interviews (Table 15, pg. 173). For example, the words 'at school' in the item 'My friends help me at school' were replaced with 'when I need it' to reflect times when YP described receiving help from friends outside of school, for example when shopping or visiting the cinema. The majority of linguistic modifications made at this stage reflected the increased maturity of YP when compared to younger children.

One new VQoL item (I feel tired because of my eyesight) was developed during the expert consensus meeting in light of codes representing the physical impact of visual

impairment, triggering achy and sensitive physical sensations. YP often described these types of feelings as fatigue, for example,

"...it's tiring at the end of the day. [I get] so tired." (Female, 17 years, VI, early onset). Although participants were probed about their experience of clinical care, and codes relating to this experience were developed, items referencing eye clinics or hospital visits were excluded from the second draft instruments based on the premise that these items measure the experience of healthcare as opposed to VQoL or FV, and are therefore better suited for a PREM (see Section 2.3.1, pg. 31).

Table 15. Minor linguistic modifications made to VQoL items during the expert consensus meeting.

Original VQoL item	Modified VQoL item
My friends help me at school	My friends help me when I need it
In spite of my eyesight, I am independent	I am independent
People give me a chance to do things on	People give me the opportunity to do
my own	things on my own
I worry about what other people think	I worry what other people think of me
about my eyes	because of my eyesight
I like to have a go at everything, despite my eyesight	I like to have a go at everything
I like being at school	I enjoy school/college

# 6.3.2 Phase 2: Pre-testing

# 6.3.2.1 Self-reported feasibility of the VQoL instrument

The majority of participants described the process of VQoL questionnaire completion as 'easy' or 'very easy'. Some participants expressed concerns that the draft instrument took too long to complete, however it was unclear in several circumstances whether this was due to the lengthy expert consultation, and participants' misunderstanding of how they would be expected to complete the instruments in clinical practice. When probed about the context in which they would prefer to complete the instruments, several participants described preference of completing the VQoL instrument when waiting for their hospital appointments:

"There are a lot of questions but I always have to wait a long time before clinics so I have time to do it then." (Female, 18 years, VI, early onset).

In each of these circumstances, efforts were made to ensure participants understood the nature of the expert consultation, that a phase of item-reduction was due to take place following pre-testing, and that when used in practice, patients would not necessarily be required to justify their answers in the same way as they were during expert consultations. Once this had been clarified, most participants agreed that the instrument was both easy and feasible for completion in clinical contexts.

#### 6.3.2.2 VQoL item modification

Qualitative feedback on individual items ensured participants interpreted each item correctly in relation to their visual impairment. Most participants took the opportunity to elaborate upon each item, relating it to aspects of everyday life and thus confirming content validity. The majority of items were endorsed as meaningful to participants, with participants giving explanations of why they had chosen their answer and factors which influenced their response.

#### 6.3.2.2.1 Linguistic modifications

The majority of modified items shown in Table 16 (pg. 176) reflect the preferred linguistic terminology used by participants during the pre-testing consultations, providing an extension and validating the linguistic changes made as a result of the expert consensus meeting conducted in phase 1 (see Section 6.3.1.1, pg. 172). The term 'tutor' was added in a number of items to reflect participants' labelling of teachers at college. The term 'opportunity' was changed to 'chance' because a number of participants needed help interpreting the meaning of the word 'opportunity' in this context.

#### 6.3.2.2.2 Modifications to aid interpretation and ease of responding

Whilst items were judged as meaningful and relevant to the everyday lives of participants, many YP had difficulty summarising their daily challenges by picking one response option. Frequently, participants asked if they were allowed to choose two response options (e.g. 'A little bit true' and 'Mostly true'), or the mid-point between two response options.

When probed as to why they felt they were unable to choose one response option, participants often indicated that their answers were context and situation dependent, varying in relation to their immediate environment and help provided by other people. For example, when asked about the item 'I have to work harder at school because of my eyesight', one participant responded:

"Obviously to a certain degree, yes. It takes me longer. I do extra at home but I like to do extra. It also depends on the subject and place." (Male, 16 years, VI, early onset).

As a result of these difficulties, a number of items were modified to ensure YP are capable of answering each item using a single response option, whilst at the same time, allowing for some degree of flexibility and application to multiple contexts and settings. This was done in some cases by reducing the breadth of the item e.g. by adding 'by my friends' in the VQoL item 'I am treated fairly'.

One item (I am independent) was modified and separated into two items to account for the different contexts in which participants described their answers (Table 16, pg. 176). Table 16. Items which were modified as a result of phase 2: pre-testing.

Original YP-VQoL item	Modified YP-VQoL item
My teachers understand how things are	My teachers and tutors understand how
for me because of my eyesight	things are for me because of my eyesight
I am independent	I am independent at home
	I am independent at school
I can do most things on my own	I can do most activities on my own
People give me the opportunity to do	People give me the chance to do things
things on my own	on my own
I have enough time to myself	I have enough private time to myself
I am treated fairly	I am treated fairly by my friends
I am confident I will be able to look after	I am confident I will be able to look after
myself when I'm older	myself in the future

A second theme arising from pre-testing the VQoL instrument was the extent to which participants required reminding that items should be answered in relation to their visual impairment. For example, when asked to justify her answer to the item 'I enjoy school/college', one participant responded:

*"I hate school. Always hated the people at school but this is not to do with my eyesight." (Female, 15 years, VI, early onset).* 

As a result of this difficulty, the phrase 'Remember to say how things are for you

because of your eyesight and how much they are true about you' was added as a

header on each page of the VQoL instrument.

### 6.3.2.2.3 Further item development

Notably, one new VQoL item was added to the item pool during phase 2. This item (I keep new friends easily) was developed initially in light of in-depth interviews with children (aged 6-9 years) which were running in parallel to the interviews with YP. However, this item was also deemed applicable to YP (aged 16-19 years) and suitable

for detecting the difficulties YP discussed when maintaining romantic relationships (see Table 13, pg. 158). Having a boy/girlfriend was discussed as a normative aspect of growing up (see Section 5.2.2.2, pg. 130), however, YP described the difficulties they had when maintaining these relationships in light of deteriorating visual function. When probed about the relevance of this item during phase 2, the majority of participants agreed it should be added to the item pool, and interpreted it with relevance to visual impairment.

#### 6.3.2.3 Changes to response options

During phase 2, a number of participants demonstrated difficulty interpreting and applying the third response option in the VQoL instrument: 'quite a bit true'. This was particularly difficult for some participants to interpret when answering items which had been reverse worded (e.g. Other young people my age pick on me because of my eyesight). When choosing option 3, one participant frequently re-phrased the response option to "mostly true" (Female, 18 years, VI, early onset). The modified wording of this response option was tested with a number participants during phase 2 and deemed an acceptable replacement.

#### 6.3.2.4 Outcomes from the expert consensus meeting concerning the

#### VQoL instrument development

Experts who took part in the consensus meeting confirmed all modifications to existing items, and the development of new items made in phase 2.

During the consensus meeting, changes to the response options of each instrument were discussed, including the modification of the VQoL response option.

## 6.3.3. Phase 3: Piloting

#### 6.3.3.1 Qualitative feedback

Of the 23 participants, 10 (43.48%) provided qualitative feedback on the instruments using the blank page which was inserted at the end of the questionnaire booklet. Most

of them took the opportunity to describe their experience of living with visual impairment. For example, one participant described their experience of mainstream and specialised-VI school, offering advice for other YP living with visual impairment:

"I think it is important to note that my experiences in a school which is specially designed for the blind are significantly better than my experience in mainstream schools due to the far higher understanding of my disabilities and visual impairment. I think blind children would gain much more confidence, teaching and support if they went to a school like mine because in mainstream schools there is simply not enough resources to build these sufficiently." (Male, 14 years, SVI/BL, early onset).

#### 6.3.3.2 Quantitative feedback

#### 6.3.3.2.1 Missing data

The 'true' missing data (i.e. no response given) per person for the YP-VQoL instrument was  $\leq 10.26\%$  and considered missing completely at random.<sup>378</sup> Similarly, the 'true' missing data per item was  $\leq 13.04\%$  with three responses missing on the VQoL items 'I am independent at home' and 'I am independent at school/college', which were missing at random. This was a result of a modification made to specify context in the original item 'I am independent' after three guestionnaire booklets had been sent out.

#### 6.3.3.2.2 Use of response categories

Participants endorsed the response options well, with the majority opting for a single response option to describe their experience of visual impairment. However, one participant demonstrated difficulty opting for a single response option to describe their experience and, on 4 occasions provided the response option 'between 2 and 3' (accounting for 10.26% of this participants' total VQoL responses), and demonstrated difficulty conceptualising the difference between 'A little bit true' and 'Mostly true'. From this point onward, phase 3 data analysis was conducted with these responses coded as missing data. No further analyses were conducted to impute missing data. No participants were removed because of missing data.

178

#### 5.3.3.2.3 Feasibility

For the VQoL instrument, 17 YP (73.91% of respondents) reported the time it had taken them to complete the instrument. This ranged from 3-60 minutes (Median = 10, IQR = 23.75).

When asked how easy it was for them to complete, 95% of YP rated completion of the VQoL instrument as 'easy' or 'very easy' and 100% rated the instructions as 'easy' or 'very easy' to understand.

#### 6.3.3.3 Response distribution in the VQoL instrument

#### 6.3.3.3.1 Skewness and kurtosis in the VQoL instrument

Seven VQoL items were flagged as problematic due to z-score for skewness and/or kurtosis outside the acceptable limits (-2.00 to +2.00).<sup>313</sup> All of these items originated from the early stages of the foundation research (I have got some good friends, Other young people my age pick on me because of my eyesight, I feel left out because of my eyesight, I get along with my family, I have enough private time to myself, I feel lonely because of my eyesight, I like to have a go at everything). One item (Other young people my age pick on me because of my eyesight) was included in the draft instrument having been excluded in the later stages of the foundation research.

#### 6.3.3.3.2 Floor and ceiling effects in the VQoL instrument

When analysed for floor and ceiling effects by calculating the percent of participants who endorsed the extreme categories (i.e. option 1 or 4), 3 VQoL items were flagged as problematic with >60% of participants choosing response option 4 (Completely true). Each item had been flagged as problematic due to high skewness and originated from the foundation research (I have got some good friends, I get along with my family, I have enough private time to myself). No participants endorsed response category 4 when answering one VQoL item (I feel lonely because of my eyesight) (see Figure 3, pg. 181).

One VQoL item was flagged as problematic due to floor effects (i.e. >60% of respondents choosing option 1 (Not at all true)). This item originated from the foundation research (I feel lonely because of my eyesight). No participants endorsed response category 1 when answering 12 VQoL items (I have got some good friends, I can stand up for myself if someone picks on me, My friends help me when I need it, My teachers and tutors understand how things are for me because of my eyesight, I get along with my family, I am independent at home, I am independent at school/college, I have enough private time to myself, I cope well with my eyesight problems, I am treated fairly by my friends, I am positive about the future, I can get around on my own).


Figure 3. Percent of YP choosing extreme response categories when answering items in the VQoL instrument (1% and 60% thresholds highlighted in red).

Because the aim of phase 3 was to identify any immediate problems with the instruments in preparation for the larger scale postal survey conducted in phase 4, and because of the small sample recruited, no items were removed as a result of response distribution.

# 6.3.4 Phase 4: Validation

Table 17 (pg. 183) shows the VQoL items entered into the questionnaire booklets which were sent to children (aged 8-12 years) and YP (aged 13-17 years) in the final validation phase. Each item was assigned a value corresponding to the order of presentation within each instrument. Items presented in the same row of Table 17 are overlapping or 'core' items which are present in instruments designed for both age groups. Items parallel to empty cells are age-specific items, presented to only one age group.

Table 17. Items included in the Child (8-12 years) and YP (13-17 years) version of the VQoL instruments entered into phase 4 (psychometric validation).

	VQoL in	strumen	t
	Child (8-12 years)		YP (13-17 years)
Item	Item	Item	Item
no.		no.	
1	I have got some good friends	1	I have got some good friends
2	I make new friends easily	2	I make new friends easily
3	I keep friends easily	3	I keep friends easily
4	I am happy with how many friends I have	4	I am happy with my social life
5	I spend enough time with my friends	5	I spend enough time with my friends
6	Other children pick on me because of my eyesight	6	Other young people my age pick on me because of my eyesight
7	I can stand up for myself if someone picks on me	7	I can stand up for myself if someone picks on me
8	My friends understand how things are for me because of my eyesight	8	My friends understand how things are for me because of my eyesight
9	My friends help me at school	9	My friends help me when I need it
		10	I get treated the same as everyone else
		11	I feel like I fit in

	VQoL ir	nstrumer	nt
	Child (8-12 years)		YP (13-17 years)
Item no.	Item	Item no.	Item
10	My friends encourage me to join in their activities	12	My friends encourage me to join in their activities
11	I feel different from other children because of my eyesight	13	I feel different from other young people because of my eyesight
12	I feel left out because of my eyesight	14	I feel left out because of my eyesight
13	My teachers understand how things are for me because of my eyesight	15	My teachers and tutors understand how things are for me because of my eyesight
14	I get along with my family	16	I get along with my family
15	I can decide things for myself		
16	I am independent at home	17	I am independent at home
17	I am independent at school	18	I am independent at school/college
		19	I am comfortable going places on my own
18	People give me a chance to do things for myself	21	People give me the chance to do things on my own
		22	People overprotect me because of my eyesight
19	I am happy asking for help	23	I am comfortable asking for help
		24	I have enough private time to myself

	VQoL ins	strumen	t
	Child (8-12 years)		YP (13-17 years)
ltem no.	Item	Item no.	Item
20	I cope well with my eyesight problems	25	I cope well with my eyesight problems
21	I feel tired because of my eyesight	26	I feel tired because of my eyesight
22	I feel frustrated because of my eyesight	27	I feel frustrated because of my eyesight
23	I feel lonely because of my eyesight	28	I feel lonely because of my eyesight
24	I feel confident	29	I feel confident
25	Other people are fair to me	30	I am treated fairly by my friends
26	I worry what other people think of me because of my eyesight	31	I worry what other people think of me because of my eyesight
		32	I am positive about the future
		33	I am confident I will be able to look after myself in the future
		34	I worry my eyesight will get worse
		35	I worry about what job I will be able to do in the future
27	I like to have a go at everything, although my eyesight isn't perfect	36	I like to have a go at everything
		37	I can get around on my own
28	I like being at school	38	I enjoy school/college

	VQoL ins	strumen	t
	Child (8-12 years)		YP (13-17 years)
Item	Item	Item	Item
no.		no.	
29	I have to work harder at school because of my eyesight	39	I have to work harder at school/college because of my eyesight
30	I can do most activities on my own	20	I can do most activities on my own

#### 6.3.4.1 Data verification

The full YP-VQoL and PedsQL dataset was verified by the same researcher (AR) to validate the data entry. A total of 4 incorrect data entries were found out of 4,331 individual entries, constituting 0.09% of the dataset. Each incorrect entry was cross-referenced with the original questionnaire booklets and corrected. Ten percent of this dataset was also verified by a second member of the research team (VT) who found 100% of the data to have been entered correctly.

With regards to the Child-VQoL dataset, 43% of data entries were verified by the same researcher who entered the data (VT). No incorrect data entries were found. Accordingly, 10% of the data entries were verified independently by a second member of the research team (AR): 100% were found to be correct.

### 6.3.4.2 Missing data

#### 6.3.4.2.1 Missing data per person

With regards to missing data, participants were excluded from further analysis if they had >25% missing data, as follows: three children were excluded, leaving 83 included in the Child-VQoL phase 4 analysis. After excluding participants on the basis of not returning completed consent forms and/or booklets (see Section 6.2, pg. 150), no YP were excluded as a result of missing data. Seventy-one YP were included in the YP-VQoL phase 4 analysis.

#### 6.3.4.2.2 Missing data per VQoL item

Missing data was also calculated per item. After removing participants based on missing data, the largest percent of missing data per item was 6.02% for Child-VQoL Item 30 (I can do most activities on my own). This data was missing at random due to addition of the item into the VQoL instrument in the later stages of phase 4 following the recruitment of 4 participants. With regards to the YP instruments, the largest percent of missing data per YP-VQoL item was 2.82% for the item 'I am independent at

school/college'. All remaining missing data was considered missing completely at random<sup>378</sup> (see Table 18, below).

	VQoL in	strument	
Child	(8-12 years)	YP (1	3-17 years)
ltem no.	Missing data, <i>n</i> (%)	ltem no.	Missing data, <i>n</i> (%)
30*	5 (6.02)	18	2 (2.82)
21	2 (2.41)	39	1 (1.41)
20	2 (2.41)	33	1 (1.41)
16	2 (2.41)	27	1 (1.41)
13	2 (2.41)	19	1 (1.41)
9	2 (2.41)	14	1 (1.41)
6	2 (2.41)	12	1 (1.41)
3	2 (2.41)	9	1 (1.41)
25	1 (1.2)	6	1 (1.41)
24	1 (1.2)		
22	1 (1.2)		
19	1 (1.2)		
18	1 (1.2)		
17	1 (1.2)		
11	1 (1.2)		
8	1 (1.2)		
7	1 (1.2)		
5	1 (1.2)		
4	1 (1.2)		
2	1 (1.2)		
1	1 (1.2)		

Table 18. Missing data per item in the VQoL instrument, separated by age-group.

\* Items with data missing at random (i.e. not included in some questionnaire booklets) versus missing completely at random (i.e. respondent did not answer).

### 6.3.4.3 Use of response categories

Similar to phase 3 results, participants appeared to endorse response categories well. However, a number of participants indicated on multiple occasions their response to be in between two categories, or a combination of responses. Multiple response options constituted 0.88% of Child-VQoL, and 0.43% of YP-VQoL data. In these cases, responses were re-coded using the values shown in Table 7 (pg. 113).

### 6.3.4.4 Response distribution

A substantial number of both child- and YP-VQoL (see Table 19, pg. 190) items were shown to have z-score skewness and/or z-score kurtosis values outside the acceptable range of -2 to +2. These items were flagged as problematic.

The distribution of responses to the child- and YP-VQoL items (see Table 19, pg. 190) demonstrated a number of items with floor/ceiling effects and redundant response categories.

Table 19. Z-score skewness, Z-score kurtosis, floor and ceiling effects for each item in

the VQoL instrument.

Item	Z-	Z-score	%	%	Item	Z-	Z-score	%	%
110.	skew	KUIIOSIS	option 1	option 4	110.	skew	KUITOSIS	option 1	option 4
		VQoL (	Child)				VQoL	(YP)	
1	-4.02	1.08	2.4	51.8	1	-3.23	-0.56	1.4	53.5
2	-0.62	-1.93	12	25.3	2	-1.27	-1.65	16.9	22.5
3	-2.33	-0.62	2.4	38.6	3	-2.69	0.58	2.8	36.6
4	-5.01	1.41	6	60.2	4	-3.17	0.04	9.9	36.6
5	-2.8	-0.48	4.8	41	5	-2.14	-1.09	12.7	32.4
6	4.63	1.58	49.4	7.2	6	4.79	2.36	54.9	4.2
7	-1.9	-1.92	10.8	39.8	7	-3.15	-0.25	8.5	42.3
8	-2.86	-0.3	6	38.6	8	-2.28	-1.24	9.9	40.8
9	-2.37	-0.26	4.8	32.5	9	-3.19	-0.55	0	57.7
10	-0.99	-1.68	8.4	27.7	10	-2.26	-0.84	7	38
11	-0.11	-2.57	19.3	28.9	11	-2.67	-1.13	9.9	43.7
12	3.37	-0.24	43.4	8.4	12	-3.19	0.13	5.6	42.3
13	-4.05	0.08	3.6	56.6	13	0.59	-1.93	22.5	18.3
14	-6.79	4.73	0	74.7	14	3.43	0.23	43.7	7
15	-3.92	1.74	2.4	48.2	15	-2.14	-1.22	5.6	42.3
16	-2.65	-0.57	2.4	36.1	16	-5.69	1.03	0	80.3
17	-0.9	-0.78	1.2	25.3	17	-2.13	0.17	2.8	32.4
18	-1.61	-0.81	1.2	33.7	18	-1.54	-0.32	4.2	23.9
19	-3.26	0.33	3.6	43.4	19	-0.52	-2.46	25.4	23.9
20	-2.99	-0.76	1.2	49.4	20	-1.36	-1.41	9.9	28.2
21	0.43	-2.37	22.9	21.7	21	-2.43	0.85	4.2	28.2
22	0.82	-1.99	36.5	13.3	22	-0.83	-1.69	22.5	15.5
23	5.7	3.06	57.8	7.2	23	-1.77	-0.99	7	31
24	-2.68	-0.12	4.8	36.1	24	-3.45	-0.5	0	59.2
25	-1.88	-0.49	3.6	30.1	25	-3.28	0.5	5.6	40.8
26	2.08	-2.12	41	16.9	26	0.86	-2.01	22.5	21.1
27	-2.78	-0.82	0	53	27	1.31	-1.57	11.3	21.1

ltem no.	Z- score skew	Z-score kurtosis	% endorsing option 1	% endorsing option 4	ltem no.	Z- score skew	Z-score kurtosis	% endorsing option 1	% endorsing option 4
		VQoL (	Child)				VQoL	(YP)	
28	-2.25	-0.5	2.4	37.3	28	6.16	4.67	62	5.6
29	-1.89	-1.85	12	37.3	29	-1.30	-1.55	11.3	26.8
30	-1.86	-0.35	3.6	27.7	30	-4.56	2.51	1.4	57.7
	1	L			31	1.13	-2.12	28.2	19.7
					32	-1.9	-1.23	11.3	31
					33	-1.68	-1.53	8.5	35.2
					34	1.24	-2.21	28.2	22.5
					35	0.53	-2.02	22.5	18.3
					36	-2.41	-0.9	8.5	38
					37	-1.15	-1.69	8.5	31
					38	-2.26	-1.09	8.5	38
					39	-1.24	-2.15	12.7	38

\*VQoL response option 1 = 'Not at all true', response option 4 = 'Completely true'

The percent of respondents endorsing extreme categories in each item was analysed in parallel to the largest, and most problematic, skew and kurtosis values and the childand YP-VQoL items shown in Table 20 (below) were removed from the subsequent stages of item reduction.

Table 20. Items excluded as a result of response distribution.

	VQoL (child)		VQoL (YP)
ltem no.	Item	ltem no.	Item
14	I get along with my family	16	I get along with my family
27	I like to have a go at everything	9	My friends help me when I need it
		24	I have enough private time to myself
		28	I feel lonely because of my eyesight

# 6.3.4.5.1 FA

The remaining items in each VQoL instrument version were entered into a formal exploratory FA to assess for unidimensionality. FA was conducted using each of the 5 datasets with imputed missing data for each instrument, and the outcomes reported are the mean values of each dataset (see Table 21, below).

Notably, the mean number of items loading <0.4 on the first/largest factor in each imputed dataset indicates some items which do not contribute to the latent variable within each instrument. Despite this, all items included in the Child-VQoL, loaded positively onto the first factor.

In the YP-VQoL dataset one item (Item 15 (My teachers and tutors understand how things are for me because of my eyesight)) did not load positively onto the first factor in any of the imputed datasets and was therefore outstanding as a potentially problematic item, as shown in Table 21 (below).

Table 21. Outcome from FA conducted to assess dimensionality in the VQoL

instruments.

Instrument version	Eigenvalue for the first (largest) factor (% of variance explained)	Variation between eigenvalues in imputed datasets	No. of eigenvalues >1	No. of items loading onto the first (largest) factor <.4	Items not Ioading positively onto the first (largest) factor
Child- VQoL	8.19 (29.26)	0.06	8.8	3	None
YP-VQoL	12.26 (35.02)	0.15	8.6	2.4	Item 15 (My teachers and tutors understand how things are for me because of my eyesight)

# 6.3.4.5.2 Parallel analysis

Consistent with the method used to handle multiple imputations of missing data when performing FA, parallel analysis was run using each of the five imputed datasets. The mean eigenvalues for the first five eigenvalues in each dataset are presented in Table 22 (below).

Table 22. Average of the first five eigenvalues produced by parallel analysis for each VQoL dataset.

Actual eigenvalue	Average eigenvalue	95 <sup>th</sup> Percentile eigenvalue
Child-VQoL		
8.19	2.27	2.47
2.15	2.06	2.2
1.87	1.91	2.03
1.68	1.78	1.88
1.48	1.66	1.76
YP-VQoL		
12.26	2.64	2.88
3.27	2.4	2.56
2.28	2.22	2.36
1.89	2.07	2.19
1.5	1.94	2.05

The matrix produced by parallel analysis can be used to assess dimensionality by judging the magnitude of the actual eigenvalue in relation to the average and 95<sup>th</sup> percentile eigenvalues. If the actual eigenvalue is found to be greater than both the average and 95<sup>th</sup> percentile eigenvalues, then the eigenvalue is interpreted as a factor. Thus, only one actual eigenvalue greater than the corresponding average and 95<sup>th</sup> percentile eigenvalue is thought to represent unidimensionality. In the Child-VQoL dataset the average and percentile eigenvalues were discordant in relation to the actual eigenvalue. In this case, the actual eigenvalue was compared to the percentile eigenvalue only as per recommendations in light of PA's tendency to overestimate the number of factors.<sup>400</sup> Thus, the Child-VQoL dataset was deemed unidimensional.

The YP-VQoL dataset was shown to be multidimensional according to the matrix. The largest actual eigenvalue for this dataset was larger than both the average and percentile eigenvalues however, the second largest actual eigenvalue was also

194

marginally larger than both alternative eigenvalues (see Table 22, pg. 200). Despite this finding, the YP-VQoL dataset was deemed suitable for Rasch analysis based on evidence suggesting unidimensionality in the corresponding Child-VQoL instrument, and aims to develop the instruments in tandem as complementary measures.

The values produced by the averaged matrix (see Table 22, pg. 194) were used to produce a scree plot for each dataset, confirming the results from the matrix (see Figure 4, pg. 196 and 5, pg. 197).



Figure 4. Screeplot showing actual, average and percentile eigenvalues in the Child-VQoL instrument.



Figure 5. Screeplot showing actual, average and percentile eigenvalues in the YP-VQoL instrument.

### 6.3.4.6 Rasch analysis

Rasch analysis was conducted using Child- and YP-VQoL datasets independently, and then using combined Child- and YP-VQoL datasets, with the aim of calibrating the instruments. Results are presented in relation to the Child-VQoL, and YP-VQoL instruments separately before combining and calibrating instruments.

# 6.3.4.7 Rasch analysis using the 28-item Child-VQoL instrument

A total of 4 iterations were required before the Child-VQoL instrument met the functional criteria of Rasch analysis, indicating good measurement capacity as follows:

## 6.3.4.7.1 Item fit to the Rasch measurement scale

Item 9 was the first item to be removed during this analysis, due to an OUTFIT meansquare value (MNSQ) which was above the threshold of 1.5 (see Table 23, below). Fit statistics for all remaining items fell within the range of 0.5 to 1.5 (see Table 24, pg. 200), demonstrating good fit of the model of measurement to the data.

Table 23. Problematic items in the Child-VQoL instrument due to fit to the Rasch measurement scale.

			INFIT	C	UTFIT
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z-standardised (ZSTD)	Mean- square value (MNSQ)	Z-standardised (ZSTD)
Iteration 1 (Excluding Item 14 (I get along with my family) and 27 (I like to have a go at everything, although my eyesight isn't perfect))	9 (My friends help me at school)	1.23	1.5	1.72	3.9

### 6.3.4.7.2 Order of person abilities

Three further items (Item 30 (I can do most activities on my own), 24 (I feel confident) and 23 (I feel lonely because of my eyesight)) emerged as problematic when assessing the observed, sample-dependent, average measure of persons (relative to each item) in this analysis who responded in each category. In each case, the average measure for a higher score value was lower than for a lower score value. This finding contradicts the assumption that a higher level of VQoL predicts a higher score in the instrument, and therefore refutes the underlying principle of additive measurement in this instrument. As a result, three further iterations were used to assess the measurement properties of the Child-VQoL instrument without these three items. Fit statistics were assessed after the removal of each item, to determine the impact upon the existing good-fitting items.

### 6.3.4.7.3 Differential item functioning (DIF)

Analysis of DIF by gender and age was performed using participants stratified by age group, using the age categories 'young' (7-9 years) and 'old' (10-13 years). This stratification was necessary to produce two age groups roughly comparable in terms of sample size, and big enough to detect realistic DIF values.

This analysis revealed no DIF contrasts above the threshold of 1.0 (see Table 24, pg. 200). Thus, the final iteration of the Child-VQoL instrument contained 24 items. The measurement properties of the final iteration were checked for conformity to the predefined thresholds, and fit statistics were within acceptable limits (see Table 24, pg. 200). The average measures of persons were ordered in each item in relation to ability.

|--|

Item code	Child-VQoL item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
1	I have got some good friends	0.84	0.75	-0.39	-0.34
2	I make new friends easily	0.89	0.87	0.23	-0.12
3	I keep friends easily	0.79	0.79	-0.15	-0.05
4	I am happy with how many friends I have	1.29	1.09	0.48	-0.28
5	I spend enough time with my friends	0.90	0.83	0.26	0
6	Other children pick on me because of my eyesight	1.05	1.06	-0.55	0.45
7	I can stand up for myself if someone picks on me	1.23	1.19	0.07	0.27
8	My friends understand how things are for me because of my eyesight	0.98	1.01	-0.25	0.24
10	My friends encourage me to join in their activities	1.22	1.45	-0.06	0.45
11	I feel different from other children because of my eyesight	0.99	1.04	-0.09	0.20
12	I feel left out because of my eyesight	0.71	0.66	0.02	0.15
13	My teachers understand how things are for me because of my eyesight	1.31	1.32	0.41	-0.21
15	I can decide things for myself	1.01	1.03	-0.80	0.28
16	I am independent at home	0.94	0.97	-0.19	0.21
17	I am independent at school	0.94	0.98	-0.05	0.31
18	People give me a chance to do things for myself	0.76	0.73	-0.26	-0.03

Item code	Child-VQoL item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
19	I am happy asking for help	1.08	0.98	0.47	-0.05
20	I cope well with my eyesight problems	0.99	0.93	0.66	-0.67
21	I feel tired because of my eyesight	1.35	1.41	-0.39	-0.56
22	I feel frustrated because of my eyesight	1	1.10	0	-0.06
25	Other people are fair to me	0.65	0.62	-0.09	0.07
26	I worry about what other people think about me because of my eyesight	1.25	1.36	0.51	0.21
28	I like being at school	0.98	0.93	0.27	-0.31
29	I have to work harder at school because of my eyesight	1.14	1.17	0	-0.31

### 6.3.4.7.4 Targeting of items to persons

The item-person map revealed good targeting of the 24 items in the scale to the responders. The difference between the item and person means were .91 which is within the acceptable limit of 1 (see Figure 6, pg. 202). The person separation and reliability for this iteration were analysed using the real root-mean-square standard error (RMSE) as an indicator of the worst case scenario but were within the acceptable limits (separation = 3.46, reliability = .92), indicating high measurement precision of the Child-VQoL instrument.



Figure 6. Item-person targeting of the final 24-item Child-VQoL scale.

#### 6.3.4.7.5 Response category function

Item category probability plots produced using Winsteps were assessed individually for both the forward and reverse worded items in the 24-item scale and the step calibrations were used to complement findings. The category probability plots for the forward worded items were well ordered and Andrich thresholds increased by at least 1.13 logits across the scale, which is marginally less that the recommended increase of1.4, showing that despite coherent order of the response categories, the 4 categoriesmay not be optimally defined.



Figure 7. Category probability curves for the 4 response categories in the forwardscored Child-VQoL instrument (1 = Not at all true, 2 = A little bit true, 3 = Mostly true, 4 = Completely true).

When analysed in relation to the reverse worded items, the category probability plots were also well ordered. However, response category 3 (Mostly true) was the least likely response category to be observed, with the majority of the curve remaining submerged beneath the curve for response option 4 (Completely true). Despite the small likelihood of respondents choosing response option 3, the increase in Andrich threshold was .16 between response option 3 and 2, indicating that the response categories are ordered

as expected. The increase in Andrich threshold between response categories 1 and 2 was .45, and between 3 and 4 was 1.03.



Figure 8. Category probability curves for the 4 response categories in the reversescored Child-VQoL items (1 = Not at all true, 2 = A little bit true, 3 = Mostly true, 4 = Completely true).

#### 6.3.4.8 Rasch analysis using the 35-item YP-VQoL instrument

After removing 4 items as a result of preliminary item removal, rendering a total of 35 items in the YP-VQoL instrument, 8 iterations were required before the items conformed to the criteria of Rasch analysis. During these iterations, items were removed sequentially. One iteration (iteration 4) was performed with one previously removed item (Item 22 (People overprotect me because of my eyesight)) retained. This was with the aim of comparing outcomes with and without Items 22 and 26 (I feel tired

because of my eyesight) which were both shown to violate the criteria of good fit to the model.

### 6.3.4.8.1 Item fit to the Rasch measurement scale

The first items which were removed from the YP-VQoL instrument were Items 15 (My teachers and tutors understand how things are for me because of my eyesight), 22 and 26. These items were removed as a result of large OUTFIT MNSQ statistics. Table 25 (pg. 206) shows the INFIT and OUTFIT statistics for problematic items during each of the early iterations (iterations 1-5). After removing Items 15, 22 and 26, a decision was made to stop removing items based purely upon FIT statistics, as the remaining FIT statistics deviated only by a small percent outside the threshold of 0.5 - 1.5.

Table 25. Problematic items in the YP-VQoL instrument due to fit to the Rasch

measurement scale.

			INFIT	OUTFIT		
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z-standardised (ZSTD)	Mean- square vale (MNSQ)	Z-standardised (ZSTD)	
Iteration 1 (Excluding Items 9 (My friends help me when I need it), 16 (I get along	15 (My teachers and tutors understand how things are for me because of my eyesight)	1.87	4.3	2.26	5.2	
family), 24 (I have enough private	22 (People overprotect me because of my eyesight)	1.36	2.3	2.1	5.1	
time to myself) and 28 (I feel lonely because of my eyesight))	26 (I feel tired because of my eyesight)	1.51	3	1.59	3.1	
Iteration 2 (Excluding	22 (see above)	1.41	2.5	2.55	6.5	
16, 24, 28 and 15)	26 (see above)	1.55	3.2	1.63	3.2	
Iteration 3 (Excluding Items 9, 16, 24, 28, 15 and 22)	26 (see above)	1.58	3.3	1.67	3.2	
Iteration 4 (Excluding	22 (see above)	1.44	2.6	2.57	6.6	
16, 24, 28, 15 and 26)	8 (My friends understand how things are for me	1.29	1.7	1.56	2.7	

		INFIT		OUTFIT		
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z-standardised (ZSTD)	Mean- square vale (MNSQ)	Z-standardised (ZSTD)	
	Iteration 4 (Excluding Items 9, 16, 24, 28, 15 and 26 continued because of my eyesight)					
	34 (I worry about what job I will be able to do in the future)	1.53	3	1.5	2.6	
Iteration 5 (Excluding	34 (see above)	1.54	3.1	1.5	2.5	
16, 24, 28, 15, 26 and 22)	8 (see above)	1.28	1.6	1.53	2.5	

## 6.3.4.8.2 Order of person abilities

After removing these three items based on FIT statistics, Items 30 (I am treated fairly by my friends), 21 (People give me the chance to do things on my own) and 12 (My friends encourage me to join in their activities) were flagged as problematic and removed due to disordering of the average measures given to participants in relation to participants' average ability. These items were removed iteratively whilst monitoring the impact of removal upon the fit statistics of the remaining items.

## 6.3.4.8.3 Differential item functioning

The remaining 29 items were entered into a DIF analysis by gender and age.

DIF analysis by gender indicated a moderate to large DIF effect for one item (Item 6 (Other young people my age pick on me because of my eyesight)) for which males

were 1.07 logits less able to endorse as true in this sample and 2 items (Item 19 (I am comfortable going places on my own) and 34 (I worry my eyesight will get worse)) for which females were over 1 logit less able to endorse as true than males.

Before removing any items based on DIF by gender, DIF analysis by age was run using participants stratified to young (13-15 years) and old (16-18 years) age groups. Like the stratification of children in Section 6.3.4.7.3 (pg. 199), this stratification was needed to ensure age-groups were large enough to be compared and produce realistic DIF statistics. The cut off of 15 years was chosen based partly on the size of participants within each age group and producing two groups which are roughly equal in size, and also on the empirical findings presented in Chapter 5 (pg. 125) with regard to changes which may occur at the age of 16 years. DIF analysis by age group indicated one item (Item 7 (I can stand up for myself if someone picks on me)) which was 1.15 logits more difficult for participants in the younger age group to endorse as true.

Three problematic items highlighted by DIF analysis (Item 19, 34 and 7) were removed iteratively and in order of magnitude of DIF contrast (ranging from largest to smallest). FIT statistics were checked after removing each item and DIF analyses were re-run using each iteration. After removing Items 19, 34 and 7, the DIF contrast belonging to Item 6 was reduced to below the threshold of 1 logit. As a result, Item 6 was retained. When checking for DIF by gender using the final iteration, one further item (Item 37 (I can get around on my own)) was flagged as problematic with a DIF contrast of 1.03, indicating greater difficulty for females in this sample. However, this DIF contrast was judged as negligible, and the item was retained in light of preserving the good fit statistics of the remaining items.

The final iteration of the YP-VQoL instrument contained 26 items. The FIT statistics and DIF contrasts of the items included in the final iteration are displayed in Table 26 (pg. 210). Notably, two items (Item 8 (My friends understand how things are for me because of my eyesight) and 39 (I have to work harder at school/college because of

208

my eyesight)) have been retained despite a small divergence in FIT statistic from the threshold of 0.5-1.5. The suitability for inclusion of these two items, despite high OUTFIT statistics was verified by two further iterations in which both items were removed, and greater fit statistics were produced by the remaining items

Table 26. Fit and DIF statistics of items in the final 26-item YP-VQoL scale.

Item code	YP-VQoL item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age	DIF contrast by
				(logits)	gender (logits)
1	I have got some good friends	0.85	0.88	0.09	-0.32
2	I make new friends easily	0.93	0.88	-0.70	0.48
3	I keep friends easily	0.77	0.86	0.03	0.19
4	I am happy with my social life	0.87	0.8	0.45	-0.29
5	I spend enough time with my friends	1.13	1.05	0.49	0.14
6	Other young people my age pick on me because of my eyesight	1.35	1.17	-0.64	-0.99
8	My friends understand how things are for me because of my eyesight	1.33	1.52	-0.38	-0.11
10	I get treated the same as everyone else	1.11	1.08	-0.54	-0.12
11	I feel like I fit in	1.02	0.9	0.10	-0.40
13	I feel different from other young people because of my eyesight	0.84	0.84	0.28	-0.53
14	I feel left out because of my eyesight	0.91	0.78	0.34	-0.31
17	I am independent at home	1.02	1.03	-0.07	-0.20
18	I am independent at school/college	0.74	0.74	0.09	0.19
20	I can do most activities on my own	0.94	0.88	-0.15	0.51
23	I am comfortable asking for help	1.04	1.05	0.08	0.18
25	I cope well with my eyesight problems	0.86	0.78	-0.13	-0.23

Item code	YP-VQoL item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
27	I feel frustrated because of my eyesight	1.21	1.31	0	-0.08
29	I feel confident	0.78	0.77	0.03	0.46
31	I worry what other people think of me because of my eyesight	1.02	0.89	-0.21	0.21
32	I am positive about the future	0.90	0.91	0	-0.12
33	I am confident I will be able to look after myself in the future	0.90	0.79	0	0
35	I worry about what job I will be able to do in the future	0.84	0.82	0	0.46
36	I like to have a go at everything	0.95	0.88	0.47	-0.28
37	I can get around on my own	1.11	1.06	-0.06	1.03
38	I enjoy school/college	1.18	1.36	0.08	-0.66
39	I have to work harder at school/college because of my eyesight	1.4	1.52	0.15	0.15

### 6.3.4.8.4 Targeting

The person-item map using the items included in the final iteration revealed good targeting of items to the abilities of participants, with a difference of .71 between the mean of participants' ability and the mean difficulty of the remaining items. Additionally, the person separation and reliability for this iteration were within acceptable limits (separation = 2.74, reliability = .88) when analysing the real RMSE. As expected, the RMSE produced by the model, which represents the reliability measures when assuming all misfit in the data is random, were slightly improved (separation = 2.86, reliability = .89).



Figure 9. Item-person targeting of the final 26-item YP-VQoL scale.

### 6.3.4.8.5 Response category function

The Andrich thresholds for the category probability plots for the forward worded items in the final 26-item YP-VQoL scale increased by at least 0.96 logits across the scale (see Figure 10, below), which is marginally less than the increase displayed in the Child-VQoL instrument, and the recommended increase of 1.4. This indicates that the response categories in this measure may not be optimally defined. Despite this, the categories are well-ordered, demonstrating strong potential for accurate measurement.



Figure 10. Category probability curves for the 4 response categories in the forwardscored YP-VQoL instrument (1 = Not at all true, 2 = A little bit true, 3 = Mostly true, 4 = Completely true).

As with the Child-VQoL instrument, when analysed in relation to the reverse worded items, the category probability plots were well ordered. However, response category 3 (Mostly true) was again, the least likely response category to be observed, with the majority of the curve remaining submerged beneath the curve for response option 4 (Completely true). The increase in Andrich threshold was .26 between response option 3 and 2, indicating that the response categories are ordered as expected. The increase in Andrich threshold between response categories 1 and 2 was .71, and between 3 and 4 was 1.61 (see Figure 11, below).



Figure 11. Category probability curves for the 4 response categories in the reversescored YP-VQoL instrument (1 = Not at all true, 2 = A little bit true, 3 = Mostly true, 4 = Completely true).

# 6.3.4.9 Calibration of the Child- and YP-VQoL instruments

Prior to further analyses, the items included in the 24-item Child-VQoL and 26-item YP-VQoL instruments were assigned a new item label corresponding to the VQoL instrument in which they appear (be it Child-VQoL, YP-VQoL or both instruments (see Table 27, pg. 216)). These new item labels were used to identify each item during the process of calibration. Items were assigned a person-factor code (C = child, YP = young people). By combining both child- and YP-VQoL datasets, the sample size used in this phases of analysis was increased to 154 participants aged 7-18 years.

Table 27. Items included in the final 24-item Child- and 26-item YP-VQoL instruments and labels used for calibration.

Child		YP		Item label for calibration
Original item no.	Label	Original item no.	Label	
1	I have got some good friends	1	I have got some good friends	1
2	I make new friends easily	2	I make new friends easily	2
3	I keep friends easily	3	I keep friends easily	3
4	I am happy with how many friends I have	4	I am happy with my social life	4
5	I spend enough time with my friends	5	I spend enough time with my friends	5
6	Other children pick on me because of my eyesight	6	Other young people my age pick on me because of my eyesight	6
7	I can stand up for myself if someone picks on me			7C
8	My friends understand how things are for me because of my eyesight	8	My friends understand how things are for me because of my eyesight	8
		10	I get treated the same as everyone else	9YP
		11	I feel like I fit in	10YP
10	My friends encourage me to join in their activities			11C
11	I feel different from other children because of my eyesight	13	I feel different from other young people because of my eyesight	12
	Child		YP	Item label for calibration
-------------------	---	-------------------	---	----------------------------
Original item no.	Label	Original item no.	Label	
12	I feel left out because of my eyesight	14	I feel left out because of my eyesight	13
13	My teachers understand how things are for me because of my eyesight			14C
15	I can decide things for myself			15C
16	I am independent at home	17	I am independent at home	16
17	I am independent at school	18	I am independent at school/college	17
		20	I can do most activities on my own	18YP
18	People give me a chance to do things on my own			19C
19	I am happy asking for help	23	I am comfortable asking for help	20
20	I cope well with my eyesight problems	25	I cope well with my eyesight problems	21
21	I feel tired because of my eyesight			22C
22	I feel frustrated because of my eyesight	27	I feel frustrated because of my eyesight	23
		29	I feel confident	24YP
25	Other people are fair to me			25C
26	I worry what other people think of me because of my eyesight	31	I worry what other people think of me because of my eyesight	26
		32	I am positive about	27YP

Child		YP		Item label for calibration
Original item no.	Label	Original item no.	Label	
			the future	
		33	I am confident I will be able to look after myself in the future	28YP
		35	I worry about what job I will be able to do in the future	29YP
		36	I like to have a go at everything	30YP
		37	I can get around on my own	31YP
28	I like being at school	38	I enjoy school/college	32
29	I have to work harder at school because of my eyesight	39	I have to work harder at school/college because of my eyesight	33

### 6.3.4.9.1 Item fit to the Rasch measurement model

When analysed as a complete instrument, the overlapping or 'core' Child-VQoL and YP-VQoL items demonstrated good fit with the Rasch measurement model and FIT statistics were all within the range of .62 to 1.48. All observed average measures for persons were ordered in the right direction depending on whether the item was forward or reverse worded.

### 6.3.4.9.2 Differential item functioning

An analysis was conducted using the overlapping or 'core' items to detect DIF between the Child (8-12 years) and YP (13-17 years) items with a view to detect the stability of the item functioning between instruments and therefore determine whether the core items can be used to fix the scale calibration of the measure in the presence of agespecific items which are presented to only one age-group. Analysis revealed no DIF contrasts greater then 1, indicating that the core items are not biased in any way to either age group (after adjusting for the overall scores of respondents (see Figure 12, pg. 220). The largest DIF value was observed in Item 4 (I am happy with how many friends I have/I am happy with my social life) which was 0.5 logits more difficult for participants in the older age group (e.g. aged 13-17 years) to endorse as true. This may be a result of the change in wording between the items, with the term 'social life' representing complex, broader aspects of QoL than quantity of 'friends'.

# PERSON DIF plot (DIF=@AGE-GROUP)



\*the baseline measure (no DIF)

Figure 12. DIF in overlapping VQoL items between children (aged 8-12 years) (C) and YP (aged 13-17 years) (Y).

### 6.3.4.9.3 Fitting FPs to the model

When analysed together, the final 24-item Child-VQoL and 26-item YP-VQoL instrument demonstrated good functional ability, conforming to the numerous criteria of Rasch analysis indicating good fit to the model of measurement, and potential to be used across the full age-range of 8-17 years. The complete score-to-measure tables (see Appendix XVI, pg. 384 and XVII, pg. 385) can be used to convert raw scores on each age-specific instrument into scores which can be compared to converted scores on the opposing instrument, and used to analyse outcomes between children and YP of different age groups with reference to the same measurement scale and latent trait.

The default equations provided by Winsteps software to convert the raw scores into measure (or logit) scores for both Child (f) and YP (g) versions of the VQoL instrument are shown below.

When fitted to FPs, a 4<sup>th</sup> order polynomial trendline was found to fit the data optimally in both Child (see Figure 13, pg. 222) and YP (see Figure 14, pg. 223) raw scores. The 4 parameters shown in Figures 13 and 14 were entered into the equations presented in Chapter 4, Section 4.9.4.4 (pg. 120) to produce a formula to be used in practice to convert raw scores into logit scores in both the Child and YP instruments (h, i).

Child VQoL Measure =  $21.0726 + Score \times .7920$ 

 $YP VQoL Measure = 21.5534 + Score \times .7180$ 

221

(f)

(g)



Figure 13. Fit of a  $4^{th}$  order FP trendline to the score-to-measure values in the Child-VQoL instrument.

Child VQoL Measure

 $= 47.31 - 48.44 \frac{1}{\sqrt{Score}} + 0.0002577 \times Score^{4} - 0.0001226 \times Score^{4} \\ \times \ln(Score) + 0.00001467 \times Score^{4} \times \ln(Score) \times \ln(Score)$ (h)



Figure 14. Fit of a 4<sup>th</sup> order FP trendline to the score-to-measure values in the YP-VQoL instrument.

$$YP - VQoL Measure$$

$$= 47.09 + Score^{4} - 0.0000904 \times Score^{4} \times \ln(Score) + 0.00001062$$

$$\times Score^{4} \times \ln(Score) - 48.40 \times \frac{1}{\sqrt{Score}}$$
(i)

Although the model produced by fitting this FP to both Child- and YP-VQoL scores was found to improve upon the linear equation provided by Winsteps, when fitted to rawand measure-VQoL scores, the model was found to give a poor estimation of extreme scores when fitted to a user-friendly scale of 0-100. Thus, the best way to accurately calculate scores which are comparable between instrument versions is to use the conversion tables produced by Winsteps (see Appendix XVI, pg. 384 and XVII, pg. 385).

#### 6.3.4.10 Testing construct validity

6.3.4.10.1 Demonstrating normality in the logit and summary scores Skew, kurtosis, and Shapiro-Wilks tests of normality were used to assess construct validity in the developed instruments. Out of a total of 24 variables (comprising VA, VQoL summary logit scores and PedsQL summary and sub-domain scores for children (aged 8-12 years), YP (aged 13-17 years) and the entire sample combined), 17 were calculated as non-normally distributed when judged by the criteria of a) Z-score skewness outside the range of -2.00 to +2.00, b) Z-score kurtosis outside the range of -2.00 to +2.00, c) S-W significance value <.05 or d) a combination of the above. As a result, Spearman's Rank<sup>401</sup> statistics were used to evaluate correlations between all variables.

6.3.4.10.2 Correlation between VQoL logit scores and PedsQL summary scores To determine construct validity of the instruments designed to measure VQoL, the correlation between VQoL logit scores and PedsQL summary scores was analysed. Seventy-nine children and 69 YP were included in this analysis after removing participants with >50% missing PedsQL scores.<sup>388</sup> When analysed separately, both the Child- and YP-VQoL logit scores correlated positively with the Child- and YP-PedsQL summary scores ( $r_s = .65$ , z = 5.71, for Child scores and  $r_s = .79$ , z = 6.48, for YP). Child- and YP-VQoL logit scores also correlated positively with the summary scores on the psychosocial subscale of the PedsQL ( $r_s = .64$ , z = 5.69, for Child scores and  $r_s = .81$  z = 6.68, for YP).

Analysis also revealed a significant positive correlation between the VQoL logit and PedsQL summary scores when combining the Child and YP datasets ( $r_s = .72$ , z = 8.76). This is expected if the VQoL instrument truly detects components of QoL. Similarly, correlation between the VQoL summary scores and summary scores based on the psychosocial subscale of the PedsQL was significant ( $r_s = .73$ , z = 8.88), indicating that participants with higher levels of psychosocial well-being as detected by

224

the PedsQL instrument will also receive higher scores on the VQoL instrument (see Table 28, below).

Table 28. Table showing correlation coefficients between VQoL logit scores, PedsQL summary and sub-scale scores for children and YP aged 8-17 years (two-tailed).

	VQoL logit score	PedsQL summary score	PedsQL psychosocial summary score
VQoL logit score	1	.72 (8.76)	.73 (8.88)
PedsQL summary score	148	1	-
PedsQL psychosocial summary score	148	148	1

\*z-scores shown in parentheses.

### 6.3.4.10.3 Correlation between VQoL logit scores and VA

Finally a correlation analysis of the relationship between VQoL logit scores and participants' latest recorded VA was undertaken. When analysed individually, neither Child- nor YP-VQoL summary scores correlated with VA ( $r_s = -.02$ , z = -0.15 for Child scores and  $r_s = -.13$ , z = -0.84 for YP).

After combining the Child and YP datasets, analyses revealed the two variables are non-significantly correlated ( $r_s = -.1$ , z = -1.18) indicating that there is no relationship between visual acuity and VQoL in this sample of children and YP (see Table 29, pg. 226). Table 29. Table showing correlation coefficients between VQoL logit scores and VA for children and YP aged 8-17 years (two-tailed).

	VQoL logit score	VA
VQoL logit score	1	1 (-1.18)
VA	154	1

\*z-scores shown in parentheses.

## 6.3.4.11 Final assessment of unidimensionality

Following formal psychometric item reduction, the 24-item Child-VQoL and 26-item YP-VQoL instruments were entered into a final FA to assess unidimensionality.

Notably, the percentage of variance explained by the first (largest) factor in both VQoL instruments increased after psychometric item reduction was performed, demonstrating improvement in the measurement of one construct. The greatest increase in variance explained was observed in the YP-VQoL instrument: 35.02% (see Table 21, pg. 193) to 39.59% (see Table 30, below).

Table 30. Eigenvalues for the largest factor, percent of variance explained, variation between eigenvalues for datasets with imputed missing data, number of eigenvalues and items not loading onto factors in the final FA.

	Eigenvalue for the first (largest) factor (percent of variance explained)	Variation between eigenvalues for the first (largest) factor in imputed datasets	No. of eigenvalues >1	No. of items loading onto Factor 1 <.4	Items not Ioading positively onto the first (largest) factor
Child- VQoL	7.13 (29.71)	0.12	7.2	2	None
YP- VQoL	10.29 (39.59)	0.12	5	0.2	None

# 6.4 Development of the functional vision (FV) instruments

# 6.4.1 Phase 1: Item development and adaptation

The *functioning* – *school, home and leisure* domain of the final hierarchy of childcentred data which was developed in the foundation research<sup>5-7</sup> was used as the foundation for developing items for the FV instrument (see Table 31, pg. 228). Table 31. Areas of overlap and discrepancy between FV reported by 16-19 year old YP and 10-15 year old children.

VQoL domain⁵	Domain-pertinent issues identified from interviews conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
Functioning – school, home and leisure	Activities at home (e.g. TV, computers, reading, cooking, music), school (e.g. sport) and leisure (e.g. shopping, cinema),	Even though I sit at the front, you still do get the odd time when you do need help seeing the board [ <i>at school</i> ]. (Male, 16, SVI/BL, early onset)	Feeling tired/having aches and pains related to visual impairment, Maintaining physical appearance (e.g. doing bair, putting on make up)	I'll get tired quicker and when I'm tired I'll focus less and then I'll get a headache [] And, then I'll come home tired so then I'll have to nap as well. (Female, 17, VI, early onset)
	VI-related activities, such as adapted sports and using technology, Level of functioning (at home, at school – lessons and break-times, in leisure, out and about),	I couldn't watch the TV from here. (Male, 17, VI, early onset) When I do the washing up, [ <i>my mum</i> ] says 'do it systematically.' (Male, 18, SVI/BL, early onset)	hair, putting on make-up), Level of functioning at break/lunchtimes, Age-specific activities (e.g. going to parties/pubs).	onset) Yeah, if I tried to do [ <i>my</i> <i>own hair</i> ], it would either be a guessing game or um, modern art! (Male, 16, SVI/BL, early onset)

VQoL domain⁵	Domain-pertinent issues identified from interviews conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
	Functioning – school,			Functioning – school,
	home and leisure			home and leisure
	continued			continued
	Restrictions and			On, yean, I can't do my
	limitations,			eyebrows, make-up, and
				my toenalis [] I have to
				l've done my make-up
	MODIIIty,			right (Female 19 SVI/BI
				late onset)
	Communication (e.g.			I just hang around with my
	reading text messages			friends. Erm, two of my
	and emails).			friends, I made friends
				with this year, we go, most
				time we sit in this quiet
				room. Which is really
				quiet. And we can just
				relax, then we just chat
				really. (Female, 17, VI,
				early onset)

VQoL domain⁵	Domain-pertinent issues identified from interviews conducted with 10-15 year old children <sup>5</sup> and overlap with 16-19 year old YP.	Quotes (16-19 years old)	Areas of discrepancy/novel issues (16-19 years old)	Quotes (16-19 years old)
				Functioning – school, home and leisure continued Obviously you play football, rugby, that sort of thing. And most of the time we just chat. Just walk around, have a conversation. Get food, that sort of thing. (Male, 16, SVI/BL, early onset) Yeah, sometimes I get lost in clubs [] it's probably because they've got strobe lighting. (Female, 17 years, VI, early onset)

The self-reported functional difficulties associated with visual impairment were comparable to those reported by younger children aged 10-15 years old.<sup>5, 6</sup> Discrepancies emerged when YP discussed functional difficulties associated with new, age-specific activities, such as visiting pubs and going to parties. The majority of new, age-appropriate issues relating to functional ability in YP emerged in relation to independence and autonomy which was largely dependent upon fewer parental restrictions and/or encouragement from parents to try new things, such as making a snack without help from others.

Codes were mapped onto 21 existing FV items. Items covered the domains shown in Table 31 (pg. 228). Fifteen existing items did not receive any new age-appropriate codes. Nineteen items which were excluded in the later stages of the foundation research were re-considered for inclusion. Codes which justified the inclusion of these original items were centred upon the functional consequences of increased independence, for example, cooking food or preparing a snack without help or learning strategies to stay safe when navigating outdoor public places when alone. Three codes that did not map onto existing items were considered for inclusion as new age-specific items (see Figure 15, below).



Figure 15. Flowchart showing the origin of items included in the first draft FV instrument.

Using the codes, a draft FV instrument for YP aged 13-17 years was developed. The first draft instrument is displayed in Table 32 (below).

Table 32. Items included in the first draft FV instrument.

	Watching TV
	Playing video and computer games
	Playing other games, e.g. board games or card games
	Using the computer for homework
	Reading food packets, tickets, labels or recipes
	Doing household chores, e.g. washing up
	Reading small print text books, worksheets and exam papers
Ę	Reading enlarged text books, worksheets and exam papers
n = 2	Seeing the board in class
) sm	Finding friends in the playground
g ite	Taking part in maths classes
tistin	Taking part in PE
۵ ۵	Taking part in English classes
	Keeping up with the teacher in lessons
	Getting around school by myself
	Getting around outdoors by myself
	Reading signs and posters at stations or shops
	Playing team sports e.g. football, without adaptations
	Watching films in the cinema
	Watching plays and shows in the theatre
	Reading price tags
eq	Telling the time on a wrist watch
-cod = 15)	Telling the time on a wall clock
un g	Using the computer for lessons
istin	Drawing or painting
Ĕ	Reading hand writing

	Recognising people e.g. in school corridors
	Recognising other people's facial expressions
	Taking part in science classes
	Taking part in geography classes
	Keeping up with other students in class
	Getting around in crowds by myself
	Seeing small moving objects, such as balls
	Seeing large moving objects, such as cars passing
	Using escalators
	Finding correct money to pay
	Getting myself a drink
	Making myself a snack
	Cooking
	Finding objects I have dropped or lost
	Getting dressed by myself
ss of	Writing
stage	Taking part in drama classes
ater s	Going up and down the stairs by myself
the la 19)	Getting around the house by myself
d from the from $n = 1$	Finding my way around an unfamiliar house or a new building
pmei	Crossing the road by myself
velo	Getting the right bus by myself
de	Using other public transport e.g. trains, by myself
on it	Reading tickets or recipes
oundati	Playing ball sports e.g. football with adaptations, such as brightly coloured or bell balls
Ŭ.	Using mobile phone for texting my friends
	Using mobile phone for phoning people
	Playing musical instruments
	Shopping by myself e.g. for food or clothes

t te	Falling over, tripping, being clumsy, having accidents
′ age opria des = 3)	Seeing far-away/distant objects
New appro co ( <i>n</i>	Telling what colour something is, colour perception

Consistent with the development of the VQoL instrument, not all of the items included in the foundation FV instrument were assigned codes. Despite this, they were entered into the first draft of the new, age-appropriate instruments, due to relevant literature and professional opinion indicating that they will likely be relevant items.

### 6.4.1.1 Outcomes from the expert consensus meeting concerning the FV

### instrument development

In addition to minor modifications made to reflect age-appropriate preferences in linguist terminology (see Table 33, below), two new FV items (Looking after my appearance, for example, doing my hair, shaving, putting on make-up, and Using a mobile phone or tablet for social networking (Facebook, Twitter, MySpace) were discussed for suitability for inclusion in the second draft FV instrument.

Table 33. Minor linguistic modifications made to FV items during the expert consensus meeting.

Original FV item	Modified FV item
Using the computer for lessons	Using the computer at school to do my schoolwork
Finding friends in the playground	Finding friends outdoors at school
Watching plays and shows in the theatre	Watching shows, such as plays, at the theatre

### 6.4.2 Phase 2: Pre-testing

#### 6.4.2.1 Self-reported feasibility of the FV instrument

Consistent with self-reported feasibility of the YP-VQoL instrument, completion of the YP-FV instrument was described by the majority of participants as 'easy' or 'very easy' with many participants expressing positive perceptions of completing the instrument, and using it as a means to inform other people about the functional impact of visual impairment. As in pre-testing of the VQoL instrument, some participants expressed concerns that the instrument would take too long to complete. These however, were mitigated when participants were told that they would not necessarily be required to justify their answers when completing the instrument in clinical contexts.

6.4.2.2 FV item modification

#### 6.4.2.2.1 Linguistic modifications

FV items referring to different lessons at school were modified as a result of YP's experience of flexibility in education, meaning that not all subjects are studied. Modifications made by removing the term 'lessons' allowed participants flexibility in answering each item, and reference to their abilities outside of formal education.

#### 6.4.2.2.2 Modifications to aid interpretation and ease of responding

Consistent with the findings from pre-testing the VQoL instrument, some participants experienced difficulty when choosing one response option, and many described the need for a response option which represented the mid-point between response option 2 (Easy) and 3 (Difficult). When probed about their feelings about this, many participants described their functional ability as context dependent, varying in relation to levels of lighting, physical surroundings and social support.

Difficulty participants had when choosing one response option to describe their functional vision was evidenced when participants described their responses to the items 'Doing household chores, for example, washing up', and 'Recognising people, for

example, in school corridors'. Participants flagged these items as dependent upon the specific activity and the VA required, and had difficulty applying the items to a broader range of activities and locations versus, specifically 'washing up' and recognising people in 'school corridors'. As a result of this, items were modified by adding a second example (e.g. '...or tidying my bedroom' and '...in shops') in which to consider functional ability.

One item (Getting around outdoors by myself) was modified and separated by adding an additional item to distinguish between difficulty getting around outdoors in daylight and when it's dark. Table 34. Items which were modified as a result of phase 2: pre-testing.

Original YP-FV item	Modified YP-FV item
Using the computer at home to do my	Using the computer at home to do my
nomework	nomework/coursework
Finding objects I have dropped	Finding objects I have dropped such as
	coins or glasses on a low contrast
	surface
Using the computer at school to do	Using the computer at school or college
schoolwork	to do schoolwork/coursework
Reading handwriting	Reading other people's handwriting
Seeing the board in the classroom	Seeing the board in the classroom when
	sitting at the front
Recognising people, for example, in	Recognising people, for example, in
school corridors	school corridors or shops
Recognising other people's facial	Recognising other people's facial
expressions	expressions when they are close to
	me/at arm's length
Finding friends outdoors at school	Finding friends in crowded areas
Taking part in Maths lessons	Doing maths
Taking part in English lessons	Doing English or literacy
Taking part in Science lessons	Doing science
Taking part in PE lessons	Doing sports at school/college
Keeping up with the teacher in lessons	Keeping up with the teacher or tutor in
	lessons
Getting around school by myself	Getting around school/college by myself
Seeing big moving objects, such as	Seeing big moving objects, such as
bicycles passing	bikes passing in day light.
Getting around outdoors by myself	Getting around outdoors e.g. shops or
	the park by myself in the daylight
	Getting around outdoors e.g. shops or
	the park by myself when it's dark

In addition to items requiring greater specificity, a number of participants had difficulty answering items, based on not knowing whether they should respond in relation to their best corrected, or uncorrected vision. As a result of this, the phrase 'Remember to tell us how things are for you when wearing your glasses (if you wear them), with your low vision aids and other devices (if you use them for these activities) and with the best lighting and contrast for you' was added as a header to each page of the FV instrument.

#### 6.4.2.2.3 Outcomes from the expert consensus meeting

The possibility of introducing a 5<sup>th</sup> response option as a mid-point was discussed during the expert consensus meeting in relation to the difficulty many participants had when choosing one response option and wanting to opt for a mid-option but being unable to do so. A decision was made to retain the 4 response options in light of the psychometric difficulties involved when using a mid-response option indicating neither easy nor difficult. Additionally, the same process of instrument development for younger children (aged <10 years) confirmed the suitability of 4 response options and, since the measures are intended to be used simultaneously in practice, 4 response options were retained.

### 6.4.3 Phase 3: Piloting

### 6.4.3.1 Qualitative feedback

In their qualitative feedback, and consistent with results from phase 2 of the FV instrument development, many participants described the different answers they would give depending on different contexts, and therefore confirmed the need for the added header inserted on each page of the FV instrument reminding participants to respond in relation to 'the best lighting and contrast for you' (see above). For example, one participant described the impact of lighting:

"Ease of doing things is very much improved if the lighting conditions are right. For instance bright sunlight makes everything practically impossible unless a room is made darker or wearing wrap-around dark glasses over correction glasses." (Female, 15 years, SVI/BL, early onset).

#### 6.4.3.2 Quantitative feedback

#### 6.4.3.2.1 Missing data

The 'true' missing data (i.e. no response given) per item for the FV instrument were ≤13.04% with three missing responses on the FV item 'Watching shows, such as play, at the theatre', and missing completely at random. Four participants had 5.13% missing data as a result of items which had been modified after the questionnaire booklets had been sent out. These items had been modified significantly and responses were coded as 'no response' (see Table 8, pg. 113).

#### 6.4.3.2.2 Use of response categories

Similar to results from piloting the VQoL instrument, a small number of participants indicated their favoured response option was missing from the FV instrument by circling two answers. The majority of these answers were reported by the same participants. For example, one participant provided the response option 'between 2 and 3' on 13 separate occasions (accounting for 31.71% of this participants' total FV responses) indicating some difficulty deciding between the response 'Easy' and 'Difficult'. These responses were coded according to the values shown in Table 8 (pg. 113), and results are presented in relation to the number of valid responses after entering these values. As in phase 3 of the VQoL instrument development, no further analyses were conducted to impute missing data and no participants were removed because of missing data.

#### 6.4.3.2.3 Feasibility

Completion time for the FV instrument ranged from 5-60 minutes (median = 14.5, IQR = 10) and 91% of participants rated completion of the YP-FV instrument as 'easy' or 'very easy'. All participants rated the instructions as 'easy' or 'very easy' to understand.

When asked about their overall opinion of the questionnaire booklet containing both VQoL and FV instruments, 56.52% of YP stated that completion of the questionnaire booklet was useful to them. Nine participants (31.13%) reported needing help from

another person to complete the full booklet. The majority of these (n = 7) reported needing help to read the items and response options.

### 6.4.3.3 Response distribution in the FV instrument

### 6.4.3.3.1 Skew and kurtosis in the FV instrument

Four FV items were flagged as problematic in terms of skewness and kurtosis outside the acceptable limits: all originated from the foundation research and three referred to reading printed text (Reading food packets, tickets, labels or recipes, Reading enlarged textbooks, worksheets and exam papers, Reading other people's handwriting, Finding friends in crowded areas).

#### 6.4.3.3.2 Floor and ceiling effects in the FV instrument

One FV item was flagged as problematic due to floor effects (e.g. >60% choosing response option 4 (Very difficult or impossible)). This item (Reading small print textbooks, worksheets or exam papers) originated from the foundation research. No participants chose response option 4 when answering 5 FV items (Making myself a snack at home, Doing English or literacy, Getting around school/college by myself, Finding correct money to pay when shopping, Using a mobile phone or tablet for social networking, for example, Facebook, Twitter or MySpace).

No FV items appeared to have ceiling effects (>60% respondents choosing option 1 (Very easy)). However, three FV items (Reading small print textbooks, worksheets and exam papers, Seeing small balls when playing games, such as tennis or cricket, Reading signs and posters at stations or shops) were not allocated response option 1 by any participants, indicating greater functional difficulty (see Figure 16, pg. 241).



Figure 16. Percent of YP choosing extreme response categories when answering items in the FV instrument (1% and 60% thresholds highlighted in red).

# 6.4.4 Phase 4: Validation

Table 35 (pg. 243) shows the FV items entered into the questionnaire booklets which were sent to children (aged 8-12 years) and YP (aged 13-17 years) in the final validation phase. As in phase 4 of the VQoL instrument development, each item was assigned a value corresponding to the order of presentation within each instrument. Items presented in the same row of Table 35 are overlapping or 'core' items which are present in instruments designed for both age groups. Items parallel to empty cells are age-specific items, presented to only one age group. Table 35. Items included in the Child (8-12 years) and YP (13-17 years) version of the FV instrument entered into phase 4 (psychometric validation).

	FV instrument							
	Child (8-12 years)	YP (13-17 years)						
Item no.	Item	ltem no.	Item					
1	Watching TV	1	Watching TV					
2	Playing video and computer games	2	Playing video and computer games					
3	Playing other indoor games, such as board games or card games	3	Playing other indoor games, such as board games or card games					
4	Playing outdoor games, such as tag or hide and seek							
5	Using the computer at home to do my school work	4	Using the computer at home to do my homework/coursework					
6	Reading small writing such as food packets or instructions for toys	5	Reading food packets, tickets, labels or recipes					
7	Doing household jobs, for example, tidying up my toys	6	Doing household chores, for example, washing up or tidying my bedroom					
		7	Looking after my appearance, for example, doing my hair, shaving, putting on make-up					
		8	Making myself a snack at home					
		9	Making myself a meal					
		10	Finding objects I have dropped such as coins or glasses on a low contrast surface					

	FV instrument							
	Child (8-12 years)	YP (13-17 years)						
Item no.	Item	ltem no.	Item					
8	Using the computer in school lessons	11	Using the computer at school or college to do schoolwork/coursework					
9	Reading small print worksheets and text books like dictionaries	12	Reading small print textbooks, worksheets and exam papers					
10	Reading enlarged worksheets and text books like dictionaries	13	Reading enlarged textbooks, worksheets and exam papers					
11	Drawing or painting	14	Drawing or painting					
12	Reading other people's handwriting	15	Reading other people's handwriting					
13	Seeing the board in the classroom	16	Seeing the board in the classroom when sitting at the front					
14	Recognising people, for example in school corridors	17	Recognising people, for example, in corridors at school/college or shops					
15	Recognising other people's facial expressions	18	Recognising other people's facial expressions when they are close to me/at arm's length					
16	Finding friends in the playground	19	Finding friends in crowded areas					
17	Doing maths in lessons	20	Doing maths					
18	Doing literacy in lessons	21	Doing English or literacy					
		22	Doing science					
19	Doing PE	23	Doing sports at school/college					

	FV instrument							
	Child (8-12 years)	YP (13-17 years)						
Item no.	Item	ltem no.	Item					
20	Keeping up with the teacher in lessons	24	Keeping up with the teacher or tutor in lessons					
21	Keeping up with other children in lessons	25	Keeping up with other students in lessons					
22	Getting around the school without someone helping me	26	Getting around school/college by myself					
23	Playing team sports without special balls	27	Playing team sports, such as football, without adaptations such as special balls					
24	Seeing small balls when playing games, such as tennis or cricket	28	Seeing small balls when playing games, such as tennis or cricket					
25	Seeing big moving objects, such as bicycles passing by	29	Seeing big moving objects, such as bikes passing, in daylight.					
26	Getting around outdoors in daytime	30	Getting around outdoors e.g. shops or the park, by myself when it's daylight					
27	Getting around outdoors when it is dark	31	Getting around outdoors e.g. shops or the park, by myself when it's dark					
		32	Getting around in crowds by myself					
		33	Finding my way around an unfamiliar house or a new building					
28	Reading signs and posters at stations or shops	34	Reading signs and posters at stations or shops					
		35	Finding correct money to pay when shopping					

	FV instrument							
	Child (8-12 years)		YP (13-17 years)					
ltem no.	Item	Item no.	Item					
29	Watching films in the cinema	36	Watching films in the cinema					
30	Watching shows at the theatre	37	Watching shows, such as plays, at the theatre					
		38	Crossing the road by myself					
		39	Using public transport, such as trains, buses or the tube by myself					
		40	Using a mobile phone to text people					
		41	Using a mobile phone or tablet for social networking, for example, Facebook, Twitter or MySpace					

### 6.4.4.1 Data verification

The FV dataset was verified by the same researcher who entered the data in the first instance (AR) to validate the data entry. A total of 12 incorrect data entries were found out of 2,911 individual entries constituting 0.27% of the dataset. Each incorrect entry was cross-referenced with the original data source and corrected. Ten percent of this dataset was verified by a second member of the research team (VT) who found 100% of the data to have been entered correctly.

With regards to the Child-FV dataset, 43% of data entries were verified by the same researcher who entered the data (VT). No incorrect data entries were found. Accordingly, 10% of the data entries were verified independently by a second member of the research team (AR): 100% were found to be correct.

### 6.4.4.2 Missing data

#### 6.4.4.2.1 Missing data per person

With regards to missing data, four children were excluded from the Child-FV instrument development based on the criteria for exclusion of >25% missing data. Thus, data from 82 children were included in the phase 4 analysis. After excluding participants on the basis of not returning completed consent forms and/or booklets, no YP were excluded as a result of missing data. Seventy-one YP were included in the YP-FV phase 4 analysis.

#### 6.4.4.2.2 Missing data per FV item

After removing participants based on missing data, the largest percent of missing data per item was 4.88% for the Child-FV items 'Playing team sports without special balls', and 'Getting around outdoors when it is dark'. With regards to the YP instrument, the largest percent of missing data was 7.04% for the item 'Watching shows, such as plays, at the theatre' (see Table 36, pg. 248). Amount of missing data was negligible, justifying inclusion of all items in the subsequent stages of analysis.

Table 36. Missing data per item in the FV instrument, separated by age-group.

FV instrument							
Child	(8-12 years)	YP (13-17 years)					
ltem	Missing	ltem	Missing				
no.	data, <i>n</i> (%)	no.	data, <i>n</i> (%)				
27	4 (4.88)	37	5 (7.04)				
23	4 (4.88)	38	3 (4.23)				
29	3 (3.66)	36	3 (4.23)				
25	2 (2.44)	29	3 (4.23)				
21	2 (2.44)	16	3 (4.23)				
18	2 (2.44)	2	3 (4.23)				
17	2 (2.44)	41	2 (2.82)				
5	2 (2.44)	26	2 (2.82)				
4	2 (2.44)	23	2 (2.82)				
1	2 (2.44)	14	2 (2.82)				
30	1 (1.22)	4	2 (2.82)				
26	1 (1.22)	3	2 (2.82)				
20	1 (1.22)	40	1 (1.41)				
16	1 (1.22)	39	1 (1.41)				
15	1 (1.22)	35	1 (1.41)				
14	1 (1.22)	33	1 (1.41)				
12	1 (1.22)	32	1 (1.41)				
10	1 (1.22)	31	1 (1.41)				
9	1 (1.22)	30	1 (1.41)				
8	1 (1.22)	28	1 (1.41)				
7	1 (1.22)	25	1 (1.41)				
6	1 (1.22)	24	1 (1.41)				

FV instrument								
Child	(8-12 years)	YP (13-17 years)						
ltem no.	Missing data, <i>n</i> (%)	ltem no.	Missing data, <i>n</i> (%)					
3	1 (1.22)	19	1 (1.41)					
2	1 (1.22)	13	1 (1.41)					
		12	1 (1.41)					
		11	1 (1.41)					
		10	1 (1.41)					
		1	1 (1.41)					

### 6.4.4.3 Use of response categories

The majority of participants appeared to endorse response categories well. However, occasions when children selected multiple response options, or indicated their favoured response option to be missing constituted 1.02% of the Child, and 1.2% of the YP-FV datasets. Responses were re-coded using the values shown in Table 8 (pg. 113).

### 6.4.4.4 Response distribution

As with the items included in phase 4 of the VQoL instrument development, a substantial number of child-FV and YP-FV items (see Table 37, pg. 250) were shown to have z-score skewness and/or z-score kurtosis values outside the acceptable thresholds of -2 to +2. These items were flagged as problematic.

Table 37. Z-score skewness, z-score kurtosis, floor and ceiling effects for each item in

the child- and YP-FV instruments.

Item no.	Z- score	Z-score kurtosis	% endorsing	% endorsing	ltem no.	Z- score	Z-score kurtosis	% endorsing	% endorsing	
	skew		option 1	option 4		skew		option 1	option 4	
	I	FV (Ch	ild)		FV (YP)					
1	-1.17	-1.6	19.5	9.8	1	1.25	-0.89	14.1	12.7	
2	-0.3	-1.52	15.9	13.4	2	-0.07	-1.64	12.7	19.7	
3	-0.27	-1.68	20.7	8.5	3	0.48	-0.91	11.3	11.3	
4	-0.51	-2.18	18.3	23.2	4	1.5	-1.1	15.5	15.5	
5	-0.35	-1.86	19.5	14.6	5	-1.95	-0.89	5.6	35.2	
6	-5.56	3.31	3.7	59.8	6	1.29	-0.96	23.9	7	
7	1.21	-1.29	31.7	1.2	7	0.59	-1.78	22.5	14.1	
8	-0.64	-1.48	13.4	17.1	8	2.52	-0.32	36.6	5.6	
9	-4.8	2.01	4.9	53.7	9	0.33	-1.94	23.9	14.1	
10	3.61	-0.35	43.9	7.3	10	-2.55	1.03	4.2	29.6	
11	1.54	-1.45	30.5	8.5	11	0.65	-1.37	16.9	12.7	
12	-3.34	1.47	4.9	34.1	12	-2.92	-0.73	1.4	52.9	
13	-4.2	1.16	7.3	46.3	13	2.95	-0.37	36.6	11.3	
14	-0.56	-1.89	12.2	24.4	14	-0.09	-2.23	25.4	16.9	
15	0.09	-2.04	17.1	20.7	15	-1.55	-0.86	1.4	35.2	
16	-3.4	-0.36	9.8	43.9	16	-1.8	-1.06	7	32.4	
17	-0.05	-1.67	22	6.1	17	-1.89	-0.85	9.9	28.2	
18	0.01	-1.03	15.9	4.9	18	0.50	-2.41	23.9	25.4	
19	-0.72	-1.61	14.6	18.3	19	-1.34	-1.5	0	39.4	
20	-0.61	-1.68	19.5	11	20	1.63	-0.8	28.2	5.6	
21	-0.57	-1.30	13.4	13.4	21	1.13	-1.02	21.1	8.5	
22	2.73	-0.6	46.3	1.2	22	0.93	-1.52	19.7	15.5	
23	-1.05	-2.01	15.9	25.6	23	-0.62	-1.57	14.1	19.7	
24	-3.1	-1.02	12.2	45.1	24	0.73	-0.82	15.5	8.5	
25	0.35	-1.49	22	6.1	25	0.66	-1.05	18.3	8.5	
26	0.84	-1.36	32.2	7.3	26	1.95	-0.33	31	4.2	
27	-3.59	-0.04	9.8	42.7	27	-1.42	-1.8	14.1	32.4	

ltem no.	Z- score skew	Z-score kurtosis	% endorsing option 1	% endorsing option 4	ltem no.	Z- score skew	Z-score kurtosis	% endorsing option 1	% endorsing option 4
	I	FV (Ch	ild)			I	FV (Y	′P)	L
28	-2.74	0.02	7.3	31.7	28	-4.15	2.4	1.4	50.7
29	1.96	-1.06	29.3	8.5	29	0.45	-1.87	21.1	15.5
30	-1	-2.19	17.1	28	30	1.97	-0.76	21.1	12.7
		L			31	-1.05	-1.67	11.3	28.2
					32	-1.66	-1.18	11.3	25.4
					33	-0.63	-1.71	12.7	23.9
					34	-3.10	0.57	4.2	42.3
					35	0.93	-1.18	22.5	8.5
					36	2.05	-0.59	18.3	12.7
					37	0.02	-1.31	2.8	19.7
					38	0.3	-1.83	19.7	15.5
					39	-1.01	-1.5	15.5	19.7
					40	2.52	-0.71	36.6	9.9
					41	2.55	-0.92	36.6	12.7

\*FV response option 1 = 'Very easy', response option 4 = 'Very difficult or impossible'.

The percent of respondents endorsing extreme categories in each item was analysed in parallel to the largest, and most problematic, skew and kurtosis values. Only one item (Item 19 (Finding friends in crowded areas)) was removed from the subsequent stages of item reduction of the YP-instrument development, based on no respondents choosing response option 1 (Very easy).

### 6.4.4.5 Assessment of unidimensionality

### 6.4.4.5.1 FA

FA was conducted using each of the 5 datasets with imputed missing data for each instrument, and the outcomes shown in Table 38 (pg. 252) are the mean values of each dataset.

Both Child- and YP-FV instruments appeared to have good unidimensionality, with all items loading positively (and >0.4) onto the first factor.

Instrument version	Eigenvalue for the first (largest) factor (percent of variance explained)	Variation between the first (largest) eigenvalues in imputed datasets	No. of eigenvalues >1	No. of items loading positive onto Factor 1 <.4	Items not loading positive onto the first (largest) factor
Child-FV	13.49 (44.95)	0.12	6	0	None
YP-FV	20.79 (51.97)	0.17	6	0	None

Table 38. Outcome from FA conducted to assess dimensionality in the FV instruments.

### 6.4.4.5.2 Parallel analysis

Parallel analysis was run using each of the 5 imputed datasets. The mean eigenvalues for the largest 5 eigenvalues in each dataset are presented in Table 39 (pg. 253).
Table 39. Average of the first five eigenvalues produced by parallel analysis for each dataset.

Actual eigenvalue	Average eigenvalue	95 <sup>th</sup> Percentile eigenvalue
Child-FV		
13.48	2.36	2.56
1.96	2.14	2.29
1.63	1.98	2.1
1.4	1.85	1.96
1.14	1.73	1.82
YP-FV		
20.79	2.81	3.05
2.35	2.56	2.73
1.92	2.38	2.52
1.42	2.22	2.35
1.29	2.09	2.2

The matrix produced by parallel analysis for each dataset provides evidence to support the finding of unidimensionality in both Child-FV and YP-FV datasets as the second largest eigenvalue is smaller than both the average eigenvalue and percentile eigenvalue in both Child- and YP-FV datasets.

The values produced by the averaged matrix (see Table 39, above) were used to produce a scree plot for each dataset. Both scree plots visually represent unidimensionality in the datasets.



Figure 17. Screeplot showing actual, average and percentile eigenvalues in the Child-FV instrument.



Figure 18. Screeplot showing actual, average and percentile eigenvalues in the YP-FV instrument.

### 6.4.4.6 Rasch analysis

Rasch analysis was performed on the Child- and YP-FV instrument datasets independently, and then, using the datasets combined with the aim of calibrating the instruments. Results from the Child-FV, YP-FV and calibration of both instruments are presented respectively.

6.4.4.7 Rasch analysis using the 30-item Child-FV instrument

All 30 items included in the draft Child-FV instrument were entered into the Rasch analysis after fulfilling the criteria specified in the preliminary item removal stage. Only two iterations of Rasch analysis were required before the items included in the instrument conformed to the multiple criteria of good fit to the model of measurement.

# 6.4.4.7.1 Item fit to the Rasch measurement model

The first item to be removed was as a result of fit statistics outside the threshold of 0.5 to 1.5 (see Table 40, below). Item 27 (Getting around outdoors when it is dark) had a large OUTFIT MNSQ indicating a high level of unexpected observations by persons on this item which is relatively very easy or very hard for them and therefore a threat to validity. After removing Item 27, all fit statistics fell within the required threshold, demonstrating good fit of the data to the Rasch measurement model.

Table 40. Problematic items in the Child-FV instrument due to fit to the Rasch measurement scale.

	INFIT		INFIT		OUTFIT
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z- standardised (ZSTD)	Mean- square value (MNSQ)	Z- standardised (ZSTD)
Iteration 1	27 (Getting around outdoors when it is dark)	1.33	1.9	1.67	3.1

# 6.4.4.7.2 Order of person abilities

After removing Item 27, all items were analysed for average measures of persons and the direction of ordering of these measures. No items were found to have person abilities which were out of order in relation to the responses on items.

# 6.4.4.7.3 Differential item functioning

DIF analyses were conducted using gender and the same stratified age groups described in Section 6.3.4.7.3 (pg. 199). Using the 29-item scale, analysis revealed no DIF contrasts above the threshold of 1.0 (see Table 41, pg. 258). The largest DIF

contrast observed between males and females was .73 in Item 11 (Drawing or painting) which was more difficult for females than males in this sample. The largest DIF contrast observed between young and old age groups was Item 23 (Playing team sports without special balls) which was .68 logits more difficult for participants in the older age group (aged 10-13 years).

The final iteration of the Child-FV instrument contained 29 items. The measurement properties of the final iteration were checked for conformity to the predefined thresholds, and fit statistics were within acceptable limits (see Table 41, pg. 258). The average measures of persons were ordered in each item in relation to ability.

Table 41. Fit and DIF statistics of items in the final 29-item Child-FV scale.

Item code	Child-FV item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
1	Watching TV	0.82	0.77	-0.43	0.18
2	Playing video and computer games	0.99	0.96	0	-0.13
3	Playing other indoor games, such as board games or card games	0.74	0.71	0.23	-0.12
4	Playing outdoor games, such as tag or hide and seek	1.03	0.99	0.11	0.02
5	Using the computer at home to do my school work	1.26	1.23	0.16	0.09
6	Reading small writing such as food packets or instructions for toys	1.03	0.94	-0.11	-0.22
7	Doing household jobs, for example tidying my toys	0.94	0.98	-0.19	0.09
8	Using the computer in school lessons	0.91	0.86	0.08	-0.28
9	Reading small print worksheets and textbooks like dictionaries	1	0.94	-0.13	0.63
10	Reading enlarged worksheets and textbooks like dictionaries	1.15	1.37	-0.07	-0.38
11	Drawing or painting	1.11	1.22	-0.35	0.73
12	Reading other people's handwriting	0.64	0.63	-0.16	0.07
13	Seeing the board in the classroom	1.10	1.06	-0.66	0.36
14	Recognising people, for example in school corridors	1.08	1.07	0	0

Item code	Child-FV item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
15	Recognising other people's facial expressions	0.94	0.91	0.09	0.62
16	Finding friends in the playground	1.07	0.96	0	0.11
17	Doing maths in lessons	1.08	1.05	0.34	-0.41
18	Doing literacy in lessons	1.07	1.11	-0.34	-0.21
19	Doing PE	1.15	1.23	0.13	-0.53
20	Keeping up with the teacher in lessons	1.24	1.26	-0.20	-0.15
21	Keeping up with other children in lessons	1	1.07	0.47	-0.08
22	Getting around the school without someone helping me	1.23	1.05	-0.25	-0.37
23	Playing team sports without special balls	1.26	1.16	0.68	-0.44
24	Seeing small balls when playing games such as tennis or cricket	1.25	1.15	0.19	-0.13
25	Seeing big moving objects, such as bicycles passing by	0.73	0.75	0.16	0.13
26	Getting around outdoors in daytime	0.75	0.74	-0.20	-0.13
28	Reading signs and posters at stations and shops	0.59	0.56	-0.25	0
29	Watching films in the cinema	0.91	0.84	0.40	0.20
30	Watching shows at the theatre	1.07	1.05	0	0.47

# 6.4.4.7.4 Targeting

The person-item map using the final 29-item scale is presented in Figure 19 (pg. 261). The map revealed good targeting of the items to ability of participants with a difference between the two means of .04. The separation and reliability values support the good targeting of this scale and are within acceptable limits (separation = 5.54, reliability = .97) when analysing the real RMSE.



Figure 19. Item-person targeting of the final 29-item Child-FV scale.

#### 6.4.4.7.5 Response category function

Because all items included in the FV instrument are worded in the same direction (with higher response options indicating lower functional ability) only one collection of category probability curves was produced by Rasch analysis. Figure 20 (below) demonstrates a distinct period in terms of the latent variable (displayed on the x axis) when each category is most probable. In terms of the Andrich thresholds for the categories, thresholds increased by at least 1.42 logits across the scale, which meets the criteria of an increase of 1.4. Thus, there is evidence to prove that the response categories in the Child-FV instrument function well in terms of measurement across the abilities of participants in this sample.



Figure 20. Category probability curves for the 29-item Child-FV instrument (1 = Very easy, 2 = Easy, 3 = Difficult, 4 = Very difficult or impossible).

### 6.4.4.8 Rasch analysis using the 40-item YP-FV instrument

Forty items were entered into Rasch analysis of the YP-FV instrument, with one item (Item 19 (Finding friends in crowded areas)) having been removed in the preliminary stage of item removal (see Section 6.4.4.4, pg. 249). During the first iteration of Rasch analysis (using the 40-item scale) all items fulfilled the criteria of good measurement in terms of fit to the Rasch measurement model and ordering of person abilities in relation to response categories. Thus, DIF analyses were conducted using the full 40-item scale.

#### 6.4.4.8.1 Differential item functioning

DIF analyses were conducted using groups of participants stratified by gender and the ages specified in Section 6.3.4.8.3 (pg. 207). When stratified by gender, three items were shown to be problematic, with DIF contrasts greater than 1. Two of these items (Items 21 (Doing English or literacy) and 40 (Using a mobile phone to text people)) were shown to be more difficult for females in the sample with DIF contrasts of 1.30 and 1.06 respectively. The remaining item (Item 38 (Crossing the road by myself)) was assigned a DIF contrast of -1.15, being more difficult for males.

In relation to DIF by age group, four items were flagged as problematic. Three items (Items 13 (Reading enlarged textbooks, worksheets and exam papers), 14 (Drawing or painting) and 27 (Playing team sports, such as football, without adaptations such as special balls)) were shown to be more difficult for participants in the younger age group (aged 13-15 years) with DIF contrasts of -1.25, -1.59 and -1.03 respectively, and one item (Item 32 (Getting around in crowds by myself)) was shown to be more difficult for participants classified as 'old' (aged 16-18 years) with a DIF contrast of 1.03.

Six problematic items were removed iteratively in the order of magnitude of DIF contrast (ranging from largest to smallest). After removing Item 13 for DIF by age group, the DIF contrast for Item 32 was no longer greater than the threshold of 1, and subsequently retained in the scale. The 34-item scale was then entered into a new

Rasch analysis and assessed for fit to the Rasch measurement scale. As a result of the removal of items due to substantial DIF, the fit statistics of three further items reached the unacceptable threshold of >1.5 and were removed (see Table 42, below).

Table 42. Problematic items in the YP-FV instrument due to fit to the Rasch

measurement scale.

			INFIT	OUTFIT	
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z- standardised (ZSTD)	Mean- square value (MNSQ)	Z- standardised (ZSTD)
Iteration 7 (Excluding Item 19 (Finding friends in crowded areas), 14 (Drawing or painting), 21 (Doing English or literacy), 13 (Reading enlarged textbooks, worksheets and exam papers), 40 (Using a mobile phone to text people) and 27 (Playing team sports, such as football, without adaptations such as special balls)	23 (Doing sports at school/ college)	1.57	3.0	1.64	3.2
Iteration 8 (Excluding Items 19, 14, 21, 13, 40, 27 and 23)	17 (Recognising people, for example, in corridors at school/college or shops)	1.58	3.1	1.43	2.3

			INFIT		DUTFIT
Iteration	Problematic item(s)	Mean- square value (MNSQ)	Z- standardised (ZSTD)	Mean- square value (MNSQ)	Z- standardised (ZSTD)
Iteration 8 (Excluding Items 19, 14, 21, 13, 40, 27, 23 and 17)	18 (Recognising other people's facial expressions when they are close to me/at arm's length)	1.59	3.1	1.49	2.6

The results of DIF analyses which were conducted using the revised 31-item scale are presented in Table 43 (pg. 266). No DIF contrasts reached the threshold of >1, and all remaining items were retained.

Table 43. Fit and DIF statistics of items in the final 31-item YP-FV scale.

Item code	YP-FV item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender
					(logits)
1	Watching TV	0.96	1.04	-0.11	0.38
2	Playing video and computer games	1.20	1.25	-0.78	0.34
3	Playing other indoor games, such as board games or card games	0.78	0.82	-0.27	0.25
4	Using the computer at home to do my homework/coursework	1.32	1.36	0.81	0.48
5	Reading food packets, tickets, labels or recipes	0.83	0.77	-0.09	0.34
6	Doing household chores, for example, washing up or tidying my bedroom	0.81	0.85	-0.30	0.05
7	Looking after my appearance, for example, doing my hair, shaving or putting on make-up	1.03	1.01	0.45	-0.32
8	Making myself a snack at home	0.74	0.69	0.61	0.36
9	Making myself a meal	0.98	0.97	0.68	0.36
10	Finding objects I have dropped such as coins or glasses on a low contrast surface	1.03	1.30	-0.65	0.33
11	Using the computer at school or college to do schoolwork/coursework	1.04	0.98	0	0
12	Reading small print worksheets, textbooks and exam papers	1.06	1.15	-0.36	-0.32
15	Reading other people's handwriting	0.91	0.92	-0.21	0.24

Item code	YP-FV item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
16	Seeing the board in the classroom when sitting at the front	1.16	1.10	-0.78	0.18
20	Doing maths	1.26	1.26	-0.11	0.16
22	Doing science	1.24	1.23	-0.28	0.52
24	Keeping up with the teacher or tutor in lessons	0.94	0.91	-0.28	0.59
25	Keeping up with other students in lessons	0.78	0.77	-0.27	-0.13
26	Getting around school/college by myself	0.81	0.75	-0.23	0
28	Seeing small balls when playing games, such as tennis or cricket	1.08	1.10	-0.44	0.23
29	Seeing big moving object, such as bikes passing, in daylight	0.97	0.94	-0.30	-0.79
30	Getting around outdoors e.g. shops or the park, by myself when it's daylight	0.58	0.56	0.53	-0.69
31	Getting around outdoors e.g. shops or the park, by myself when it's dark	1.17	1.10	0.51	-0.79
32	Getting around in crowds by myself	1.11	1.02	0.96	0.85
33	Finding my way around an unfamiliar house or a new building	1.04	0.97	0.62	-0.62
34	Reading signs and posters at stations or shops	0.79	0.70	0	0.07
35	Finding correct money to pay when shopping	1.14	1.17	0.25	0.39
36	Watching films in the cinema	0.83	0.81	0.03	-0.42

Item code	YP-FV item	INFIT MNSQ	OUTFIT MNSQ	DIF contrast by age (logits)	DIF contrast by gender (logits)
37	Watching shows, such as plays, at the theatre	0.99	1.05	-0.13	0.42
39	Using public transport, such as trains, buses or the tube by myself	1.10	1.07	0.41	-0.84
41	Using a mobile phone or tablet for social networking, for example, Facebook, Twitter or MySpace	1.17	1.08	-0.58	0.05

# 6.4.4.8.2 Targeting

The person-item map using the final 31-item scale is presented in Figure 21 (pg. 269). Concordant with the previous three analyses, the difficulty of the final 31 items was well targeted to the functional ability of the participants. The difference between the person and item means was .03 and separation and reliability values were high when using the real RMSE values (5.40 and .97 respectively).



Figure 21. Item-person targeting of the final 31-item YP-FV scale.

### 6.4.4.8.3 Response category function

Each of the four response categories can be seen to peak, representing a period along the latent trait when each response category is the most likely to be chosen (see Figure 22, pg. 270). The category probability curves are supported by the increase in Andrich threshold which was  $\geq$ 2.22 between each response category, indicating that each response category is well defined and likely to be observed in future applications of the instrument.



Figure 22. Category probability curves for the 4 response categories in the 31-item YP-FV instrument (1 = Very easy, 2 = Easy, 3 = Difficult, 4 = Very difficult or impossible).

### 6.4.4.9 Calibration of the Child- and YP-FV instruments

The 29-item Child- and 31-item YP-FV scales were used in the calibration of the instruments. All items were assigned a new item label indicating which instrument they appear in (see Table 44, pg. 271). The sample size of the dataset containing both child and YP responses was n = 153.

Table 44. Items included in the final 29-item Child- and 31-Item YP-FV instruments and labels used for calibration of the two instruments.

	Child		YP	Item label for calibration
Original item no.	Label	Original item no.	Label	
1	Watching TV	1	Watching TV	1
2	Playing video and computer games	2	Playing video and computer games	2
3	Playing other indoor games, such as board games or card games	3	Playing other indoor games, such as board games or card games	3
4	Playing outdoor games, such as tag or hide and seek			4C
5	Using the computer at home to do my school work	4	Using the computer at home to do my homework/coursework	5
6	Reading small writing such as food packets or instructions for toys	5	Reading food packets, tickets, labels or recipes	6
7	Doing household jobs, for example, tidying up my toys	6	Doing household chores, for example, washing up or tidying my bedroom	7
		7	Looking after my appearance, for example, doing my hair, shaving or putting on make-up	8YP
		8	Making myself a snack at home	9YP
		9	Making myself a meal	10YP
		10	Finding objects I have dropped such as coins or glasses on a low contrast surface	11YP

	Child		YP	Item label for calibration
Original item no.	Label	Original item no.	Label	
8	Using the computer in school lessons	11	Using the computer at school or college to do schoolwork/coursework	12
9	Reading small print worksheets and textbooks like dictionaries	12	Reading small print textbooks, worksheets and exam papers	13
10	Reading enlarged worksheets and textbooks like dictionaries			14C
11	Drawing or painting			15C
12	Reading other people's handwriting	15	Reading other people's handwriting	16
13	Seeing the board in the classroom	16	Seeing the board in the classroom when sitting at the front	17
14	Recognising people, for example in school corridors			18C
15	Recognising other people's facial expressions			19C
16	Finding friends in the playground			20C
17	Doing maths in lessons	20	Doing maths	21
18	Doing literacy in lessons			22C
		22	Doing science	23YP
19	Doing PE			24C
20	Keeping up with the teacher in lessons	24	Keeping up with the teacher or tutor in lessons	25

	Child		YP	Item label for calibration
Original item no.	Label	Original item no.	Label	
21	Keeping up with other children in lessons	25	Keeping up with other students in lessons	26
22	Getting around the school without someone helping me	26	Getting around school/college by myself	27
23	Playing team sports without special balls			28C
24	Seeing small balls when playing games such as tennis or cricket	28	Seeing small balls when playing games such as tennis or cricket	29
25	Seeing big moving objects, such as bicycles passing by	29	Seeing big moving objects, such as bikes passing, in daylight	30
26	Getting around outdoors in daytime	30	Setting around outdoors e.g. shops or the park, by myself when it's daylight	31
		31	Getting around outdoors e.g. shops or the park, by myself when it's dark	32YP
		32	Getting around in crowds by myself	33YP
		33	Finding my way around an unfamiliar house or a new building	34YP
28	Reading signs and posters at stations or shops	34	Reading signs and posters at stations or shops	35
		35	Finding correct money to pay when shopping	36YP
29	Watching films in the cinema	36	Watching films in the cinema	37

Child		YP		Item label for calibration
Original item no.	Label	Original item no.	Label	
30	Watching shows at the theatre	37	Watching shows, such as plays, at the theatre	38
		39	Using public transport, such as trains, buses or the tube by myself	39YP
		41	Using a mobile phone or tablet for social networking, for example, Facebook, Twitter or MySpace	40YP

#### 6.4.4.9.1 Item fit to the measurement model

When analysed together, the final 29-item Child-FV and 31-item YP-FV instruments demonstrated good functional ability in terms of fit to the Rasch model of measurement. All item fit statistics were within the range of .61 to 1.44 and all observed average measures for persons were ordered in the right direction in terms of the difficulty of items.

### 6.4.4.9.2 Differential item functioning

DIF analyses were conducted between the Child (8-12 years) and YP (13-17 years) age groups and on all the 'core' overlapping items shown in Table 44 (pg. 271). Analyses revealed two items (Item 6 (Reading small writing such as food packets or instructions for toys/Reading food packets, tickets, labels or recipes) and 29 (Seeing small balls when playing games, such as tennis or cricket) with DIF contrasts greater than the threshold of 1 and classified as moderate to large.<sup>402</sup> Item 6 was shown to be 1.21 logits more difficult for YP (aged 13-17 years) included in this analysis to perform. In contrast, Item 29 was shown to be 1.02 logits more difficult for the children (aged 8-12 years) to perform. Both items were significant using the Rasch-Welch statistic (t = 4.25, p = <.001 for Item 6, t = -3.65, p = .004 for Item 29). After removing Item 6, the

DIF contrast for Item 29 decreased to an acceptable value of -0.96 and was therefore retained. Figure 23 (pg. 276) shows the DIF contrasts for the remaining 'core' items.

# PERSON DIF plot (DIF=@AGE-GROUP)



\*the baseline measure (no DIF)

Figure 23. DIF in overlapping FV items between children (aged 8-12 years) (C) and YP (aged 13-17 years) (Y).

#### 6.4.4.9.3 Fitting FPs to the model

The complete score-to-measure tables for the FV instruments (see Appendix XVIII, pg. 386 and XIX, pg. 387) displayed equations provided by Winsteps software which can be used to convert the raw scores into measure (or logit) scores for both Child (j) and YP (k) versions of the FV instrument.

Child FV Measure  
= 
$$19.3516 + Score \times .7353$$
 (j)

YP FV Measure

$$= 19.6981 + Score \times .6791$$
 (k)

When fitted to FPs, using a 4<sup>th</sup> order polynomial trendline, fit of the raw scores and the logit scores to the model was found to improve in both Child (see Figure 24, pg. 278) and YP (see Figure 25, pg. 279) scores. The 4 parameters used are shown in Figure 24 and 25. The equations produced by fitting the trendlines in Figures 24 and 25 are presented for scores of the Child (I) and YP (m) instrument versions.



Figure 24. Fit of a 4<sup>th</sup> order FP trendline to the score-to-measure values in the Child-FV instrument.

```
Child FV Measure = 2.983 + 11.08 \times \ln(Score)
= 0.00009009 \times Score^4 - 0.00004156 \times Score^4 \times \ln(Score)
+ 0.00000483 \times Score^4 \times \ln(Score) \times \ln(Score) (1)
```



Figure 25. Fit of a 4<sup>th</sup> order polynomial FP trendline to the score-to-measure values in the YP-FV instrument.

```
YP FV Measure
```

```
= 3.042 + 10.87 \times \ln(Score) + 0.000076 \times Score^{4} - 0.00003213 
\times Score^{4} \times \ln(Score) + 0.000003675 \times Score^{4} \times \ln(Score) 
\times \ln(Score) (m)
```

Although the FP model was found to improve upon the linear equation produced automatically by Winsteps, it was also found to inaccurately predict scores at the extreme end of the scale when manipulated to fit the scale of 0-100. Thus, the optimal method for converting raw scores into measure scores which can be compared between age-groups is to use the score-to-measure table produced by Winsteps (see Appendix XVIII, pg. 386 and XIX, pg. 387).

## 6.4.4.10 Testing construct validity

### 6.4.4.10.1 Demonstrating normality in the logit scores

Out of a total of 18 variables (comprising VA and FV logit scores for children (aged 8-12 years), YP (aged 13-17 years) and the entire sample combined), 13 were calculated as non-normally distributed judged using the criteria described in Section 6.3.4.10.1 (pg. 224). As a result, Spearman's Rank<sup>401</sup> statistics were used to evaluate correlations between all variables.

#### 6.4.4.10.2 Correlation between FV summary scores and VA

When analysed separately, both the Child- and YP-FV logit scores based on the measures correlated positively with participants' VA ( $r_s = .49$ , z = 4.37, (one-tailed) for Child scores and  $r_s = .42$ , z = 3.53, (one-tailed) for YP), indicating construct validity of the individual Child- and YP-FV instruments.

A correlation was also found when analysing the relationship between FV logit scores and VA when combining the logit scores of both children and YP. This relationship was significant ( $r_s = .44$ , z = 5.42, (one-tailed)), signalling that higher levels of VA are associated with lower FV summary scores based on the measure and demonstrating construct validity of the entire suite of core and age-specific items (see Table 45, below).

Table 45. Correlation coefficients comparing FV logit scores and VA for children and YP aged 8-17 years (one-tailed).

	FV logit score	VA
FV logit score	1	.44 (5.42)
VA	153	1

6.4.4.11 Final assessment of unidimensionality

Following Rasch analysis, the 28-item Child-FV and 30-item YP-FV instruments were entered into a final FA to assess unidimensionality. Notably, the percent of variance explained by the first (largest) factor increased in the YP-FV instrument after psychometric item reduction had been performed, demonstrating improvement in the measurement of FV as one construct. However, in the Child-FV instrument, the percent of variance explained decreased marginally: from 44.95 (see Table 38, pg. 252) to 44.69 (see Table 46, below).

Table 46. Eigenvalues for the largest factor, percent of variance explained, variation between eigenvalues for datasets with imputed missing data, number of eigenvalues and items not loading onto factors.

	Eigenvalue for the first (largest) factor (percent of variance explained)	Variation between eigenvalues for the first (largest) factor in imputed datasets	No. of eigenvalues >1	No. of items loading onto Factor 1 <.4	Items not Ioading positively onto the first (largest) factor
Child-FV	12.51 (44.69)	0.17	5.2	0	None
YP-FV	16.28 (54.28)	0.14	6	0	None

### 6.5 Directions for use of instruments

To ensure the developed instruments can be readily applied to both research and clinical ophthalmology contexts, we have developed guidelines for clinical practitioners and other health professionals. Specifically, a higher summary and measure score on the child- and YP-VQoL instruments denotes higher VQoL, and a higher summary and measure score on both FV instruments denotes a higher level of functional difficulty. Both instruments have been developed so that scores across the entire questionnaire must be added to make up a valid and reliable summary score.

If missing responses are found, it is important that researchers and clinicians explore the reason for missing responses (e.g. by addressing any instances where a respondent has written notes on the instrument, or indicated that some items are not relevant or very difficult to answer). In the context of research, if responses are found to be missing at random, we suggest they are coded using an improbable value (e.g. 999) and estimated or imputed using statistical techniques such as the ones we have used (see Section 4.9.4.2, pg. 115).<sup>378, 379</sup> If used in a clinical setting, the available scores may be used to derive a pro-rated summary score (i.e. by replacing the missing scores for a given patient with their observed mean score on other items on the scale). Specifically, this may be done using the following formula:

Where **S** = the observed summary score,

L = the number of items with complete responses (i.e. with scores not missing) and,M = the number of items with missing responses:

$$\frac{S}{L-M} \times M \tag{(n)}$$

And adding the resulting value to the summary score. The decimal-score point in the pro-rated summary score may be interpolated for a more accurate estimation. However, we recommend, for ease and consistency, that the final summary score is rounded to the closest integer. This may be used to locate the corresponding measure score when using Child- and YP- instruments simultaneously.

# 6.6 Summary

Following development which is grounded in the perspectives of children and YP living with visual impairment in the UK, and analysis using robust psychometric methods, the final Child-VQoL, YP-VQoL, Child-FV and YP-FV comprise 24, 26, 28 and 30 items respectively, and may be suitable for application within clinical contexts to capture the self-reported impact of visual impairment during childhood and adolescence. The final Child- and YP-version of each instrument are also valid for use sequentially or in parallel, rendering the instruments suitable for measuring the impact of visual impairment longitudinally throughout the course of childhood and adolescence (i.e. throughout the age range of 8-17 years) or across a population with age-range of 8-17 years, whilst maintaining an age-appropriate approach and capturing age-specific issues and challenges.

### Chapter 7 Discussion

Key findings, strengths and limitations of the study are presented in this chapter. The self-reported experience of visual impairment, from the perspectives of children and YP is discussed in relation to the format and implementation of the final instruments and in relation to existing generic- and vision-specific PROMs. Results from both streams of the project (i.e. qualitative investigation and instrument development) are considered in relation to the extant generic literature and the implications for policy and practice within paediatric ophthalmology services in the UK are discussed.

# 7.1 Strengths

To our knowledge, this study is the first in-depth exploration of the experience of visual impairment during childhood and adolescence applied to the development of an ageappropriate suite of PROMs designed to capture the impact of visual impairment during everyday life. Using a pre-established methodological framework, the instrument development has been consistently grounded in the perspectives of children and YP living with visual impairment. This renders the project an important methodological advance on previous efforts to develop vision-specific PROMs for which input from children and YP has often been supplemented or replaced by input from adults living, or working with children and YP.<sup>261, 265, 266, 271, 274, 278, 282, 285, 286</sup> During each stage of development, we took precautions to encourage independent self-report from participants, by interviewing them directly, engaging them in expert consultations, and providing materials designed for parents and guardians alongside those designed for children and YP. These attempts are reflected in the content and format of the finalised child- and YP-centred instruments.

### 7.1.1 Instrument calibration

The majority of existing vision-specific PROMs designed for children and YP are designed for use by individuals within a broad age-range i.e. 8-18 years<sup>213, 268-270, 276, 283,</sup>

<sup>284, 403-405</sup> and therefore are not sensitive to aspects of developmental change and capturing the impact of issues and challenges which are age-specific. The defining feature of the current development is that two complementary age-appropriate suites of PROMs have been developed for children aged 8-12 years and YP aged 13-17 years. Because these age-appropriate PROMs were developed in parallel, within the same research team and using the same theoretical and methodological approach, they can be used collaboratively within clinical practice to capture the impact of visual impairment across the age-range of 8-17 years, whilst maintaining specificity to the developmental nuances of particular age groups. The method of development we used is comparable to the Wolfe and Chiu approach<sup>406</sup> in that we have ensured that the shared, or overlapping, items in each instrument are stable in terms of measurement function i.e. we have developed a common scale calibration. To our knowledge, this approach has been applied in the past to the development of a PROM designed to capture the impact of spectacle correction on refractive error (i.e. pre- and postintervention),<sup>261</sup> but not in relation to the calibration of age-specific instruments as we have done.

# 7.1.2 Age-range of users of the developed PROMs

The finalised instruments were developed for children aged older than 8 years specifically, based on a) our findings from the early stages of instrument development for children younger than 10 years which will not be discussed in detail in this thesis and b) literature stemming from cognitive development and ability to self-report during childhood.<sup>137</sup> This decision echoes the fact that the majority of existing child-centred, vision-specific PROMs have a minimum age-threshold of 8 years: namely the EYE-Q,<sup>276,403</sup> IVI\_C,<sup>213,268</sup> LVP-FVQ,<sup>269,404</sup> LVP-FVQ II<sup>270</sup> and the PREP<sup>283,284,405</sup> (see Table 2, pg. 54). As previously described (see Section 2.5.1, pg. 42), varying degrees of developmental delay are often reported in samples of children living with visual impairment. Therefore, we would suggest a fluid approach to administering the developed instruments within clinical practice to children within the age-thresholds we

have defined, as some children aged over 8 years may experience difficulty conceptualising some items. Additionally, some children younger than 8 years may be cognitive capable of completing the instruments in a meaningful way.

To date, only one existing vision-specific PROM specifies 17 years as the upper age threshold of respondents,<sup>279, 281, 407</sup> with the majority extending this to 18 years. Our decision to specify 17 years as the upper age threshold is, however, well validated by our findings from the early phases of instrument development. We found that the selfreported impact of visual impairment in relation to everyday life was prone to change after participants had reached their 18<sup>th</sup> birthday. Changes were often a result of YP entering further education or formal employment and triggered new vision-specific challenges which were difficult to capture using the same questions as those used to capture aspects of VQoL in younger children. For example, asking YP to differentiate between their home life and education was difficult when YP had moved away from their family home, and were living in student accommodation at a college or university. Similarly, YP experienced difficulty answering questions and probes relating to specific aspects of education (such as seeing the board in the classroom) after they had transitioned to further education. However, consistent with our recommendations for the minimum age threshold, we would also recommend flexibility in the implementation of the maximum age threshold when the developed instruments are used within clinical practice and that practitioners consider the contextual circumstances of the patient. This is in keeping with the 'stage'-appropriate approach YP emphasised in our investigation of the experience of transition from paediatric- to adult-ophthalmology services.

# 7.1.3 Psychometric properties of the final instruments

As previously discussed (see Section 3.3, pg. 73), Rasch analysis can be used for instrument development to construct linearity by fitting the data (in this case the instrument items) to the model of measurement. This is done whilst considering the difficulty of each item in relation to the ability of respondents. Using Rasch analysis,

therefore, we can be sure the final instruments are capable of additive measurement. Calibration of the two age-specific instruments using the method we used confirms that the principles of additive measurement are met when using instruments independently, as well as collaboratively to compare outcomes of individuals at different stages in the trajectory of childhood. This is an important methodological advance on earlier developments of vision-specific PROMs, for which FA has been widely implemented.

The early, exploratory phases confirmed content validity of the final instruments as children and YP had an active role in contributing to the individual items. Efforts were made to incorporate elements of participants' speech in the item development and several items were worded verbatim based on spontaneous speech from children and YP. Qualitative analysis ensured the wording of each item was optimal to capture the underlying theme or experience. Thus, content and face validity in each instrument can be confirmed.

Using Rasch analysis, we were able to validate the instruments in terms of a) item fit, b) targeting of persons to items, c) item-category thresholds, d) response scale functioning and e) item bias, and, in doing so, provide a detailed description of how each instrument functions in relation to the outcomes it is designed to measure as well as the ability of the sample.

### 7.2 Limitations

## 7.2.1 Participation rate and sample size

Participation rate of YP living with visual impairment was low ( $\approx$ 30%) throughout the study, but is consistent with previous studies recruiting children and YP living with visual impairment.<sup>5, 240</sup> This was expected at the outset of the study and we addressed this by sampling patients cautiously throughout the 4 phases, inviting only those needed to fulfil the requirements of analyses, and preserving the majority of the sample for the final phase of validation. Efforts were made throughout the study to a) locate all

YP living with visual impairment in England, Ireland and Wales using recruitment from multiple sources, b) appeal to YP and their families using age-appropriate recruitment materials, and c) contact patients at multiple times, whilst sending reminder and confirmation letters. The resulting sample size of the first three phases of development was large enough to fulfil the requirements of the qualitative and early quantitative analyses and represented children and YP with a range of demographic characteristics, living in different communities and with different manifestations of visual impairment. In-depth interviews and expert consultations produced informative and detailed qualitative data, suggesting participants were eager to share their experience of visual impairment and happy to take part in the research study.

Although the participation rate of YP was consistent throughout the phases of development, the obtained sample size of the dataset in the final phase of development was particularly small in light of the planned psychometric development. A relatively small sample size was also attained in the parallel development of instruments for children. This posed a number of difficulties when validating each instrument using Factor and Rasch analyses. With regard to FA, the obtained sample size was well below the advised 'rule of thumb' ratio of at least 10-15 participants per variable.<sup>408</sup> This may have increased the errors in correlation coefficients and hence factor loadings, suggesting a greater contribution of individual items to factors. However, the importance of applying this 'rule of thumb' has been disputed as some authors have found that changes in the ratio of participants to variables make little difference to the stability of the factor solution.<sup>409, 410</sup> In this study, we used FA for the sole purpose of screening the datasets for unidimensionality prior to Rasch analysis, and thus, applied criteria to the first, most dominant factor in each analysis. In the FV instruments, all items loaded well onto the dominant factor, and it was only in the VQoL instruments that we observed some items which did not load heavily onto the dominant factor. Thus it is likely that an increase in sample size would have resulted in stronger evidence to
support unidimensionality in the child- and YP-VQoL instruments rather than alternative findings per se.

With regards to Rasch analysis, it is recommended that sample size should equal at least 10 observations in each response category for each item to ensure that the step calibration between response options is stable.<sup>317</sup> The sample size of the individual child and YP datasets which were used to analyse the final response scale functioning would have been large enough to fulfil this criterion if there had been an even spread of responses. However, in each instrument, responses to some items were skewed, with the majority of participants choosing responses at one end of the scale. This reflected minor variation in participants' well-being and ability at the time of recruitment. The likely impact of this skew would manifest itself in the strength of step calibrations between response options, rendering these less precise and stable than they might have been in a larger scale of development with less skewness. As with any other statistical analyses, the small sample size may have also produced less precise estimates, less powerful fit statistics, and less robust estimates which are sensitive to extreme responses.<sup>316</sup> However, it should be noted that the development of ageappropriate instruments in parallel revealed some consistency in the functioning of response categories between two independent groups and provides some evidence to support the final response scale in this development.

Findings regarding the overall participation rate in the current study has implications for the feasibility of future studies recruiting children and YP with complex health conditions, such as visual impairment, and performing psychometric analyses such as Rasch analysis. For future studies, we would suggest that sample size could be improved by recruiting children and YP during school holidays or other anticipated leisure times. This strategy was impractical to implement in the current study given the time frame of the project but we suggest this may be beneficial for future studies, particularly since many potential participants with more severe manifestations of visual impairment we contacted were attending specialist schools away from their family

homes, and were therefore uncontactable. Given that approximately one third of children and YP living in the UK with visual impairment attend specialist schools for pupils with learning, physical and/or sensory impairments,<sup>411</sup> future studies could address this by gaining support from visual impairment specialist residential institutions for children and YP at the outset. However, variation in the functional, emotional and psychosocial outcomes of children and YP attending these institutions in comparison to those attending mainstream, non-specialist education must be considered.

Contact details of potential participants were collected using electronic patient management systems at the two primary recruitment centres. These systems were not suitable for systematically searching patient records, and this had to be done manually throughout the project and was extremely time-consuming. One of the primary recruitment centres was a children's hospital and may have been the first institution involved in the specialist ophthalmic care of the majority of children and YP who have visual impairment onset before the age of 16 years. Thus the contact details of potential participants aged 13 years and older at the time of the project may have been out-of-date. This was particularly likely for individuals who no longer required routine specialist follow-up care and had early onset visual impairment which was nonprogressive. Although the recruitment sources restricted the number of patients we were able to contact, we were able to recruit a good distribution of patients with varying manifestations of visual impairment (i.e. including those with early and late onset visual impairment which was stable and progressive in nature of deterioration). Given that some instrument items were taken from instruments developed in the foundation research which also recruited a good, representative sample of children and YP<sup>6,7</sup> we are confident that instrument items are meaningful and valid for application in the population of children and YP living with visual impairment in the UK.

A final factor influencing the small sample size used in the current study is the restriction of the study sample to children and YP with visual impairment and no other significant physical, sensory or learning impairments. This was a unique aspect of our

study, and essential to achieve a focus on visual impairment *per se*, whilst ensuring participants were cognitively capable of reflecting on the impact of visual impairment. However, we acknowledge that the majority (77%) of children and YP living with visual impairment which is classified as severe in the UK do have additional sensory, physical or psychological impairments,<sup>412</sup> and were excluded from the sampling framework. Thus, recruitment of children and YP with visual impairment may be more successful in studies which do not require the same focus on visual impairment.

## 7.2.1.1 Predictors of participation: Index of multiple deprivation and severity of visual impairment

Participation analysis revealed that higher socioeconomic status was associated with a greater likelihood of participation in any of the four phases of instrument development. This finding is consistent with literature demonstrating similar participation trends in children aged 10-15 years<sup>413</sup> and the families of younger children living with a visual impairment<sup>414</sup> who were recruited through the primary recruitment site(s) used in the current study. However, severity of visual impairment was also found to predict participation in the current study: those with worse vision were less likely to participate. Thus, despite efforts to ensure all research materials were accessible e.g. by providing alternative electronic formats and contacting families by telephone to discuss the requirements of participation, it is possible that some YP living with blindness experienced some response burden,<sup>415, 416</sup> and perceived the activities involved in the latter phases of development (i.e. completing consent forms and questionnaire booklets) as too difficult and/or time consuming.

In contrast with literature demonstrating participation trends in similar samples<sup>413, 414</sup> ethnicity was not found to predict participation in the current study. Throughout the study, we recruited YP with a range of ethnic backgrounds. However, only a minority of participants came from Black Caribbean/African/other, 'Other', and 'Mixed' backgrounds. Because ethnicity was found to significantly predict participation in the early, univariate analyses, it is hypothesised that a larger sample size (and wider

distribution of participants with a range of ethnic backgrounds) would have diluted this effect.

Determining the impact of socioeconomic status (SES) upon outcomes in the current study is complicated because there are multiple levels of IMD influences (e.g. community, neighbourhood, and family) which can either be transitory or persistent. However, to date, research has demonstrated an association between SES and cognitive development emerging early in life course.<sup>417</sup> Family income has been shown to be a powerful correlate of IQ at the age of 5 years, and neighbourhood income differences may be determinant of IQ and externalising behaviour.<sup>418</sup> Living in a high-SES neighbourhood has positive benefits for school readiness and achievement.<sup>419</sup> With regards to health, research demonstrating the dynamic impact of SES upon outcomes during childhood<sup>420</sup> has shown disparities in access to resources such as nutrition,<sup>421</sup> cognitively stimulating experiences, and healthcare.<sup>422</sup> Evidence also shows that children from lower SES groups are less likely to see an eve-care specialist.<sup>423, 424</sup> However, we tried to negate any kind of bias caused by the participation trend in the current study (i.e. individuals with lower socioeconomic deprivation being more likely to participate) by grounding the instrument development in the perspectives of YP with a range of socioeconomic and ethnic backgrounds, and manifestations of visual impairment, and using qualitative techniques and expert consensus versus statistical analyses. We are therefore confident that the final instrument items are valid and reliable in terms of measurement precision for the national population of children and YP living with visual impairment.

#### 7.2.2 Influence of others upon self-report

Interviews conducted with YP were lengthy (lasting on average over 1 hour), and the majority of YP were extremely willing to discuss their experience of visual impairment. A one-to-one context in which attempts were made to exclude other family members enhanced perceptions of confidentiality, allowing participants to feel secure and open to describing their true feelings to the researcher. Efforts were also made to ensure

participants would be capable, in terms of visual function, to participate independently, particularly in the latter stages of development when they were required to complete printed questionnaire booklets. We encouraged independent self-report by providing parents and caregivers their own questionnaire booklet to report on their perceptions of their child/young person's VQoL or FV. However, it became apparent in the early phases of development, that it was not always possible to exclude parents, guardians, siblings or other family members from influencing YP's self-reported impact of visual impairment. At times, parents encouraged other family members, such as siblings to attend and contribute to interviews which were conducted with children and YP. Thus, presence of others may have limited the content of the developed instrument items and some items detecting more sensitive components of VQoL or FV may have been omitted.

The dominating role of parents also emerged in a number of everyday domains which were explored in this study: in the way YP formed future ambitions and goals, developed and asserted independence, and perceived the benefits of forming romantic relationships, and can be likened to traditional models of disability in which disabled children are seen as somewhat incapable of communicating or performing activities themselves (see Section 2.2, pg. 26). In the past, research exploring the role of parents in the broader context of childhood disability has linked parents' responses to their children's symptoms to children/YP's own experiences of pain. Incorporating elements of social learning theory<sup>425, 426</sup> such as positive and negative reinforcement, this perspective explains how protective, or dominating parenting can inadvertently result in an increase in the child/YP's self-reported symptoms and functional disability.427-431 Research has yet to explore this association in the context of visually impaired children and adolescents, however, given that visual impairment is a sensory impairment associated with reduced ability to complete activities independently, it may be that children/YP's perceptions of their functional impairment rely, somewhat, on the responses of parents, or significant others, who view the child/YP as incapable.

Findings from the current study mirror those in a study exploring peer and adult relationships in adolescents with restricted mobility<sup>432</sup> in which adolescents perceived their parents as controlling, and an obstacle to their attainment of independence. Participants described their relationships with parents as asymmetrical where only the parents were involved in decision-making processes.<sup>432</sup>

#### 7.3 The final suite of vision-specific PROMs

One aim of the current project was to apply the knowledge gained through the exploratory, qualitative investigation as to the impact of visual impairment during childhood and adolescence to develop a suite of age-appropriate, vision-specific measurement instruments.

# 7.3.1 The Child (8-12 years) and YP (13-17 years) versions of the Vision-Related Quality of Life (VQoL) instrument

The finalised child- and YP-centred VQoL instruments contain 24 items and 26 items respectively and are valid for use with children aged 8-12 years and YP aged 13-17 years. The instruments contain 17 overlapping items, 7 items designed specifically for children and 9 items designed for YP. These items span five of the six domains of VQoL specified in the foundation research: 1) social relationships, acceptance and participation, 2) independence and autonomy, 3) psychological and emotional well-being, 4) future – aspirations and fears and 5) functioning – school, home and leisure.<sup>5</sup> Items corresponding to the sixth domain: treatment of eye condition, were excluded in the early stages of development based on the premise that these items would be better suited for a patient-reported experience measure (PREM) and one item related to the YP's concerns that their eyesight might deteriorate (I worry my eyesight will get worse) was excluded during Rasch analysis because a gender bias was detected. Thus, the final instruments cover all domains of VQoL which were expected, and are appropriate, for inclusion in the PROMs. Items relating to the 5<sup>th</sup> component of VQoL (functioning –

school, home and leisure) were worded carefully and with emphasis on a broader social context, rendering them appropriate for inclusion in the VQoL instrument, as distinct from the FV instrument which focuses on a level of difficulty in performing a particular activity. For example, five items (I feel tired because of my eyesight, I have to work harder at school/college because of my eyesight, I can do most activities on my own, and I can get around on my own) were designed to pick up on the underlying psychosocial burden of visual impairment and effort to complete tasks, as distinct from reality of functional difficulty.

## 7.3.1.1 Definition of VQoL (as defined by children and YP with visual impairment)

In our study, factors defining VQoL in children and YP were largely similar, comprising values, goals and concerns, with emphasis on social environment and other people's perceptions and understanding of visual impairment. These findings can be described in relation to broader, well-supported definitions of QoL which emphasise the role of experience in constructing a mental representation of current QoL, such as that provided by the WHO Quality of Life Group (see Section 2.3, pg. 28). However, we also found that some aspects of VQoL were age-specific. For example, it is only during adolescence that participants in this study explained concerns about the impact of visual impairment in the future and the consequences this could have upon their future lifestyle and, specifically, what kind of job they might be able to do. Whilst all participants described the impact of their impairment upon their social life, age-specific VQoL items developed for children emphasise social inclusion and fairness, which hold parallel to social models of disability,<sup>433, 434</sup> showing individuals' awareness of social discrimination related to their disability emerging at a young age.

## 7.3.1.2 Comparison to existing vision-specific PROMs designed for children and YP

The number of items in both the final child- and YP-centred VQoL instruments are comparable to the number of items in existing vision-specific PROMs designed for children which range from 3<sup>272, 435</sup> to 39.<sup>259</sup> In our experience, the numbers of items in PROMs need to reflect a comprehensive coverage of the measurement construct, whilst considering the condition-specific needs of the intended users, such as response burden. During the second phase of development, some participants described the burden of having to complete lengthy self-report measures, particularly since this traditionally takes place within clinical contexts which are often perceived as time-consuming and arduous.<sup>436, 437</sup> Thus, we would conclude that the number of items in each of the developed instruments is optimal for capturing outcomes whilst minimising effort needed from respondents.

With regards to format, the final VQoL instrument (comprising statements for which respondents are required to indicate as true or not true) is comparable to three existing vision-specific PROMs designed for children: namely the IXTQ,<sup>279, 281, 407</sup> SREEQ,<sup>261</sup> and PREP,<sup>283, 284, 405</sup> and two vision-specific PROMs designed for parents i.e. proxy reports (the CVFQ<sup>265, 266</sup> and ATI<sup>271</sup>). With regards to response options, the VQoL instruments feature a 4 Likert-type response scale which is in keeping with the average number of response options featured in existing vision-specific PROMs designed for children (ranging from 2 – 6 response options). Three existing vision-specific PROMs designed for children include a 'not applicable' response option<sup>213, 258, 268, 271</sup>: a feature that a number of YP who took part in the current development favoured. Despite this, we decided to exclude a 5<sup>th</sup> 'not applicable' response option based on the premise that this would involve adding the response option to the corresponding child instrument, and may exceed the cognitive capacity required of children and YP to complete the final instruments in the absence of full functional vision.<sup>137</sup> We also made efforts throughout the development to include only items which are meaningful and relevant to

children and YP of varying ages. Inclusion of a 'not applicable' response option would confound this, indicating to users of the instruments that not all items may be applicable and/or important. Additionally, a 'not applicable' response option was implemented in the foundation research and led to some methodological difficulties, potentially confounding participants' endorsement of the 'Very difficult or impossible' response category preceding it.<sup>6</sup>

In keeping with the format of the final VQoL instruments, and participants' tendency to refer to factors other than visual impairment, such as personality when answering items, one existing vision-specific PROM designed for children (the IVI\_C)<sup>213, 268</sup> contains a statement which is comparable to the one we have implemented with the aim of reminding participants that 'The questions are all about how things are for you because of your eyesight'.

#### 7.3.1.3 Construct validity of the VQoL instrument(s)

We correlated participants' logit scores in each VQoL instrument to other outcomes: namely VA, PedsQL summary scores, and PedsQL psychosocial summary scores with a view to determine convergent validity as a form of criterion validity. Analysis of associations between novel instruments and existing generic instruments, such as the PedsQL is an approach used widely throughout existing developments of novel PROMs and applied to a range of patient populations including children born preterm,<sup>438</sup> living with physical disabilities such as cerebral palsy<sup>439</sup> and cardiac disease,<sup>440</sup> and to measurement of personality-based constructs such as selfstigmatization.<sup>441</sup> Consistent with the majority of existing developments, and as hypothesised, VQoL scores were significantly associated with PedsQL summary and sub-scale scores, indicating that the instruments capture similar constructs. Additionally, and in keeping with findings from the foundation research,<sup>7</sup> VQoL was not associated with VA and therefore we provide further evidence of the disability paradox<sup>88</sup> (see Section 2.3, pg. 28) in the sample. This finding points towards the utility of developing two complementary PROMs distinguishing between aspects of VQoL

which may not be related to each other and/or traditional objective outcomes such as VA.

# 7.3.2 The Child (8-12 years) and YP (13-17 years) versions of the Functional Vision (FV) instrument

The child- and YP-FV instruments contain 28 and 30 items respectively, of which 19 were overlapping. The final version of the Child-FV instrument contains 9 ageappropriate items, and the YP-FV instrument contains 11 age-appropriate items.

Each item in the final FV instruments is expressed as a verb phrase in the statement 'Because of my eyesight I find...' This format is comparable to that of one existing instrument (the CVAQC)<sup>262</sup> which was designed to capture visual ability and includes a reminder for participants to consider their ability in relation to their best corrected visual function. Our development of FV age-appropriate instruments advances upon the CVAQC in terms of sensitivity of the developed items to age-appropriate activities, such as playing in the playground, getting around outdoors when it's dark, and looking after physical appearances, which were found to be meaningful to only one age-group of children and YP.

## 7.3.2.1 Definition of FV (as defined by children and YP with visual impairment)

Unlike definitions of QoL, HRQoL and VQoL which are controversial but nevertheless receive empirical attention, FV is yet to be formally defined by extant literature. However, there is a broad understanding that FV occupies the gap between personoutcomes and symptom-outcomes as specified by the ICF model. Thus, FV can be defined as a self-reported evaluation of ability to complete meaningful daily activities in real everyday environments, incorporating elements of VA and broader physical and social surroundings. In this development, FV items which were developed for children largely span activities which are performed at home and in school. Age-appropriate items developed for the YP-version refer to activities which are performed independently, and reflect a primary outcome from the in-depth exploratory phase of the project: emergence of independence and autonomy during adolescence.

#### 7.3.2.2 Construct validity of the FV instrument(s)

When correlated with VA; an objective clinical measurement of visual function, participants' logit scores on the finalised child- and YP-FV instruments were positively associated, demonstrating good convergent validity in the final instruments, and indicating some consistency between objective and subjective assessments of visual function.

# 7.4 Experience of visual impairment during childhood and adolescence: Key findings in relation to existing knowledge

Use of in-depth, semi-structured interviews in the early stages of instrument development in this study afforded the opportunity to explore and understand children and YP's experience of growing up with a visual impairment, and thus capture novel data. To date, there has been limited investigation of children and adolescents' lived *experience* of visual impairment, with most accounts focused on describing the everyday difficulties or challenges, and using cross-sectional approaches. This form of research has shown that children and adolescents living with visual impairment want to be independent,<sup>231</sup> and perceive not being able to drive as a significant barrier to this,<sup>212, 231</sup> are worried about the future<sup>231</sup> and may experience embarrassment and isolation in social contexts.<sup>213</sup> We have extended this evidence base to identify when age- and vision-specific challenges and changes were experienced, and how children and YP overcame these.

We found that children develop and refine a number of coping strategies which are used to negate the negative impact of visual impairment in varying contexts, and when faced with varying challenges. Existing literature aimed at identifying the range and effectiveness of coping strategies used by individuals with visual impairment is largely restricted to adult populations<sup>442-444</sup> and cannot be readily applied to children and YP who have different developmental capacities and needs and, more importantly, grow up without sight as opposed to having had vision earlier. Where literature focusing on children or YP does exist, it is centred upon capturing the burden of caring for children and YP with visual impairment upon their parents and families,<sup>445-448</sup> and ignores the voices of children and YP themselves. To date, one empirical account examines coping strategies developed by YP aged 16-24 years with congenital visual impairment in relation to resisting the impact of visual impairment.<sup>449</sup> Findings are somewhat comparable to the coping strategies identified in the current study. For example, advocacy which was described by authors as 'actions of disempowered people to achieve justice, equity or inclusion'<sup>447(p196)</sup> can be compared to times in the current study when participants described their attempts to prove others wrong about their ability to participate in physical activities, be included in social activities, and develop friendships in which they were viewed as equal to their sighted peers. Similarly, aspects of 'passing' i.e. minimising the impact of visual impairment by refusing to use aids or identify as disabled, were discussed by some participants in their use of humour as an attempt to avoid the negative social stigma related to visual impairment.

A pertinent finding from the early, qualitative stages of instrument development was variation between the experience of those with early onset and late onset visual impairment, particularly with regard to acceptance of and adaptation to visual impairment. This finding supports the 'early timing hypothesis'<sup>450</sup> in the context of childhood visual impairment as those who experienced the impact of visual impairment early in childhood, appeared to better able to cope (both psychologically and functionally) with their impairment at a later stage. To date, existing literature examining differences between individuals with early and late onset visual impairment is centred upon identifying neuroanatomical differences and, in particular, sensitive or critical periods for visual input in light of cortical plasticity.<sup>451, 452</sup> Our findings extend the current evidence base, demonstrating differences in the subjective, self-reported experience of

visual impairment among those with varying manifestations of vision disorders. In the future, these findings can be combined using mixed-method studies which explore possible interactions between physiological (i.e. neuroanatomical) and psychological (i.e. self-reported) impacts of visual impairment during childhood and use techniques such as fMRI imaging alongside self-report. Findings may be used to clarify the psychosocial and emotional impact of visual impairment which manifests in late childhood and early adolescence.

# 7.4.1 Age- and vision-specific transitions and changes during childhood and early adolescence

A number of formal and informal transitions during childhood and early adolescence were described by participants living with visual impairment and often in relation to intrinsic changes in personality, attitudes, and beliefs, such as emerging perceptions of responsibility, and increased awareness of other people's reactions to impairment and the stigma associated with disability. Transitions which took place within education were described as particularly memorable and meaningful in terms of social contexts, as friendship groups were established, and adults were increasingly appreciated as role models. This finding is supported by literature investigating shifts in social identity among healthy adolescents, showing that social identity effects occur alongside experiences of change.<sup>453</sup> The school to adult-world transition has been viewed as a process involving changes in roles (i.e. socially expected behaviour or patterns of meaningful activities that are expected of individuals in various contexts<sup>454</sup>). These changes are thought to trigger further changes in how adolescents make sense of themselves and their world,<sup>455, 456</sup> suggesting that the changes children and YP described in the current study would have long-term effects upon identity as being visually impaired or disabled.

Whilst many of the transitions YP described were normative, and likely to be experienced by all YP regardless of visual impairment, participants often described vision-specific challenges, rendering the impact of visual impairment fluid and dynamic during childhood and adolescence. These findings speak to the value of longitudinal use of the developed instruments within paediatric ophthalmology. It may be valuable to monitor changes in the impact of visual impairment which are triggered by transitions in education, for example, as changes to a new educational environment may alter the daily impact of visual impairment, in addition to deeper perceptions about the future. Several YP who took part in the early phases of instrument development described transitioning from specialist to mainstream (or vice versa) schools or colleges, and the changes they experienced in terms of physical environments, vision-specific support and social surroundings. Again, these findings speak to targeting vision-specific support services and treatments to children and YP who are most in need of intervention, which can be done by administering the developed PROMs at critical time points during childhood and adolescence, and in contexts outside of paediatric ophthalmology settings, such as education.

7.4.2 Transition from paediatric- to adult-ophthalmology services: Implications for clinical practice

One transition we took the opportunity to explore in detail was the transition from paediatric to adult ophthalmology services. We designed the age threshold for participation at the outset of the current study to reflect the gap in VQoL and FV instrument provision for YP older than 15 years. However, we also noted that recruitment of YP aged 16-19 years in interviews would be optimal for exploring participants' experience of the transition from paediatric to adult ophthalmology services, since participants were on the conventional transition age-threshold. The importance of ensuring a timely and successful transition from child- to adult services is recognised internationally,<sup>396-398</sup> particularly for those who progress into adulthood with rare childhood onset conditions and complex health needs.<sup>457</sup> A growing literature has identified the impact of a timely and successful transition in terms of secure disease-related knowledge,<sup>458</sup> high self-efficacy and good confidence for self-management of

health,<sup>459</sup> and assessed the effectiveness of technology-based systems and methods such as use of the internet and mobile phones to provide YP with access to transitionrelated information and increased control over their transition.<sup>459-461</sup> Despite this understanding, there is currently very scant literature to inform transition planning and provision in ophthalmology.<sup>399</sup>

YP who took part in this stage of development described pertinent differences between clinical settings which subsequently impacted their feelings of being in control of their healthcare and confidence to manage their care independently in the future. Emphasis was placed upon provisions which are patient-centred when transitioning from paediatric to adult-centred care, and findings largely support application of the generic NHS transition guidelines<sup>396</sup> to ophthalmology services. The value YP placed upon an appropriate peer group being served by the same services confirms the key importance of considering 'stage' as opposed to 'age' in timing of transition to ensure it occurs after the developmental tasks of adolescence have been completed.<sup>462</sup> Transition that takes place too early in this trajectory risks feelings of insecurity in the new environment.<sup>399</sup>

In 2015, 74.7% (5.2 of 7.07 million) of out-patient appointments in adult ophthalmology services in the UK were attended by patients over the age of 50 years. This is in contrast to 10.5% (764,400), which were attended by infants and children aged 0-15 years and a mere 0.9% (65,200), which were attended by YP aged 16-19 years.<sup>463</sup> Given this skewed age distribution and the specific needs of YP, it is arguable that transition from paediatric ophthalmology should ideally be into specialist adolescent/young adult services.<sup>399</sup> Models of this provision exist in other areas of child health, for example, endocrinology services for the late sequelae of childhood cancer,<sup>464</sup> which have promising patient-reported outcomes.<sup>465</sup> This would address the challenges of the 'no-man's land' that lies between child and adult ophthalmology services.<sup>399</sup>

Good and consistent communication between patients and their families and their managing clinicians lies at the heart of effective paediatric ophthalmology services.

Implementation of vision-specific, age-appropriate PROMs may constitute a means in which to enhance age-appropriate communication, specifically when YP are at either side of the age-threshold of transition in ophthalmology services, by ensuring clinicians address the factors which are most important to patients during clinical consultations. Consistent use of the age-appropriate PROMs within the latter stages of paediatric, and early stages of adult-ophthalmology services may address patients' perceptions of clinicians' knowledge about their specific needs: a factor that we found to influence patients' perceptions of less effective communication after transitioning.

#### 7.6 Application of the finalised PROMs

The value of measuring the impact of a disability or illness over time is reflected in the number of clinical trials which have implemented PROMs in the past with a view to assess the impact of health-related interventions or treatments within a wide range of clinical specialities.<sup>116, 118, 466-469</sup> Most long-term follow-up studies using PROMs have administered the same instruments at different time points with the aim of ensuring measurements are reliable and that change in outcomes reflect true change in the construct of interest.<sup>470-472</sup> However, when children are recruited as participants in these studies, this approach may be conceptually problematic in that fluctuations in the experience of disability or impairment may be related to growth and development, and changing attitudes, values and priorities, as opposed to the intervention of interest. In a review of measures of HRQoL suitable for long-term follow-up in children who have experienced major trauma,<sup>473</sup> authors identified only 3 out of 14 HRQoL measures that met the criteria for psychometric validation and were also suitable for an age-range of at least 10 years: the DISABKIDS,<sup>474, 475</sup> KIDSCREEN 52<sup>476, 477</sup> and PedsQL.<sup>478</sup> Authors speculate that, because the number of items and scoring systems in different ageversions of the instruments are similar, different versions of these instruments can be considered as one measure. Indeed, there are two versions of the DISABKIDS instrument (suitable for children aged 4-7 years and 8-16 years) and four versions of

the PedsQL (suitable for parent proxy report for children aged 2-4 years, and selfreport for children aged 5-7 years, 8-12 years, and 13-18 years) and the format of response options and scoring systems are similar in each age-version. However, unlike the approach used in the work reported in this thesis, neither of these instruments were developed using psychometric analyses to validate the function of each age-version independently, and in collaboration with the alternative age-versions. Justification of the suitability for different ages is based on subtle changes in terminology, for example in the PedsQL the term 'kid' is modified to 'teen' and in the DISABKIDS, the Likert scale is presented as series of smiley faces. To our knowledge, the KIDSCREEN, which was described by authors of this review<sup>473</sup> as suitable for an age-range of 10 years, is not available in age-specific formats, but instead short- and long-versions comprising 27 and 52 items respectively, which are both described as suitable for children and YP aged from 8-18 years. Therefore, it is likely that the majority of existing PROMs designed to capture HRQoL among children and YP omit important age-specific contributors to QoL,<sup>479-481</sup> particularly since definitions of QoL emphasise life experience (see Section 7.3.1.1, pg. 295). Despite this, we chose to administer the child (8-12 years) and teen (13-17 years) versions of the PedsQL in the current study because it has been validated, and used widely in a range of paediatric health conditions.482,483

7.7 The role of PROMs within the current NHS: Policy and practice PROMs have been developed and used within clinical practice over the last 30 years but are increasingly recognised as valuable tools with potential to trigger changes in clinical practice.<sup>123</sup> The current NHS emphasises a patient-centred approach<sup>484</sup> and PROMs can provide valuable insight for provider trusts and clinical commissioning groups and for individual clinicians through their ability to evaluate the impact of clinical interventions upon outcomes which are traditionally non-quantifiable. In parallel to the emergence of psychometric understanding and application using varying statistical

approaches, we have now reached a stage where it is possible to quantify complex psychologically grounded concepts, such as QoL and, in doing so, enable clinical practitioners and trusts to develop patient-centred approaches. PROMs play an important role in NHS England's Five Year Forward View objective which was put forth in 2014 and emphasises patient empowerment through allowing them greater access to information, and greater control over their healthcare.<sup>485</sup> The patients' organisation, National Voice, describes how important it is for statutory services to recognise patients' own life goals, wellbeing and independence as the first steps in developing personalised care,<sup>486</sup> and thereby emphasises the role of PROMs in fulfilling the future goals of the NHS.

Despite an increase in life expectancy over time,<sup>95</sup> and a population which is becoming increasingly healthier, demand for convenient, effective and available NHS services is continuing to increase. The impact of this is largely financial and may be explained partly by the cost of caring for elderly patients.<sup>487</sup> At the same time, funding throughout the NHS has reduced dramatically.<sup>488</sup> A framework for understanding the different ways in which access to high-quality care can be limited by commissioners and providers emphasises six types of rationing: deflection, delay, denial, selection, deterrence and dilution.<sup>489</sup> Rationing by dilution, specifically, refers to the process in which budgets are cut, resources are spread thinly, and the quality of healthcare is reduced. Dilution in the current NHS climate is evidenced by staff shortages<sup>490</sup> which can have negative impacts on patients who may experience greater delays, increased cancellations and appointments needing to be rescheduled as a result.<sup>489, 491, 492</sup> Currently, in NHS adult ophthalmology services an alarming number of patients appear to be losing their sight needlessly due to delayed diagnosis which is caused by capacity problems,<sup>493</sup> such as those associated with rationing by dilution. Although this finding is yet to be applied to paediatric services, it is clear that financial demand and capacity problems in ophthalmology services in the NHS are having a negative impact upon patient outcomes. In these contexts, PROMs can also be used to evaluate quality and

consistency of healthcare over time: outcomes which reflect the impact of budget cuts upon patient-reported outcomes, and the NHS's goals of prioritising patient empowerment.

In the current political climate, the NHS is facing unprecedented financial and operational pressures.<sup>488</sup> Despite an increase of around 8000 new doctors and nurses working in the NHS since 2014,<sup>494</sup> there are some gaps in availability of specialist professionals. In a recent survey, only 52% of current NHS staff self-reported satisfaction with opportunities for flexible working and 36.7% report feeling unwell as a result of work-related stress.<sup>495</sup> Because PROMs can be used flexibly, administered in waiting rooms or at patients' family homes prior to consultations, and come in multiple, cost-effective formats, it is possible that PROMs constitute an affordable means to alleviate some of the demands placed upon NHS staff, by reducing the need for lengthy consultations, and allowing paediatric ophthalmologists to capture, quickly and accurately, patients' primary concerns. Additionally, caring for children and YP who are losing their eyesight is a challenging, and at times emotionally-fuelled task.<sup>496, 497</sup> Conversations can be particularly distressing when they involve discussions about possible future deterioration of vision. The vision-specific PROMs we have developed provide a platform in which clinicians may breach sensitive topics within clinical consultations, and the first step in understanding patients' individual needs.

As previously discussed (see Section 7.2.1, pg. 287), children and YP living with visual impairment and an additional associated non-ophthalmic disorder were excluded from the current development, based on theoretical considerations. However, the prevalence of additional, non-ophthalmic conditions speak to the implementation of the developed PROMs within clinical practice, as children and YP visiting ophthalmic services may likely also attend a range of alternative specialist clinics. The extant generic child health literature indicates that transitions from paediatric to adult services are particularly successful when YP's health needs are predominantly due to one condition,<sup>498</sup> and that successful transition for those with a number of comorbid health

conditions relies on excellent communication and organisation between caregivers, specialities/departments and institutions.<sup>499</sup> Thus, the developed instruments may be used to enhance consistent communication between clinical teams, promoting multidisciplinary approaches to the treatment of childhood visual impairment.

#### 7.8 Summary

We suggest that, following further empirical investigation into feasibility of routine implementation, the developed PROMs should be implemented into routine clinical practice, using an approach which takes account of patients' developmental 'stage' (versus 'age'). Routine application of the instruments at a nationwide level can result in benefits for both patients and clinical professionals, allowing effective and timely targeting of vision-specific interventions, whilst reducing the workload of clinical professionals and addressing economic factors prioritised by higher level commissioning bodies. Having explored the dynamic impact of visual impairment throughout childhood and early adolescence, we recommend that, at a minimum, the instruments are used with patients on either side of the age threshold of transitions which take place within ophthalmology services, but to optimise clinical outcomes, are used regularly as patients progress through childhood and adolescence, and face a number of age-related transitions.

#### Chapter 8 Future Work

As previously discussed (see Section 2.3.2, pg. 31 and Section 7.7, pg. 305), use of PROMs can facilitate patient-centeredness: a focus of several NHS frameworks. Patient-reported outcomes such as VQoL and FV may be collected and fed back to care providers longitudinally to track patient's outcomes over time, evaluate whether treatments are effective and facilitate treatment modifications as needed.<sup>500</sup>

At a higher level of practices, policies and guidelines, PROMs may improve patient care via two pathways. Firstly, a 'change' pathway in which feedback from patients encourages practices to take steps to change clinical care, and secondly a 'selection' pathway, whereby patients and higher organizations such as commissioning and regulatory bodies choose high-performing providers over low-performing providers.<sup>501-503</sup>

Because a large proportion of children and YP living with visual impairment in the UK also suffer from additional sensory, physical or psychological problems (see Section 7.2.1.1, pg. 291), and in keeping with our recommendations about the timing of transition from paediatric- to adult-ophthalmology services (see Section 7.6, pg. 304) we advocate a 'stage' versus 'age'-approach to implementing the developed instruments with children/YP of varying ages, and that clinicians consider patients' living conditions and developmental capacity to self-report when selecting a PROM. In relation to children and YP living with visual impairment, the 'one size fits all' methodological model which currently prevails in the context of PROM administration may not be optimal.<sup>504</sup> There is also a need to assess the feasibility of using PROMs within routine paediatric ophthalmology services.

#### 8.1 Preferences and attitudes towards using PROMs

There is a difference between collecting data using PROMs as part of a scientific study (as we have done) and as part of routine clinical care. In our study, efforts were made

to engage with children and YP and their families throughout the 4 phases of instrument development, emphasising the value of the research project, using ageappropriate materials, and ongoing communication and contact with patients and their families. However, in routine clinical practice, PROMs must be administered and results collected, scored and interpreted (often prior to doctor-patient interaction) without disrupting the clinical workflow. Successful implementation of PROMs relies on excellent co-operation between patients, their parents/guardians, and clinicians. Assessing these stakeholders' preferences is integral to integrating the developed PROMs into routine practice.

#### 8.1.1 Patient preferences

Patients need to be happy to complete PROMs for them to be successfully incorporated into routine ophthalmology services. Where it does exist, research focused on children and YP's perceptions of using PROMs is largely confined to mental health settings and the extent to which this can be extrapolated to children/YP living with visual impairment is unclear. However, in the past, children and YP aged 8-19 years visiting mental health services have described their support for PROMs, and preference to complete written measures versus talking to a clinician about difficult issues.<sup>505</sup> However, the main objection to using PROMs in this context was patients' perceptions of the ability of instruments to reflect complex and dynamic outcomes such as well-being. For example, some participants described day-to-day fluctuations in outcomes and concerns that this would make self-reporting impossible. Many participants expressed concerns about the practicality of using PROMs within mental health services, indicating that they wanted to be better informed about, and have more control over, the assessment process so that they were comfortable with the timing of the data collection and presentation of results.

In the UK, specifically, YP aged 9-17 years visiting mental health and diabetes services were shown to be worried that, using PROMs, the scope of clinical discussions may be widened, and outcomes may attract unwarranted professional attention to emotional

issues.<sup>506</sup> Many YP found questions to be personal, and felt that the information should be used only with discretion. Some had worries about the confidentiality of their selfreport. In these cases, YP did not want information divulged outside of their immediate clinical team.

Findings from these studies point towards the impact of patients' preferences upon obtaining valid and reliable outcomes. If a child/YP has a negative attitude to completing a PROM, they may become tired or bored and, as a result, skip certain questions. Within paediatric ophthalmology settings, specifically, children and YP may be required to have a number of assessments/procedures conducted at each outpatient visit (e.g. imaging which involves pupil dilation) which may affect their ability and willingness to complete the instruments at particular time points during their clinic visits.

In the current study, YP described parents/carers as playing an active role in managing clinical care and, at times, dominating discussions during consultations. The perceived need for parents/carers to dominate within clinical contexts was often magnified by the functional difficulties YP described when attending hospital visits independently and navigating their way around clinical environments, or accessing written information, meaning that they were unable to attend hospital appointments without a parent or carer accompanying them.

Input from parents during clinical consultations when a patient is under the age of 15 years is required to some degree for legal reasons,<sup>507</sup> but may affect children/YP's self-reported outcomes, particularly since the VQoL instruments we designed contain some items which reflect complex psychological states such as coping and life satisfaction. In a study investigating the views of adolescents living with mobility restrictions,<sup>432</sup> participants raised concerns that they might come into conflict with their parents if they did not openly express their appreciation and satisfaction with the help they receive from them. Adolescence is frequently described as a time of heightened awareness of the perceptions of others<sup>21, 508-510</sup> and children/YP living with visual impairment who

took part in the current study described their increasing concerns about the reactions of other people (e.g. siblings, parents, and peers) towards their impairment. Thus, it may be the case that, as children develop into adolescence, it is increasingly difficult to capture true self-report which is not influenced by the presence of others when completing or, specifically, patients' perceptions about how they *should* (versus *want*) to respond.

Consideration must therefore be given to the setting, presence of others, and method in which the developed instruments are administered in clinical ophthalmology contexts, especially if, even after transitioning from paediatric to adult-centred services, YP are likely to be accompanied to their clinical visits, and may have few opportunities for confidential disclosure with clinical professionals.

#### 8.1.2 Parent preferences

At the level of family members such as parents/guardians, implementing PROMs within paediatric ophthalmology services may serve some benefits, namely that responses from child self-report can provide insight as to the subjective impact of visual impairment, and the specific problems their child might be facing.<sup>505</sup> However, parents of children with mental health issues have been shown to be worried of professional scrutiny of the quality of care they are providing for their child when completing PROMs.<sup>506</sup> Parents may also be concerned that PROMs are too restrictive<sup>505</sup> and can replace (versus complement) important discussions about clinical assessments.<sup>506</sup>

The items included in the final suite of vision-specific PROMs we have developed are designed to reflect children/YP's self-reported difficulties, challenges and daily struggles in light of living with visual impairment. Thus, users of the instruments are required to reflect on some sensitive, psychological issues and it is possible that completion of the PROMs may cause some degree of distress for some children/YP. If it is the case that parents/guardians of these users do not understand the value of patient-reported outcomes, or how PROMs will be used within clinical contexts, and

perceive instrument completion to be potentially distressing, they will likely object to their child completing the instrument. Given that consent from someone with parental responsibility is essential before a child under the age of 16 is treated within clinical contexts,<sup>507</sup> it is imperative to promote parent/guardians' perceptions of the utility of PROMs, with emphasis on positive patient outcomes, to ensure they can be implemented successfully in routine ophthalmology services.

#### 8.1.3 Clinician preferences

Clinicians may be reluctant to use PROMs if they do not perceive the data as credible or understand the application to improving quality of care.<sup>503</sup> Extant generic literature reflects findings from adult healthcare settings and populations, and may not be readily applicable to paediatric ophthalmology contexts. However, as a foundation, some considerations may be made.

Consultants may be worried that implementing PROMs will lead to an increase in workload associated with collecting and analysing data, and communicating outcomes to patients.<sup>511-514</sup> This attitude may be reinforced by perceptions that mandatory public reporting programmes initiated by regulators or national governments are driven solely by political motives and do not impact upon routine clinical practice or patient outcomes.<sup>503</sup> Clinicians have also raised concerns that they do not understand the clinical importance of the data collected, and what to do when validity of the measurement is compromised i.e. when participants did not complete the measures accurately.<sup>515</sup> Preferences have been shown to parallel those of parents, as some clinicians have expressed concerns that PROMs may constrain the patient-clinician relationship because they trivialise patients' emotions.<sup>503</sup>

However, in a RCT evaluating improvements in individual patient care in adult oncology services in the UK,<sup>516</sup> PROMs were shown to lead to improvements in patient-doctor communication, with increased discussion of non-specific chronic symptoms. Implementing PROMs in this trial did not significantly increase the duration of

consultations, but instead led to greater patient engagement and empowerment through improved communication about aspects of health which are rated as important by patients. Thus, there may be some dissonance between the evidence-base and the preferences/attitudes of healthcare professionals.

If clinicians maintain negative attitudes towards implementing PROMs within routine clinical practice, or have concerns about their relevance or impact, it is possible that instrument validity may be compromised, and incorrect data captured. For example, clinicians have been found, in the past, to adapt PROMs to make them more suitable for use with individual patients.<sup>503</sup> Although we advocate a 'one size does not fit all' approach,<sup>504</sup> making changes by altering the PROMs to suit individual patients will compromise validity of the data captured.

## 8.2 Promoting positive attitudes to using PROMs within paediatric ophthalmology contexts

Common barriers towards implementing PROMs within routine clinical practice largely stem from lack of understanding on behalf of all stakeholders as to the utility of PROMs, the confidential nature of the data collected, and the potential benefits at the levels of individuals (i.e. patients and clinicians). Thus, education for both patients and their parents/guardians about how patient-reported outcomes are going to be used and who is going to look at the results is an important first step towards promoting positive attitudes towards using the developed PROMs.

In the past, scientific attention has focused upon improving health literacy: the ability to read, understand and act upon health information, with an aim to empower patients to take control of their healthcare.<sup>517</sup> Patient information materials (printed and electronic) have emerged as practical methods to enhance patients' knowledge and understanding of their condition.<sup>518</sup> The positive impact of these materials has been shown to be greater when information is personalized,<sup>519-521</sup> well targeted to patients

and available at the right time.<sup>522</sup> Therefore, it is possible that patient information materials comprising leaflets, information booklets, or online resources may be used to increase patients' and their parents' understanding about why PROMs are important outcomes. However, the best method to do this is yet to be explored within paediatric ophthalmology contexts and may be complicated by factors such as the nature of visual impairment and developmental delay in cognitive ability.

In the past, 2 potential misuses of PROMs have been demonstrated at the level of clinical professionals: in using PROMs to evaluate the value of treatment/intervention upon individual patient outcomes *before* administering it, and considering only short-term outcomes when deciding on competing demands between funding for treatments.<sup>103</sup> Therefore, education is needed to promote awareness among clinicians of the benefits and clinical value (as discussed in Section 8.1.3, pg. 313), and psychometric properties of PROMs.<sup>523</sup>

An intervention incorporating educational, epidemiological, marketing, behavioural, organisational, coercive, and social interaction approaches, involving a multidisciplinary training team, evidence based support for the values of PROMs, familiarization and interaction prior to use, and decision-support<sup>524</sup> has been applied to changing practice within paediatric clinical contexts. PROMs were administered online and paediatricians received background knowledge about the PROMs and their potential benefits prior to data collection.<sup>525</sup> This study highlighted the value of consultants observing (via DVD) others using PROMs, and the positive effect of evidence upon ideas of how to use patient-reported outcomes, and incorporate PROMs into routine clinical care.<sup>526</sup> If the 'one size does not fit all' approach<sup>504</sup> we advocate is used as a foundation for developing ophthalmic-specific guidelines, studies assessing the feasibility of using this type of intervention are needed, as learning through observing others, and empirical evidence-based ideas may be complicated when patients' physical and developmental capacities vary.

#### 8.3 Electronic methods to administer PROMs

In the past, a number of theoretical benefits have been associated with implementing PROMs administered using electronic formats, such as smartphones, tablets and touch screens within generic adult settings.<sup>527</sup> Not only are electronic methods financially optimal (in comparison with pen-and-paper and postal methods which require printing, processes of data entry and postage)<sup>528</sup> but they can be completed by patients in flexible locations without adversely affecting clinic workflow or visit length.<sup>516, 529</sup> Using electronic methods, data can also be entered automatically, and outcomes can be assessed quickly. Fewer missing data have been reported when using electronic (versus pen-and-paper) methods, thereby improving measurement accuracy.<sup>530-532</sup> Objective and subjective outcomes may be integrated into one system, and visual representations, such as graphs, can allow professionals to compare outcomes quickly and accurately.

Another benefit of using electronic methods of administration is the potential for realtime reporting and alerts.<sup>529</sup> Systems capable of real-time reporting have been developed within the field of adult oncology services,<sup>533</sup> and allow patients to enter and monitor their own outcomes and generate longitudinal reports which can be made available to staff. When applied to paediatric ophthalmology contexts, real-time reporting may provide early warning signs about potentially concerning issues and be particularly valuable given the progressive nature of some forms of visual impairment. Integration of PROMs into routine electronic health records could also allow practitioners to identify patients who need to complete PROMs, who can be contacted ahead of their clinic appointment and asked to complete PROMs at home before they come to clinics, and therefore reduce the administrative burden of having to complete PROMs within clinical settings.<sup>534</sup> A method to alert clinicians of issues requiring immediate attention may also be incorporated, and provide information directing clinicians to targeted resources (e.g. teachers or support workers) available to address any self-reported problems which may be resolved outside of clinical settings.<sup>535</sup>

The quantity and quality of electronic software designed specifically for users with visual impairment has expanded in the recent years, comprising screen readers, video magnifiers and note-takers.<sup>536</sup> To date, some studies have evaluated the feasibility of administering electronic questionnaires among child populations, showing that electronic methods are feasible, reliable and valid.<sup>537-539</sup> However, if these methods are going to be integrated into routine ophthalmology practice then software needs to be programmed to allow children and YP living with visual impairment to access the instruments, understand page layouts, save their data when sessions are interrupted and move through items.

Although this approach has immense potential to improve the administration and completion of PROMs within paediatric ophthalmology settings, which are located within an NHS framework which emphasises better use of data and technology in the next 2 years,<sup>540-542</sup> there is currently limited understanding as to the preferences of children/YP living with visual impairment, and their vision-related ability to use electronic methods. In the final phase of the current study, all participants were given the choice to complete the developed instruments using either pen-and-paper or electronic formats designed specifically for individuals with reduced visual function. However, only two YP chose to complete the electronic versions. Thus, electronic methods of PROM administration need to be designed carefully, and with respect to children/YP's preferences and a range of complex and dynamic functional difficulties.

## 8.3.1 Combining auditory and visual techniques: A feasible method of administering PROMs?

In 1997, Scott and colleagues<sup>543</sup> discussed the feasibility of using auditory devices such as Walkmans to ensure children could respond to questionnaires without needing help from others, and have their answers remain confidential. In that study, Walkmans were shown to have two advantages: no one other than the respondent could hear the questions being asked and, because the answer booklet contained only response

options, no one could tell which answers corresponded to which questions. Additionally, users were able to control the questionnaire administration process by pausing or rewinding the audio tape, and subsequently avoid the potential embarrassment of having to ask an interviewer to repeat certain questions. Although research has yet to explore the suitability of this technique for children/YP living with restricted functional vision, a combination of similar auditory and visual methods may be optimal for capturing true self-report in light of the potential influence of others whilst completing PROMs (see Section 8.1.1, pg. 310).

However, there are functional-related issues which need to be investigated before the Walkman technique can be implemented within paediatric ophthalmology settings, including what voice type, volume, and speed of reading is optimal. One difficulty of administering instruments using Walkmans was children's literal interpretation of questions when read aloud by an automated voice, as some participants tried to guess the state of the script reader rather than consider the items in relation to themselves.<sup>543</sup> This may pose difficulties among the population of children and YP living with visual impairment, especially when completing items included in the finalised FV instruments which are formatted as simple verb phrases. Thus, further multidisciplinary investigation combining input from information technology, paediatric ophthalmology and psychology specialities is required to develop sophisticated, user-friendly devices which can be used for PROM administration and completion.

## 8.4 Capacity of services to address psychologically-related

#### outcomes

Implementing PROMs within paediatric ophthalmology practices speaks to ensuring vision-specific services are able to address outcomes. Although the primary aim of implementing PROMs is to evaluate the quality of ophthalmic care and treatment, it is likely that, using PROMs as a platform in which to self-report, some children/YP may report that they are experiencing profound psychological/social/emotional problems

associated with their visual impairment, such as depression or anxiety. Using the developed instruments, respondents may report information which they are reluctant to communicate in other ways. This places some responsibility on those gathering, and analysing outcomes, to identify any responses indicating some concern, and how these should be dealt with.

In the past, clinicians have raised concerns that the information gathered using PROMs can interfere with treatment, forcing discussions into areas which are not clinically relevant and for which clinicians have little control over.<sup>123, 503</sup> Therefore it is important to ensure services are in place to respond to outcomes, and alleviate some of the burden placed upon ophthalmic consultants to address aspects of health which may be perceived to be beyond their clinical expertise. The population of children and YP living with visual impairment is complex in terms of its health-related needs and, as a result, may be receiving support from a number of vision-specific services within education, social, and mobility contexts. The capacity of these services for addressing patient-reported outcomes needs to be evaluated (and promoted) before a shared model of decision making (a component of the NHS Five Year Forward View <sup>485</sup>) can be developed.

#### 8.5 Summary

Combining the findings from this study with the future work outlined in this chapter would allow for incorporation of the developed instruments within paediatric ophthalmology clinical services throughout the UK to measure, with accuracy, sensitivity, and meaning, the subjective, self-reported impact of visual impairment during childhood and adolescence. Integration of the instruments into routine ophthalmology services within the NHS would be the first step to incorporating the instruments within a range of support services, including education and social support services, encouraging a multidisciplinary approach to supporting children and YP living with visual impairment, whilst ensuring consistency between institutions and clinical teams. Once this is complete it will be possible to systematically develop, and evaluate the impact of, a range of clinical and/or non-clinical vision-specific interventions, and target specifically what matters most to children and YP living with visual impairment.

### Chapter 9 References

1. World Health Organization. *Definition of Key Terms.* Available from: http://www.who.int/hiv/pub/guidelines/arv2013/intro/keyterms/en/ [Accessed 5<sup>th</sup> December 2015].

2. Haymes SA, Johnston AW, Heyes AD. The development of the Melbourne lowvision ADL index: a measure of vision disability. Invest Ophth Vis Sci. 2001;42(6):1215-25.

3. Frost N, Sparrow J, Durant J, Donovan J, Peters T, Brookes S. Development of a questionnaire for measurement of vision-related quality of life. Neuro-Ophthalmology. 1998;5(4):185-210.

4. Misajon R, Hawthorne G, Richardson J, Barton J, Peacock S, Iezzi A, et al. Vision and quality of life: the development of a utility measure. Invest Ophth Vis Sci. 2005;46(11):4007-15.

5. Rahi JS, Tadić V, Keeley S, Lewando-Hundt G. Capturing children and young people's perspectives to identify the content for a novel vision-related quality of life instrument. Ophthalmology. 2011;118(5):819-24.

6. Tadić V, Cooper A, Cumberland P, Lewando-Hundt G, Rahi JS. Development of the functional vision questionnaire for children and young people with visual impairment: The FVQ\_CYP. Ophthalmology. 2013;120(12):2725-32.

7. Tadić V, Cooper A, Cumberland P, Lewando-Hundt G, Rahi JS. Measuring the quality of life of visually impaired children: First stage psychometric evaluation of the novel VQoL\_CYP instrument. PLoS One. 2016;11(2):10.1371/journal.pone.0146225.

8. Gogtay N, Giedd JN, Lusk L, Hayashi KM, Greenstein D, Vaituzis AC, et al. Dynamic mapping of human cortical development during childhood through early adulthood. Proceedings of the National Academy of Sciences of the United States of America. 2004;101(21):8174-9.

9. Dobbing J, Sands J. Quantitative growth and development of human brain. Arch Dis Child. 1973;48(10):757-67.

10. Malina RM, Bouchard C, Bar-Or O. Growth, maturation, and physical activity, 2<sup>nd</sup> ed. Champaign (IL): Human Kinetics; 2004.

11. Thelen E, Fisher DM, Ridley-Johnson R. The relationship between physical growth and a newborn reflex. Infant Behav Dev. 1984;7(4):479-93.

12. Viru A, Loko J, Harro M, Volver A, Laaneots L, Viru M. Critical periods in the development of performance capacity during childhood and adolescence. European Journal of Physical Education. 1999;4(1):75-119.

13. James A, Prout, A. Constructing and reconstructing childhood: Contemporary issues in the sociological study of childhood. Oxon: Routledge; 1997.

14. Holloway SL, Valentine G. Children's geographies and the new social studies of childhood. In: Jenks C. (Ed.), Childhood: Critical concepts in sociology. Oxon: Routledge; 2005.

15. Robson E. Working girls and boys: children's contributions to household survival in West Africa. Geography: Journal of the Geographical Association. 1996;81(4):403.

16. Gelman SA. The essential child: Origins of essentialism in everyday thought. USA: Oxford University Press; 2003.

17. Lerner RM. Concepts and theories of human development, 3<sup>rd</sup> ed. Mahwah (NJ): Lawrence Erlbaum Associates; 2002.

18. Freud S. Three essays on the theory of sexuality. S.E. 1905;7:125-243.

19. Watson JB, editor. Behaviorism. Revised ed. New York: WW Norton & Co.; 1930.

20. Watson JB, Rayner R. Conditioned emotional reactions. J Exp Psychol. 1920;3(1):1.

21. Selman RL. The growth of interpersonal understanding. New York: Academic Press; 1980.

22. Piaget J, Cook M, Norton W. The origins of intelligence in children. New York: International Universities Press; 1952.

23. Inhelder B, Piaget J. The early growth of logic in the child: Classification and seriation. London: Routledge and Kegan Paul; 1964.

24. Benson PL, Scales P. The definition and preliminary measurement of thriving in adolescence. J Posit Psychol. 2009;4(1):85-104.

25. Christie D, Viner R. Adolescent development. Brit Med J. 2005;330(7486):301.

26. Piaget J, Inhelder B, Piaget J. The growth of logical thinking from childhood to adolescence: An essay on the construction of formal operational structures. Oxon: Routledge; 2013.

27. Piaget J. Intellectual evolution from adolescence to adulthood. Hum Dev. 1972;15(1):1-12.

28. Spear HJ, Kulbok P. Autonomy and adolescence: A concept analysis. Public Health Nurs. 2004;21(2):144-52.

29. Coleman JC. Independence in adolescence. British Journal of Projective Psychology & Personality Study. 1968;13(2):7-10.

30. Adelson J, Green B, O'Neil RP. Growth of the idea of law in adolescence. Dev Psychol. 1969;1(4):327.

31. Adelson J, O'Neil RP. Growth of political ideas in adolescence: The sense of community. J Pers Social Psychol. 1966;4(3):295-306.

32. Rogol AD, Clark PA, Roemmich JN. Growth and pubertal development in children and adolescents: effects of diet and physical activity. Am J Clin Nutr. 2000;72(2):521s-8s.

33. Rogol AD, Roemmich JN, Clark PA. Growth at puberty. J Adolescent Health. 2002;31(6):192-200.

34. Marshall WA, Tanner JM. Variations in pattern of pubertal changes in girls. Arch Dis Child. 1969;44(235):291.

35. Marshall WA, Tanner JM. Variations in the pattern of pubertal changes in boys. Arch Dis Child. 1970;45(239):13-23.

36. United Nations, UNICEF. The state of the world's children 2011: Adolescence - an age of opportunity. Geneva: UNICEF; 2011.

37. Ng ES, Schweitzer L, Lyons ST. New generation, great expectations: A field study of the millennial generation. J Bus Psychol. 2010;25(2):281-92.

38. Singh S, Darroch JE. Adolescent pregnancy and childbearing: Levels and trends in developed countries. Fam Plann Perspect. 2000:14-23.

39. Atwater E. Adolescence. Englewood Cliffs (NJ): Simon & Schuster Company; 1992.

40. Rauer AJ, Pettit GS, Lansford JE, Bates JE, Dodge KA. Romantic relationship patterns in young adulthood and their developmental antecedents. Dev Psychol. 2013;49(11):2159.

41. Patton H, Fuchs A, Hille B, Scher A, Steiner R. Textbook of Physiology, vol. 1. Philadelphia: WB Saunders; 1989.

42. Griffin JE, Ojeda SR. Textbook of endocrine physiology. New York: Oxford University Press; 1992.

43. Nottelmann E, Susman E, Inoff-Germain G, Cutler G, Loriaux D, Chrousos G. Developmental processes in early adolescence: Relationships between adolescent adjustment problems and chronologic age, pubertal stage, and puberty-related serum hormone levels. J Pediatr. 1987;110(3):473-80.

44. Beach FA. Behavioral endocrinology: An emerging discipline: How hormones affect—and are affected by—behavior is a subject of increasing interest to students of human and animal behavior. Am Sci. 1975;63(2):178-87.

45. Buchanan CM, Eccles JS, Becker JB. Are adolescents the victims of raging hormones? Evidence for activational effects of hormones on moods and behavior at adolescence. Psychol Bull. 1992;111(1):62.

46. Popma A, Vermeiren R, Geluk CA, Rinne T, van den Brink W, Knol DL, et al. Cortisol moderates the relationship between testosterone and aggression in delinquent male adolescents. Biol Psychiat. 2007;61(3):405-11. 47. Yi-Zhen Y, Jun-Xia S. Relationship between levels of testosterone and cortisol in saliva and aggressive behaviors of adolescents. Biomed Environ Sci. 2009;22(1):44-9.

48. Marceau K, Ram N, Susman EJ. Development and lability in the parent–child relationship during adolescence: Associations with pubertal timing and tempo. J Res Adolescence. 2015;25(3):474-89.

49. Vermeersch H, T'sjoen G, Kaufman J-M, Vincke J. The role of testosterone in aggressive and non-aggressive risk-taking in adolescent boys. Horm Behav. 2008;53(3):463-71.

50. de Macks ZAO, Moor BG, Overgaauw S, Güroğlu B, Dahl RE, Crone EA. Testosterone levels correspond with increased ventral striatum activation in response to monetary rewards in adolescents. Dev Cogn Neuros-Neth. 2011;1(4):506-16.

51. Haber SN, Knutson B. The reward circuit: linking primate anatomy and human imaging. Neuropsychopharmacol. 2010;35(1):4-26.

52. Giedd JN, Blumenthal J, Jeffries NO, Castellanos FX, Liu H, Zijdenbos A, et al. Brain development during childhood and adolescence: a longitudinal MRI study. Nat Neurosci. 1999;2(10):861-3.

53. Citizens Advice. *Compulsory School Age.* Available from: https://www.citizensadvice.org.uk/wales/family/education/school-education/compulsoryschool-age-w/ [Accessed 9<sup>th</sup> February 2017].

54. The Headmasters' & Headmistresses' Conference. *The British Education System.* Available from: http://www.hmc.org.uk/about-hmc/projects/the-british-education-system/ [Accessed 9<sup>th</sup> February 2017].

55. Kirchler E, Palmonari A, Pombeni ML. Developmental tasks and adolescents' relationships with their peers and their family. In: Rodriguez-Tomé SJH (Ed.), Adolescence and its social worlds. Hillsdale (NJ): Lawrence Erlbaum Associates; 1993:pp. 145-67.

56. Poulin F, Pedersen S. Developmental changes in gender composition of friendship networks in adolescent girls and boys. Dev Psychol. 2007;43(6):1484-96.
57. Skinner EA, Zimmer-Gembeck MJ, Connell JP, Eccles JS, Wellborn JG. Individual differences and the development of perceived control. Monogr Soc Res

Child. 1998:i-231.

58. Jones DC. Social comparison and body image: Attractiveness comparisons to models and peers among adolescent girls and boys. Sex Roles. 2001;45(9-10):645-64.
59. Hall GS. Adolescence its psychology and its relations to physiology,

antropology, sociology, sex, crime, religion, and education. Englewood Cliffs (NK): Prentice-Hall; 1904.

60. Lazarus RS, Kanner AD, Folkman S. Emotions: A cognitive-phenomenological analysis. In: Plutchik R, Kellerman H (Eds.), Theory, research, and experience. New York: Academic; 1980:pp. 189-217.

61. Mandler G. Recognizing: The judgment of previous occurrence. Psychol Rev. 1980;87(3):252.

62. Alsaker FD. Psychosocial disturbances in young people: Challenges for prevention. Cambridge (UK): Cambridge University Press; 1995.

63. Arnett JJ. Adolescent storm and stress, reconsidered. Am Psychol. 1999;54(5):317.

64. Disability Discrimination Act. *Disability Discrimination Act.* Available from: http://www.legislation.gov.uk/ukpga/1995/50/contents [Accessed 30th September 2015].

65. IDEA Partnership. *Definition of Child with a Disability.* Available from: http://www.ideapartnership.org/topics-database/idea-2004/idea-2004-part-b/1396-definition-of-child-with-a-disability.html [Accessed 9th February 2017].

66. Crow L. Including all of our lives: Renewing the social model of disability. In: Morris J (Ed.), Encounters with strangers: Feminism and disability. London: Women's Press; 1996: pp.206-26.

67. Oliver M. The individual and social models of disability. Paper presented at Joint Workshop of the Living Options Group and the Research Unit of the Royal

College of Physicians; 1990. Available from:

https://files.eric.ed.gov/fulltext/ED510913.pdf [Accessed 30<sup>th</sup> September 2015]. 68. Baldwin S. The costs of caring: Families with disabled children. Oxon: Routledge & Kegan Paul; 1985.

69. French S. What's so great about independence? In Swain J, Finkelstein F, French S, Oliver M. (Eds.), Disabling barriers–enabling environments. London: Sage; 1993:pp. 44-8.

70. Morris J. Pride against prejudice: A personal politics of disability. London: Women's Press; 1991.

71. World Health Organization. International classification of functioning, disability and health: ICF. Geneva: World Health Organization; 2001.

72. World Health Organization. *Towards a Common Language for Functioning, Disability and Health: ICF.* Available from:

http://www.who.int/classifications/icf/training/icfbeginnersguide.pdf [Accessed 11<sup>th</sup> December 2014].

73. Shakespeare T. The social model of disability. In: Davis LJ (Ed.), The disability studies reader. New York: Routledge; 2006:pp.197-204.

74. Department of Education. *Removing barriers to achievement: The Government's strategy for special educational needs.* Available from:

http://www.advanced-training.org.uk/resources/Universal/removing\_barriers.pdf [Accessed 1<sup>st</sup> October 2015].

75. Wood P. Appreciating the consequences of disease: The international classification of impairments, disabilities, and handicaps. WHO Chron. 1980;34(10):376-80.

76. Callahan D. The WHO definition of 'health'. Stud Hastings Cent. 1973;1(3):77-87.

77. Gill TM, Feinstein AR. A critical appraisal of the quality of quality-of-life measurements. JAMA-J Am Med Assoc. 1994;272(8):619-26.

78. Farquhar M. Definitions of quality of life: a taxonomy. J Adv Nurs. 1995;22(3):502-8.

79. Eiser C, Morse R. The measurement of quality of life in children: Past and future perspectives. J Dev Behav Pediatr. 2001;22(4):248-56.

80. WHOQOL Group. The World Health Organization quality of life assessment (WHOQOL): Position paper from the World Health Organization. Soc Sci Med. 1995;41(10):1403-9.

81. Maguire P, Brooke M, Tait A, Thomas C, Sellwood R. The effect of counselling on physical disability and social recovery after mastectomy. Clin Oncol. 1983;9(4):319-24.

82. Greer S, Moorey S, Baruch J, Watson M, Robertson BM, Mason A, et al. Adjuvant psychological therapy for patients with cancer: A prospective randomised trial. Brit Med J. 1992;304(6828):675-80.

83. Macmillan Cancer Support. *About us.* Available from:

https://www.macmillan.org.uk/about-us [Accessed 19<sup>th</sup> October 2017].

84. Samaritans. *How we can help you*. Available from:

https://www.samaritans.org/how-we-can-help-you [Accessed 19<sup>th</sup> October 2017].
85. Goode D. The national quality of life for persons with disabilities project: A quality of life agenda for the United States. Cambridge (MA): Brookline Press; 1994.
86. Leplege A, Hunt S. Defining and measuring quality of life in medicine. JAMA-J

Am Med Assoc. 1998;279(6):431.

87. Albrecht GL, Higgins PC. Rehabilitation success: the interrelationships of multiple criteria. J Health Soc Behav. 1977:36-45.

88. Albrecht GL, Devlieger PJ. The disability paradox: High quality of life against all odds. Soc Sci Med. 1999;48(8):977-88.

89. Lerner DJ, Levine S. Health-related quality of life: Origins, gaps, and directions. In: Albrecht GL, Fitzpatrick R (Eds.), Advances in medical sociology: Quality of life in health care. Greenwich (CT): JAI Press; 1994:pp. 43-65.
90. Albrecht GL. Using subjective health assessments in practice and policymaking. Health Care Anal. 1996;4(4):284-92.

91. Taillefer M-C, Dupuis G, Roberge M-A, Le May S. Health-related quality of life models: Systematic review of the literature. Soc Indic Res. 2003;64(2):293-323.

92. King S, Schwellnus H, Russell D, Shapiro L, Aboelele O. Assessing quality of life of children and youth with disabilities: A review of available measures. Hamilton, Ontario: McMaster University, CanChild: Centre for Childhood Disability Research; 2005.

93. Spieth LE, Harris CV. Assessment of health-related quality of life in children and adolescents: An integrative review. J Pediatr Psychol. 1996;21(2):175-93.

94. McDougall J, Wright V, Rosenbaum P. The ICF model of functioning and disability: incorporating quality of life and human development. Dev Neurorehabil. 2010;13(3):204-11.

95. World Health Organization. *Life Expectancy.* Available from: http://www.who.int/gho/mortality\_burden\_disease/life\_tables/situation\_trends/en/ [Accessed 25<sup>th</sup> October 2017].

96. Darzi A. High quality care for all: NHS next stage review final report. London: Department of Health; 2008.

97. Health and Social Care Information Centre. *Background Information about PROMs.* Available from: http://www.hscic.gov.uk/proms-background [Accessed 1<sup>st</sup> October 2015].

98. Browne J, Jamieson L, Lawsey J, Van der Meulen J, Black N, Cairns J. Patient reported outcome measures (PROMs) in elective surgery. Report to the Department of Health. London: London School of Hygiene and Tropical Medicine, Royal College of Surgeons of England; 2007.

99. Timmins N. Assessing patient care: NHS goes to the PROMS. Brit Med J. 2008;336(7659):1464.

100. Devlin N, Appleby J. Getting the most out of PROMs: Putting health outcomes at the heart of NHS decision-making. London: The King's Fund; 2010.

101. Department of Health. *NHS Outcomes Framework 2012/13.* Available from https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/213711/ dh\_131723.pdf [Accessed 1<sup>st</sup> October 2017].

102. Weenink J-W, Braspenning J, Wensing M. Patient reported outcome measures (PROMs) in primary care: An observational pilot study of seven generic instruments. BMC Fam Pract. 2014;15(1):88.

103. Black N. Patient reported outcome measures could help transform healthcare. Brit Med J. 2013;346:10.1136/bmj.f167.

104. Department of Health. *Patient Repported Outcome Measures (PROMs) in England: A Methodology for Identifying Outliers.* Available from:

https://www.gov.uk/government/publications/patient-reported-outcome-measuresproms-in-england-a-methodology-for-identifying-potential-outliers--2 [Accessed 24<sup>th</sup> April 2017].

105. Janssens A, Thompson Coon J, Rogers M, Allen K, Green C, Jenkinson C, et al. A systematic review of generic multidimensional patient-reported outcome measures for children, part I: Descriptive characteristics. Value Health. 2015;18(2):315-33.

106. Feinstein AR. Clinimetric perspectives. J Chron Dis. 1987;40(6):635-40.

107. Eiser C, Morse R. Quality-of-life measures in chronic diseases of childhood. Health Tech Assess. 2001;5:1-158.

108. Efficace F, Baccarani M, Breccia M, Saussele S, Abel G, Caocci G, et al. International development of an EORTC questionnaire for assessing health-related quality of life in chronic myeloid leukemia patients: The EORTC QLQ-CML24. Qual Life Res. 2014;23(3):825-36.

109. Lyons GJ. The "PRESTON profile"--the first disease-specific tool for assessing quality of life in patients with malignant glioma. Disabil Rehabil. 1996;18(9):460-8.

110. O'Leary E, Drummond FJ, Gavin A, Kinnear H, Sharp L. Psychometric evaluation of the EORTC QLQ-PR25 questionnaire in assessing health-related quality of life in prostate cancer survivors: A curate's egg. Qual Life Res. 2015;24(9):2219-30.
111. Garin O, Ferrer M, Pont A, Rue M, Kotzeva A, Wiklund I, et al. Disease-specific health-related quality of life questionnaires for heart failure: A systematic review with

meta-analyses. Qual Life Res. 2009;18(1):71-85.. 112. Ulla W, Anna S, Arestedt K. Development and validation of an arrhythmiaspecific scale in tachycardia and arrhythmia with focus on health-related quality of life. J Cardiovasc Nurs. 2015;30(2):98-108.

113. White J, Withers KL, Lencioni M, Carolan-Rees G, Wilkes AR, Wood KA, et al. Cardiff cardiac ablation patient-reported outcome measure (C-CAP): Validation of a new questionnaire set for patients undergoing catheter ablation for cardiac arrhythmias in the UK. Qual Life Res. 2016;25(6):1571-83.

114. Schulz K-H, Kroencke S, Ewers H, Schulz H, Younossi ZM. The factorial structure of the Chronic Liver Disease Questionnaire (CLDQ). Qual Life Res. 2008;17(4):575-84.

115. van der Plas SM, Hansen BE, de Boer JB, Stijnen T, Passchier J, de Man RA, et al. The Liver Disease Symptom Index 2.0: Validation of a disease-specific questionnaire. Qual Life Res. 2004;13(8):1469-81.

116. Silverman LR, Demakos EP, Peterson BL, Kornblith AB, Holland JC, Odchimar-Reissig R, et al. Randomized controlled trial of azacitidine in patients with the myelodysplastic syndrome: A study of the cancer and leukemia group B. J Clin Oncol. 2002;20(10):2429-40.

117. Grant AM, Wileman SM, Ramsay CR, Bojke L, Sculpher MJ, Kilonzo MM, et al. The effectiveness and cost-effectiveness of minimal access surgery amongst people with gastro-oesophageal reflux disease–a UK collaborative study. Health Technol Asses. 2008; 12(31).

118. Grant AM, Wileman SM, Ramsay CR, Mowat NA, Krukowski ZH, Heading RC, et al. Minimal access surgery compared with medical management for chronic gastrooesophageal reflux disease: UK collaborative randomised trial. Brit Med J. 2008;337:10.1136/bmj.a2664.

119. Williams O, Fitzpatrick R, Hajat S, Reeves BC, Stimpson A, Morris RW, et al. Mortality, morbidity, and 1-year outcomes of primary elective total hip arthroplasty. J Arthroplasty. 2002;17(2):165-71.

120. Malchau H, Garellick G, Eisler T, Kärrholm J, Herberts P. Presidential guest address: The Swedish Hip Registry: Increasing the sensitivity by patient outcome data. Clin Orthop Relat R. 2005;441:19-29.

121. Appleby J, Devlin N, Maynard A, Scott H, Vallance-Owen A. Measuring success in the NHS. London: Kings Fund and City University; 2004.

122. Dawson J, Doll H, Fitzpatrick R, Jenkinson C, Carr AJ. The routine use of patient reported outcome measures in healthcare settings. Brit Med J. 2010;340.

123. Valderas J, Kotzeva A, Espallargues M, Guyatt G, Ferrans C, Halyard MY, et al. The impact of measuring patient-reported outcomes in clinical practice: A systematic review of the literature. Qual Life Res. 2008;17(2):179-93.

124. Detmar SB, Muller MJ, Schornagel JH, Wever LD, Aaronson NK. Health-related quality-of-life assessments and patient-physician communication: A randomized controlled trial. JAMA-J Am Med Assoc. 2002;288(23):3027-34.

125. Greenhalgh J, Meadows K. The effectiveness of the use of patient-based measures of health in routine practice in improving the process and outcomes of patient care: A literature review. J Eval Clin Pract. 1999;5(4):401-16.

126. Meadows K, Rogers D, Greene T. Attitudes to the use of health outcome questionnaires in the routine care of patients with diabetes: A survey of general practitioners and practice nurses. Brit J Gen Pract. 1998;48(434):1555-9.

127. Morris J, Perez D, McNoe B. The use of quality of life data in clinical practice. Qual Life Res. 1997;7(1):85-91.

128. Christie MJ, French D, Sowden A, West A. Development of child-centered disease-specific questionnaires for living with asthma. Psychosom Med. 1993;55(6):541-8.

129. Modi AC, Quittner AL. Validation of a disease-specific measure of healthrelated quality of life for children with cystic fibrosis. J Pediatr Psychol. 2003;28(8):535-46.

130. Schneider JW, Gurucharri LM, Gutierrez AL, Gaebler-Spira DJ. Health-related quality of life and functional outcome measures for children with cerebral palsy. Dev Med Child Neurol. 2001;43(9):601-8.

131. Todorov BK. Validity and reliability of the adolescent versions of the Migraine Specific Quality of Life Questionnaire and the Headache Disability Inventory. Dissertation Abstracts International: Section B: The Sciences and Engineering. 2015;76(5-B(E)):No Pagination Specified.

132. Uwe N, Eva O, Ann-Charlotte E, Gun N, Ann G. Validation of a disease-specific questionnaire for measuring parent-reported health-related quality of life in children with allergies. Scand J Caring Sci. 2012;26(4):679-87.

133. Brouwer CN, Schilder AG, van Stel HF, Rovers MM, Veenhoven RH, Grobbee DE, et al. Reliability and validity of functional health status and health-related quality of life questionnaires in children with recurrent acute otitis media. Qual Life Res. 2007;16(8):1357-73.

134. Eiser C, Mohay H, Morse R. The measurement of quality of life in young children. Child Care Hlth Dev. 2000;26(5):401-13.

135. García-Morales LM, Berber A, Macias-Lara CC, Lucio-Ortiz C, Del-Rio-Navarro BE, Dorantes-Alvárez LM. Use of sibutramine in obese mexican adolescents: A 6-month, randomized, double-blind, placebo-controlled, parallel-group trial. Clin Ther. 2006;28(5):770-82.

136. Eiser C, Morse R. Can parents rate their child's health-related quality of life? Results of a systematic review. Qual Life Res. 2001;10:347-57.

137. Varni JW, Limbers CA, Burwinkle TM. How young can children reliably and validly self-report their health-related quality of life?: An analysis of 8,591 children across age subgroups with the PedsQL<sup>™</sup> 4.0 Generic Core Scales. Health Qual Life Out. 2007;5(1):1.

138. Landgraf JM, Abetz L. Influences of sociodemographic characteristics on parental reports of children's physical and psychosocial well-being: Early experiences with the Child Health Questionnaire. In: Drotar D (Ed.), Measuring health related quality of life in children and adolescents. Mahwah (NJ): Lawrence Erlbaum Associates; 1998:pp.105-30.

139. Upton P, Lawford J, Eiser C. Parent–child agreement across child healthrelated quality of life instruments: A review of the literature. Qual Life Res. 2008;17(6):895-913.

140. Phipps S, Dunavant M, Jayawardene D, Srivastiva DK. Assessment of healthrelated quality of life in acute in-patient settings: Use of the BASES instrument in children undergoing bone marrow transplantation. Int J Cancer. 1999;83(S12):18-24.

141. Eiser C, Havermans T, Craft A, Kernahan J. Development of a measure to assess the perceived illness experience after treatment for cancer. Arch Dis Child. 1995;72(4):302-7.

142. Youngstrom E, Loeber R, Stouthamer-Loeber M. Patterns and correlates of agreement between parent, teacher, and male adolescent ratings of externalizing and internalizing problems. J Consult Clin Psych. 2000;68(6):1038.

143. Vogels T, Verrips G, Verloove-Vanhorick S, Fekkes M, Kamphuis R, Koopman H, et al. Measuring health-related quality of life in children: The development of the TACQOL parent form. Qual Life Res. 1998;7(5):457-65.

144. Varni JW, Katz ER, Colegrove Jr R, Dolgin MJ. Adjustment of children with newly diagnosed cancer: Cross-informant variance. J Psychosoc Oncol. 1996;13(4):23-38.

145. Matza LS, Swensen AR, Flood EM, Secnik K, Leidy NK. Assessment of healthrelated quality of life in children: A review of conceptual, methodological, and regulatory issues. Value Health. 2004;7(1):79-92.

146. Matza LS, Patrick DL, Riley AW, Alexander JJ, Rajmil L, Pleil AM, et al. Pediatric patient-reported outcome instruments for research to support medical product labeling: Report of the ISPOR PRO good research practices for the assessment of children and adolescents task force. Value Health. 2013;16(4):461-79.

147. Streiner DL, Norman GR, Cairney J. Health measurement scales: A practical guide to their development and use. Oxford (UK): Oxford University Press; 2014.
148. Oppenheim AN. Questionnaire design, interviewing and attitude measurement. New York: Bloomsbury Publishing; 2000.

149. Willms DG, Johnson NA. Essentials in qualitative research: A notebook for the field. Unpublished manuscript. Hamilton, Ont: McMaster University; 1993.

150. Mokkink LB, Terwee CB, Knol DL, Stratford PW, Alonso J, Patrick DL, et al. Protocol of the COSMIN study: COnsensus-based Standards for the selection of health Measurement INstruments. BMC Med Res Methodol. 2006;6(1):2.

151. Mokkink LB, Terwee CB, Knol DL, Stratford PW, Alonso J, Patrick DL, et al. The COSMIN checklist for evaluating the methodological quality of studies on measurement properties: a clarification of its content. BMC Med Res Methodol. 2010;10(1):22.

152. Fitzpatrick R, Davey C, Buxton MJ, Jones DR. Evaluating patient-based outcome measures for use in clinical trials. Health Technol Asses. 1998;2(14).

153. DeVellis RF. Classical test theory. Med Care. 2006;44(11):S50-S9.

154. Lord FM. Applications of item response theory to practical testing problems. Mahwah (NJ): Routledge; 1980.

155. Weiss D, Yoes M. Item Response Theory. In: Hambleton R, Zaal J (Eds.), Advances in educational and psychological testing: Theory and applications. New York: Kluwer Academic Publishers; 1991:pp. 69-95.

156. Rasch G. Probabilistic models for some intelligence and attainment tests. Chicago (IL): MESA Press; 1993.

157. Turner RR, Quittner AL, Parasuraman BM, Kallich JD, Cleeland CS. Patientreported outcomes: Instrument development and selection issues. Value Health. 2007;10(s2):S86-S93.

158. Spearman C. The proof and measurement of association between two things. Am J Psychol. 1904;15(1):72-101.

159. Bunge WW. The geography. Prof Geogr. 1973;25(4):331-7.

160. Tadić V, Cumberland PM, Lewando-Hundt G, Rahi JS, Dale N, Khaw PT, et al. Do visually impaired children and their parents agree on the child's vision-related quality of life and functional vision? Brit J Ophthalmol. 2016:10.1136/bjophthalmol-2016-308582.

161. National Institutes of Health. *NIH Human Subjects Policies and Guidance.* Available from: https://humansubjects.nih.gov/nih-human-subjects-policies-guidance [Accessed 15<sup>th</sup> March 2017].

162. World Medical Association. *WMA Declaration of Helsinki - Ethical Principles for Medical Research Involving Human Subjects.* Available from:

https://www.wma.net/policies-post/wma-declaration-of-helsinki-ethical-principles-formedical-research-involving-human-subjects/ [Accessed 15<sup>th</sup> March 2017].

163. British Psychological Society. Code of human research ethics. Leicester (UK): The British Psychological Society; 2010.

164. Einarsdóttir J. Research with children: Methodological and ethical challenges. Eur Early Child Educ. 2007;15(2):197-211.

165. Irwin LG, Johnson J. Interviewing young children: Explicating our practices and dilemmas. Qual Health Res. 2005;15(6):821-31.

166. Tisdall EKM. The challenge and challenging of childhood studies? Learning from disability studies and research with disabled children. Child Soc. 2012;26(3):181-91.

167. Walker EA, Newman E, Koss M, Bernstein D. Does the study of victimization revictimize the victims? Gen Hosp Psychiat. 1997;19(6):403-10.

168. Larson R, Asmussen L. Anger, worry, and hurt in early adolescence: An enlarging world of negative emotions. In: Colten ME, Gore S (Eds.), Adolescent stress: Causes and consequences. New York: Aldine De Gruyter; 1991:pp. 21-41.

169. Larson R, Ham M. Stress and "storm and stress" in early adolescence: The relationship of negative events with dysphoric affect. Dev Psychol. 1993;29(1):130.
170. Moolchan ET, Mermelstein R. Research on tobacco use among teenagers: Ethical challenges. J Adolescent Health. 2002;30(6):409-17.

171. Beresford B. Personal accounts: Involving disabled children in research. London: Social Policy Research Unit, The Stationary Office; 1997.

172. Pascolini D, Mariotti SP. Global estimates of visual impairment: 2010. Brit J Ophthalmol. 2011;96:614-18.

173. Rahi JS, Cable N. Severe visual impairment and blindness in children in the UK. Lancet. 2003;362(9393):1359-65.

174. Rahi JS, Dezateux C. Epidemiology of visual impairment in Britain. Arch Dis Child. 1998;78(4):381-6.

175. World Health Organization. International statistical classification of diseases, injuries and causes of death. Can Med Assoc J; 1961;85(5):266-267.

176. Dandona L, Dandona R. Revision of visual impairment definitions in the International Statistical Classification of Diseases. BMC Med. 2006;4(1):7.

177. World Health Organization. *Fact Sheet no. 282 - Visual impairment and blindness.* Available from: http://www.who.int/mediacentre/factsheets/fs282/en/ [Accessed 12<sup>th</sup> November 2014].

178. World Health Organization. *International Classification of Diseases (ICD)*. Available from: http://www.who.int/classifications/icd/en/ [Accessed 30<sup>th</sup> November 2015].

179. Mitry D, Bunce C, Wormald R, Leamon S, Simkiss P, Cumberland P, et al.
Causes of certifications for severe sight impairment (blind) and sight impairment (partial sight) in children in England and Wales. Brit J Ophthalmol. 2013;97(11):1431-6.
180. Macleod S, Appleton R. Neurological disorders presenting mainly in

adolescence. Arch Dis Child. 2007;92(2):170-5.

181. Fazzi E, Signorini SG, Bova SM, La Piana R, Ondei P, Bertone C, et al. Spectrum of visual disorders in children with cerebral visual impairment. J Child Neurol. 2007;22(3):294-301.

182. Trevarthen C. Communication and cooperation in early infancy: A description of primary intersubjectivity. In: Bullowa M (Ed.), Before speech: The beginning of interpersonal communication. Cambridge (UK): Cambridge University Press; 1979:pp. 530-71.

183. Mitchell SA. Relationality: From attachment to intersubjectivity. New York: Psychology Press; 2014.

184. Trevarthen C. (1978) Secondary intersubjectivity: Confidence, confiding and acts of meaning in the first year In: Lock A (Ed.), Action, gesture and symbol: The emergence of language. London: Academic press; 1978:pp. 183-229.

185. Baillargeon R, Spelke ES, Wasserman S. Object permanence in five-month-old infants. Cognition. 1985;20(3):191-208.

186. Bruce SM, Vargas C. Assessment and instruction of object permanence in children with blindness and multiple disabilities. J Visual Impair Blin. 2012;106(11):717-27.

187. Fazzi E, Signorini SG, Bomba M, Luparia A, Lanners J, Balottin U. Reach on sound: A key to object permanence in visually impaired children. Early Hum Dev. 2011;87(4):289-96.

188. Hobson RP. On the origins of self and the case of autism. Dev Psychopathol. 1990;2(2):163-81.

189. Hobson RP. Autism and the development of mind. East Sussex: Lawrence Erlbaum Associates; 1995.

190. Hobson RP. The cradle of thought: Exploring the origin of thinking. Oxford (UK): Oxford University Press; 2002.

191. Urwin C. The development of communication between blind infants and their parents. In Lock A (Ed.), Action, gesture and symbol: The emergence of language. London: Academic Press; 1978:pp. 79-108.

192. Bigelow AE. Locomotion and search behavior in blind infants. Infant Behav Dev. 1992;15(2):179-89.

193. Bigelow AE. The development of joint attention in blind infants. Dev Psychopathol. 2003;15(2):259-75.

194. Dale N, Salt A. Social identity, autism and visual impairment (VI) in the early years. Br J Vis Impair. 2008;26(2):135-46.

195. Premack D, Woodruff G. Does the chimpanzee have a theory of mind? Behav Brain Sci. 1978;1(4):515-26.

196. Leslie AM. Pretense and representation: The origins of "theory of mind". Psychol Rev. 1987;94(4):412.

197. Peterson CC, Peterson JL, Webb J. Factors influencing the development of a theory of mind in blind children. Brit J Dev Psychol. 2000;18(3):431-47.

198. Green S, Pring L, Swettenham J. An investigation of first-order false belief understanding of children with congenital profound visual impairment. Brit J Dev Psychol. 2004;22(1):1-17.

199. Pring L, Dewart H, Brockbank M. Social cognition in children with visual impairments. J Visual Impair Blin. 1998;92:754-68.

200. Mundy P, Fox N, Card J. EEG coherence, joint attention and language development in the second year. Developmental Sci. 2003;6(1):48-54.

201. Pérez-Pereira M, Conti-Ramsden G. Language development and social interaction in blind children. East Sussex: Psychology Press; 2013.

202. Andersen ES, Dunlea A, Kekelis LS. Blind children's language: Resolving some differences. J Child Lang. 1984;11(3):645-64.

203. Fraiberg S. Insights from the blind: Comparative studies of blind and sighted infants. New York: Basic Books; 1977.

204. Andersen ES, Dunlea A, Kekelis L. The impact of input: Language acquisition in the visually impaired. First Lang. 1993;13(37):23-49.

205. Sacks S, Kekelis L. The development of social skills by blind and visually impaired students: Exploratory studies and strategies. New York: American Foundation for the Blind; 1992.

206. Greenaway R, Pring L, Schepers A, Isaacs D, Dale N. Neuropsychological presentation and adaptive skills in high-functioning adolescents with visual impairment: A preliminary investigation. Appl Neuropsych-Chil. 2017;6(2):145-57.

207. Harris P. Desires, beliefs, and language. In: Carruthers P, Smith PK (Eds.), Theories of theories of mind. Cambridge (UK): Cambridge University Press; 1996:pp. 200-221).

208. Dunn J. Changing minds and changing relationships. In: Lewis C, Mitchell P (Eds.), Children's early understanding of mind: Origins and development. East Sussex: Lawrence Erlbaum Associates; 1994:pp. 297-310.

209. Amedi A, Merabet LB, Bermpohl F, Pascual-Leone A. The occipital cortex in the blind: Lessons about plasticity and vision. Curr Dir Psychol Sci. 2005;14(6):306-11.
210. Pring L. Psychological characteristics of children with visual impairments: Learning, memory and imagery. British J Vis Impair. 2008;26(2):159-69.

211. Dale N, Sonksen P. Developmental outcome, including setback, in young children with severe visual impairment. Dev Med Child Neurol. 2002;44(9):613-22.
212. Khadka J, Ryan B, Margrain TH, Woodhouse JM, Davies N. Listening to voices of children with a visual impairment: A focus group study. Brit J Vis Impair. 2012;30(3):182-96.

213. Cochrane G, Lamoureux E, Keeffe J. Defining the content for a new quality of life questionnaire for students with low vision (the Impact of Vision Impairment on Children: IVI\_C). Ophthal Epidemiol. 2008;15(2):114-20.

214. McGaha CG, Farran DC. Interactions in an inclusive classroom: The effects of visual status and setting. J Visual Impair Blin. 2001;95(2):80-94.

215. D'Allura T. Enhancing the social interaction skills of preschoolers with visual impairments. J Visual Impair Blin. 2002;96(8):576-84.

216. Buhrow MM, Hartshorne TS, Bradley-Johnson S. Parents' and teachers' ratings of the social skills of elementary-age students who are blind. J Visual Impair Blin. 1998;92:503-11.

217. Dormal V, Crollen V, Baumans C, Lepore F, Collignon O. Early but not late blindness leads to enhanced arithmetic and working memory abilities. Cortex. 2016;83:212-21.

218. Tao Q, Chan CC, Luo Y-j, Li J-j, Ting K-h, Wang J, et al. How does experience modulate auditory spatial processing in individuals with blindness? Brain Topogr. 2015;28(3):506-19.

219. Burton H, Snyder AZ, Conturo TE, Akbudak E, Ollinger JM, Raichle ME. Adaptive changes in early and late blind: A fMRI study of Braille reading. Journal of Neurophysiol. 2002;87(1):589-607.

220. Burton H. Visual cortex activity in early and late blind people. J Neurosci. 2003;23(10):4005-11.

221. Büchel C, Price C, Frackowiak R, Friston K. Different activation patterns in the visual cortex of late and congenitally blind subjects. Brain. 1998;121(3):409-19.

222. Wan CY, Wood AG, Reutens DC, Wilson SJ. Early but not late-blindness leads to enhanced auditory perception. Neuropsychologia. 2010;48(1):344-8.

223. Kujala T, Alho K, Huotilainen M, Ilmoniemi R, Lehtokoski A, Leinonen A, et al. Electrophysiological evidence for cross-modal plasticity in humans with early-and late-onset blindness. Psychophysiology. 1997;34(2):213-6.

224. Burton H, Snyder A, Diamond J, Raichle M. Adaptive changes in early and late blind: A FMRI study of verb generation to heard nouns. J Neurophysiol. 2002;88(6):3359-71.

225. Roder B, Rosler F. Memory for environmental sounds in sighted, congenitally blind and late blind adults: Evidence for cross-modal compensation. Int J Psychophysiol. 2003;50(1-2):27-39.

226. Cohen LG, Weeks RA, Sadato N, Celnik P, Ishii K, Hallett M. Period of susceptibility for cross-modal plasticity in the blind. Ann Neurol. 1999;45(4):451-60. 227. Rutter M. Psychosocial resilience and protective mechanisms. Am Orthopsychiat. 1987;57(3):316.

228. Haase JE, Heiney SP, Ruccione KS, Stutzer C. Research triangulation to derive meaning-based quality-of-life theory: Adolescent Resilience Model and instrument development. Int J Cancer. 1999;83(S12):125-31.

229. Dodds A, Ferguson E, Ng L, Flannigan H, Hawes G, Yates L. The concept of adjustment: A structural model. J Visual Impair Blin. 1994;88:487.

230. Hewett R, Douglas G, Keil S. Post-16 transition experience of visually impaired young people in England and Wales: Early findings from a longitudinal study. Brit J Vis Impair. 2014;32(3):211-22.

231. Rainey L, Elsman EBM, Nispen RMA, Leeuwen LM, Rens GHMB.

Comprehending the impact of low vision on the lives of children and adolescents: A qualitative approach. Qual Life Res. 2016;25(10):2633-2643.

232. Cimera RE, Rumrill PD, Chan F, Kaya C, Bezyak J. Vocational rehabilitation services and outcomes for transition-age youth with visual impairments and blindness. Journal of Vocational Rehabilitation. 2015;43(2):103-11.

233. McDonnall MC. Factors predicting post-high school employment for young adults with visual impairments. Rehabil Couns Bull. 2010;54(1):36-45.

234. Kelly SM, Wolffe KE. Internet use by transition-aged youths with visual impairments in the United States: Assessing the impact of postsecondary predictors. J Visual Impair Blin. 2012;106(10):597-608.

235. Appelhans P, Rath W. Counselling and support of visually handicapped students regarding choice of occupation and vocational training during transition from school to work. Int J Rehabil Res. 1988;11(1):93-4.

236. Douglas G, Hewett R. Views of independence and readiness for employment amongst young people with visual impairment in the UK. Australian Journal of Rehabilitation Counselling. 2014;20(2):81-99.

237. Mitchell PJ. Beyond the classroom: A summer transition program for blind and visually impaired adolescents. RE:view: Rehabilitation and Education for Blindness and Visual Impairment. 2001;32(4):151-5.

238. Keil S, Crews N. Post-16 and post-18 transitions of young people with visual impairment in Wales. Brit J Vis Impair. 2008;26(2):190-201.

239. Pfeiffer JP, Pinquart M. Goal engagement and goal attainment in adolescents with and without visual impairment. J Adolescence. 2012;35(4):909-16.

240. Tadić V, Hundt GL, Keeley S, Rahi JS. Seeing it my way: living with childhood onset visual disability. Child Care Hlth Dev. 2014;41(2):239-48.

241. Musch DC, Farjo AA, Meyer RF, Waldo MN, Janz NK. Assessment of health-related quality of life after corneal transplantation. Am J Ophthalmol. 1997;124(1):1-8.
242. Globe DR, Levin S, Chang TS, Mackenzie PJ, Azen S. Validity of the SF-12 quality of life instrument in patients with retinal diseases. Ophthalmology. 2002;109(10):1793-8.

243. Buck D, Clarke MP, Powell C, Tiffin P, Drewett RF. Use of the PedsQL in childhood intermittent exotropia: Estimates of feasibility, internal consistency reliability and parent-child agreement. Qual Life Res. 2012;21(4):727-36.

244. Rajagopalan K, Abetz L, Mertzanis P, Espindle D, Begley C, Chalmers R, et al. Comparing the discriminative validity of two generic and one disease-specific health-related quality of life measures in a sample of patients with dry eye. Value Health. 2005;8(2):168-74.

245. Mangione CM, Lee PP, Pitts J, Gutierrez P, Berry S, Hays RD. Psychometric properties of the National Eye Institute visual function questionnaire (NEI-VFQ). Arch Ophthalmol-Chic. 1998;116(11):1496-504.

246. Margolis MK, Coyne K, Kennedy-Martin T, Baker T, Schein O, Revicki DA. Vision-specific instruments for the assessment of health-related quality of life and visual functioning. Pharmacoeconomics. 2002;20(12):791-812.

247. Vitale S, Schein OD, Meinert CL, Steinberg EP. The refractive status and vision profile: A questionnaire to measure vision-related quality of life in persons with refractive error. Ophthalmology. 2000;107(8):1529-39.

248. Keeffe J, McCarty C, Hassell JB, Gilbert A. Description and measurement of handicap caused by vision impairment. Aust N Z J Ophthalmol. 1999;27(3-4):184-6.
249. Chadha RK, Subramanian A. The effect of visual impairment on quality of life of children aged 3-16 years. Brit J Ophthalmol. 2011;95(5):642-5.

250. Bleything WB, Landis SL. The College of Optometrists in Vision Development -QOL Questionnaire in a socially at-risk population of youth. Optometry and Vision Development. 2008;39(2):82-90.

251. Vitale MG, Levy DE, Johnson MG, Gelijns AC, Moskowitz AJ, Roye BP, et al. Assessment of quality of life in adolescent patients with orthopaedic problems: Are adult measures appropriate? J Pediatr Orthoped. 2001;21(5):622-8.

252. Borgers N, De Leeuw E, Hox J. Children as respondents in survey research: Cognitive development and response quality. Bulletin de methodologie Sociologique. 2000;66(1):60-75.

253. Mangione CM, Lee PP, Gutierrez PR, Spritzer K, Berry S, Hays RD. Development of the 25-list-item national eye institute visual function questionnaire. Arch Ophthalmol-Chic. 2001;119(7):1050-8.

254. Steinberg EP, Tielsch JM, Schein OD, Javitt JC, Sharkey P, Cassard SD, et al. The VF-14: An index of functional impairment in patients with cataract. Arch Ophthalmol-Chic. 1994;112(5):630-8.

255. Mozlin R. Quality-of-life outcomes assessment. J Optom Vis Dev. 1995;26(4):194-9.

256. Wolffsohn JS, Cochrane AL. Design of the low vision quality-of-life questionnaire (LVQOL) and measuring the outcome of low-vision rehabilitation. Am J Ophthalmol. 2000;130(6):793-802.

257. Tadić V, Hogan A, Sobti N, Knowles RL, Rahi JS. Patient-reported outcome measures (PROMs) in paediatric ophthalmology: A systematic review. Brit J Ophthalmol. 2013;0:10.1136/bjophthalmol-2013-303350.

258. Elsman EBM, van Nispen RMA, van Rens G. Feasibility of the Participation and Activity Inventory for Children and Youth (PAI-CY) and Young Adults (PAI-YA) with a visual impairment: A pilot study. Health Qual Life Out. 2017;15(1):98.

259. Okumura T, Miura T, Nakanishi M, Uno M, Wakamiya E, Tamai H. Development of a vision-related symptom and performance checklist for children. Brain Dev-Jpn. 2013;45(5):360-5.

260. Bokhary KA, Suttle C, Alotaibi AG, Stapleton F, Boon MY. Development and validation of the 21-item children's vision for living scale (CVLS) by Rasch analysis. Clin Exp Optom. 2013;96(6):566-76.

261. Crescioni M, Messer DH, Warholak TL, Miller JM, Twelker JD, Harvey EM. Rasch analysis of the Student Refractive Error and Eyeglass Questionnaire. Optometry Vision Sci. 2014;91(6):624-33.

262. Khadka J, Ryan B, Margrain TH, Court H, Woodhouse JM. Development of the 25-item Cardiff Visual Ability Questionnaire for Children (CVAQC). British J Ophthalmol. 2010;94(6):730-5.

263. Huang J, Khadka J, Gao R, Zhang S, Dong W, Bao F, et al. Validation of an instrument to assess visual ability in children with visual impairment in China. Brit J Ophthalmol. 2016;101(4):475-80.

264. Ozen Tunay Z, Caliskan D, Oztuna D, Idil A. Validation and reliability of the Cardiff Visual Ability Questionnaire for Children using Rasch analysis in a Turkish population. Brit J Ophthalmol. 2016;100(4):520-4.

265. Birch EE, Cheng CS, Felius J. Validity and reliability of the Children's Visual Function Questionnaire (CVFQ). J AAPOS. 2007;11(5):473-9.

266. Felius J, Stager DR, Berry PM, Fawcett SL, Salomão SR, Berezovsky A, et al. Development of an instrument to assess vision-related quality of life in young children. Am J Ophthalmol. 2004;138(3):362-72.

267. Farin E, Metten M, Nagl M, Lagreze WA, Pieh-Beisse C. Psychometric properties of the FKS: Reliability, validity, Rasch analysis, and descriptive results of the German version of the Childrens Visual Function Questionnaire. Ophthalmologe. 2011;108(7):644-50.

268. Cochrane GM, Marella M, Keeffe JE, Lamoureux EL. The Impact of Vision Impairment for Children (IVI\_C): Validation of a vision-specific pediatric quality-of-life questionnaire using Rasch analysis. Invest Ophth Vis Sci. 2011;52(3):1632-40.

269. Gothwal VK, Lovie-Kitchin JE, Nutheti R. The development of the LV Prasad-Functional Vision Questionnaire: A measure of functional vision performance of visually impaired children. Invest Ophth Vis Sci. 2003;44(9):4131-9.

270. Gothwal VK, Sumalini R, Bharani S, Reddy SP, Bagga DK. The second version of the L. V. Prasad-functional vision questionnaire. Optometry Vision Sci. 2012;89(11):1601-10.

271. Cole SR, Beck RW, Moke PS, Celano MP, Drews CD, Repka MX, et al. The Amblyopia Treatment Index. J AAPOS. 2001;5(4):250-4.

272. Carlton J. Developing the draft descriptive system for the child amblyopia treatment questionnaire (CAT-QoI): A mixed methods study. Health Qual Life Outcomes. 2013;11:174.

273. Carlton J. Development of the Child Amblyopia Treatment Questionnaire (CAT-QoL): A disease-specific health related quality of life (HRQOL) measure for amblyopia. Discussion Paper. (Unpublished).

274. Borsting EJ, Rouse MW, Mitchell GL, Scheiman M, Cotter SA, Cooper J, et al. Validity and reliability of the revised convergence insufficiency symptom survey in children aged 9 to 18 years. Optometry Vision Sci. 2003;80(12):832-8.

275. Angeles-Han ST, Yeh S, McCracken C, Jenkins K, Stryker D, Myoung E, et al. Using the effects of youngsters' eyesight on quality of life questionnaire to measure visual outcomes in children with uveitis. Arthrit Care Res. 2015;67(11):1513-20.

276. Angeles-Han ST, Griffin KW, Harrison MJ, Lehman TJA, Leong T, Robb RR, et al. Development of a vision-related quality of life instrument for children ages 8-18 years for use in juvenile idiopathic arthritis-associated uveitis. Arthrit Care Res. 2011;63(9):1254-61.

277. Angeles-Han ST, Griffin KW, Lehman TJ, Rutledge JR, Lyman S, Nguyen JT, et al. The importance of visual function in the quality of life of children with uveitis. J AAPOS. 2010;14(2):163-8.

278. Hrisos S, Clarke M, Wright C. The emotional impact of amblyopia treatment in preschool children: Randomized controlled trial. Ophthalmology. 2004;111(8):1550-6.
279. Bian W, Wu Y, Liu Y, Wan J, Xu Y, Tan M. Development and initial validation of the child intermittent Exotropia Questionnaire among child strabismus patients. International Journal of Nursing Sciences. 2015;2(1):66-72.

280. Hatt SR, Leske DA, Yamada T, Bradley EA, Cole SR, Holmes JM. Development and initial validation of quality-of-life questionnaires for intermittent exotropia. Ophthalmology. 2010;117(1):163-8.

281. Leske DA, Holmes JM, Michele Melia B, Miller A, Jackson J, Suh D, et al. Evaluation of the intermittent exotropia questionnaire using rasch analysis. JAMA Ophthalmol. 2015;133(4):461-5.

282. Holmes JM, Leske DA, Cole SR, Chandler DL, Repka MX. A symptom survey and quality of life questionnaire for nasolacrimal duct obstruction in children. Ophthalmology. 2006;113(9):1675-80.

283. Jones-Jordan LA, Chitkara M, Coffey B, Jackson JM, Manny RE, Rah MJ, et al. A comparison of spectacle and contact lens wearing times in the ACHIEVE study. Clin Exp Optom. 2010;93(3):157-63.

284. Walline JJ, Gaume A, Jones LA, Rah MJ, Manny RE, Berntsen DA, et al. Benefits of contact lens wear for children and teens. Eye Contact Lens. 2007;33(6):317-21.

285. Lukman H, Choong Y, Martins S, Laws D. Perceived Psychosocial Questionnaire (PPQ): Developing a measure of psychosocial impact of amblyopia treatment. Invest Ophth Vis Sci. 2002;43(13):4693.

286. Choong Y, Lukman H, Martin S, Laws D. Childhood amblyopia treatment:
Psychosocial implications for patients and primary carers. Eye. 2004;18(4):369-75.
287. Sabri K, Knapp CM, Thompson JR, Gottlob I. The VF-14 and psychological impact of amblyopia and strabismus. Invest Ophth Vis Sci. 2006;47(10):4386-92.

288. Sacchetti M, Baiardini I, Lambiase A, Aronni S, Fassio O, Gramiccioni C, et al. Development and testing of the quality of life in children with vernal keratoconjunctivitis questionnaire. Am J Ophthalmol. 2007;144(4):557-63.

289. Declich S, Carter AO. Public health surveillance: Historical origins, methods and evaluation. B World Health Organ. 1994;72(2):285.

290. Champion JA. Epidemic disease in London. London: Centre for Metropolitan History; 1993.

291. NHS Choices. Your health and care records. Available from:

http://www.nhs.uk/NHSEngland/thenhs/records/healthrecords/Pages/overview.aspx [Accessed 4<sup>th</sup> April 2017].

292. Centres for Disease Control and Prevention. *Electronic Medical* 

Records/Electronic Health Records (EMRs/EHRs). Available from:

https://www.cdc.gov/nchs/fastats/electronic-medical-records.htm [Accessed 4<sup>th</sup> April 2017].

293. NHS Digital. *Hospital Episode Statistics.* Available from:

http://content.digital.nhs.uk/hes [Accessed 4<sup>th</sup> April 2017].

294. Kermack WO, McKendrick AG, McKinlay PL. Death-rates in Great Britain and Sweden some general regularities and their significance. Lancet. 1934;223(5770):698-703.

295. Office for National Statistics. *Mortality Statistics: Metadata*. Available from: http://webarchive.nationalarchives.gov.uk/20160105160709/http://www.ons.gov.uk/ons/guide-method/user-guidance/health-and-life-events/mortality-metadata.pdf [Accessed 24<sup>th</sup> March 2017].

296. Mallinson T. Why measurement matters for measuring patient vision outcomes. Optometry Vision Sci. 2007;84(8):675-82.

297. Campbell NR. Physics: the elements. Cambridge (UK): Cambridge University Press; 2013.

298. Stevens SS. On the theory of scales of measurement. Science. 1946;103(2684):677-80.

299. Novick MR. The axioms and principal results of classical test theory. J Math Psychol. 1966;3(1):1-18.

300. Birnbaum A. Some latent trait models and their use in inferring an examinee's ability. In Lord FM, Novick MR (Eds.), Statistical theories of mental test scores. Reading (MA): Addison-Wesley; 1968:pp. 397-472.

301. Rasch G. Some probabilistic models for intelligence and attainment tests. Chicago: University of Chicago Press; 1980.

302. Traub RE. Classical test theory in historical perspective. Educ Meas-Issues Pra. 1997;16(4):8-14.

303. Cattell R. The scientific use of factor analysis in behavioral and life sciences. New York: Plenum Press; 1978.

304. Comrey AL, Lee HB. A first course in factor analysis. New York: Psychology Press; 2009.

305. Thurstone LL. Multiple factor analysis. Chicago: University of Chicago Press; 1947.

306. Gorsuch RL. Factor analysis, 2nd ed. Hillsdale (NJ): Lawrence Erlbaum Associates; 1983.

307. Hambleton RK, Swaminathan H, Rogers HJ. Fundamentals of item response theory. California: Sage; 1991.

308. Sijtsma K. On the use, the misuse, and the very limited usefulness of Cronbach's alpha. Psychometrika. 2009;74(1):107.

309. Pesudovs K. Patient-centred measurement in ophthalmology – a paradigm shift. BMC Ophthalmol. 2006;6(1):25.

310. Cano S, Klassen A, Scott A, Thoma A, Feeny D, Pusic A. Health outcome and economic measurement in breast cancer surgery: Challenges and opportunities. Expert Rev Pharm Out. 2010;10(5):583-94.

311. Andrich D. Rating scales and Rasch measurement. Expert Rev Pharm Out. 2011;11(5):571-85.

312. Petrillo J, Cano SJ, McLeod LD, Coon CD. Using Classical Test Theory, Item Response Theory, and Rasch Measurement Theory to evaluate patient-reported outcome measures: A comparison of worked examples. Value Health. 2015;18(1):25-34.

313. Pesudovs K, Burr JM, Harley C, Elliott DB. The development, assessment, and selection of questionnaires. Optometry Vision Sci. 2007;84(8):663-74.

314. Andrich D. Controversy and the Rasch model: A characteristic of incompatible paradigms? Med Care. 2004;42(1):I-7.

315. Masters GN. A Rasch model for partial credit scoring. Psychometrika. 1982;47(2):149-74.

316. Linacre J. Sample size and item calibration stability. Rasch Measurement Transactions. 1994;7:28.

317. Linacre JM. Optimizing rating scale category effectiveness. Journal of Applied Measurement. 2002;3(1):85-106.

318. Guest G, Bunce A, Johnson L. How many interviews are enough? An experiment with data saturation and variability. Field Method. 2006;18(1):59-82.
319. Garrard J. Health sciences literature review made easy. Burlington (MA):Jones & Bartlett Publishers; 2013.

320. Chavda S, Hodge W, Si F, Diab K. Low-vision rehabilitation methods in children: A systematic review. Can J Ophthalmology. 2014;49(3):71-3.

321. Thomas R, Barker L, Rubin G, Dahlmann-Noor A. Assistive technology for children and young people with low vision. Cochrane Db Syst Rev. 2015:10.1002/14651858.CD011350.pub2.

322. Musch DC, Gardner TW. Diabetes and nonrefractive visual impairment: The young have it. JAMA-J Am Med Assoc. 2012;308(22):2403-4.

323. Wagner RS. Smartphones, video display terminals, and dry eye disease in children. J Pediat Ophth Strab. 2014;51(2):76.

324. Kim M. Quality of life in teenagers with strabismus. Patient Prefer Adher. 2016;10:2277-2278.

325. Heppe EC, Kef S, Schuengel C. Testing the effectiveness of a mentoring intervention to improve social participation of adolescents with visual impairments: Study protocol for a randomized controlled trial. Trials. 2015;16:506.

326. Atasavun Uysal S, Duger T. Visual perception training on social skills and activity performance in low-vision children. Scand J Occup Ther. 2012;19(1):33-41.
327. Bogdanici ST, Costin D, Bogdanici CM. Quality of life for amblyopic children and their parents. Revista medico-chirurgicala a Societatii de Medici si Naturalisti din Iasi. 2015;119(1):214-20.

328. Huurre T, Aro H. Psychosocial development among adolescents with visual impairment. Eur Child Adoles Psy. 1998;7(2):73-8.

329. Huurre TM, Komulainen EJ, Aro HM. Social support and self-esteem among adolescents with visual impairments. J Visual Impair Blin. 1999;93:26-37.

330. Jones-Jordan LA, Sinnott LT, Graham ND, Cotter SA, Kleinstein RN, Manny RE, et al. The contributions of near work and outdoor activity to the correlation between siblings in the Collaborative Longitudinal Evaluation of Ethnicity and Refractive Error (CLEERE) Study. Invest Ophth Vis Sci. 2014;55(10):6333-9.

331. Kef S. The personal networks and social supports of blind and visually impaired adolescents. J Visual Impair Blin. 1997;91(3):236-44.

332. Maca SM, Amirian A, Prause C, Gruber K, Mejdoubi L, Barisani-Asenbauer T. Understanding the impact of uveitis on health-related quality of life in adolescents. Acta Ophthalmol. 2013;91(3):219-24.

333. Menon V, Saha J, Tandon R, Mehta M, Khokhar S. Study of the psychosocial aspects of strabismus. J Pediat Ophth Strab. 2002;39(4):203-8.

334. Papadopoulos K, Metsiou K, Agaliotis I. Adaptive behavior of children and adolescents with visual impairments. Res Dev Disabil. 2011;32(3):1086-96.

335. Rosenblum LP. Perceptions of the impact of visual impairment on the lives of adolescents. J Visual Impair Blin. 2000;94(7):434-45.

336. Sacks SZ, Wolffe K. Lifestyles of adolescents with visual impairments: An ethnographic analysis. J Visual Impair Blin. 1998;92:7-17.

337. Salminen AL, Karhula ME. Young persons with visual impairment: Challenges of participation. Scand J Occup Ther. 2014;21(4):267-76.

338. Taras ME, Matson JL, Felps JN. Using independence training to teach independent living skills to children and young men with visual impairments. Behav Modif. 1993;17(2):189-208.

339. Uttermohlen TL. On 'passing' through adolescence. J Visual Impair Blin. 1997;91(3):309-14.

340. Wolffe K, Sacks SZ. The lifestyles of blind, low vision, and sighted youths: A quantitative comparison. J Visual Impair Blin. 1997;91(3):245-57.

341. Hess I. Students with visual impairments in Israel: Quality of life as a subjective experience. Int J Adolesc Med Health. 2011;23(3):257-62.

342. Kroksmark U, Nordell K. Adolescence: The age of opportunities and obstacles for students with low vision in Sweden. J Visual Impair Blin. 2001;95(4):213-25.

343. Pinquart M, Pfeiffer JP. Perceived social support in adolescents with and without visual impairment. Res Dev Disabil. 2013;34(11):4125-33.

344. Shin HS, Park SC, Park CM. Relationship between accommodative and vergence dysfunctions and academic achievement for primary school children. Ophthal Physl Opt. 2009;29(6):615-24.

345. Van Dijk J, Oostrom KJ, Huisman J, Moll AC, Cohen-Kettenis PT, Ringens PJ, et al. Restrictions in daily life after retinoblastoma from the perspective of the survivors. Pediatr Blood Cancer. 2010;54(1):110-5.

346. Chen CC, Lin SY. The impact of rope jumping exercise on physical fitness of visually impaired students. Res Dev Disabil. 2011;32(1):25-9.

347. Dursun OB, Erhan SE, Ibis EO, Esin IS, Keles S, Sirinkan A, et al. The effect of ice skating on psychological well-being and sleep quality of children with visual or hearing impairment. Disabil Rehabilitation. 2015;37(9):783-9.

348. Jones-Jordan LA, Mitchell GL, Cotter SA, Kleinstein RN, Manny RE, Mutti DO, et al. Visual activity before and after the onset of juvenile myopia. Invest Ophth Vis Sci. 2011;52(3):1841-50.

349. Lieberman LJ, Stuart ME, Hand K, Robinson B. An investigation of the motivational effects of talking pedometers among children with visual impairments and deaf-blindness. J Visual Impair Blin. 2006;100(12):726-36.

350. Baines C. Finding out what disabled children with communication impairments want. In: Burns C (Ed.), Disabled children living away from home in foster care and residential settings. London: Mac Keith Press; 2009:pp. 87-99.

351. Hatt SR, Leske DA, Adams WE, Kirgis PA, Bradley EA, Holmes JM. Quality of life in intermittent exotropia: Child and parent concerns. Arch Ophthalmol-Chic. 2008;126(11):1525-9.

352. Huurre T, Aro H. The psychosocial well-being of Finnish adolescents with visual impairments versus those with chronic conditions and those with no disabilities. J Visual Impair Blin. 2000;94(10):625-37.

353. Kef S. Psychosocial adjustment and the meaning of social support for visually impaired adolescents. J Visual Impair Blin. 2002;96(1):22-37.

354. Kef S, Bos H. Is love blind? Sexual behavior and psychological adjustment of adolescents with blindness. Sex Disabil. 2006;24(2):89-100.

355. Kef S, Deković M. The role of parental and peer support in adolescents wellbeing: A comparison of adolescents with and without a visual impairment. J Adolescence. 2004;27(4):453-66.

356. Kef S, Hox J, Habekothe H. Social networks of visually impaired and blind adolescents. Structure and effect on well-being. Soc Networks. 2000;22(1):73-91.
357. Metsiou K, Papadopoulos K, Agaliotis I. Adaptive behavior of primary school students with visual impairments: The impact of educational settings. Res Dev Disabil. 2011;32(6):2340-5.

358. Odedra N, Wedner SH, Shigongo ZS, Nyalali K, Gilbert C. Barriers to spectacle use in Tanzanian secondary school students. Ophthal Epidemiol. 2008;15(6):410-7.
359. Pinquart M, Pfeiffer JP. Bullying in German adolescents: Attending special school for students with visual impairment. Brit J Vis Impair. 2011;29(3):163-76.
360. Pinquart M, Pfeiffer JP. Psychological adjustment in adolescents with vision impairment. Int J Disabil Dev Ed. 2012;59(2):145-55.

361. Pinquart M, Pfeiffer JP. What is essential is invisible to the eye: Intimate relationships of adolescents with visual impairment. Sex Disabil. 2012;30(2):139-47. 362. Pinquart M, Pfeiffer JP. Worry in adolescents with visual impairment. Brit J Vis Impair. 2014;32(2):94-107.

363. Pinquart M, Pfeiffer JP. Change in psychological problems of adolescents with and without visual impairment. Eur Child Adoles Psy. 2014;23(7):571-8.

364. Wee R, Van Gelder RN. Sleep disturbances in young subjects with visual dysfunction. Ophthalmology. 2004;5(3):132-3.

365. Zylka J, Lach U, Rutkowska I. Functional balance assessment with pediatric balance scale in girls with visual impairment. Pediatr Phys Ther. 2013;25(4):460-6. 366. Bomba M, Meini A, Molinaro A, Cattalini M, Oggiano S, Fazzi E, et al. Body

experiences, emotional competence, and psychosocial functioning in juvenile idiopathic arthritis. Rheumatol Int. 2013;33(8):2045-52.

367. Garaigordobil M, Bernarás E. Self-concept, self-esteem, personality traits and psychopathological symptoms in adolescents with and without visual impairment. Span J Psychol. 2009;12(1):149-60.

368. Halder S, Datta P. An exploration into self concept: A comparative analysis between the adolescents who are sighted and blind in India. Brit J Vis Impair. 2012;30(1):31-41.

369. Pierce JW, Wardle J. Body Size, parental appraisal, and self-esteem in blind children. J Child Psychol Psyc. 1996;37(2):205-12.

370. Pinquart M, Pfeiffer JP. Psychological well-being in visually impaired and unimpaired individuals: A meta-analysis. Brit J Vis Impair. 2011;29(1):27-45.

371. Pinquart M, Pfeiffer JP. Identity development in German adolescents with and without visual impairments. J Visual Impair Blin. 2013;107(5):338.

372. Yamada T, Hatt SR, Leske DA, Holmes JM. Specific health-related quality of life concerns in children with intermittent exotropia. Strabismus. 2012;20(4):145-51.
373. Hennink M, Hutter I, Bailey A. Qualitative research methods. London: Sage;

2010.

374. Strauss A, Corbin J. Basics of qualitative research: Techniques and procedures for developing grounded theory. London: Sage; 1998.

375. Boeije H. A purposeful approach to the constant comparative method in the analysis of qualitative interviews. Quality Quant. 2002;36(4):391-409.

376. Glaser B, Strauss A. The discovery grounded theory: Strategies for qualitative inquiry. Chicago: Aldin; 1967.

377. Varni J. PedsQL pediatric quality of life inventory, version 4.0, teen report. San Diego, CA: Centre for Child Health Outcomes, Children's Hospital and Health Centre; 1998.

378. Sterne JA, White IR, Carlin JB, Spratt M, Royston P, Kenward MG, et al. Multiple imputation for missing data in epidemiological and clinical research: Potential and pitfalls. Brit Med J. 2009;338:10.1136/bmj.b2393.

379. Royston P. Multiple imputation of missing values: Update of ice. Stata J. 2005;5(4):527.

380. O'Connor BP. SPSS and SAS programs for determining the number of components using parallel analysis and Velicer's MAP test. Behav Res Methods. 2000;32(3):396-402.

381. Linacre J. WINSTEPS. Version 3.93.2. Perth, Australia: Winsteps; 2017.

382. Wright BD, Linacre JM, Gustafson J, Martin-Lof P. Reasonable mean-square fit values. Rasch Measurement Transactions. 1994;8(3):370.

383. Wright B, Masters G. Computation of OUTFIT and INFIT statistics. Rasch Measurement Transactions. 1990;3(4):84-5.

384. Khadka J, Pesudovs K, McAlinden C, Vogel M, Kernt M, Hirneiss C. Reengineering the glaucoma quality of life-15 questionnaire with rasch analysis. Invest Ophth Vis Sci. 2011;52(9):6971-7.

385. Gothwal VK, Wright TA, Lamoureux EL, Pesudovs K. Rasch analysis of visual function and quality of life questionnaires. Optometry Vision Sci. 2009;86(10):1160-8. 386. Dorans NJ, Holland PW. DIF detection and description: Mantel-Haenszel and standardization. In Holland PW, Wainer H (Eds.), Differential item functioning. Hillsadle (NJ): Lawrence Erlbaum Associates; 1993:pp. 35-66.

387. Royston P, Altman DG. Regression using fractional polynomials of continuous covariates: Parsimonious parametric modelling. Appl Stat-J Roy St C. 1994:429-67. 388. Varni JW. *Scaling and scoring of the Pediatric Quality of Life Inventory PedsQL.* Available from: http://www.pedsql.org/PedsQL-Scoring.pdf [Accessed 23<sup>rd</sup> August 2017].

389. Cronbach LJ, Meehl PE. Construct validity in psychological tests. Psychol Bull. 1955;52(4):281.

390. Snedecor GW, Cochran WG. Statistical Methods. 7<sup>th</sup> ed. Ames: Iowa State University Press; 1967.

391. Fisher RA. Statistical methods for research workers. 13<sup>th</sup> ed. New York: Hafner Publishing Co.; 1925.

Bider GH. The life course as developmental theory. Child Dev. 1998;69(1):1-12.
Bronfenbrenner U. Ecological models of human development. In: Husten T,
Poslethwaite TN (Eds.), International Encyclopedia of Education. 2<sup>nd</sup> ed. New York:
Elsevier; 1994:pp. 37-43.

394. Baltes PB. Theoretical propositions of life-span developmental psychology: On the dynamics between growth and decline. Dev Psychol. 1987;23(5):611.

395. Elder JR, Glen, H. Children of the Great Depression: Social Change in Life Experience.: Chicago: University of Chicago Press; 1974.

396. National Institute for Health and Care Excellence (NICE). Transition from children's to adults' services for young people using health or social care services. London: NICE; 2016.

397. Kennedy I. Getting it right for children and young people: overcoming cultural barriers in the NHS so as to meet their needs. London: Department of Health; 2010.
398. American Academy of Pediatrics; American Academy of Family Physicians; American College of Physicians-American Society of Internal Medicine. A consensus statement on health care transitions for young adults with special health care needs. Pediatrics. 2002;110(6):1304-6.

399. Robertson AO, Tadić V, Rahi JS. Transition from paediatric to adult ophthalmology services: what matters most to young people with visual impairment. Eye. 2017:10.1038/eye.2017.203.

400. Hayton JC, Allen DG, Scarpello V. Factor retention decisions in exploratory factor analysis: A tutorial on parallel analysis. Organ Res Methods. 2004;7(2):191-205.
401. Spearman C. Correlation calculated from faulty data. Brit J Psychol. 1910;3(3):271-95.

402. Linacre J. *Table 30.1 Differential item functioning DIF pairwise*. Available from: https://www.winsteps.com/winman/table30\_1.htm [Accessed 2<sup>nd</sup> August 2017].

403. Angeles-Han ST, McCracken C, Yeh S, Jenkins K, Stryker D, Rouster-Stevens K, et al. Characteristics of a cohort of children with juvenile idiopathic arthritis and JIA-associated uveitis. Pediatr Rheumatol. 2015:10.1186/s12969-015-0018-8.

404. Gothwal VK, Sumalini R, Irfan SM, Giridhar A. Rasch analysis of Impact of
Vision Impairment for Children questionnaire. Optometry Vision Sci. 2013;90(8):820-7.
405. Walline JJ, Jones LA, Chitkara M, Coffey B, Jackson JM, Manny RE, et al. The
Adolescent and Child Health Initiative to Encourage Vision Empowerment (ACHIEVE)
study design and baseline data. Optometry Vision Sci. 2006;83(1):37-45.

406. Wolfe EW, Chiu CW. Measuring pretest-posttest change with a Rasch rating scale model. J Outcome Meas. 1999;3:134-61.

407. Hatt SR, Leske DA, Holmes JM. Comparison of quality-of-life instruments in childhood intermittent exotropia. J AAPOS. 2010;14(3):221-6.

408. Nunnally JC. Psychometric theory. New York: McGraw-Hill; 1967.

409. Arrindell WA, Van der Ende J. An empirical test of the utility of the observationsto-variables ratio in factor and components analysis. App Psych Meas. 1985;9(2):165-78.

410. Barrett PT, Kline P. The observation to variable ratio in factor analysis. Personality study and Group Behavior. 1981;1(1):23-33.

411. Keil S. RNIB survey of VI services in England and Wales 2012: Report for England. London: Royal National Institute of Blind People (RNIB); 2012.

412. Solebo AL, Rahi JS. Epidemiology, aetiology and management of visual impairment in children. Arch Dis Child. 2014;99(4):375-9.

413. Tadić V, Hamblion EL, Keeley S, Cumberland P, Lewando Hundt G, Rahi JS. 'Silent voices' in health services research: Ethnicity and socioeconomic variation in participation in studies of quality of life in childhood visual disability. Invest Ophth Vis Sci. 2010;51(4):1886-90.

414. Rahi JS, Manaras I, Tuomainen H, Lewando Hundt G. Engaging families in health services research on childhood visual impairment: Barriers to, and degree and nature of bias in, participation. Brit J Ophthalmol. 2004;88(6):782-7.

415. Banteka M, Tailor V, Dahlmann-Noor A. Delivering clinical trials and observational studies in child eye health: A nationwide survey in the United Kingdom. J Pediat Ophth Strab. 2015;52(2):106-12.

416. Galea S, Tracy M. Participation rates in epidemiologic studies. Ann Epidemiol. 2007;17(9):643-53.

417. McCall RB. Nature-nurture and the two realms of development: A proposed integration with respect to mental development. Child Dev. 1981;52(1):1-12.

418. Duncan GJ, Brooks-Gunn J, Klebanov PK. Economic deprivation and early childhood development. Child Dev. 1994;65(2):296-318.

419. Leventhal T, Brooks-Gunn J. The neighborhoods they live in: The effects of neighborhood residence on child and adolescent outcomes. Psychol Bull. 2000;126(2):309.

420. McLoyd VC. Socioeconomic disadvantage and child development. Am Psychol. 1998;53(2):185.

421. Klerman LV. The health of poor children: Problems and programs. In: Huston AC (Ed.), Children in poverty: Child development and public policy. Cambridge (UK): Cambridge University Press; 1991:pp. 136-57.

422. Bradley RH, Corwyn RF. Socioeconomic status and child development. Annu Rev Psychol. 2002;53(1):371-99.

423. Majeed M, Williams C, Northstone K, Ben-Shlomo Y. Are there inequities in the utilisation of childhood eye-care services in relation to socio-economic status? Evidence from the ALSPAC cohort. Brit J Ophthalmol. 2008;92(7):965-9.

424. Knight A, Lindfield R. The relationship between socio-economic status and access to eye health services in the UK: A systematic review. Public Health. 2015;129(2):94-102.

425. Bandura A, McClelland DC. Social learning theory. Englewood Cliffs (NJ): Prentice-Hall; 1977.

426. Bandura A. The self system in reciprocal determinism. Am Psychol. 1978;33(4):344.

427. Guite JW, McCue RL, Sherker JL, Sherry DD, Rose JB. Relationships among pain, protective parental responses, and disability for adolescents with chronic musculoskeletal pain: The mediating role of pain catastrophizing. Clin J Pain. 2011;27(9):775-81.

428. Peterson CC, Palermo TM. Parental reinforcement of recurrent pain: The moderating impact of child depression and anxiety on functional disability. J Pediatr Psychol. 2004;29(5):331-41.

429. Langer SL, Romano JM, Levy RL, Walker LS, Whitehead WE. Catastrophizing and parental response to child symptom complaints. Child Health Care. 2009;38(3):169-84.

430. Claar RL, Simons LE, Logan DE. Parental response to children's pain: The moderating impact of children's emotional distress on symptoms and disability. Pain. 2008;138(1):172-9.

431. Welkom JS, Hwang W-T, Guite JW. Adolescent pain catastrophizing mediates the relationship between protective parental responses to pain and disability over time. J Pediatr Psychol. 2013;38(5):541-50.

432. Skär RL. Peer and adult relationships of adolescents with disabilities. J Adolescence. 2003;26(6):635-49.

433. Morris C. Measuring participation in childhood disability: How does the capability approach improve our understanding? Dev Med Child Neurol. 2009;51(2):92-4.

434. Prout A, James A. A new paradigm for the sociology of childhood? Provenance, promise and problems. In: James A, Prout A (Eds.), Constructing and reconstructing childhood: Contemporary issues in the sociological study of childhood. Oxon: Routledge; 1997:pp. 6-28.

435. Carlton J. Identifying potential themes for the child amblyopia treatment questionnaire. Optometry Vision Sci. 2013;90(8):867-73.

436. Farrant B, Watson P. Health care delivery: Perspectives of young people with chronic illness and their parents. J Paediatr Child Health. 2004;40(4):175-9.

437. Garwick AW, Kohrman C, Wolman C, Blum RW. Families' recommendations for improving services for children with chronic conditions. Arch Pediat Adol Med. 1998;152(5):440-8.

438. Olivieri I, M Bova S, Fazzi E, Ricci D, Tinelli F, Montomoli C, et al. Patientreported outcomes measure for children born preterm: Validation of the SOLE VLBWI Questionnaire, a new quality of life self-assessment tool. Dev Med Child Neurol. 2016;58(9):957-64.

439. Davis E, Mackinnon A, Davern M, Boyd R, Waters E, Graham HK, et al.
Description and psychometric properties of the CP QOL-Teen: A quality of life questionnaire for adolescents with cerebral palsy. Res Dev Disabil. 2013;34(1):344-52.
440. Marino BS, Tomlinson RS, Wernovsky G, Drotar D, Newburger JW, Mahony L, et al. Validation of the pediatric cardiac quality of life inventory. Pediatrics. 2010;126(3):498-508.

441. Kaushik A, Papachristou E, Dima D, Fewings S, Kostaki E, Ploubidis G, et al. Measuring stigma in children receiving mental health treatment: Validation of the Paediatric Self-Stigmatization Scale (PaedS). Euro Psychiat. 2017;43:1-8.

442. Boerner K, Wang SW. Targets for rehabilitation: An evidence base for adaptive coping with visual disability. Rehabil Psychol. 2012;57(4):320.

443. Weber JA, Wong KB. Older adults coping with vision loss. Home Health Care Serv Q. 2010;29(3):105-19.

444. Lee E-KO, Brennan M. 'I cannot see flowers but I can smell them' The relation of age and gender to self-reported coping strategies among older adults with visual impairment. Qual Soc Work. 2002;1(4):389-411.

445. Leyser Y, Heinze T. Perspectives of parents of children who are visually impaired: Implications for the field. RE:view. 2001;33(1):37.

446. Leyser Y, Heinze A, Kapperman G. Stress and adaptation in families of children with visual disabilities. Families in Society: The Journal of Contemporary Social Services. 1996;77(4):240-9.

447. Hancock K, Wilgosh L, McDonald L. Parenting a visually impaired child: The mother's perspective. J Vis Impair Blin. 1990;84(8):411-413.

448. Kirkham MA, Schilling RF, Norelius K, Schinke SP. Developing coping styles and social support networks: An intervention outcome study with mothers of handicapped children. Child Care Hlth Dev. 1986;12(5):313-23.

449. Jessup GM, Bundy AC, Cornell E. To be or to refuse to be? Exploring the concept of leisure as resistance for young people who are visually impaired. Leisure Stud. 2013;32(2):191-205.

450. Elder GH. Life trajectories in changing societies. In: Bandura A (Ed.), Selfefficacy in changing societies. Cambridge (UK): Cambridge University Press; 1995:pp. 46-68.

451. Bedny M, Pascual-Leone A, Dravida S, Saxe R. A sensitive period for language in the visual cortex: Distinct patterns of plasticity in congenitally versus late blind adults. Brain Lang. 2012;122(3):162-70.

452. Noppeney U. The effects of visual deprivation on functional and structural organization of the human brain. Neurosci Biobehav R. 2007;31(8):1169-80.

453. Tanti C, Stukas AA, Halloran MJ, Foddy M. Social identity change: Shifts in social identity during adolescence. J Adolescence. 2011;34(3):555-67.

454. Warda M. The family and chronic sorrow: Role theory approach. J Pediatr Nurs. 1992;7(3):205-10.

455. King GA, Baldwin PJ, Currie M, Evans J. Planning successful transitions from school to adult roles for youth with disabilities. Child Health Care. 2005;34(3):193-216. 456. Roth J, Brooks-Gunn J. What do adolescents need for healthy development? Implications for youth policy. Ann Arbor (MI): Society for Research in Child Development; 2000.

457. McDonagh JE, Kaufman M. Transition from pediatric to adult care after solid organ transplantation. Curr Opin Organ Tran. 2009;14(5):526-32.

458. Mackie AS, Islam S, Magill-Evans J, Rankin KN, Robert C, Schuh M, et al. Healthcare transition for youth with heart disease: A clinical trial. Heart. 2014:10.1136/heartjnl-2014-305748.

459. Huang JS, Terrones L, Tompane T, Dillon L, Pian M, Gottschalk M, et al. Preparing adolescents with chronic disease for transition to adult care: A technology program. Pediatrics. 2014;133(6):1639-46. 460. Breakey VR, Ignas DM, Warias AV, White M, Blanchette VS, Stinson JN. A pilot randomized control trial to evaluate the feasibility of an internet-based selfmanagement and transitional care program for youth with haemophilia. Haemophilia. 2014;20(6);784-93.

461. Korus M, Cruchley E, Stinson JN, Gold A, Anthony SJ. Usability testing of the internet program: "Teens Taking Charge: Managing My Transplant Online". Pediatr Transplant. 2015;19(1):107-17.

462. Viner R. Transition from paediatric to adult care. Bridging the gaps or passing the buck? Arch Dis Child. 1999;81(3):271-5.

463. Hospital Episode Statistics. *Hospital Outpatient Activity - 2014-15.* Available from: http://content.digital.nhs.uk/article/2021/Website-

Search?productid=19879&q=title%3a+%22hospital+outpatient+activity%22&sort=Most +recent&size=10&page=1&area=both#top [Accessed 16<sup>th</sup> November 2016].

464. Kirk J, Clayton P. Specialist services and transitional care in paediatric endocrinology in the UK and Ireland. Clin Endocrinol. 2006;65(1):59-63.

465. Viner RM. Do adolescent inpatient wards make a difference? Findings from a national young patient survey. Pediatrics. 2007;120(4):749-55.

466. Barfar E, Sharifi V, Amini H, Mottaghipour Y, Yunesian M, Tehranidoost M, et al. Cost-effectiveness analysis of an aftercare service vs. treatment-as-usual for patients with severe mental disorders. J Ment Health Policy. 2017;20(3):101-10.

467. Malmstrom M, Ivarsson B, Klefsgard R, Persson K, Jakobsson U, Johansson J. The effect of a nurse led telephone supportive care programme on patients' quality of life, received information and health care contacts after oesophageal cancer surgery-a six month RCT-follow-up study. Int J Nurs Stud. 2016;64:86-95.

468. Gibson PG, Yang IA, Upham JW, Reynolds PN, Hodge S, James AL, et al. Effect of azithromycin on asthma exacerbations and quality of life in adults with persistent uncontrolled asthma (AMAZES): A randomised, double-blind, placebo-controlled trial. Lancet. 2017;390(10095):659-68.

469. Devlin N, Herdman M, Pavesi M, Phung D, Naidoo S, Beer TM, et al. Healthrelated quality of life effects of enzalutamide in patients with metastatic castrationresistant prostate cancer: An in-depth post hoc analysis of EQ-5D data from the PREVAIL trial. Health Qual Life Out. 2017;15(130):10.1186/s12955-017-0704-y.

470. Luckett T, Butow P, King M. Improving patient outcomes through the routine use of patient-reported data in cancer clinics: Future directions. Psycho-Oncology. 2009;18(11):1129-38.

471. Boyer L, Lançon C, Baumstarck K, Parola N, Berbis J, Auquier P. Evaluating the impact of a quality of life assessment with feedback to clinicians in patients with schizophrenia: Randomised controlled trial. Brit J Psychiat. 2013;202(6):447-53.

472. Wolfe J, Orellana L, Cook EF, Ullrich C, Kang T, Geyer JR, et al. Improving the care of children with advanced cancer by using an electronic patient-reported feedback intervention: Results from the PediQUEST randomized controlled trial. J Clin Oncol. 2014;32(11):1119-26.

473. Janssens L, Gorter JW, Ketelaar M, Kramer WL, Holtslag HR. Health-related quality-of-life measures for long-term follow-up in children after major trauma. Qual Life Res. 2008;17(5):701-13.

474. Simeoni M-C, Schmidt S, Muehlan H, Debensason D, Bullinger M, Group D. Field testing of a European quality of life instrument for children and adolescents with chronic conditions: The 37-item DISABKIDS Chronic Generic Module. Qual Life Res. 2007;16(5):881-93.

475. Schmidt S, Debensason D, Mühlan H, Petersen C, Power M, Simeoni MC, et al. The DISABKIDS generic quality of life instrument showed cross-cultural validity. J Clin Epidemiol. 2006;59(6):587-98.

476. Robitail S, Simeoni M-C, Erhart M, Ravens-Sieberer U, Bruil J, Auquier P, et al. Validation of the European proxy KIDSCREEN-52 pilot test health-related quality of life questionnaire: First results. J Adolescent Health.

2006;39(4):10.1016/j.jadohealth.2006.01.009.

477. Ravens-Sieberer U, Gosch A, Rajmil L, Erhart M, Bruil J, Duer W, et al. KIDSCREEN-52 quality-of-life measure for children and adolescents. Expert Rev Pharm Out. 2005;5(3):353-64.

478. Varni JW, Seid M, Rode CA. The PedsQL<sup>™</sup>: Measurement model for the pediatric quality of life inventory. Med Care. 1999;37(2):126-39.

479. Petersson C, Simeonsson RJ, Enskar K, Huus K. Comparing children's selfreport instruments for health-related quality of life using the International Classification of Functioning, Disability and Health for Children and Youth (ICF-CY). Health Qual Life Out. 2013;11(1):75.

480. Morris C, Janssens A, Shilling V, Allard A, Fellowes A, Tomlinson R, et al. Meaningful health outcomes for paediatric neurodisability: Stakeholder prioritisation and appropriateness of patient reported outcome measures. Health Qual Life Out. 2015;13(1):87.

481. Morris C, Janssens A, Allard A, Thompson Coon J, Shilling V, Tomlinson R, et al. Informing the NHS Outcomes Framework: Evaluating meaningful health outcomes for children with neurodisability using multiple methods including systematic review, qualitative research, Delphi survey and consensus meeting. Health Serv Deliv Res. 2014;2(15):10.3310/hsdr02150.

482. Varni JW, Limbers CA, Burwinkle TM. Impaired health-related quality of life in children and adolescents with chronic conditions: A comparative analysis of 10 disease clusters and 33 disease categories/severities utilizing the PedsQL<sup>™</sup> 4.0 Generic Core Scales. Health Qual Life Out. 2007;5(1):43.

483. Varni JW, Seid M, Smith Knight T, Burwinkle T, Brown J, Szer IS. The PedsQL<sup>™</sup> in pediatric rheumatology: Reliability, validity, and responsiveness of the Pediatric Quality of Life Inventory<sup>™</sup> Generic Core Scales and Rheumatology Module. Arthritis Rheumatol. 2002;46(3):714-25.

484. The King's Fund. *How to deliver high-quality, patient-centred, cost-effective care.* Available from: https://www.kingsfund.org.uk/sites/default/files/how-to-deliver-high-quality-patient-centred-cost-effective-care-16-september-2010-kings-fund.pdf [Accessed 25<sup>th</sup> October 2017].

485. NHS England. *Five Year Forward View.* Available from:

https://www.england.nhs.uk/wp-content/uploads/2014/10/5yfv-web.pdf [Accessed 25<sup>th</sup> October 2017].

486. National Voices. *Person-centred care in 2017*. Available from: https://www.nationalvoices.org.uk/sites/default/files/public/publications/personcentred\_care\_in\_2017\_-\_national\_voices.pdf [Accessed 25<sup>th</sup> October 2017].

487. Licchetta M, & Stelmach, M. *Fiscal sustainability analytical paper: Fiscal sustainability and public spending on health.* Available from:

http://budgetresponsibility.org.uk/docs/dlm\_uploads/Health-FSAP.pdf [Accessed 25<sup>th</sup> October 2017].

488. Maguire D, Dunn, P., & McKenna, H. *How hospital activity in the NHS England has changed over time.* Available from:

https://www.kingsfund.org.uk/publications/hospital-activity-funding-changes [Accessed 25<sup>th</sup> October 2017].

489. Robertson R, Wenzel L, Thompson J, Charles A. *Understanding NHS financial pressures. How are they affecting patient care?* Available from:

https://www.kingsfund.org.uk/sites/default/files/field/field\_publication\_file/Understanding %20NHS%20financial%20pressures%20-%20full%20report.pdf [Accessed 25<sup>th</sup> October 2017].

490. Addicott R, Maguire D, Honeyman M, Jabbal J. *Workforce planning in the NHS.* Available from: https://www.kingsfund.org.uk/publications/workforce-planning-nhs [Accessed 25<sup>th</sup> October 2017].

491. NHS England. *Delayed Transfers of Care Data 2017-18*. Available from: https://www.england.nhs.uk/statistics/statistical-work-areas/delayed-transfers-of-care/delayed-transfers-of-care-data-2017-18/ [Accessed 25<sup>th</sup> October 2017].

492. NHS England. *Consultant-led Referral to Treatment Waiting Times*. Available from: https://www.england.nhs.uk/statistics/statistical-work-areas/rtt-waiting-times/ [Accessed 25<sup>th</sup> October 2017].

493. RNIB. Saving money, losing sight. RNIB campaign report executive summary November 2013. Available from: http://www.rnib.org.uk/campaigning-policy-and-reports-hub-eye-health/eye-health-reports [Accessed 25<sup>th</sup> October 2017].

494. NHS Hospital & Community Health Service (HCHS). *Provisional Statistics, calculate for November 2014 to November 2016.* Available from:

https://digital.nhs.uk/article/191/Find-data-and-publications [Accessed 25<sup>th</sup> October 2017].

495. NHS England. *NHS England published latests staff survey results*. Available from: https://www.england.nhs.uk/2017/03/staff-survey/ [Accessed 25<sup>th</sup> October 2017]. 496. Cole-Hamilton I, McBride S. Taking the time: Telling parents their child is blind or partially sighted. London, United Kingdom: Royal National Institute for the Blind; 1996.

497. Youngson-Reilly S, Tobin MJ, Fielder AR. Patterns of professional involvement with parents of visually impaired children. Dev Med Child Neurol. 1994;36(5):449-58. 498. Care Quality Commission. *From the pond into the sea. Children's transition to adult health services.* Available from:

https://www.cqc.org.uk/sites/default/files/CQC\_Transition%20Report.pdf [Accessed 6<sup>th</sup> August 2015].

499. Scal P, Ireland M. Addressing transition to adult health care for adolescents with special health care needs. Pediatrics. 2005;115(6):1607-12.

500. Snyder CF, Aaronson NK, Choucair AK, Elliott TE, Greenhalgh J, Halyard MY, et al. Implementing patient-reported outcomes assessment in clinical practice: A review of the options and considerations. Qual Life Res. 2012;21(8):1305-14.

501. Hibbard JH, Stockard J, Tusler M. Hospital performance reports: Impact on quality, market share, and reputation. Health Affair. 2005;24(4):1150-60.

502. Berwick DM, James B, Coye MJ. Connections between quality measurement and improvement. Med Care. 2003;41(1):i30-8.

503. Greenhalgh J DS, Dalkin S, Gooding K, Gibbons E, Wright J, Meads D, et al. Functionality and feedback: A realist synthesis of the collation, interpretation and utilisation of patient-reported outcome measures data to improve patient care. Health Serv Deliv Res. 2017;5(2):10.3310/hsdr05020.

504. Tadić V, Rahi JŠ. One size doesn't fit all: Time to revisit patient-reported outcome measures (PROMs) in paediatric ophthalmology? Eye. 2017;31:511-8. 505. Stasiak K, Parkin A, Seymour F, Lambie I, Crengle S, Pasene-Mizziebo E, et al.

Measuring outcome in child and adolescent mental health services: Consumers' views of measures. Clin Child Psychol P. 2013;18(4):519-35.

506. Wolpert M, Curtis-Tyler K, Edbrooke-Childs J. A qualitative exploration of patient and clinician views on patient reported outcome measures in child mental health and diabetes services. Adm Policy Ment Hlth. 2016;43(3):309-15.

507. NHS Choices. *Children and Young People.* Available from:

https://www.nhs.uk/conditions/consent-to-treatment/children/ [Accesssed 6<sup>th</sup> November 2017].

508. Gottman JM, Mettetal G. Speculations about social and affective development: Friendship and acquaintanceship through adolescence. In Gottman JM, Parker, JG (Eds.), Studies in emotion and social interaction. Conversations of friends: Speculations on affective development. New York: Cambridge University Press; 1986:pp. 192-237.

509. Way N, Greene ML. Trajectories of perceived friendship quality during adolescence: The patterns and contextual predictors. J Res Adolescence. 2006;16(2):293-320.

510. Hartup WW, Stevens N. Friendships and adaptation across the life span. Curr Dir Psychol Sci. 1999;8(3):76-9.

511. Callaly T, Hyland M, Coombs T, Trauer T. Routine outcome measurement in public mental health: Results of a clinician survey. Aust Health Rev. 2006;30(2):164-73.

512. Hughes R, Aspinal F, Addington-Hall J, Chidgey J, Drescher U, Dunckley M, et al. Professionals' views and experiences of using outcome measures in palliative care. Int J Palliat Nurs. 2003;9(6):234-8.

513. Dunckley M, Aspinal F, Addington-Hall J, Hughes R, Higginson I. A research study to identify facilitators and barriers to outcome measure implementation. Int J Palliat Nurs. 2005;11(5):218-25.

514. Hughes R, Aspinal F, Addington-Hall JM, Dunckley M, Faull C, Higginson I. It just didn't work: The realities of quality assessment in the English health care context. Int J Palliat Nurs. 2004;41(7):705-12.

515. Boyce MB, Browne JP, Greenhalgh J. The experiences of professionals with using information from patient-reported outcome measures to improve the quality of healthcare: A systematic review of qualitative research. BMJ Qual Saf. 2014:10.1136/bmjgs-2013-002524.

516. Velikova G, Booth L, Smith AB, Brown PM, Lynch P, Brown JM, et al.
Measuring quality of life in routine oncology practice improves communication and patient well-being: A randomized controlled trial. J Clin Oncol. 2004;22(4):714-24.
517. Coulter A. Patient engagement—what works? J Ambul Care Manage. 2012;35(2):80-9.

518. Kinnersley P, Edwards AG, Hood K, Cadbury N, Ryan R, Prout H, et al. Interventions before consultations for helping patients address their information needs. Cochrane Db Syst Rev. 2007:10.1002/14651858.CD004565.pub2.

519. Clement S, Ibrahim S, Crichton N, Wolf M, Rowlands G. Complex interventions to improve the health of people with limited literacy: A systematic review. Patient Educ Couns. 2009;75(3):340-51.

520. Haynes RB, Ackloo E, Sahota N, McDonald HP, Yao X. Interventions for enhancing medication adherence. Cochrane Db Syst Rev.

2008:10.1002/14651858.CD000011.pub3.

521. Jennett P, Hall LA, Hailey D, Ohinmaa A, Anderson C, Thomas R, et al. The socio-economic impact of telehealth: A systematic review. J Telemed Telecare. 2003;9(6):311-20.

522. Murray E, Burns J, See Tai S, Lai R, Nazareth I. Interactive health communication applications for people with chronic disease. Cochrane Db Syst Rev. 2005:10.1002/14651858.CD004274.pub4.

523. Gilbert A, Sebag-Montefiore D, Davidson S, Velikova G. Use of patient-reported outcomes to measure symptoms and health related quality of life in the clinic. Gynecol Oncol. 2015;136(3):429-39.

524. Grol R. Personal paper. Beliefs and evidence in changing clinical practice. Brit Med J. 1997;315(7105):418-21.

525. Santana MJ, Haverman L, Absolom K, Takeuchi E, Feeny D, Grootenhuis M, et al. Training clinicians in how to use patient-reported outcome measures in routine clinical practice. Qual Life Res. 2015;24(7):1707-18.

526. Santana M-J, Feeny DH. Using the Health Utilities Index in routine clinical care: Process, feasibility, and acceptability. Patient. 2009;2(3):159-67.

527. Coons SJ, Gwaltney CJ, Hays RD, Lundy JJ, Sloan JA, Revicki DA, et al. Recommendations on evidence needed to support measurement equivalence between electronic and paper-based patient-reported outcome (PRO) measures: ISPOR ePRO Good Research Practices Task Force report. Value Health. 2009;12(4):419-29.

528. Morris C, Gibbons, E., Fitzpatrick, R. Child and parent reported outcome measures: A scoping report focusing on feasibility for routine use in the NHS. A report to the Department of Health. Oxford: Patient-reported Outcomes Measurement Group, Dept. of Public Health, University of Oxford; 2009.

529. Jensen RE, Snyder CF, Abernethy AP, Basch E, Potosky AL, Roberts AC, et al. Review of electronic patient-reported outcomes systems used in cancer clinical care. J Oncol Pract. 2013;10(4):215-22.

530. Bliven BD, Kaufman SE, Spertus JA. Electronic collection of health-related quality of life data: Validity, time benefits, and patient preference. Qual Life Res. 2001;10(1):15-21.

531. Allenby A, Matthews J, Beresford J, McLachlan S. The application of computer touch-screen technology in screening for psychosocial distress in an ambulatory oncology setting. Eur J Cancer Care. 2002;11(4):245-53.

532. Kongsved SM, Basnov M, Holm-Christensen K, Hjollund NH. Response rate and completeness of questionnaires: A randomized study of internet versus paper-and-pencil versions. J Med Internet Res. 2007;9(3):25.

533. Basch E, Artz D, Dulko D, Scher K, Sabbatini P, Hensley M, et al. Patient online self-reporting of toxicity symptoms during chemotherapy. J Clin Oncol. 2005;23(15):3552-61.

534. Bilimoria KY, Cella D, Butt Z. Current challenges in using patient-reported outcomes for surgical care and performance measurement: everybody wants to hear from the patient, but are we ready to listen? JAMA Surg. 2014;149(6):505-6.

535. Snyder CF, Jensen R, Courtin SO, Wu AW. PatientViewpoint: A website for patient-reported outcomes assessment. Qual Life Res. 2009;18(7):793-800.

536. RNIB. Beginner's guide to assistive technology. Available from:

http://www.rnib.org.uk/information-everyday-living-using-technology-beginnersguides/beginners-guide-assistive-technology [Accessed 6th November 2017].

537. Mangunkusumo RT, Duisterhout JS, De Graaff N, Maarsingh EJ, De Koning HJ, Raat H. Internet versus paper mode of health and health behavior questionnaires in elementary schools: Asthma and fruit as examples. J School Health. 2006;76(2):80-6.

538. Mussaffi H, Omer R, Prais D, Mei-Zahav M, Weiss-Kasirer T, Botzer Z, et al. Computerised paediatric asthma quality of life questionnaires in routine care. Arch Dis Child. 2007;92(8):678-82.

539. Raat H, Mangunkusumo RT, Landgraf JM, Kloek G, Brug J. Feasibility, reliability, and validity of adolescent health status measurement by the Child Health Questionnaire Child Form (CHQ-CF): Internet administration compared with the standard paper version. Qual Life Res. 2007;16(4):675-85.

540. McCluggage B. A paperless NHS by 2018 is possible. Health Serv J. 2013;123(6338):18.

541. Waring N. To what extent are practices' paperless' and what are the constraints to them becoming more so? Br J Gen Pract. 2000;50(450):46-7.

542. Allan R. Paperless NHS hospital medical records. Clinical Med. 2010;10(4):315. 543. Scott J. Children as respondents: Methods for improving data quality. In: Lyberg L, Biemer P, Collins M, de Leeuw C, Dippo N, Trewin D (Eds.), Survey Measurement and Process Quality. New York: Wiley; 1997:pp. 331-50.

## Chapter 10 Appendices

## 10.1 Appendix I – External NHS Trusts included as patient

## identification centres (PICs) in phase 3 and phase 4

East Lancashire Hospitals NHS Trust University Hospital Southampton NHS Foundation Trust West Suffolk NHS Foundation Trust Royal Cornwall Hospitals NHS Trust Hampshire Hospitals NHS Foundation Trust Countess of Chester Hospital NHS Foundation Trust Hinchingbrooke Health Care NHS Trust Mid Cheshire Hospitals NHS Foundation Trust University Hospitals of North Midlands NHS Trust The Queen Elizabeth Hospital, Kings Lynn, NHS Foundation Trust **Royal Devon & Exeter NHS Foundation Trust** University Hospital of Wales, Cardiff and Vale University Health Board Birmingham Children's Hospital NHS Foundation Trust Epsom and St Helier University Hospitals NHS Trust Leeds Teaching Hospitals NHS Trust Portsmouth Hospitals NHS Trust Bristol Eye Hospital, University Hospitals Bristol NHS Foundation Trust Bradford Royal Infirmary, Bradford Teaching Hospitals NHS Foundation Trust Addenbrooke's Hospital, Cambridge University Hospitals NHS Trust Manchester Royal Eye Hospital, Central Manchester University Hospitals NHS Foundation Trust

# 10.2 Appendix II – GP letter

[Printed on UCL headed paper]

### Dear [GP name]

### Re:

**Research Project:** Functional Vision (FV) and Vision-related Quality of Life (VQoL) of children and young people with visual impairment: development of age-appropriate patient reported outcome measures (PROMs) for routine use in paediatric ophthalmology

### Patient: [patient name, date of birth, address]

Our research team at UCL Institute of Child Health are working on a project, funded by the Fight for Sight charity, to develop age-appropriate patient reported outcome measures of vision related quality of life and functional vision for children and young people who are visually impaired. The study has been approved by the National Research Ethics Service [REC reference number: 12-EE-0455].

We are writing to inform you that we are about to write to the family of your patient [*patient name*] who has previously been treated at [*hospital name*] to ask whether they would like to participate in the study. This will involve [to be amended as per below in accordance to the relevant study phase:

- Letter for phase 1: discussing with me whether and how his/her eye-sight affects different aspects of his/her life.
- Letter for phase 2: discussing with me whether the draft questions we have developed are meaningful and relevant and what we can do (e.g. in terms of format) to improve our draft questionnaires.
- Letter for phase 3: completing two self-report questionnaires, one asking questions about his/her functional vision and the other asking questions about his/her quality of life.
- Letter for phase 4: completing three self-report questionnaires, one asking questions about his/her functional vision and two asking questions about his/her quality of life.]

It would be very helpful if you would indicate on the attached reply slip whether we should or should not contact this family. We have enclosed a project information sheet for your reference but please let us know if you would like any further information regarding the study.

We are very grateful for your help with this study. Please do not hesitate to contact us if you have any questions or comments about it.

Yours sincerely,

Miss Alexandra Robertson (PhD student),

Dr Valerie Tadić (Research Associate),

Professor Jugnoo Rahi (Study Chief Investigator),

\_\_\_\_\_

## Reply Slip

### **Research Project:**

Functional Vision (FV) and Vision-related Quality of Life (VQoL) of children and young people with visual impairment: development of age-appropriate patient reported outcome measures (PROMs) for routine use in paediatric ophthalmology

Name of GP: [GP name]

Name of Child: [Patient name]

Please tick the box that applies:

I would contact this family.

Signature:	 	 
Date:	 	 

Yes

No

# 10.3 Appendix III - Parent information sheets

# 10.3.1 Parents of children aged 13-15 years

### Information for parents

### Study title: 'Children and young people's views on living with impaired sight'

We would like to ask you and your son/daughter [to be written as is genderappropriate and amended for the rest of the document] to take part in this study.

If you or your child wish to have this document in Braille or have it read to you or him/her by a researcher, please let us know using the contact details at the end of this letter.

### 1. What is the aim of the study?

The aim of the study is to find out more about the quality of life and visual ability of children and young people who have problems with their eyesight. We know that poor vision can affect children and young people in a number of ways as they develop and grow, especially at school or college. But we do not know what children and young people themselves think about how their eyesight and any treatment they may have for it, affects their day-to-day lives.

### 2. Why is the study being done?

We are developing a series of age-appropriate questionnaires that children and young people with visual impairment can fill in and which measure their quality of life (e.g. how they feel about their eyesight) and visual functioning (e.g. difficulty reading from the computer or with independent mobility). We hope that these questionnaires will be used in the future by those who provide medical, educational or social services for children and young people with visual loss and those who carry out research into the causes of visual loss. We hope that this will help to give children and young people a say in all these areas.

### 3. Why are we being asked to take part?

With the help of other children and young people we have already developed questionnaires of quality of life and visual functioning that are suitable for children and young people with visual impairment aged 10-15 years. We now wish to develop questionnaires that are suitable for younger children (aged 8-12 years) and also for older teenagers (aged 13-17 years). We are asking you to take part because we would like to include all children and young people with visual impairment who are in that age range and who have attended an Eye Department in the UK.

### 4. What will happen if we decide to take part?

This is a large study, which involves different phases, but we would **only** ask you and your son/daughter to take part in the first phase of the study. This would involve the following:

- a) If you and your son/daughter agree to take part, we would like you to complete the attached consent forms to let us know. You can use the freepost envelopes provided to post the forms back.
- b) We would then like to [to be amended as per below in accordance to the relevant study phase:
  - Letter for phase 2: ask your son/daughter to talk to a researcher about some questions that we have developed with the help of other children and young people with visual impairment. We will meet your son/daughter in your home at a time convenient for your family. We would ask your son/daughter to give us advice regarding the individual questions, for example, if they are important or easy to understand, and whether the instructions for completing them are clear. We would also show some of these to him/her on a laptop to ask for his/her opinion on the electronic version of our questionnaire. This will help us ensure our questionnaire covers all the important information and is child/young person-friendly. This will take about 1 hour.
  - Letter for phase 3: ask you and your son/daughter to complete 2 forms we have developed with the help of other children and young people with visual impairment. One asks about his/her quality of life and the other about his/her visual ability. We will also ask you to provide some feedback about the questionnaire. You and your son/daughter can complete these in print or electronically, using a Word version on a CD or online. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with the questions (e.g. most children and young people skipping one particular question will tell us that this is not a particularly relevant question for this age group). It will also help us obtain the different perspectives that affected children or young people and their parents may have on all this. This will take between 10 and 45 minutes.
  - Letter for phase 4: We would like to ask you and your son/daughter to complete 3 forms asking questions about his/her quality of life and his/ her visual ability. Two of these we have developed with the help of other children and young people with visual impairment. The third is a general questionnaire that asks questions that should be suitable for all children and young people of the same age as your son/daughter, whether or not they have a health condition. You and your son/daughter can complete these in print or electronically, using a Word version on a CD or online version. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with the questions. It will also help us obtain the different perspectives that affected children or young people and their parents may have on all this. This will take between 15 minutes and 1 hour.]
- c) We would also like you to complete the enclose short form (see the enclosed Family Background Questionnaire) which asks some questions

about you, your child and family so that we can learn more about your family background. This will take around 3 minutes. This also helps us to make sure that we are including as many different types of families in our study as possible so we can get as representative a point of view as possible.

We hope that taking part in our study should be a positive experience for your son/daughter allowing him/her to tell us in detail what it means for him/her to live with an eyesight problem. However, we recognise that there is the potential for you or your son/daughter to find some aspects of this difficult. We would like to reassure you that our study team is experienced and appropriately trained and able to deal sensitively with any difficulties that might arise. We also remind you that you are free to withdraw from the study at any point.

With your permission, we may also contact your family GP to inform them of any health or care related concerns that arise from your and your child's participation in this study.

### 5. Who will have access to my child's records?

Only the research team involved in this study and representatives from Regulatory Authorities or from the NHS Trust where it is relevant to you and your child taking part in this research would have access to your child's medical notes and the actual data collected during the study.

### 6. Who are the researchers involved in this study?

- Professor Jugnoo Rahi, Study Chief Investigator, Professor of Ophthalmic Epidemiology and Honorary Consultant Ophthalmologist, UCL Institute of Child Health and Great Ormond Street Hospital for Children NHS Foundation Trust
- Miss Alexandra Robertson, Primary Researcher, PhD Student, UCL Institute of Child Health
- Dr Val Tadić, Research Associate, UCL Institute of Child Health
- Mrs Phillippa Cumberland, Senior Research Associate in Biostatistics, UCL Institute of Child Health
- Professor Gillian Hundt, Professor of Social Sciences in Health, University of Warwick
- Dr Alki Liasis, Clinical Scientist and Head of Ophthalmology Department, Great Ormond Street Hospital for Children NHS Foundation Trust
- Ms Lisa Davies, Clinical Outcomes Lead, Great Ormond Street Hospital for Children NHS Foundation Trust
- Mrs Corie Brown, a parent of NHS user
- Professor Peng Khaw, the Director of the NIHR Biomedical Research Centre and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology
- Professor Anthony Moore, Duke Elder Professor of Ophthalmology and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology

### 7. What will happen with the findings of the study?

We plan to publish the findings of the study in scientific journals so that we can let other professionals know what we have learned from this study. However, we will pool information in our reports, so that **it will not be possible to identify any individual person who takes part. Everything you and your son/daughter tell us will be strictly confidential.** We will also write to you at a later date to let you know about the study findings.

### 8. What are the potential benefits?

This study is unlikely to bring any immediate benefits to your son/daughter, although many children and young people have found it a very positive experience to share their thoughts about their life with us. However, we hope that it will improve our understanding of children and young people's perspectives of their visual loss and our ability to measure their quality of life and vision in order to improve the treatment and services provided for them at this hospital and elsewhere in the country in the future.

## 9. Do I have to take part in this study?

If you or your son/daughter decide, now or at a later stage, that you do **not** wish to take part in this research project, that is entirely your right. This **will not** affect any present or future treatment of your son/daughter in any way.

## 10. Who do I speak to if problems arise?

If you have any questions or complaints about the way in which this study has been, or is being conducted, please, in the first instance, discuss them with the researchers named below. For independent advice you can contact the Patient Advice and Liaison Services (PALS) (http://www.pals.nhs.uk).

### 11. Details of how to contact the researchers:

If you have any questions or concerns that you would like to discuss before agreeing to participate in this study and signing the consent form you can contact any of the main researchers by letter, email or phone below.

Miss Alexandra Robertson (PhD Student),

Dr Valerie Tadić (Study Research Associate),

Professor Jugnoo Rahi (Study Chief Investigator),

# 10.3.2 Parents of YP aged 16-19 years

### Information for parents

### Study title: 'Children and young people's views on living with impaired sight'

We would like to ask you and your son/daughter [to be written as is genderappropriate and amended for the rest of the document] to take part in this study.

If you or your child wish to have this document in Braille or have it read to you or him/her by a researcher, please let us know using the contact details at the end of this letter.

### 1. What is the aim of the study?

The aim of the study is to find out more about the quality of life and visual ability of children and young people who have problems with their eyesight. We know that poor vision can affect children and young people in a number of ways as they develop and grow, especially at school or college. But we do not know what children and young people themselves think about how their eyesight and any treatment they may have for it, affects their day-to-day lives.

### 2. Why is the study being done?

We are developing a series of age-appropriate questionnaires that children and young people with visual impairment can fill in and which measure their quality of life (e.g. how they feel about their eyesight) and visual functioning (e.g. difficulty reading from the computer or with independent mobility). We hope that these questionnaires will be used in the future by those who provide medical, educational or social services for children and young people with visual loss and those who carry out research into the causes of visual loss. We hope that this will help to give children and young people a say in all these areas.

### 3. Why are we being asked to take part?

With the help of other children and young people we have already developed questionnaires of quality of life and visual functioning that are suitable for children and young people with visual impairment aged 10-15 years. We now wish to develop questionnaires that are suitable for younger children (aged 6-9 years) and also for older teenagers (aged 16-18 years). We are asking you to take part because we would like to include all children and young people with visual impairment who are in that age range and who have attended an Eye Department in the UK.

### 4. What will happen if we decide to take part?

This is a large study, which involves different phases, but we would **<u>only</u>** ask you and your son/daughter to take part in one phase of the study. This would involve the following:

a) If you and your son/daughter agree to take part, we would like you to complete the attached consent forms to let us know. You can use the freepost envelopes provided to post the forms back.

- b) We would then like to [to be amended as per below in accordance to the relevant study phase:
  - Letter for phase 1: ask your son/daughter to talk to a researcher about how his/her eyesight affects different aspects of his/her life. For this, we will meet your son/daughter in your home at a time convenient for your family. This could take between 30 min and 1 ½ hours. All the information we gather from these discussions will be pooled together anonymously. This will then be used to decide on the individual questions that will make up the questionnaires we aim to develop. We would also like to be able to quote anonymously comments you or your son/daughter make if they identify important issues in a particularly powerful way. In all our scientific papers or other reports about the study we will ensure that neither you nor your son/daughter can be identified.
  - Letter for phase 2: ask your son/daughter to talk to a researcher about some questions that we have developed with the help of other children and young people with visual impairment. We will meet your son/daughter in your home at a time convenient for your family. We would ask your son/daughter to give us advice regarding the individual questions, for example, if they are important or easy to understand, and whether the instructions for completing them are clear. This will help us ensure our questionnaire covers all the important information and is child/young person-friendly. This will take about 1 hour.
  - Letter for phase 3: ask you and your son/daughter to complete 2 forms we have developed with the help of other children and young people with visual impairment. One asks about his/her quality of life and the other about his/her visual ability. We will also ask you to provide some feedback about the questionnaire. You and your son/daughter can complete these in print or electronically, using a Word version on a CD or online. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with the questions (e.g. most children and young people skipping one particular question will tell us that this is not a particularly relevant question for this age group). It will also help us obtain the different perspectives that affected children or young people and their parents may have on all this. This will take between 10 and 45 minutes.
  - Letter for phase 4: We would like to ask you and your son/daughter to complete 3 forms asking questions about his/her quality of life and his/her visual ability. Two of these we have developed with the help of other children and young people with visual impairment. The third is a general questionnaire that asks questions that should be suitable for all children and young people of the same age as your son/daughter, whether or not they have a health condition. You and your son/daughter can complete these in print or electronically, using a Word version on a CD or online version. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with

the questions. It will also help us obtain the different perspectives that affected children or young people and their parents may have on all this. This will take between 15 minutes and 1 hour.]

c) We would also like you to complete the enclosed short form (see the enclosed Family Background Questionnaire) which asks some questions about you, your child and family so that we can learn more about your family background. This will take around 3 minutes. This also helps us to make sure that we are including as many different types of families in our study as possible so we can get as representative a point of view as possible.

We hope that taking part in our study should be a positive experience for your son/daughter allowing him/her to tell us in detail what it means for him/her to live with an eyesight problem. However, we recognise that there is the potential for you or your son/daughter to find some aspects of this difficult. We would like to reassure you that our study team is experienced and appropriately trained and able to deal sensitively with any difficulties that might arise. We also remind you that you are free to withdraw from the study at any point.

With your permission, we may also contact your family GP to inform them of any health or care related concerns that arise from your and your child's participation in this study.

### 5. Who will have access to my child's records?

Only the research team involved in this study and representatives from Regulatory Authorities or from the NHS Trust where it is relevant to you and your child taking part in this research would have access to your child's medical notes and the actual data collected during the study.

### 6. Who are the researchers involved in this study?

- Professor Jugnoo Rahi, Study Chief Investigator, Professor of Ophthalmic Epidemiology and Honorary Consultant Ophthalmologist, UCL Institute of Child Health and Great Ormond Street Hospital for Children NHS Foundation Trust
- Miss Alexandra Robertson, Primary Researcher, PhD Student, UCL Institute of Child Health
- Dr Val Tadić, Research Associate, UCL Institute of Child Health
- Mrs Phillippa Cumberland, Senior Research Associate in Biostatistics, UCL Institute of Child Health
- Professor Gillian Hundt, Professor of Social Sciences in Health, University of Warwick
- Dr Alki Liasis, Clinical Scientist and Head of Ophthalmology Department, Great Ormond Street Hospital for Children NHS Foundation Trust
- Ms Lisa Davies, Clinical Outcomes Lead, Great Ormond Street Hospital for Children NHS Foundation Trust
- Mrs Corie Brown, a parent of NHS user

- Professor Peng Khaw, the Director of the NIHR Biomedical Research Centre and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology
- Professor Anthony Moore, Duke Elder Professor of Ophthalmology and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology

## 7. What will happen with the findings of the study?

We plan to publish the findings of the study in scientific journals so that we can let other professionals know what we have learned from this study. However, we will pool information in our reports, so that **it will not be possible to identify any individual person who takes part. Everything you and your son/daughter tell us will be strictly confidential.** We will also write to you at a later date to let you know about the study findings.

### 8. What are the potential benefits?

This study is unlikely to bring any immediate benefits to your son/daughter, although many children and young people have found it a very positive experience to share their thoughts about their life with us. However, we hope that it will improve our understanding of children and young people's perspectives of their visual loss and our ability to measure their quality of life and vision in order to improve the treatment and services provided for them at this hospital and elsewhere in the country in the future.

### 9. Do I have to take part in this study?

If you or your son/daughter decide, now or at a later stage, that you do **not** wish to take part in this research project, that is entirely your right. This **will not** affect any present or future treatment of your son/daughter in any way.

### 10. Who do I speak to if problems arise?

If you have any questions or complaints about the way in which this study has been, or is being conducted, please, in the first instance, discuss them with the researchers named below. For independent advice you can contact the Patient Advice and Liaison Services (PALS) (http://www.pals.nhs.uk).

### 11. Details of how to contact the researchers:

If you have any questions or concerns that you would like to discuss before agreeing to participate in this study and signing the consent form you can contact any of the main researchers by letter, email or phone below.

Miss Alexandra Robertson (PhD Student),

Dr Valerie Tadić (Study Research Associate),

Professor Jugnoo Rahi (Study Chief Investigator),

# 10.4 Appendix IV – Parent invitation letter (for children aged 13-15

## years and YP aged 16-19 years)

\*NOTE: The starred sentence is for the letters to parents of young people aged 16-18 years only, as they will also receive an invitation letter.

### [Printed on UCL headed paper]

### Dear [name of parents],

We are writing to you as the parent of [*child/young person name*] who has attended the Ophthalmology Department of [*hospital name*]. We would like to ask you and [*child/young person name*] to take part in our study. \*We have also written to [*young person name*] so that you can discuss this as a family before you decide whether to take part.\*

If you or your child wish to have this document in Braille or have it read to you or him/her [amend as gender appropriate, also for the rest of the document] by a researcher, please let us know using the contact details at the end of this letter.

The purpose of the study is to try to find out more about the quality of life and visual ability of children and young people who have problems with their eyesight. To do this, we are developing questionnaires for children and young people with visual impairment, which they can fill in themselves and which ask questions about how their eyesight problems affect their daily life. We hope that these questionnaires will be used by people who work with children and young people with visual loss in hospitals, schools or colleges and social services, so that they can take the children's and young people's perspectives into account when providing care and services.

To help us develop these questionnaires we would like to [to be amended as per below in accordance to the relevant study phase:

- Letter for phase 1: talk to your son/daughter about his/her daily life so that we know what kind of questions we should include in these questionnaires.
- Letter for phase 2: ask your son/daughter what he/she thinks about the questionnaires that we have developed to make sure that they are easily understood, meaningful and that we have included the right kind of questions. This will help us improve our questionnaires and make sure that they are easy to complete and comprehensible to children and young people.
- Letter for phase 3: ask you and your son/daughter to complete 2 questionnaires that we have already developed with the help of other children and young people with visual impairment, and to let us know his/her thoughts on these. This will help us improve our questionnaires and make sure that they are easy to complete and comprehensible to children and young people.
- Letter for phase 4: ask you and your son/daughter to complete 2 draft questionnaires that we have already developed, as well as another general health questionnaire. This will help us make sure that our new questionnaires are working as they should and are user friendly.]

The information sheet enclosed with this letter offers a more detailed summary of our study and what is required of you if you decide to participate.

We would be grateful if you would discuss this with your son/daughter and let us know whether you would be willing to take part. You can let us know by completing and signing the enclosed consent form. If you decide <u>not</u> to take part, this will not affect your son/daughter's medical care in any way.

We can send you and your son/daughter all the documents in large font or electronically (via email or on a CD), if you or your son/daughter prefer this.

If we do not hear from you, we will phone you in the next couple of weeks to ensure that you have received this letter. In the meantime, if you need more information or you would like some help in filling in the forms or if you would prefer to have the study explained in another language, please feel free to phone or email us using our contact details below.

Thank you for considering taking part in our study.

Yours sincerely,

Miss Alexandra Robertson (PhD Student),

Dr Valerie Tadić (Research Associate),

Professor Jugnoo Rahi (Professor of Ophthalmic Epidemiology),

On behalf of the research group:

- Miss Alexandra Robertson, Primary Researcher, PhD Student, UCL Institute of Child Health
- Professor Jugnoo Rahi, Study Chief Investigator, Professor of Ophthalmic Epidemiology and Honorary Consultant Ophthalmologist, UCL Institute of Child Health and Great Ormond Street Hospital for Children NHS Foundation Trust
- Dr Valerie Tadić, Research Associate, UCL Institute of Child Health
- Mrs Phillippa Cumberland, Senior Research Associate in Biostatistics, UCL Institute of Child Health
- Professor Gillian Hundt, Professor of Social Sciences in Health, University of Warwick
- Dr Alki Liasis, Clinical Scientist and Head of Ophthalmology Department, Great Ormond Street Hospital for Children NHS Foundation Trust
- Ms Lisa Davies, Clinical Outcomes Lead, Great Ormond Street Hospital for Children NHS Foundation Trust
- Mrs Corie Brown, a parent of NHS user
- Professor Peng Khaw, the Director of the NIHR Biomedical Research Centre and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology
- Professor Anthony Moore, Duke Elder Professor of Ophthalmology and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology

# 10.5.1 Parents of children aged 13-15 years

[Printed on UCL headed paper]

## **Consent form for parents**

Study title: Children and young people's views on living with impaired sight

**Sponsor Protocol No**: NRES Committee East of England - Essex REC ref: 12-EE-0455

Investigator: Jugnoo Rahi Contact details: 2020 7905 2250 💻: j.rahi@ucl.ac.uk

Person study ID: [PID]

### Please circle YES or NO below so we know if you agree with these statements:

5	concerns that are identified from our participation in this study. I agree to take part in this study.	Yes	No
4	I agree to my family GP being informed of any health or care related	Yes	No
3	I understand that relevant sections of my child's medical notes and data collected during the study may be looked at by the researchers involved in this study and the representatives from Regulatory Authorities or from the NHS Trust where it is relevant to our participation in this research. I give permission for these individuals to have access to these records.	Yes	No
2	I understand that my participation and that of my child is voluntary and that we are free to withdraw from the study at any time, without giving any reason and without my child's medical care or legal rights being affected.	Yes	No
1	I confirm that I have read and understood the information sheet for the above study. I have had the opportunity to consider the information, ask questions and have had these questions answered satisfactorily.	Yes	No

Child's Name	Date	
Parent Name	Date	Signature
Name of Person taking consent	Date	Signature
### 10.5.2 Parents of YP aged 16-19 years

[Printed on UCL headed paper]

#### Consent form for parents

Study title: Children and young people's views on living with impaired sight

**Sponsor Protocol No**: NRES Committee East of England - Essex REC ref: 12-EE-0455

Investigator: Jugnoo Rahi Contact details: 2020 7905 2250 💻: j.rahi@ucl.ac.uk

#### Person study ID: [PID]

#### Please circle YES or NO below so we know if you agree with these statements:

1	I confirm that I have read and understood the information sheet for the above study. I have had the opportunity to consider the information, ask questions and have had these questions answered satisfactorily.	Yes	No
2	I understand that my participation and that of my child is voluntary and that we are free to withdraw from the study at any time, without giving any reason and without my child's medical care or legal rights being affected.	Yes	No
3	I understand that relevant sections of my child's medical notes and data collected during the study may be looked at by the researchers involved in this study and the representatives from Regulatory Authorities or from the NHS Trust where it is relevant to our participation in this research. I give permission for these individuals to have access to these records.	Yes	No
4	I agree to my family GP being informed of any health or care related concerns that are identified from our participation in this study.	Yes	No
5	I agree to take part in this study.	Yes	No

Child's Name	Date	
Parent Name	Date	Signature
Name of Person taking consent	Date	Signature

### 10.6.1 Children aged 13-15 years

### Information for children and young people

#### Study title: Children and young people's views on living with impaired sight

We would like to ask you to take part in this study.

We want to learn more about what it's like to be a child or a young person with eyesight problems. We want to make forms that children and young people like you can fill in and which ask questions about your daily life. Doctors and other people who look after you in hospital and in school can then use this to understand more about you and your life.

Once you have talked about the study with your parents, they will let us know if you are happy to take part in it.

We would like you to [to be amended as per below in accordance to the relevant study phase:

- Letter for phase 2: meet a member of our research team. She will show you 2 forms that we have made with the help of other children and young people like you. She will ask you to fill in these forms. She will also ask you what you think about them, for example, whether the questions are easy to understand or how you could change them so they are more child-friendly. She would come to talk to you about this in your house. This may take up to 1 hour. Your answers will help us to make these forms better and more child- and young person-friendly.
- Letter for phase 3: We would like you to fill in 2 forms that have been made with help of other young people your age. They ask questions about how your eyesight affects your everyday life and daily activities. You can either fill in the forms on printed paper or on the computer, and you can let you parents know which one would be easier for you to do. This may take up to 1 hour. Your answers will help us make these forms better and more young person-friendly.
- Letter for phase 4: fill in 3 forms that have been made with help of other children and young people your age. They ask questions about how your eyesight affects your everyday life and daily activities. You can either fill in the forms on printed paper or on the computer, and you can let your parents know which one would be easier for you to do. This may take up to 1 hour. Your answers will help us make these forms better and more child and young people friendly.]

We cannot promise that taking part in our study will help you now. But it will help us learn how we can give better care to children and young people like you in our hospital and other hospitals in the future. We will not talk about what you tell us with anyone who is not part of our study team.

If you decide that you do **not** want to help with this study, you can just tell your parents and it will not be a problem.

If you want to read this letter or the lists of questions in Braille or have them read to you by the researcher please let you parents know and they will call us to arrange this.

#### Information for young people

# Study title: 'Children and young people's views on living with impaired sight'

We would like to ask you to take part in this study.

If you wish to have this document in Braille or have it read to you by a researcher, please let us know using the contact details at the end of this information sheet.

#### 1. What is the aim of the study?

The aim of the study is to find out more about the quality of life and visual ability of children and young people who have problems with their eyesight. We especially want to learn what children and young people themselves think about their eye condition and how it might affect everyday life.

#### 2. Why is the study being done?

We are developing questionnaires that young people can fill in to tell us more about their quality of life (e.g. how they feel about their eyesight) and their vision (e.g. difficulty reading from the computer or completing homework). We want these questionnaires to be used in the future by professionals in hospitals, schools or social services who work with children and young people with visual impairment. This will make sure that they take children and young people's views into account when providing medical care.

#### 3. Why am I being asked to take part?

With the help of other young people we have already developed questionnaires that are suitable for children and young people with visual impairment aged 10-15 years. We now wish to develop questionnaires that are suitable for younger children (aged 6-9 years) and older teenagers (aged 16-18 years). This is why we are asking for your help.

#### 4. What will happen if I decide to take part?

Taking part would involve the following:

- a) First, we would like you to discuss the study with your parents and decide whether you wish to take part. If you <u>do</u> want to take part, we would like you to complete the attached consent forms and send them back to us using the prepaid return envelope we have given you.
- b) We would then like to [to be amended as per below in accordance with the relevant study phase:
  - Letter for phase 1: talk to you in more detail about how your eyesight affects different aspects of your life. To do this, we will visit your home at a time that suits you. This may take between half an hour and 1 and a half hours. We would need to audiorecord your conversation with the researcher. All the information

you give us will be added to what other young people with eyesight problems have told us. This will then be used to design questions for the questionnaire we are developing. We may use what you tell us word-for-word in our future reports but your name will never be used and if we do include anything you have said, we will make sure you remain anonymous.

- Letter for phase 2: ask you to talk to a researcher about some questions that we have developed with the help of other young people with visual impairment. For this, we will meet you in your home at a time that suits you. We would ask you to give us advice regarding the individual questions, for example, if they are important or easy to understand and if the instructions for completing them are clear. We would also show you some of these on a laptop to ask for your opinion on the electronic version of our questionnaires. This will take about 1 hour. This will help us ensure our questionnaire covers all the important information and is young person-friendly.
- Letter for phase 3: ask you to complete 2 forms that we have developed with the help of other young people with visual impairment. One asks about your quality of life and the other about your vision. We will also ask you to provide some feedback about the questionnaires. You can complete these in print or electronically, using a Word version on a CD. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with the questions. This will take between 10 and 45 min.
- Letter for phase 4: ask you to complete 3 forms asking questions about your quality of life and vision. Two of these we have developed with the help of other young people with visual impairment. The third is a general questionnaire that asks questions about your life and health. You can complete these in print or electronically, using a Word version on a CD or online version. This will help us ensure that our questionnaires are working as they should and highlight if there are any problems with the questions. This will take between 15 minutes and 1 hour.]

We hope that taking part in this study will provide a chance for you to talk about what it means for you to live with an eyesight problem. However, we understand if you find some of this a little difficult. We would like you to know that you can drop out of the study at any point if you feel you need to. This would not affect your medical care in any way.

With your permission, we may also contact your GP to let him/her know if any health or care related concerns arise from your participation in this study.

#### 5. Who will be allowed to look at my notes?

The only people allowed to look at your research and medical notes will be the research team who carry out this study. People from the hospital where you have received your eye treatment and those from the 'research ethics committee' may

also need to look at these notes as they have to make sure that the study is done safely and properly.

#### 6. Will taking part in this study make a difference to my life?

We cannot promise that the study will make a difference to you now, but we hope it will help us understand more how young people feel about their eyesight problems. This will help us improve how we treat other young people at this hospital and elsewhere in the country in the future.

#### 7. Do I have to take part in this study?

This is entirely up to you. If you decide now or later that you do not want to take part, this would not affect your medical care or your rights in any way.

#### 8. What will happen with the findings of the study?

We plan to write reports and scientific papers from this study so that we can let other professionals know what we have learned from young people like you. However, we will put all the information together in our reports in a way that **it will not be possible for anyone to recognise you or any other young person in this study. We will not share anything you say with anybody outside of our study group.** We will also write to you at a later date to let you know our findings.

#### 9. Who can I talk to if there is a problem?

If you have any questions or complaints about the study you can tell the researcher talking to you straight away. You can also talk about it to your parents and they can take it further with our research team.

#### 10. Details of how to contact the researchers:

If you have any questions or concerns that you would like to discuss before agreeing to participate in this study and signing the consent form you can contact any of the main researchers by letter, email or phone below.

Miss Alexandra Robertson, Primary Researcher (PhD Student),

Dr Valerie Tadić, Study Research Associate,

Professor Jugnoo Rahi, Study Chief Investigator,

### 10.7 Appendix VII – Invitation letters for YP aged 16-19 years

#### [Printed on UCL headed paper]

#### Dear [Name of young person],

We are writing to you as a patient who has attended the Eye Department of *[hospital name]*. We would like to ask you to take part in our study. We have also written about this to your parents so you can talk it over with them before you decide whether you want to take part.

If you wish to have this document in Braille or have it read to you by a researcher, please let us know using the contact details at the end of this letter.

The purpose of the study is to learn more about what life is like for young people who have problems with their eyesight. To do this, we are developing questionnaires, which young people with impaired vision can fill in to explain how their eyesight problems affect their daily life. We hope that these questionnaires will be used in hospitals, schools and social services, so that the views of young people are taken into account when providing care and services.

To help us develop these questionnaires, we would like to [to be amended as per below in accordance to the relevant study phase:

- Letter for phase 1: talk to you about your day-to-day life so that we know what kind of questions we should include in these questionnaires.
- Letter for phase 2: ask you what you think about the questionnaires that we have developed to make sure that they are easily understood, meaningful and that we have included the right kind of questions.
- Letter for phase 3: ask you to fill in two questionnaires that we have developed with the help of other young people with eyesight problems, and to let us know what you think about them. This will help us improve our questionnaires and make sure that they are easy to complete and understand.
- Letter for phase 4: ask you to fill in two draft questionnaires that we have developed, as well as another general health questionnaire. This will help us make sure that our new questionnaires are working as they should and that they are user-friendly.]

The information sheet enclosed with this letter explains the study in more detail and what we would ask of you if you decide to participate.

Please discuss this with your parents and let us know whether you would be willing to take part. You can let us know by completing and signing the enclosed consent form, which can be posted back to us in the prepaid envelope provided. We have asked your parents to do the same. If you decide <u>not</u> to take part, this will not affect your medical care in any way.

We can send you all the documents in large font or electronically (via email or on a CD), if you prefer this.

If we do not hear from you, we will phone you in the next couple of weeks to ensure that you have received this letter. If you need more information now or you would like some help in filling in the forms or if you would prefer to have the study explained in another language, please feel free to telephone or email us using our contact details below. Thank you for considering taking part in our study.

Yours sincerely,

Miss Alexandra Robertson (PhD Student),

Dr Valerie Tadić (Study Research Associate),

Professor Jugnoo Rahi (Study Chief Investigator),

On behalf of the research group:

- Miss Alexandra Robertson, Primary Researcher, PhD Student, UCL Institute of Child Health
- Professor Jugnoo Rahi, Study Chief Investigator, Professor of Ophthalmic Epidemiology and Honorary Consultant Ophthalmologist, UCL Institute of Child Health and Great Ormond Street Hospital for Children NHS Foundation Trust
- Dr Valerie Tadić, Research Associate, UCL Institute of Child Health
- Mrs Phillippa Cumberland, Senior Research Associate in Biostatistics, UCL Institute of Child Health
- Professor Gillian Hundt, Professor of Social Sciences in Health, University of Warwick
- Dr Alki Liasis, Clinical Scientist and Head of Ophthalmology Department, Great Ormond Street Hospital for Children NHS Foundation Trust
- Ms Lisa Davies, Clinical Outcomes Lead, Great Ormond Street Hospital for Children NHS Foundation Trust
- Mrs Corie Brown, a parent of NHS user
- Professor Peng Khaw, the Director of the NIHR Biomedical Research Centre and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology
- Professor Anthony Moore, Duke Elder Professor of Ophthalmology and Consultant Ophthalmologist, Moorfields Eye Hospital NHS Foundation Trust and UCL Institute of Ophthalmology

### 10.8 Appendix VIII - Participant consent forms

### 10.8.1 Children aged 13-15 years

#### [Printed on UCL headed paper]

#### **Child and Young Person Assent Form**

#### STUDY TITLE:

Children and young people's views on living with impaired sight

#### Please circle:

Have your parents or carers talked to you about this study?	YES / NO
Do you understand why you are being asked to take part?	YES / NO
Have you asked all the questions you want to?	YES / NO
Do you understand that it's OK to stop taking part at any time?	YES/NO
Are you happy to take part?	YES / NO

If you <u>do</u> want to take part please write your name below Your Name..... Date....

Researcher's name: Alexandra Robertson

Signature:					
------------	--	--	--	--	--

Date: .....

Thank you very much for your help!

#### [Printed on UCL headed paper]

#### Young Person Assent Form

Study title: Children and young people's views on living with impaired sight

**Sponsor Protocol No**: NRES Committee East of England - Essex REC ref: 12-EE-0455

Investigator: Jugnoo Rahi Contact details: 2020 7905 2250 💻: j.rahi@ucl.ac.uk

#### Person study ID: [PID]

Please circle YES or NO below so we know if you agree with these statements:

1	I confirm that I understand the information sheet for this study.	Yes	No
2	I have had the chance to consider the information, ask questions and have had these questions answered.	Yes	No
3	I understand that I am free to choose whether I take part in this study or not, that I am free to withdraw from this study at any time, without giving any reason, and that this will not affect my medical care or my rights.	Yes	No
4	I understand that some sections of my medical notes and information collected during the study may be looked at by researchers doing this study and also the people who are making sure that the study is done safely and properly. I am happy for these people to have access to these records.	Yes	No
5	I agree to my family GP being told of any health or care related concerns that are identified from my participation in this study.	Yes	No
6	I agree for my interview with the researcher to be audio-recorded and for anonymised direct quotes from these interviews to be used [for participants in phase 1 only]	Yes	No
7	I agree to take part in this study.	Yes	No

Name of person taking Date Signature	Signature

# 10.9 Appendix IX – Family Background Questionnaire

#### FAMILY BACKGROUND QUESTIONNAIRE FOR PARENTS

This form asks for background information about you and your family, which we hope you will be able to provide. This will help us check if a broad range of families have been included in the study.

All the information you provide will be treated in strict confidence.

ABOUT YOU:							
What is your relationship to your child with an eyesight problem?							
Natural Mother	Stepmother	Adoptive Mother	Foster Mother				
Natural Father	Stepfather	Adoptive Father	Foster Father				
Other guardian, p	lease specify:						
now old are you?							
Less than 20 year	s 21 to 30 years	31 to 40 years	41 to 50 years				
More than 51 year	rs						
3. Please tell us to v	vhich group(s) you fee	l you belong:					
White - English, S	Scottish, Welsh, Irish	White other, p	lease specify:				
Black Caribbean	Black African	Black other, ple	ease specify:				
🗌 Indian	Pakistani	Bangladeshi					
Chinese							
Mixed							
Asian other, plea	se specify:						
Other, please spe	ecify:						

4. Please describe your education or training:
No qualifications or training
A level
City and Guilds Certificate
□ NVQs
Degree
Professional qualification
Other, please specify:
5. Please describe your job:
(e.g. housewife, nurse, bus driver, engineer, accounting clerk, or tell us if you are unemployed)
<ol> <li>Is the house/flat where you live:</li> </ol>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> </ul>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> <li>Privately rented</li> </ul>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> <li>Privately rented</li> <li>Rented from a housing association</li> </ul>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> <li>Privately rented</li> <li>Rented from a housing association</li> <li>Rented from the local authority</li> </ul>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> <li>Privately rented</li> <li>Rented from a housing association</li> <li>Rented from the local authority</li> <li>Other, please specify:</li> </ul>
<ul> <li>6. Is the house/flat where you live:</li> <li>Owned by you</li> <li>Privately rented</li> <li>Rented from a housing association</li> <li>Rented from the local authority</li> <li>Other, please specify:</li> </ul>

□ None

🗌 One

Two or more

### ABOUT YOUR CHILD WITH AN EYESIGHT PROBLEM:

8. Ple	ase tell us to which	group(s) your child b	elongs to:				
🗌 W	hite - English, Scott	ish, Welsh, Irish	White other, please specify:				
🗌 BI	ack Caribbean	Black African	Black other, please specify:				
🗌 In	dian	Pakistani	Bangladeshi				
	hinese						
□ M	ixed						
🗌 As	sian other, please s	pecify:					
0	ther, please specify	:					
9. Do	es your child have a	any <u>other</u> problems a	ffecting development?  Yes No				
If YES	S, please tick any bo	oxes which apply.					
	Movement disorde	er (e.g. cerebral palsy	/, spina bifida, muscular dystrophy)				
	Communication disorder (e.g. autism)						
	Language disorder (e.g. speech not understandable by others)						
	Behaviour disorder						
	Developmental delay						
	Epilepsy or seizur	e disorder					
	Hearing impairment						
	Eating disorder						
	Other, please spe	cify:					
10. W	hat type of school/o	ollege/university doe	s your child attend?				
	Mainstream						
	Specialist VI scho	ol/college/university					
	Community specia	alist school/college/ur	niversity				
	Home educated						
	Other, please spe	cify					

#### ABOUT YOUR FAMILY:

11. How would you describe your fa	amily struct	ture?				
Two-parent family						
Single parent family						
Other, please specify:						
12. Does your child with eyesight p	roblem hav	ve any siblings? 🗌 Yes	No			
(please provide the details of their g	ender and	ages, e.g. brother aged 12	2)			
13. Do any of your CHILD'S SIBLIN	VGS have:					
an eyesight problem?	🗌 No	Yes, please specify:				
any other serious medical problems	;? 🗌 No	Yes, please specify:				
14. Do any of your CHILD'S PAREN	ITS have:					
an eyesight problem?	🗌 No	Yes, please specify:				
any other serious medical problems	? 🗌 No	Yes, please specify:				
15. PLEASE LET US KNOW IF YOU HAVE ANY COMMENTS						

THANK YOU VERY MUCH FOR FILLING IN THIS FORM

### 10.10 Appendix X – Systematic review search terms

((((("vision disorders"[MeSH Terms] OR ("vision"[All Fields] AND "disorders"[All Fields]) OR "vision disorders"[All Fields] OR ("visual"[All Fields] AND "impairment"[All Fields]) OR "visual impairment"[All Fields])) OR ("eye diseases"[MeSH Terms] OR ("eye"[All Fields] AND "diseases"[All Fields]) OR "eye diseases"[All Fields] OR ("eye"[All Fields] AND "disease"[All Fields]) OR "eye disease"[All Fields])))

(((("adolescent"[MeSH Terms] OR "adolescent"[All Fields] OR "teen"[All Fields])) OR (young[All Fields] AND ("persons"[MeSH Terms] OR "persons"[All Fields] OR "people"[All Fields]))) OR ("young adult"[MeSH Terms] OR ("young"[All Fields] AND "adult"[All Fields]) OR "young adult"[All Fields])))

((((("quality of life"[MeSH Terms] OR ("quality"[All Fields] AND "life"[All Fields]) OR "quality of life"[All Fields])) OR ("activities of daily living"[MeSH Terms] OR ("activities"[All Fields] AND "daily"[All Fields] AND "living"[All Fields]) OR "activities of daily living"[All Fields])) OR "gratification"[All Fields]) OR ("personal satisfaction"[MeSH Terms] OR ("personal"[All Fields] AND "satisfaction"[All Fields]) OR "personal satisfaction"[All Fields] OR "satisfaction"[All Fields]))

### 10.11 Appendix XI – Feasibility questions and probes used in phase

### 2 (expert consultations)

#### Feedback on the instruction page:

1. The instructions for this questionnaire are...

#### [Researcher reads instructions aloud]

- 2. When we go through these questions, I'd like to remind you that there are no right or wrong answers and we want to know how true each statement is for you.
- 3. Do you think these instructions are clear?
- 4. Is there anything you would do to make the instructions more clear?

#### Feedback on the response options:

5. The response options are...

[Researcher reads response options aloud]

- 6. What do you think about the response choices?
- 7. How would you make the response options easier to understand?

#### Feedback on individual items:

[Researcher reads aloud Item 1]

- 8. What answer would you give to this question?
- 9. Why would you choose that answer?
- 10. OK, so you would pick [researcher reads aloud the chosen response option]? Why not this answer [researcher reads aloud an alternative response option] instead?
- 11. Can you give me an example of why you chose that answer?
- 12. What does this question mean to you?
- 13. Was this question easy to understand/Were there any specific words that are difficult to understand?
- 14. Was this question easy to answer? If not, why not?

[Researcher repeats steps 8-14 with all subsequent items. When all items have been answered, researcher repeats steps 1-14 with all items in the alternative questionnaire]

#### Feedback on the overall questionnaire:

15. What did you think about that questionnaire?

- 16. Is there anything you would change about the questionnaire?
- 17. If we gave it to you when you came to hospital, how would you feel about filling it in?
- 18. Would it be best if we gave it to you on paper? [*If not, researcher probes alternative methods of administration*]
- 19. Are there any things that we forgot to ask you about that you think are important in your life?

### 10.12 Appendix XII - Feasibility questions included in phase 3

### questionnaire booklets (presented at the end of each instrument)

We would also like you to answer some short questions about this form. This will help us understand how easy or difficult you found completing it so that we make it easier for young people to fill in.

How long did it take you to complete this form?

.....

#### How easy was it?

- Very easy
- Easy
- Difficult
- Very difficult

How easy did you find the instructions?

- Very easy
- Easy
- Difficult
- Very difficult

10.13 Appendix XIII – Paediatric Quality of Life Inventory Version

4.0 – UK English, Child Report (ages 8-12)<sup>©</sup>

On the following pages is a list of things that might be a problem for you.

Please tell us **how much of a problem** each one has been for you during the **<u>past</u>** <u>**ONE month**</u> by circling:

0 if it is never a problem

1 if it is almost never a problem

2 if it is **sometimes** a problem

3 if it is often a problem

4 if it is **almost always** a problem

There are no right or wrong answers.

If you do not understand a question, please ask for help.

In the **past month**, how much of a problem has this been for you...

ABOUT MY HEALTH AND ACTIVITIES (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. It is hard for me to walk more than					
a couple of streets (about 100 metres)	0	1	2	3	4
2. It is hard for me to run	0	1	2	3	4
3. It is hard for me to do sports					
activities or exercise	0	1	2	3	4
4. It is hard for me to lift heavy things	0	1	2	3	4
5. It is hard for me to have a bath or shower by myself	0	1	2	3	4
6. It is hard for me to do chores around the house	0	1	2	3	4
7. I have aches and pains	0	1	2	3	4
8. I feel tired	0	1	2	3	4

ABOUT MY FEELINGS (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. I feel afraid or scared	0	1	2	3	4
2. I feel sad	0	1	2	3	4
3. I feel angry	0	1	2	3	4
4. I have trouble sleeping	0	1	2	3	4
5. I worry about what will happen to me	0	1	2	3	4

HOW I GET ON WITH OTHERS Never Almost Some- Often Almost
--

(problems with)		Never	times		Always
<ol> <li>I have trouble getting on with other children</li> </ol>	0	1	2	3	4
<ol><li>Other children do not want to be my friend</li></ol>	0	1	2	3	4
3. Other children tease me	0	1	2	3	4
4. I cannot do things that other children my age can do	0	1	2	3	4
5. It is hard to keep up when I play with other children	0	1	2	3	4

ABOUT SCHOOL (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. It is hard to pay attention in class	0	1	2	3	4
2. I forget things	0	1	2	3	4
3. I have trouble keeping up with my schoolwork	0	1	2	3	4
4. I miss school because of not feeling well	0	1	2	3	4
5. I miss school to go to the doctor or hospital	0	1	2	3	4

Copyright  $\ensuremath{\mathbb{C}}$  1998-2018 James W. Varni, Ph.D. All rights reserved.

10.14 Appendix XIV – Paediatric Quality of Life Inventory Version

# 4.0 – UK English, Teenager Report (ages 13-17)<sup>©</sup>

On the following pages is a list of things that might be a problem for you.

Please tell us **how much of a problem** each one has been for you during the <u>past</u> <u>ONE month</u> by circling:

0 if it is never a problem

1 if it is almost never a problem

2 if it is **sometimes** a problem

3 if it is often a problem

4 if it is almost always a problem

There are no right or wrong answers.

If you do not understand a question, please ask for help.

In the **past month**, how much of a problem has this been for you...

ABOUT MY HEALTH AND ACTIVITIES (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. It is hard for me to walk more than					
a couple of streets (about 100 metres)	0	1	2	3	4
2. It is hard for me to run	0	1	2	3	4
3. It is hard for me to do sports					
activities or exercise	0	1	2	3	4
4. It is hard for me to lift heavy things	0	1	2	3	4
5. It is hard for me to have a bath or shower by myself	0	1	2	3	4
6. It is hard for me to do chores	0	1	2	2	1
around the house	0	I	2	5	
7. I have aches and pains	0	1	2	3	4
8. I feel tired	0	1	2	3	4

ABOUT MY FEELINGS (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. I feel afraid or scared	0	1	2	3	4
2. I feel sad	0	1	2	3	4
3. I feel angry	0	1	2	3	4
4. I have trouble sleeping	0	1	2	3	4
5. I worry about what will happen to me	0	1	2	3	4

HOW I GET ON WITH OTHERS	Never	Almost	Some-	Often	Almost
(problems with)		Never	times		Always
1. I have trouble getting on with other	0	1	2	3	4

teenagers					
2. Other teenagers do not want to be my friend	0	1	2	3	4
3. Other teenagers tease me	0	1	2	3	4
4. I cannot do things that other teenagers my age can do	0	1	2	3	4
5. It is hard to keep up with other teenagers my age	0	1	2	3	4

ABOUT SCHOOL/COLLEGE (problems with)	Never	Almost Never	Some- times	Often	Almost Always
1. It is hard to pay attention in class	0	1	2	3	4
2. I forget things	0	1	2	3	4
3. I have trouble keeping up with my school/college work	0	1	2	3	4
4. I miss school/college because of not feeling well	0	1	2	3	4
<ol> <li>I miss school/college to go to the doctor or hospital</li> </ol>	0	1	2	3	4

Copyright © 1998-2018 James W. Varni, Ph.D. All rights reserved.

### 10.15 Appendix XV – Questionnaire assessing participants'

### preferred method of administration (presented at the end of phase 4

### questionnaire booklets)

How easy did you find it to complete the booklet?

- 1. Very easy
- 2. Easy
- 3. A bit difficult
- 4. Very difficult or impossible

If you were to do this again, how would you prefer to complete the forms?

- 1. Using pen and paper
- 2. Using a computer, laptop, tablet or mobile phone

When you are reading or using a computer, do you use any assistive technologies to help you?

	YES/NO
Which assistive technologies do you use?	
Screen reader e.g. JAWS	YES/NO
Screen magnifier	YES/NO
Speech software	YES/NO
Any other assistive devices?	
When do you use assistive technologies?	
At home?	YES/NO
At school?	YES/NO
Any other times?	

Score	Measure	SE	Score	Measure	SE	Score	Measure	SE
0	.00	16.34	25	43.49	2.26	50	57.23	2.36
1	10.84	8.99	26	44.05	2.24	51	57.86	2.39
2	17.15	6.40	27	44.61	2.23	52	58.51	2.42
3	20.89	5.26	28	45.16	2.21	53	59.17	2.46
4	23.58	4.59	29	45.71	2.20	54	59.86	2.50
5	25.71	4.14	30	46.25	2.19	55	60.58	2.55
6	27.47	3.81	31	46.79	2.18	56	61.32	2.60
7	28.98	3.55	32	47.32	2.18	57	62.09	2.66
8	30.31	3.35	33	47.85	2.17	58	62.91	2.73
9	31.51	3.19	34	48.38	2.17	59	63.77	2.81
10	32.60	3.05	35	48.90	2.17	60	64.68	2.90
11	33.60	2.94	36	49.43	2.17	61	65.65	3.00
12	34.54	2.84	37	49.96	2.17	62	66.70	3.12
13	35.41	2.75	38	50.49	2.18	63	67.83	3.26
14	36.23	2.68	39	51.02	2.18	64	69.08	3.42
15	37.02	2.61	40	51.55	2.19	65	70.47	3.63
16	37.77	2.56	41	52.09	2.19	66	72.04	3.88
17	38.48	2.51	42	52.63	2.20	67	73.87	4.21
18	39.17	2.46	43	53.18	2.22	68	76.07	4.67
19	39.84	2.42	44	53.73	2.23	69	78.85	5.34
20	40.49	2.39	45	54.29	2.24	70	82.68	6.47
21	41.11	2.35	46	54.86	2.26	71	89.09	9.04
22	41.73	2.33	47	55.43	2.28	72	100.00	16.37
23	42.33	2.30	48	56.02	2.30			
24	42.91	2.28	49	56.62	2.33			

10.16 Appendix XVI – Child VQoL scaled score-to-measure table

Score	Measure	SE	Score	Measure	SE	Score	Measure	SE
0	.00	16.17	27	43.52	2.13	54	56.93	2.23
1	10.71	8.89	28	44.03	2.12	55	57.50	2.26
2	16.94	6.32	29	44.54	2.11	56	58.09	2.29
3	20.63	5.19	30	45.04	2.09	57	58.69	2.32
4	23.28	4.53	31	45.53	2.08	58	59.30	2.36
5	25.36	4.08	32	46.02	2.07	59	59.94	2.40
6	27.09	3.75	33	46.50	2.07	60	60.60	2.44
7	28.57	3.49	34	46.99	2.06	61	61.29	2.49
8	29.87	3.29	35	47.47	2.06	62	62.01	2.54
9	31.03	3.13	36	47.94	2.05	63	62.76	2.61
10	32.09	2.99	37	48.42	2.05	64	63.54	2.67
11	33.06	2.87	38	48.90	2.05	65	64.38	2.75
12	33.96	2.77	39	49.37	2.05	66	65.26	2.84
13	34.80	2.69	40	49.85	2.05	67	66.21	2.94
14	35.60	2.61	41	50.32	2.06	68	67.23	3.06
15	36.35	2.55	42	50.80	2.06	69	68.33	3.20
16	37.06	2.49	43	51.28	2.07	70	69.55	3.37
17	37.75	2.43	44	51.77	2.07	71	70.91	3.57
18	38.41	2.39	45	52.26	2.08	72	72.45	3.82
19	39.04	2.35	46	52.75	2.09	73	74.24	4.15
20	39.65	2.31	47	53.25	2.10	74	76.40	4.60
21	40.25	2.27	48	53.75	2.11	75	79.13	5.26
22	40.82	2.24	49	54.26	2.13	76	82.90	6.38
23	41.39	2.22	50	54.77	2.15	77	89.22	8.94
24	41.94	2.19	51	55.30	2.16	78	100.00	16.20
25	42.47	2.17	52	55.83	2.18			
26	43.00	2.15	53	56.38	2.21			

10.17 Appendix XVII – YP VQoL scaled score-to-measure table

Score	Measure	SE	Score	Measure	SE	Score	Measure	SE
0	.00	13.49	29	43.11	2.06	58	59.12	2.11
1	9.10	7.52	30	43.69	2.04	59	59.82	2.12
2	14.56	5.45	31	44.25	2.03	60	60.44	2.14
3	17.91	4.54	32	44.82	2.02	61	61.07	2.16
4	20.39	4.02	33	45.37	2.02	62	61.72	2.18
5	22.39	3.66	34	45.92	2.01	63	62.38	2.21
6	24.09	3.40	35	46.47	2.00	64	63.05	2.24
7	25.57	3.20	36	47.02	2.00	65	63.74	2.27
8	26.90	3.04	37	47.56	1.99	66	64.45	2.30
9	28.11	2.91	38	48.10	1.99	67	65.18	2.33
10	29.22	2.80	39	48.64	1.99	68	65.94	2.38
11	30.25	2.71	40	49.18	1.99	69	66.72	2.42
12	31.22	2.62	41	49.72	1.99	70	67.54	2.47
13	32.13	2.55	42	50.26	1.99	71	68.39	2.53
14	33.00	2.49	43	50.79	1.99	72	69.29	2.60
15	33.83	2.44	44	51.33	1.99	73	70.24	2.67
16	34.62	2.39	45	51.87	1.99	74	71.24	2.76
17	35.38	2.34	46	52.41	1.99	75	72.33	2.87
18	36.12	2.30	47	52.96	2.00	76	73.50	3.00
19	36.83	2.27	48	53.50	2.00	77	74.79	3.16
20	37.52	2.24	49	54.05	2.01	78	76.23	3.35
21	38.20	2.21	50	54.60	2.02	79	77.88	3.61
22	38.86	2.18	51	55.16	2.02	80	79.83	3.97
23	39.50	2.16	52	55.72	2.03	81	82.25	4.50
24	40.13	2.14	53	56.28	2.04	82	85.55	5.40
25	40.74	2.12	54	56.86	2.05	83	90.95	7.49
26	41.35	2.10	55	57.43	2.06	84	100.00	13.47
27	41.95	2.08	56	58.02	2.08			
28	42.53	2.07	57	58.61	2.09			

10.18 Appendix XVIII – Child FV scaled score-to-measure table

Score	Measure	SE	Score	Measure	SE	Score	Measure	SE
0	.00	13.35	31	43.17	1.97	62	59.07	2.01
1	8.99	7.44	32	43.70	1.96	63	59.63	2.03
2	14.38	5.38	33	44.23	1.95	64	60.20	2.04
3	17.67	4.48	34	44.75	1.94	65	60.78	2.06
4	20.11	3.96	35	45.26	1.93	66	61.37	2.08
5	22.07	3.60	36	45.78	1.92	67	61.98	2.10
6	23.72	3.34	37	46.29	1.92	68	62.59	2.13
7	25.17	3.15	38	46.79	1.91	69	63.22	2.15
8	26.46	2.99	39	47.29	1.91	70	63.87	2.18
9	27.64	2.86	40	47.80	1.91	71	64.53	2.21
10	28.72	2.75	41	48.30	1.90	72	65.21	2.24
11	29.72	2.65	42	48.79	1.90	73	65.91	2.28
12	30.66	2.57	43	49.29	1.90	74	66.64	2.32
13	31.55	2.50	44	49.79	1.90	75	67.40	2.37
14	32.39	2.44	45	50.28	1.90	76	68.18	2.42
15	33.19	2.38	46	50.78	1.90	77	69.01	2.48
16	33.96	2.34	47	51.28	1.90	78	69.87	2.54
17	34.69	2.29	48	51.78	1.90	79	70.79	2.62
18	35.40	2.25	49	52.28	1.91	80	71.77	2.71
19	36.09	2.22	50	52.78	1.91	81	72.82	2.81
20	36.76	2.18	51	53.28	1.91	82	73.96	2.94
21	37.40	2.15	52	53.79	1.92	83	75.21	3.10
22	38.03	2.13	53	54.29	1.92	84	76.61	3.29
23	38.65	2.10	54	54.81	1.93	85	78.22	3.55
24	39.25	2.08	55	55.32	1.94	86	80.13	3.91
25	39.84	2.06	56	55.84	1.95	87	82.50	4.43
26	40.42	2.04	57	56.36	1.95	88	85.74	5.33
27	40.98	2.02	58	56.89	1.96	89	91.06	7.40
28	41.54	2.01	59	57.43	1.97	90	100.00	13.33
29	42.09	1.99	60	57.97	1.99			
30	42.64	1.98	61	58.51	2.00			

10.19 Appendix XIX – YP FV scaled score-to-measure table

# 10.20 Appendix XX – Peer-reviewed paper

Robertson, AO; Tadić, V, Rahi JS on behalf of the Child Vision Patient-Reported Outcomes (PROMs) Group. Transition from paediatric to adult ophthalmology services: what matters most to young people with visual impairment. Eye, 2017; 10.1038/eye.2017.203.